UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, DC 20549

FORM 8-K

Current Report Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported):October 26, 2009

CEDAR SHOPPING CENTERS, INC.

(Exact Name of Registrant as Specified in its Charter)

Maryland

(State or Other Jurisdiction of Incorporation)

001-31817 (Commission File Number)

42-1241468 (IRS Employer Identification No.)

44 South Bayles Avenue Port Washington, New York 11050

(Address of Principal Executive Offices) (Zip Code)

(516) 767-6492

(Registrant's Telephone Number, Including Area Code)

Not Applicable (Former Name or Former Address, if Changed Since Last Report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

- O Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
- Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
- O Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
- O Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Item 1.01Entry into a Material Definitive Agreement.

On October 26, 2009, the Company entered into a Securities Purchase Agreement (the "SPA") with RioCan Holdings USA Inc. (the "Purchaser") and RioCan Real Estate Investment Trust, providing for, among other matters, the sale by the Company to the Purchaser of 6,666,666 shares of Common Stock of the Company at a price of \$6.00 per share and the grant of a Warrant to purchase 1,428,570 shares of Common Stock of the Company. The Warrant will be exercisable for a two year period at a price of \$7.00 per share.

Pursuant to the SPA, the Company's Board of Directors agreed to waive the prohibition contained in the Company's Articles of Incorporation with respect to any person owning more than 9.9% of the Company's outstanding Common Stock so as to permit the Purchaser to acquire up to 16% of the Company's outstanding Common Stock. The Purchaser has agreed that for a period of three years after closing, except as otherwise provided in the SPA, it will not without the prior consent of the Company's Board of Directors (a) acquire, directly or indirectly, any additional securities of the Company, (b) directly or indirectly or through any other person, solicit proxies with respect to securities under any circumstance or become a "participant" in any "election contest" relating to the election of directors of the Company (as such terms are used in Rule 14a-11 of Regulation 14A under the Securities Exchange Act of 1934); provided, that Purchaser may vote its shares in any manner it deems appropriate; (c) deposit any securities in a voting trust, or subject any securities to a voting or similar agreement; (d) directly or indirectly or through or in conjunction with any other person, engage in a tender or exchange offer for the Company's securities made by any other person or entity without the prior approval of the Company, or engage in any proxy solicitation or any other activity with any other person or entity relating to the Company without the prior approval of the Company; or (e) take any action alone or in concert with any other person to acquire or change the control of the Company or participate in any group that is seeking to obtain or take control of the Company.

Under the SPA, the Company has agreed with the Purchaser that, subject to certain exceptions, it will not issue any new shares of Common Stock unless it offers to the Purchaser the right to purchase its pro rata portion of such new securities. This right will end when the Purchaser owns less than 9.9% of the Common Stock of the

Company.

Pursuant to the SPA, the Purchaser has the right to designate one person to be a director of the Company and has designated Rags Davloor, its CFO, to be a director. The Board of Directors of the Company has approved Mr. Davloor as a director to be effective upon closing of the SPA.

Within one year after the closing, the Company has agreed to register for resale under the Securities Act of 1933 the shares of Common Stock to be acquired by the Purchaser, together with shares of Common Stock issuable upon exercise of the Warrant, pursuant to a registration rights agreement in the form attached as an exhibit to the SPA. The Purchaser has agreed that without the prior consent of the Company it will not sell, assign, transfer or otherwise dispose of the shares of Common Stock or Warrant acquired pursuant to the SPA for a period of one year, subject to certain exceptions.

RioCan Real Estate Investment Trust has guaranteed the performance of the Purchaser's obligations under the SPA

Closing under the SPA is subject to the satisfaction of various conditions, including approval of the listing of the shares of Common Stock on the New York Stock Exchange, and is expected to occur within two weeks.

On October 26, 2009, the Company also entered into an agreement regarding purchase of partnership interests (the "Purchase Agreement") with the Purchaser to form a joint venture with respect to seven supermarket-anchored properties presently owned and managed by the Company. The Company will hold a 20% interest in the joint venture and the Purchaser will acquire the remaining 80% interest. The properties consist of supermarket-anchored shopping centers in Connecticut, Massachusetts and Pennsylvania. Closing for all but two of the properties are subject to receipt of lender consents to the transfer of properties to the joint venture. Closings of the joint venture arrangements are expected to be completed in the first quarter of 2010, except for two properties that are expected to close in November 2009.

The Company expects to receive approximately \$100 million in net cash proceeds from the sale of the common stock and the seven properties to the joint venture. The net proceeds will be used to reduce amounts outstanding under the Company's secured revolving credit facility for stabilized properties and secured revolving credit facility for development properties.

In addition to the above described transactions, the Company and the Purchaser have agreed to enter into 80% Purchaser and 20% Company joint ventures to acquire primarily supermarket-anchored shopping centers in the northeastern United States during the next two years in amounts anticipated at up to \$500 million. Related to the future acquisitions, the Company has granted to the Purchaser a right of first refusal for two years in the same joint venture format on primarily supermarket-anchored properties and other properties in excess of 50,000 square feet to be acquired by the Company in the states of New York, New Jersey, Pennsylvania, Massachusetts, Connecticut, Maryland and Virginia.

In both the existing and future joint ventures, the Company will provide property management, leasing, construction management and financial management services at standard rates. The Company will also be entitled to certain fees on acquisitions, dispositions, financings and refinancings.

The foregoing descriptions are summaries of the SPA and Purchase Agreement and are qualified in their entirety by reference to the SPA and Purchase Agreement that are filed hereto as Exhibits 10.1 and 10.2.

Item 2.06 Material Impairments.

As the result of entering into the Purchase Agreement with the Purchaser summarized in Item 1.01 of this Form 8-K relating to the formation of a joint venture with respect to seven supermarket anchored properties presently owned and managed by the Company, effective October 26, 2009, the Company has determined that the joint venture transaction should be treated as a "sale" for accounting purposes and, accordingly, will be recording a charge for impairment that it presently estimates will be \$23 million. This amount is subject to adjustment after the final determination of the ultimate sales price of the seven properties and final determination of costs and expenses. This charge results from the sale of the seven properties to the joint venture at a sales price that is less than the carrying amounts of such assets on the books of the Company.

Item 3.02 Unregistered Sales of Equity Securities

Reference is made to Item 1.01 for a description of the sale of shares of Common Stock and a Warrant to the Purchaser. The consideration to be received by the Company will be \$39,999,996 in cash. The sale will be exempt from registration pursuant to Section 4(2) of the Securities Act of 1933 based on representations and warranties of the Purchaser contained in the SPA. The proceeds from this sale will be used to repay amounts outstanding under the Company's stabilized credit facility.

Item 9.01 Financial Statements and Exhibits.

(d) Exhibits

Exhibit No.	Description
10.1	Securities Purchase Agreement dated as of October 26, 2009, by and among the Company, Cedar Shopping Centers Partnership L.P., RicoCan Holdings USA Inc. and RioCan Real Estate Investment Trust.
10.2	Agreement regarding purchase of Partnership Interests dated October 26, 2009 between the Company and RioCan Holdings USA Inc.
99.1	Press release issued by the Company on October 26, 2009.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Dated: October 30, 2009

CEDAR SHOPPING CENTERS, INC.

By: /s/ Leo S. Ullman

Leo S. Ullman

Chairman of the Board, President and CEO

Exhibit Index

Exhibit No.	<u>Description</u>
10.1	Securities Purchase Agreement dated as of October 26, 2009, by and among the Company, Cedar Shopping Centers Partnership L.P., RicoCan Holdings USA Inc. and RioCan Real Estate Investment Trust.
10.2	Agreement regarding purchase of Partnership Interests dated October 26, 2009 between the Company and RioCan Holdings USA Inc.
99.1	Press release issued by the Company on October 26, 2009.

SECURITIES PURCHASE AGREEMENT

Securities Purchase Agreement dated as of the 26th day of October, 2009, by and among Cedar Shopping Centers, Inc., a Maryland corporation (the "Company"), Cedar Shopping Centers Partnership L.P., a Delaware limited partnership (the "Operating Partnership"), RioCan Holdings USA Inc., a Delaware corporation (the "Purchaser"), and RioCan Real Estate Investment Trust, an unincorporated closed-end trust constituted in accordance with the laws of the Province of Ontario ("RioCan").

WITNESSETH:

WHEREAS, the Company desires to issue and sell to the Purchaser shares (the 'Shares') of Common Stock, \$.06 par value, of the Company (the "Common Stock"), and a stock purchase warrant (the "Warrant") to purchase shares of Common Stock of the Company;

WHEREAS, the Purchaser desires to purchase the Shares and the Warrant from the Company;

WHEREAS RioCan desires to guarantee the performance by the Purchaser of its covenants, agreements and obligations under this Agreement pursuant to a guaranty in the form of Exhibit A attached hereto (the "Guaranty");

NOW, THEREFORE, in consideration of the premises and the mutual agreements contained herein, the parties hereby agree as follows:

ARTICLE I PURCHASE

1.1 Purchase.

Subject to the terms and conditions set forth herein, the Company hereby agrees to issue and sell to the Purchaser, and the Purchaser hereby agrees to subscribe for and purchase from the Company, on the Closing Date set forth in Article IV, 6,666,666 Shares at a purchase price of \$6.00 per Share, which price was established on September 15, 2009, plus the Warrant to purchase 1,428,570 shares of Common Stock for which no additional consideration is being paid. The Warrant shall be in the form of Exhibit B attached hereto.

1.2 Purchase Price.

The purchase price payable by the Purchaser hereunder for the Shares and Warrant is \$39,999,996, which will be paid by the Purchaser to the Company on the Closing Date by means of a wire transfer to an account and depository designated by the Company to the Purchaser in writing at least three business days prior to the Closing Date.

ARTICLE II REPRESENTATIONS AND WARRANTIES OF THE COMPANY AND THE OPERATING PARTNERSHIP

The Company and the Operating Partnership hereby jointly and severally represent and warrant to the Purchaser, as of the date hereof, as follows and acknowledge and confirm that the Purchaser is relying upon such representations and warranties in connection with the purchase by the Purchaser of the Shares and the Warrant:

2.1 Due Organization and Qualification.

The Company is a corporation duly organized, validly existing and in good standing under the laws of the State of Maryland. The Operating Partnership is a limited partnership, duly organized and validly existing and in good standing under the laws of the State of Delaware. Each of the Company's other subsidiaries is duly organized and validly existing in good standing under the laws of the state of its incorporation or organization, except where the failure to be organized and existing will not have a Material Adverse Effect. The Company and each of the subsidiaries has all requisite corporate, partnership or entity power and authority, as the case may be, to conduct its business as it is now being conducted and to own or lease the properties and assets it now owns or holds under lease. Each of the Company and its subsidiaries is duly qualified as a foreign corporation to do business and is in good standing in every jurisdiction in which the nature of the business conducted by it makes such qualification necessary, except to the extent that the failure to be so qualified or be in good standing would not have a Material Adverse Effect. For purposes of this Agreement, "Material Adverse Effect' shall mean any condition, circumstance, or situation that could reasonably be expected to result in (i) a material adverse effect on the legality, validity or enforceability of this Agreement or the transactions contemplated herein, (ii) a material adverse effect on the results of operations, assets, business or condition (financial or otherwise) of the Company and its subsidiaries, taken as a whole, or (iii) a material adverse effect on the Company's ability to perform in any material respect on a timely basis its obligations under this Agreement.

2.2 Authorization.

The Company has the full corporate power and authority to enter into this Agreement and the Registration Rights Agreement between the Company and the Purchaser in the form of Exhibit C attached hereto (the "Registration Rights Agreement"), and to perform all of its obligations hereunder and thereunder. The Operating Partnership has the full partnership power and authority to enter into this Agreement and to perform all of its obligations hereunder. The execution, delivery and performance of this Agreement by the Company and the Operating Partnership and the Registration Rights Agreement by the Company have been duly authorized by all necessary action, and no further consent or authorization is required by the Company, its Board of Directors or its stockholders or the Operating Partnership, its general partner or its unitholders, as applicable. This Agreement and the Registration Rights Agreement constitute legal, valid and binding obligations of the Company, enforceable against the Company in accordance with their respective terms. This Agreement constitutes the legal, valid and binding obligation of the Operating Partnership, enforceable against the Operating Partnership in accordance with its terms. The issuance of the Shares and the Warrant and of the shares of Common Stock issuable upon exercise of the Warrant (the "Warrant Shares") pursuant to the provisions of

2.3 Capitalization.

The Company's authorized capitalization consists of 150,000,000 shares of Common Stock, par value \$.06 per share, and 12,500,000 shares of Preferred Stock, par value \$.01 per share. All of such outstanding shares have been validly issued and are fully paid and non-assessable and have been issued in compliance with applicable United States federal and state securities laws. The Common Stock is currently quoted on the New York Stock Exchange ("NYSE") under the trading symbol "CDR". At the date of this Agreement, there were 45,236,144 shares of Common Stock issued and outstanding and 3,550,000 shares of Series A Preferred Stock issued and outstanding. Except for (a) options to purchase 13,332 shares of Common Stock at an exercise price of \$10.50 per share (expiring July 10, 2011), (b) 2,005,888 shares of Common Stock issuable upon the redemption of outstanding units (the "Units") in the Operating Partnership, (c) warrants to purchase 83,333 Units at an exercise price of \$13.50 per Unit (expiring May 2012) and (d) the Standby Equity Purchase Agreement between the Company and YA Global Master SPV Ltd. dated September 21, 2009, there are no outstanding options, warrants or other rights, commitments or arrangements, written or oral to which the Company is a party or by which it is bound, to purchase or otherwise acquire any authorized but unissued shares of capital stock of the Company or any security convertible into or exchangeable or exercisable for any capital stock of the Company. There are no securities or instruments containing anti-dilution or similar provisions that will be triggered by this Agreement or any related agreement or the consummation of the transactions described herein or therein. The Company has furnished to the Purchaser true and correct copies of the Company's Articles of Incorporation, as amended and as in effect on the date hereof (the "Articles of Incorporation"), and the Company's By-laws, as in effect on the date hereof (the "By-laws").

2.4 Valid Issuance.

- (a) The Shares and the Warrant, when delivered in accordance with the terms of this Agreement, will be duly and validly issued, fully paid and non-assessable and will be issued free and clear of any liens or encumbrances, subject to any restrictions imposed by the Securities Act (as defined herein) and applicable state securities laws. The Warrant Shares, when issued and delivered upon exercise of the Warrant, will be duly and validly issued, fully paid and nonassessable and issued free and clear of any liens or encumbrances, subject to any restrictions imposed by the Securities Act and applicable state securities laws. The Shares, the Warrant and the Warrant Shares will be issued in compliance with all applicable United States federal and state securities laws and regulations and rules of the NYSE. Neither the stockholders of the Company, nor any other person or entity have any preemptive rights or rights of first refusal with respect to the Shares, the Warrant or the Warrant Shares or other rights to purchase or receive any of the Shares, the Warrant or the Warrant Shares or any other securities or, except as set forth on Schedule 2.4 hereto, assets of the Company, and no person has the right, contractual or otherwise, to cause the Company to issue to it, or register pursuant to the Securities Act, any shares of Common Stock or other securities or assets of the Company upon the issuance or sale of the Shares, the Warrant or the Warrant Shares. The Company is not obligated to offer the Shares, the Warrant or the Warrant Shares on a right of first refusal basis or otherwise to any third parties including, but not limited to, current or former shareholders of the Company, underwriters, brokers, or agents.
- (b) The 6,666,666 limited partnership units of the Operating Partnership (the 'OP Units') and the warrant to purchase 1,428,570 OP Units (the "OP Warrant") to be issued to the Company pursuant to Section 4.5(g) of the Agreement of Limited Partnership of Cedar Shopping Centers Partnership, L.P., a Delaware limited partnership, dated as of June 25, 1998, as amended on October 9, 2003, July 26, 2004 and March 30, 2005 (the "OP Partnership Agreement"), concurrently with the issuance of the Shares and the Warrant to the Purchaser at the Closing will be duly and validly issued, fully paid and non-assessable and will be issued free and clear of any liens or encumbrances, subject to any restrictions imposed by the Securities Act and applicable state securities laws.

2.5 SEC Documents; Financial Statements.

- (a) The Common Stock is registered pursuant to Section 12(b) of the Securities Exchange Act of 1934, as amended (together with the rules and regulations promulgated thereunder, the "Exchange Act"). The Company has filed all reports, schedules, forms, statements and other documents required to be filed by it with the Securities and Exchange Commission ("SEC") within the two years preceding the date hereof (such documents, as they may be amended after the date hereof and all exhibits included therein and financial statements and schedules thereto and documents incorporated by reference therein, being hereinafter referred to as the "SEC Documents"). The Company is current with its filing obligations under the Exchange Act and all SEC Documents have been filed on a timely basis or the Company has received a valid extension of such time of filing and has filed any such SEC Document prior to the expiration of any such extension. As of their respective dates, each SEC Document complied in all material respects as to form with the requirements of the Exchange Act and the rules and regulations of the SEC promulgated thereunder applicable to such SEC Document, and except to the extent that information contained in any SEC Document has been revised or superseded by a later filed SEC Document, none of the SEC Documents contains any untrue statement of a material fact or omits to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. As of their respective dates, the financial statements of the Company included in the SEC Documents complied as to form in all material respects with applicable accounting requirements and the published rules and regulations of the SEC with respect thereto. Such financial statements have been prepared from the books and records of the Company and its subsidiaries in accordance with U.S. generally accepted accounting principles ("GAAP"), consistently applied, during the periods involved (except (i) as may be otherwise indicated in such financial statements or the notes thereto, or (ii) in the case of unaudited interim statements, to the extent they may exclude footnotes or may be condensed or summary statements) and fairly present the financial position of the Company as of the dates thereof and the results of its operations, cash flows and changes in stockholders' equity for the periods then ended (subject, in the case of unaudited statements, to normal year-end audit adjustments).
- (b) Except (i) as set forth in the SEC Documents, (ii) for liabilities or obligations incurred in the ordinary course of business since June 30, 2009, and (iii) for mortgages placed on properties of the Company since June 30, 2009, copies of which have been provided to the Purchaser, neither the Company nor any of its subsidiaries has any material liabilities or obligations of any nature (whether accrued, absolute, contingent or otherwise) required by GAAP to be set forth on a consolidated balance sheet of the Company or in the notes thereto.
- (c) None of the Company's subsidiaries is, or has at any time within the two years preceding the date hereof been, subject to the reporting requirements of Sections 13(a) and 15(d) of the Exchange Act.

2.6 No Violations.

None of the sale or issuance of the Shares or Warrant (or the issuance and delivery of the Warrant Shares), the execution and delivery by the Company of this Agreement and the Registration Rights Agreement, or the fulfillment by the Company of the terms set forth in this Agreement and the Registration Rights Agreement and the consummation of the transactions contemplated by this Agreement and the Registration Rights Agreement (including, without limitation, the granting by the Company to the Purchaser of the preemptive rights in Section 9.3 hereof), will (i) except as disclosed in writing to the Purchaser on October 21, 2009, violate, conflict with or constitute a breach of, or constitute a default under or an event which, with or without notice or lapse of time or both, would be a breach of or default under or violation of the Articles of Incorporation or By-laws or would be a breach of or default under or violation of, or give to others any rights of termination, amendment, acceleration or cancellation of, any agreement, document, indenture, mortgage or other instrument or undertaking by which the Company or any of its subsidiaries is bound or to which any of its or their properties are subject, or would be a violation of any law, rule, administrative regulation, judgment, order or decree applicable to the Company or any of its subsidiaries, (ii) except as disclosed in writing to the Purchaser on October 21, 2009, require the consent of any person or entity under any agreement, indenture, mortgage, document or other instrument or undertaking by which the Company or any of its subsidiaries is bound or to which any of its or their properties are subject, (iii) except as specifically contemplated by this Agreement, as required under the NYSE, as required under the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder (the "Securities Act") and any applicable state securities laws, require the Company or any of its subsidiaries to obtain any consent, authorization or order of, or make any filing or registration with, any court or governmental agency in order for it to execute, deliver or perform any of its obligations under or contemplated by this Agreement in accordance with the terms hereof, except as such consent, authorization or order has been obtained prior to the date hereof, except for approval of the NYSE or registration of the Shares and Warrant Shares as required under the Registration Rights Agreement, (iv) require the consent, approval or authorization of the stockholders of the Company under the rules and regulations of the NYSE, or (v) require any action under the terms of Subtitle 7 of Title 3 of the Maryland General Corporation Law (the "MGCL") or Section 3-602 of the MGCL. Except as disclosed in the SEC Documents, neither the Company nor its subsidiaries is in violation of any term of or in default under its Articles of Incorporation or Bylaws or their organizational charter or by-laws, respectively, or any material contract, agreement, mortgage, indebtedness, indenture, instrument, judgment, decree or order or any statute, rule or regulation applicable to the Company or its subsidiaries. The business of the Company and its subsidiaries is not being conducted in violation of any material law, ordinance or regulation of any governmental entity. All the properties and assets owned by the Company or any of its subsidiaries that are subject to the Joint Venture Purchase Agreement (as defined below) constitute exempt assets for purposes of the Hart-Scott-Rodino Antitrust Improvements Act of 1976 (the "HSR Act") and Rules 802.2(h) and 802.5 thereunder, and, in connection with the contemplated joint venture involving the Operating Partnership and the Purchaser (the "Rapids JV"), the Company shall use its commercially reasonable efforts to cause the Rapids JV to use such properties and assets that are subject to the Joint Venture Purchase Agreement for investment or rental purposes only in compliance with the exemption provided by Rule 802.5 under the HSR Act.

The Company and its subsidiaries own or possess adequate rights or licenses to use all material trademarks, trade names, service marks, service mark registrations, service names, patents, patent rights, copyrights, inventions, licenses, trade secrets and rights necessary to conduct their respective businesses as now conducted. The Company and its subsidiaries do not have any knowledge of any infringement by the Company or its subsidiaries of trademarks, trade name rights, patents, patent rights, copyrights, inventions, licenses, service names, service marks, service mark registrations, trade secrets or other similar rights of others, and to the knowledge of the Company, there is no claim, action or proceeding being made or brought against or being threatened against, the Company or its subsidiaries regarding trademarks, trade names, patents, patent rights, inventions, copyrights, licenses, service names, service marks, service mark registrations, trade secrets or other infringement.

2.8 Employee Relations.

Neither the Company nor any of its subsidiaries is involved in any labor dispute nor, to the knowledge of the Company or any of its subsidiaries, is any such dispute threatened, in each case which would have a Material Adverse Effect. Except for matters which would not, individually or in the aggregate, have a Material Adverse Effect: (i) neither the Company nor any of its subsidiaries is engaged in any unfair labor practice; (ii) there is (A) no unfair labor practice complaint pending or, to the Company's or the Operating Partnership's knowledge, threatened against the Company or any of its subsidiaries before the National Labor Relations Board, and no grievance or arbitration proceeding arising out of or under collective bargaining agreements is pending or to the knowledge of the Company threatened, (B) no strike, labor dispute, slowdown or stoppage is pending or, to the Company's or the Operating Partnership's knowledge, threatened against the Company or any of its subsidiaries, and (C) to the knowledge of the Company no union representation dispute currently existing concerning the employees of the Company or any of its subsidiaries; and (iii) to the Company's knowledge, (A) no union organizing activities are currently taking place concerning the employees of the Company or any of its subsidiaries and (B) there is no violation of any federal, state, local or foreign law relating to discrimination in the hiring, promotion or pay of employees, any applicable wage or hour laws or any provision of the Employee Retirement Income Security Act of 1974 ("ERISA") or the rules and regulations promulgated thereunder concerning the employees of the Company or any of its subsidiaries.

The Company and its subsidiaries are in compliance with wage and hour determinations issued by the U.S. Department of Labor under the Service Contract Act of 1965 and the Fair Labor Standards Act in paying its employees' salaries, fringe benefits and other compensation for the performance of work or other duties in connection with contracts with the U.S. government, and are in compliance with the requirements of the Americans with Disabilities Act of 1990, the Family and Medical Leave Act of 1993, the Civil Rights Act of 1964 (Title VII), the National Labor Relations Act, the Vietnam Era Veteran's Readjustment Act, the Age Discrimination in Employment Act, as amended by the Older Workers' Benefit Protection Act, and federal, state and local labor laws, each as amended except in each case where the failure to comply with any such requirements has not, and will not, have a Material Adverse Effect.

2.9 Environmental Laws.

Except as disclosed in the SEC Documents, the Company and its subsidiaries (i) are in compliance with any and all material foreign, federal, state and local laws and regulations relating to the protection of human health and safety, the environment or hazardous or toxic substances or wastes, pollutants or contaminants ("Environmental Laws") applicable to its or their use or control of the Owned Real Property and the Leased Real Property (as such terms are defined herein), (ii) have received all permits, licenses or other approvals required of them under applicable Environmental Laws to conduct their respective businesses and (iii) are in compliance with all terms and conditions of any such permit, license or approval, in each case except where any noncompliance or non-receipt would not, individually or in the aggregate, have a Material Adverse Effect. Except as disclosed in the SEC Documents or as would not, singly or in the aggregate, result in a Material Adverse Effect, there are no pending or, to the knowledge of the Company, threatened administrative, regulatory or judicial actions, suits, demands, demand letters, claims, liens, notices of noncompliance or violation, investigation or proceedings relating to any Environmental Law against the Company or any of its subsidiaries with respect to its or their use or control of the Owned Real Property and the Leased Real Property. Neither the Company nor any of its subsidiaries has been named as a "potentially responsible party" under the Comprehensive Environmental Response Compensation and Liability Act of 1980, as amended. To the knowledge of the Company, neither the Company nor any of its subsidiaries owns, leases or occupies any property that appears on any list of hazardous sites compiled by any state or local government agency.

2.10 Title

Except as disclosed in the SEC Documents, the Company, each of its subsidiaries and any joint ventures in which the Company or any of its subsidiaries owns an interest, as the case may be, has good and marketable fee title to all material real property and material tangible assets owned by it that are used or held for use in, or that are necessary to conduct its business as conducted on the date hereof (each an "Owned Real Property"), free and clear of any pledge, lien, security interest, encumbrance or claim (each of the foregoing a "Lien"), other than (a) Liens permitted under any of the loan and credit facilities disclosed in the SEC Documents, (b) Liens permitted under the terms hereof or any agreement executed and delivered in connection herewith, and (c) Liens that are not material to the business of the Company, and (d) mortgages placed on properties subsequent to the date of the latest SEC Document, copies of which have been provided to the Purchaser. Except as disclosed in the SEC Documents, (i) any material real property held under lease by the Company and its subsidiaries (each a "Leased Real Property") are held by them under valid, subsisting and enforceable leases and (ii) no notice of default or termination thereunder is outstanding. Except as disclosed in the SEC Documents, each of the Owned Real Properties and Leased Real Properties complies with all applicable building, zoning, subdivision and other land use codes, laws and regulations, except for such failures to comply that would not in the aggregate have a Material Adverse Effect. Except as disclosed in the SEC Documents, the Company does not have any knowledge of any pending or threatened condemnation proceedings affecting the Owned Real Property, except such condemnation proceedings that would not have a Material Adverse Effect.

2.11 Insurance.

The Company and each of its subsidiaries are insured against such losses and risks and in such amounts as management of the Company believes to be prudent and customary in the businesses in which the Company

and its subsidiaries are engaged. Neither the Company nor any such subsidiary has any reason to believe that it will not be able to renew its existing insurance coverage as and when such coverage expires or to obtain similar coverage from similar insurers as may be necessary to continue its business at a cost that would not have a Material Adverse Effect.

2.12 Title Insurance.

Title insurance in favor of the Company and its subsidiaries, as the case may be, has been obtained with respect to each material property owned by any such entity in an amount at least equal to the cost of acquisition of such property, except where the failure to obtain such title insurance would not have a Material Adverse Effect.

2.13 REIT Status.

The Company qualifies as a real estate investment trust (including its organization and method of operations and those of its subsidiaries) for U.S. federal income tax purposes.

2.14 Regulatory Permits.

The Company and its subsidiaries possess all material certificates, authorizations and permits issued by the appropriate federal, state or foreign regulatory authorities necessary to conduct their respective businesses, and neither the Company nor any such subsidiary has received any notice of proceedings relating to the revocation or modification of any such certificate, authorization or permit, except in each instance which would not have a Material Adverse Effect.

2.15 Internal Accounting Controls.

The Company and each of its subsidiaries maintain an effective system of internal control over financial reporting.

2.16 No Material Adverse Breaches, etc.

Except as disclosed in the SEC Documents, neither the Company nor any of its subsidiaries is subject to any charter, corporate or other legal restriction, or any judgment, decree, order, rule or regulation which in the judgment of the Company's officers has or is expected in the future to have a Material Adverse Effect.

2.17 Absence of Litigation.

Except as disclosed in the SEC Documents, there is no action, suit, proceeding, inquiry or investigation before or by any court, public board, government agency, self-regulatory organization or body pending against or affecting the Company, the Common Stock or any of the Company's subsidiaries, wherein an unfavorable decision, ruling or finding would have a Material Adverse Effect.

2.18 Subsidiaries.

Except as disclosed in the SEC Documents, the Company does not presently own or control, directly or indirectly, any interest in any other material corporation, partnership, association or other business entity, except for those acquired or formed in the ordinary course of business since December 31, 2008. The outstanding equity interests of, or other ownership interests in, each of the Company's subsidiaries have been duly authorized and validly issued, are fully paid and, except as to subsidiaries that are partnerships or limited liability companies, nonassessable, and are owned by the Company, directly or indirectly, free and clear of any security interest, lien, encumbrance or claim, except for Joint Ventures listed on Schedule 2.32.

2.19 Tax Status.

Except as disclosed in the SEC Documents and subject to valid extensions, the Company and each of its subsidiaries has made or filed all material federal and state income and all other material tax returns, reports and declarations required by any jurisdiction to which it is subject and (unless and only to the extent that the Company and each of its subsidiaries has set aside on its books provisions reasonably adequate for the payment of all unpaid and unreported taxes) has paid all taxes and other governmental assessments and charges that are material in amount, shown or determined to be due on such returns, reports and declarations, except those being contested in good faith and has set aside on its books provision reasonably adequate for the payment of all taxes for periods subsequent to the periods to which such returns, reports or declarations apply. There are no unpaid taxes by the Company or its subsidiaries in any material amount claimed to be due by the taxing authority of any jurisdiction.

2.20 Certain Transactions.

Except as disclosed in the SEC Documents or the recent leasing of additional office space at the Company's executive offices, the particulars of which have been fully disclosed in writing to the Purchaser, none of the officers or directors of the Company is presently a party to any transaction with the Company which individually or in the aggregate would be material (other than for services as officers and directors), including any contract, agreement or other arrangement providing for the furnishing of services to or by, providing for rental of real or personal property to or from, or otherwise requiring payments to or from any officer or director or, to the knowledge of the Company, any corporation, partnership, trust or other entity in which any officer or director has a substantial interest or is an officer, director, trustee or partner.

2.21 Waiver of Ownership Limit.

The Board of Directors of the Company has waived the Ownership Limit (as defined in the Articles of Incorporation) to permit the Purchaser to acquire and hold an ownership position in the Company in an amount not to exceed 16% of the issued and outstanding Common Stock (provided, however, that the Purchaser shall not be in breach of the Ownership Limit if its percentage ownership increases above 16% solely as the result of the

repurchase by the Company of its Common Stock), and such waiver is in full force and effect. A true and complete copy of such waiver has been delivered to the Purchaser prior to the execution hereof.

2.22 Absence of Manipulations.

The Company has not taken and will not take, directly or indirectly, any action designed to cause or result in, or that constitutes or might reasonably be expected to result in or constitute, the stabilization or manipulation of the price of any security of the Company or which caused or resulted in, or which would in the future reasonably be expected to cause or result in, stabilization or manipulation of the price of any security of the Company.

2.23 Investment Company.

Neither the Company nor any of its subsidiaries is an "investment company" or an "affiliated person" of, or "promoter" or "principal underwriter" for, an "investment company," as such terms are defined in the Investment Company Act of 1940, as amended, nor is the Company nor any of its subsidiaries, directly or indirectly, controlled by or acting on behalf of any person that is an "investment company" within the meaning of such act

2.24 Disclosure Controls and Procedures.

The Company has established and maintains disclosure controls and procedures (as such term is defined in Rule 13a-15 and 15d-15 under the Exchange Act) in accordance with the rules and regulations under the Sarbanes-Oxley Act, of 2002, the Securities Act and the Exchange Act.

2.25 Sarbanes-Oxley.

There is and has been no failure on the part of the Company or any of the officers and directors of the Company, in their capacities as such, to comply in all material respects with the provisions of the Sarbanes-Oxley Act of 2002 and the rules and regulations promulgated thereunder.

2.26 Maryland Business Combination Act; Maryland Control Share Acquisition Act.

Pursuant to the Articles of Incorporation, Section 3-602 of the MGCL is inapplicable to any business combination between the Company and any other person. Pursuant to the Articles of Incorporation, Subtitle 7 of Title 3 of the MGCL is inapplicable to any acquisition of Control Shares (as defined in the MGCL) that is not prohibited by the terms of Article IV of the Articles of Incorporation. To the extent that any acquisition of shares of Common Stock by the Purchaser pursuant to this Agreement would constitute an acquisition of Control Shares, such acquisition has been exempted from Subtitle 7 of Title 3 of the MGCL pursuant to the waiver of the Ownership Limit referred to in Section 2.21 above.

2.27 Patriot Act.

No portion of the purchase price received by the Company hereunder will be used to further any action in violation or contravention of the U.S.A. Patriot Act or otherwise violate or contravene the rules, regulations or policies of the U.S. Office of Foreign Assets Control.

2.28 FCPA.

Neither the Company nor any of its subsidiaries nor, to the knowledge of the Company or the Operating Partnership, any director, officer, agent or employee of the Company or any of its subsidiaries is aware of or has taken any action, directly or indirectly, that would result in a violation by such persons of the Foreign Corrupt Practices Act of 1977, as amended, and the rules and regulations thereunder (the "FCPA"), including, without limitation, making use of the mails or any means or instrumentality of interstate commerce corruptly in furtherance of an offer, payment, promise to pay or authorization of the payment of any money, or other property, gift, promise to give, or authorization of the giving of anything of value to any "foreign official" (as such term is defined in the FCPA) or any foreign political party or official thereof or any candidate for foreign political office, in contravention of the FCPA; and the Company and its subsidiaries have conducted their businesses in compliance with the FCPA and have instituted and maintain policies and procedures designed to ensure, and which are reasonably expected to continue to ensure, continued compliance therewith.

2.29 Money Laundering Laws.

The operations of the Company and its subsidiaries are and have been conducted at all times in compliance in all material respects with applicable financial recordkeeping and reporting requirements and the money laundering statutes and the rules and regulations thereunder and any related or similar rules, regulations or guidelines, issued, administered or enforced by any governmental agency applicable to the Company or its subsidiaries (collectively, the "Money Laundering Laws") and no action, suit or proceeding by or before any court or governmental agency, authority or body or any arbitrator involving the Company or any of its subsidiaries with respect to the Money Laundering Laws is pending or, to the best knowledge of the Company, threatened.

2.30 Private Offering.

The Company has not issued, sold or offered, or solicited any offers to acquire, any security of the Company to or from any person under circumstances that would cause the sale of the Shares, the Warrant or the Warrant Shares as contemplated by this Agreement, to be subject to the registration requirements of the Securities Act. Assuming the representations of Purchaser contained in Sections 3.4, 3.5, 3.6 and 3.7 are true and correct, the sale and delivery of the Shares, the Warrant and the Warrant Shares hereunder are exempt from the registration and prospectus delivery requirements of the Securities Act.

2.31 No Material Adverse Change.

Since June 30, 2009, there has not been any adverse change in the Company's results of operations, assets, business or condition (financial or otherwise) that may reasonably be expected to result in a Material Adverse Effect.

2.32 Joint Ventures.

All of the joint ventures in which the Company or any subsidiary owns an interest of greater than 5.0% and that are currently conducting business (the "Joint Ventures") are listed on Schedule 2.32 hereto. The Company's (or subsidiary's, as the case may be) ownership interest in such Joint Venture is as set forth on Schedule 2.32. To the knowledge of the Company and the Operating Partnership, each of the Joint Ventures possesses such certificates, authorizations or permits issued by the appropriate state, federal or foreign regulatory agencies or bodies necessary to conduct the business now being conducted by it, and none of the Joint Ventures has received notice of any proceedings relating to the revocation or modification of any such certificate, authority or permit which singly or in the aggregate, if the subject of an unfavorable ruling or decision, would have a Material Adverse Effect.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE PURCHASER

The Purchaser hereby represents and warrants to the Company, as of the date hereof as follows and hereby acknowledges and confirms that the Company is relying upon such representations and warranties in connection with the issuance to the Purchaser of the Shares and the Warrant:

3.1 Due Organization.

The Purchaser is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware. The Purchaser has all requisite power and authority to conduct its business as it is now being conducted and to own or lease the properties and assets it now owns or holds under lease.

3.2 Authorization.

The Purchaser has the full power and authority to enter into this Agreement and the Registration Rights Agreement and to perform all of its obligations hereunder and thereunder. The execution, delivery and performance of this Agreement and the Registration Rights Agreement by the Purchaser have been duly authorized by all necessary action. This Agreement and the Registration Rights Agreement constitute legal, valid and binding obligations of the Purchaser, enforceable against the Purchaser in accordance with their respective terms

3.3 No Violations.

None of the purchase of the Shares or Warrant (or the issuance and delivery of the Warrant Shares), the execution and delivery of this Agreement and the Registration Rights Agreement by the Purchaser, nor the fulfillment by the Purchaser of the terms set forth in this Agreement and the Registration Rights Agreement and the consummation of the transactions contemplated by this Agreement and the Registration Rights Agreement, will (i) violate, conflict with or constitute a breach of, or constitute a default under or an event which, with or without notice or lapse of time or both, would be a breach of or default under or violation of the organizational documents of the Purchaser or would be a breach of or default under or violation of any agreement, document, indenture, mortgage or other instrument or undertaking by which the Purchaser or any of its affiliates (as such term is defined in Rule 12b-2 of Regulation 12b under the Exchange Act) is bound or to which any of its or their properties are subject, or would be a violation of any law, rule, administrative regulation, judgment, order or decree applicable to the Purchaser or any of its affiliates, (ii) require the consent of any person or entity under any agreement, indenture, mortgage, document or other instrument or undertaking by which the Purchaser or any of its affiliates is bound or to which any of its or their properties are subject, or (iii) except as specifically contemplated by this Agreement or as required under the Securities Act, require the Purchaser or any of its affiliates to obtain any consent, authorization or order of, or make any filing or registration with, any court or governmental agency in order for it to execute, deliver or perform any of its obligations under or contemplated by this Agreement in accordance with the terms hereof, except as such consent, authorization or order has been obtained prior to the date hereof.

3.4 Investment Intent.

The Purchaser is acquiring the Shares and Warrant (and Warrant Shares) for its own account, for investment purposes only, and not with a view to, or in connection with, any resale or other distribution thereof in violation of any of the registration requirements of the Securities Act. Other than the Shares to be acquired on the Closing Date pursuant hereto, the Purchaser does not beneficially own any shares of Common Stock of the Company.

3.5 No Registration under Federal or State Securities Laws.

The Purchaser acknowledges that the Shares, Warrant and Warrant Shares have not been registered under the Securities Act or the securities laws of any state by reason of a specific exemption or exemptions from registration under the Securities Act and applicable state securities laws, and that the Company's reliance on such exemptions is predicated on the accuracy and completeness of the Purchaser's representations, warranties, acknowledgments and agreements herein. Accordingly, the Shares, Warrant and Warrant Shares may not be offered, sold, transferred, pledged or otherwise disposed of by the Purchaser without an effective registration statement under the Securities Act and any applicable state securities laws or an opinion of counsel acceptable to the Company that the proposed transaction will be exempt from registration. The Purchaser understands that the Shares, Warrant and Warrant Shares will bear a legend substantially to the effect of the following:

"The securities represented by this certificate have not been registered under the Securities Act of 1933, as amended (the "Act"), or the securities laws of any state. The securities may not be

offered, sold, transferred, pledged or otherwise disposed of without an effective registration statement under the Act and under any applicable state securities laws, receipt of a no-action letter issued by the Securities and Exchange Commission (together with either registration or an exemption under applicable state securities laws) or an opinion of counsel acceptable to the Company that the proposed transaction will be exempt from registration under the Act and applicable state securities laws."

and that the Company will instruct its transfer agent to place a stop order against the transfer of the certificates representing the Shares and Warrant Shares and refuse to effect any transfers thereof in the absence of satisfying the conditions contained in the foregoing legend.

3.6 Investment Experience.

The Purchaser has such knowledge and experience in financial and business matters that the Purchaser is capable of evaluating the merits and risks of its investment in the Company and of protecting its own interests in connection therewith. The Purchaser is an "accredited investor" within the meaning of Rule 501(a) promulgated under the Securities Act.

3.7 Investment Risks.

The Purchaser acknowledges that an investment in the Company involves substantial risks. The Purchaser is able to bear the economic risk of its investment for an indefinite period of time.

3.8 Commissions.

The Purchaser has not paid or given any commission or other remuneration in connection with the purchase of the Shares, other than the payment of fees to RBC Capital Markets in connection with advisory services provided by it to the Purchaser and its affiliates in connection with the transactions contemplated by this Agreement.

3.9 Tax Representations.

Capitalized terms used in this Section 3.9 but not otherwise defined in this Agreement shall have the meanings assigned to such terms in the Articles of Incorporation of the Company.

- (a) No "individual" who Beneficially Owns for Tax Purposes any of the outstanding Common Stock of the Company held by the Purchaser does or will Beneficially Own for Tax Purposes in the aggregate more than 9.9% of the value of the outstanding Common Stock of the Company by reason of (i) a direct or indirect ownership interest in the Purchaser (or any direct or indirect member or owner thereof) and/or (ii) a direct or indirect interest in the Company. The determination by the Purchaser of the number in clause (ii) of the preceding sentence is or will be based solely on information disclosed in writing to the Purchaser by either the Company or a shareholder of Purchaser (or a direct or indirect member or owner thereof).
- (b) Neither the Purchaser nor any person on behalf of whom the Purchaser owns Common Stock will actually own or Constructively Own an interest in a tenant of the Company (or a tenant of an entity owned or controlled by the Company) that would cause the Company to Constructively Own in the aggregate more than a 9.9% interest (as set forth in Section 856(d) (2)(B) of the *Internal Revenue Code of 1986*, as amended (the "Code")) in such tenant.
- (c) No person or entity owns more than fifty percent of the equity interest in RioCan.
- (d) For purposes of these tax representations, "individual" has the same meaning provided in Section 542(a)(2) of the Code, "Beneficially Owns for Tax Purposes" means direct, indirect, or constructive ownership through the application of Section 544 of the Code, as modified by Section 856(h)(1) of the Code, and "Constructively Own" means direct, indirect, or constructive ownership through the application of Section 318 of the Code, as modified by Section 856(d)(5) of the Code.
- (e) The Purchaser agrees that any violation or attempted violation of the representations (i) set forth in Section 3.9(a) solely by reason of an ownership interest described in clause (i) thereof, or (ii) if such violation or attempted violation causes or would cause the Company to lose its status as a real estate investment trust for U.S. federal income tax purposes, set forth in any other provision of Section 3.9, will result in all or a portion of the shares of Common Stock of the Company owned by the Purchaser being automatically transferred to a Charitable Trust in accordance with Section B (4)(c)(i) of Article IV of the Articles of Incorporation
- (f) The Purchaser expressly permits the Company and Stroock & Stroock & Lavan LLP, as counsel to the Company, to rely on the representations set forth in this Section 3.9 as if the Purchaser made such representations directly to both the Company and Stroock & Stroock & Lavan LLP.

(g) The Purchaser agrees and acknowledges that the continued truth and accuracy of the representations set forth in this Section 3.9 is a condition precedent to the validity and effectiveness of the waiver of the ownership limit granted in connection herewith and that the Purchaser will, upon request by the Company, promptly deliver written confirmation of such representations

3.10 Financial Resources.

The Purchaser has sufficient funding, and will have sufficient funds available at the Closing, to enable it to purchase the Shares on the terms hereof and to undertake the agreements and obligations contemplated by this Agreement.

3.11 Opportunity for Independent Investigation.

Except for the representations and warranties made by the Company in Article II, the Purchaser is relying on its own investigation of the Company in making its decision to execute this Agreement and to purchase the Shares. The Purchaser has conducted its own independent investigation, review and analysis of the business, operations, assets, liabilities, results of operations, financial condition, and prospects of the Company, which investigation, review and analysis was conducted by or on behalf of the Purchaser. In entering into this Agreement, the Purchaser acknowledges that it has relied solely upon such investigation, review and analysis and not on any factual representations of the Company, other than the specific representations and warranties of the Company set forth in Article II. Except as provided in Section 7.1, nothing in this Section 3.11 shall diminish or limit in any manner the effect of the representations and warranties of the Company set forth in Article II or the indemnity of the Company provided in Article VI.

3.12 HSR Act.

In connection with its participation in the Rapids JV, the Purchaser shall use its commercially reasonable efforts to cause the Rapids JV to use the properties and assets that are subject to the Joint Venture Purchase Agreement for investment or rental purposes only in compliance with the exemption provided by Rule 802.5 under the HSR Act.

ARTICLE IV CLOSING DATE

4.1 Closing Date.

The closing (the "Closing") of the transactions contemplated by this Agreement shall take place at the offices of Stroock & Stroock & Lavan LLP, 180 Maiden Lane, New York, New York 10038 at 10:00 a.m. on the second business day after the satisfaction or waiver of the conditions contained in Article VII and Article VIII, except for any conditions that by their nature can only be satisfied on the Closing Date, but subject to the satisfaction of such conditions or waiver by the party entitled to waive such conditions (the "Closing Date") or at such other time, place or date as the parties may mutually agree. At the Closing, (i) the Purchaser shall deliver to the Company the purchase price as set forth in Section 1.2, (ii) the Company shall deliver to the Purchaser the Shares and the Warrant and (iii) the Purchaser and the Company shall execute and deliver the Registration Rights Agreement and the Joint Venture Purchase Agreement.

ARTICLE V COVENANTS

5.1 Completion of Transaction.

- (a) The Company covenants and agrees with the Purchaser to perform all obligations required to be performed by the Company under this Agreement, and to use commercially reasonable efforts to do such acts and things as may be necessary in order to complete the transactions contemplated by this Agreement as soon as reasonably practicable following the satisfaction of the conditions thereto, including taking or causing to be taken all commercially reasonable steps necessary to satisfy the conditions set forth in Article VII.
- (b) The Purchaser covenants and agrees with the Company to perform all obligations required to be performed by the Purchaser under this Agreement, and to use commercially reasonable efforts to do such acts and things as may be necessary in order to complete the transactions contemplated by this Agreement as soon as reasonably practicable following the satisfaction of the conditions thereto, including taking or causing to be taken all commercially reasonable steps necessary to satisfy the conditions set forth in Article VIII.

5.2 NYSE Approval.

Within two business days following the date hereof, the Company shall file with the NYSE a supplemental listing application with respect to the Shares and the Warrant Shares, and the Company shall use commercially reasonable efforts to cause the listing of the Shares and Warrant Shares to be approved by the NYSE, subject to official notice of issuance.

5.3 RioCan Guarantee.

RioCan shall unconditionally and irrevocably guarantee the due and punctual payment and performance by the Purchaser of all of its covenants, agreements and obligations under this Agreement pursuant to the Guaranty.

6.1 Indemnity.

The Company and the Operating Partnership agree, jointly and severally, to indemnify, release, defend and hold harmless the Purchaser and its affiliates, nominees, successors and agents and the respective directors, trustees, officers, agents and employees of each of the foregoing (the "Indemnified Parties") from and against all losses, claims, damages, liabilities, penalties, fines, judgments, suits, awards, sanctions, fees, whatsoever or other costs or expenses to which an Indemnified Party may become subject, other than those relating to such Indemnified Party's fraud, willful misconduct or gross negligence or such Indemnified Party's breach of, default under or non-compliance in any material respect with any material representation, warranty, term, condition or covenant of this Agreement, which arise out of or relate to or result from, directly or indirectly, any breach or default of or under any representation, warranty, covenant or agreement of the Company or the Operating Partnership contained herein. In addition to the foregoing, the Company and the Operating Partnership agree, jointly and severally, to reimburse each Indemnified Party for all reasonable legal or other expense incurred in connection with investigating, defending or participating in any action or other proceeding relating to any such losses or liabilities (whether or not such Indemnified Party is a party to such action or proceeding).

6.2 Indemnity by the Purchaser.

The Purchaser agrees to indemnify, release, defend and hold harmless the Company and its subsidiaries, affiliates, nominees, successors and agents and the respective directors, trustees, officers, agents and employees of each of the foregoing (the "Company Indemnified Parties") from and against all losses, claims, damages, liabilities, penalties, fines, judgments, suits, awards, sanctions, fees whatsoever or other costs or expenses to which a Company Indemnified Party may become subject, other than those relating to such Company Indemnified Party's fraud, willful misconduct or gross negligence or such Company Indemnified Party's breach of, default under or non compliance in any material respect with any material representation, warranty, term, condition or covenant of this Agreement, which arise out of or relate to or result from, directly or indirectly, any breach or default of or under any representation, warranty, covenant or agreement of the Purchaser contained herein. In addition to the foregoing, the Purchaser agrees to reimburse each Company Indemnified Party for all reasonable legal or other expense incurred in connection with investigating, defending or participating in any action or other proceeding relating to any such losses or liabilities (whether or not such Company Indemnified Party is a party to such action or proceeding).

6.3 Limitations on Indemnification.

- (a) No claim for indemnification may be made with respect to a representation and warranty after the expiration of 18 months from the Closing Date.
- (b) Except for fraud, willful misconduct or gross negligence, neither the Company nor the Operating Partnership shall under any circumstances be liable to the Purchaser for consequential, incidental or punitive damages, including damages for lost profits or other speculative damages.
- (c) This Article VI shall be the exclusive remedy of the parties hereto following the Closing for any losses arising out of any misrepresentation or breach of the representations or warranties of the parties contained in this Agreement, except for fraud, willful misconduct or gross negligence. In furtherance of the foregoing, each of the parties hereto hereby waives, to the fullest extent permitted under applicable law, any and all rights, claims and causes of action it may have against the other party hereto, arising under or based upon any law, other than the right to seek indemnity pursuant to this Article VI or as provided in the following sentence. The parties acknowledge that the failure to comply with a covenant or obligation contained in this Agreement may give rise to an irreparable injury to a party inadequately compensable in damages. Accordingly, a party may seek to enforce the performance of this Agreement by injunction or specific performance upon application to a court of competent jurisdiction.

ARTICLE VII CLOSING CONDITIONS OF PURCHASER

The obligations of the Purchaser under this Agreement are subject to the satisfaction of the following conditions on or prior to the Closing Date, any of which may be waived in whole or in part by the Purchaser:

7.1 Representations and Warranties.

The representations and warranties of the Company contained in this Agreement shall be true and correct as of the date hereof and as of the Closing Date with the same force and effect as if made on the Closing Date, except to the extent that any such representation and warranty is made as of a specific date, in which case such representation and warranty shall be true and correct on such specific date. If prior to the Closing Date the Purchaser becomes aware of the breach of or inaccuracy in any representation or warranty made by the Company in this Agreement, then the Purchaser (a) shall promptly notify the Company thereof and (b) if the Purchaser elects to consummate this Agreement, will release the Company from any liability for any loss or damage which may be sustained by reason of such breach or inaccuracy.

7.2 Covenants.

The Company will have performed, satisfied and complied in all material respects with all covenants, agreements and conditions required by this Agreement to be performed, satisfied or complied with by the Company on or before the Closing Date.

7.3 Director.

Rags Davloor, as the initial designee of the Purchaser, shall have been appointed as a director of the

Company as provided in Section 9.1.

7.4 Litigation.

No action, suit or proceeding shall have been instituted by or before any court or governmental body or instituted or threatened by any governmental agency or body to restrain or prevent the carrying out of the transactions contemplated hereby.

7.5 Registration Rights Agreement.

The Company shall have executed and delivered the Registration Rights Agreement.

7.6 Certificates.

The Purchaser shall have received a stock certificate representing the Shares purchased by the Purchaser, plus the Warrant.

7.7 Joint Venture.

The Operating Partnership, or one or more subsidiaries of the Operating Partnership, as the case may be, shall have executed and delivered the purchase and sale agreement between the Operating Partnership and the Purchaser providing for the purchase by the Purchaser of a direct or indirect 80% interest from the Operating Partnership, or one or more subsidiaries of the Operating Partnership, as the case may be, of seven existing properties in the form attached hereto as Exhibit D (the "Joint Venture Purchase Agreement").

7.8 NYSE Matters.

The NYSE shall have approved the listing of the Shares and the Warrant Shares, subject to official notice of issuance. The Common Stock (other than the Shares and the Warrant Shares) shall be quoted on the NYSE, and the Company shall not have received any notice threatening the continued listing of the Common Stock on the NYSE or notice that the issuance of the Shares, the Warrant or the Warrant Shares will violate the shareholder approval requirements of the NYSE.

7.9 Credit Facility.

The Company shall have used its commercially reasonable efforts to obtain commitments to renew its Loan Agreement dated January 30, 2004, as amended, in an amount that would not be less than \$196,000,000.

7.10 Opinions of Counsel.

The Purchaser shall have received opinion letters from Stroock & Stroock & Lavan LLP in the forms of Schedules A and B attached hereto.

ARTICLE VIII CLOSING CONDITIONS OF THE COMPANY

The obligations of the Company under this Agreement are subject to the satisfaction of the following conditions on or prior to the Closing Date, any of which may be waived in whole or in part by the Company:

8.1 Representations and Warranties.

The representations and warranties of the Purchaser contained in this Agreement shall be true and correct as of the date hereof and as of the Closing Date with the same force and effect as if made on the Closing Date.

8.2 Covenants.

The Purchaser will have performed, satisfied and complied in all material respects with all covenants, agreements and conditions required by this Agreement to be performed, satisfied or complied with by the Purchaser on or before the Closing Date.

8.3 Litigation.

No action, suit or proceeding shall have been instituted by or before any court or governmental body or instituted or threatened by any governmental agency or body to restrain or prevent the carrying out of the transactions contemplated hereby.

8.4 NYSE Approval.

The NYSE shall have approved the listing of the Shares and the Warrant Shares, subject to official notice of issuance

8.5 Joint Venture.

The Purchaser shall have executed and delivered the Joint Venture Purchase Agreement.

8.6 Purchase Price.

The Company shall have received the purchase price payable by the Purchaser as provided in Section

8.7 Registration Rights Agreement.

The Purchaser shall have executed and delivered the Registration Rights Agreement.

8.8 Guaranty.

The Guaranty shall remain in full force and effect.

ARTICLE IX POST CLOSING COVENANTS

9.1 Directors.

Commencing on the Closing Date and as long as the Purchaser and its affiliates continue to beneficially own more than 9.9% of the outstanding Common Stock of the Company, the Company shall use its commercially reasonable efforts to have its Board of Directors include a designee of the Purchaser, such designee to be either a senior management officer or a member of the board of trustees of RioCan and who initially shall be Rags Davloor. Without limiting the generality of the foregoing, the Company shall nominate a designee of the Purchaser and solicit proxies for the election or re-election, as the case may be, of such designee to the Board of Directors of the Company at each subsequent annual meeting of the Company or other election of directors. In the event of the death or resignation of the Purchaser's designee (or the rejection of such designee by the Company's Board of Directors), the Purchaser shall be entitled to designate a substitute senior management officer or member of the board of trustees of RioCan to serve as the Purchaser's designee on the Company's Board of Directors. Any such designee of the Purchaser must be approved by the Board of Directors of the Company, acting reasonably and in good faith; provided that in the event the Board of Directors rejects a designee put forward by the Purchaser, it shall provide a written summary of its good faith reasons for just such rejection to the Purchaser. Such designee shall not be deemed independent under the rules of the NYSE. Such designee shall be entitled to receive all written materials furnished to each committee of the Board of Directors of the Company and to attend as an observer all such committee meetings.

9.2 Reservation of Common Stock.

The Company shall reserve and keep available out of its authorized but unissued Common Stock the number of shares of Common Stock required for issuance upon exercise of the Warrant, including any additional shares of Common Stock that may become so issuable by reason of the operation of the anti-dilution provisions of the Warrant.

9.3 Preemptive Rights.

- The Company agrees that it will not offer or sell any New Securities (as defined herein) unless it first offers to the Purchaser the right to purchase a portion of such New Securities in accordance with, and subject to, the provisions of this Section 9.3. If the Company proposes to sell any New Securities, other than pursuant to a Non-Eligible Public Offering (as defined herein), then the Company shall provide the Purchaser with written notice of the terms or proposed terms for the sale of the New Securities (the "Notice"). If the Purchaser wishes to purchase New Securities pursuant to the Notice, it shall notify the Company by written notice within five business days after the Notice is delivered how many of such New Securities it desires to purchase. The Purchaser shall be entitled to purchase up to a number of New Securities equal to its pro rata percentage ownership of the Company on the Closing Date, calculated on a fully diluted basis (the "Percentage Interest"). If the Purchaser does not elect to purchase all of its Percentage Interest, the Company shall have the right to complete the sale of any New Securities offered to but not purchased by the Purchaser upon terms no less favorable to the Company than those specified in the Notice. The Purchaser shall not have any preemptive rights with respect to any New Securities sold by the Company in a Public Offering (as defined herein) unless in such Public Offering (i) the New Securities are anticipated to be priced at more than a 10% discount to the volume weighted average closing market price of the Common Stock on the NYSE during the three-trading day period immediately prior to the date of the Notice, based on the indicative range advised by the underwriters of such Public Offering or (ii) the Company sells more than 20% of its outstanding Common Stock in a Public Offering (any Public Offering other than a Public Offering described in (i) or (ii) above, a "Non-Eligible Public Offering"). In either of such events, the Purchaser shall have the right to subscribe for up to its Percentage Interest; provided, however, that it must exercise this right prior to the anticipated pricing of such Public Offering. The Company agrees to provide written notice to the Purchaser of a Public Offering in which the Purchaser has preemptive rights as promptly as practicable under the circumstances, and in any event prior to the pricing of such Public Offering.
- (b) The closing of the purchase of New Securities by the Purchaser pursuant to this Section 9.3 shall occur concurrently with the closing of the sale of the New Securities and at the same place as such sale or at such other time and place as the Company and the Purchaser may mutually determine. At such closing, the Company shall deliver to the Purchaser the certificates representing such New Securities, free and clear of all liens and encumbrances caused or created by the Company, and the Purchaser shall pay to the Company the purchase price for such New Securities by wire transfer of funds to the account and depository designated by the Company.

- (c) The preemptive rights set forth in this Section 9.3 shall terminate and be of no further force or effect if the Purchaser owns less than 9.9% of the outstanding Common Stock of the Company for more than 60 consecutive days following the earlier of (i) the Purchaser's receipt of written notice from the Company that the Shares held by the Purchaser and its affiliates at such time represent less than 9.9% of the outstanding Common Stock, and (ii) the public disclosure by the Company of its outstanding share data demonstrating that the Shares held by the Purchaser and its affiliates at such time represent less than 9.9% of the outstanding Common Stock; provided, however, that the 60-consecutive day period requirement shall only apply if the Purchaser owns less than 9.9% of the outstanding Common Stock as the result of (x) a Non-Eligible Public Offering or (y) any of the events described in clauses (i) through (iii) of the definition of New Securities and provided further that if, as a result of restrictions imposed under applicable United States securities laws (it being understood that a requirement to disgorge profits under Section 16(b) of the Exchange Act shall not be deemed or construed to prevent the acquisition by the Purchaser of additional shares of Common Stock for purposes of this proviso), the Company's Code of Business Conduct and Ethics or otherwise imposed by the Company, the Purchaser is prevented from acquiring additional shares of Common Stock during the 60-consecutive day period referred to above, such period shall be extended by that number of days equal to the number of days during such 60-day period the Purchaser was prevented from acquiring additional shares, with such extension to start on the trading day following the later of (i) the last day of such 60-day period and (ii) the last day that such trading restrictions are applicable to the Purchaser.
- (d) The Purchaser shall be entitled to assign its rights under this Section 9.3 to an affiliate of the Purchaser.
- (e) "New Securities" means shares of Common Stock of the Company, or any securities that are convertible into or exchangeable for Common Stock, or any right, option or warrant to acquire any Common Stock of the Company, except for the following: (i) the issuance, grant or sale of Common Stock, options to purchase Common Stock or Common Stock issuable upon the exercise of options or other equity awards pursuant to any stock option, stock bonus or other stock plan or arrangement adopted by the Company, (ii) the issuance of securities by the Company in connection with an acquisition, merger, joint venture, or sale or purchase of assets, (iii) any Common Stock issuable upon the redemption of outstanding Units in the Operating Partnership, or (iv) Warrant Shares.
- (f) "Public Offering" means a distribution of New Securities in an underwritten public offering pursuant to a registration statement filed with and declared effective by the SEC pursuant to the Securities Act.

9.4 Hold Period.

The Purchaser agrees that without the prior written consent of the Company, which consent shall not be unreasonably withheld, it will not be permitted to sell, assign, transfer, encumber, pledge or otherwise dispose of (collectively, a "Transfer") any Shares, the Warrant or the Warrant Shares for a period of one year after the Closing Date other than (i) to one or more affiliates of the Purchaser, or (ii) pursuant to a tender or exchange offer for the Common Stock. Thereafter, subject to compliance with applicable federal and state securities laws no other restrictions on Transfer shall apply except that the Purchaser shall provide to the Company at least 30 days' notice of its intent to Transfer any Shares, Warrant or Warrant Shares (other than any Transfer through the facilities of the NYSE (for which no prior notice shall be required). The obligation not to Transfer for a period of one year after the Closing Date shall not apply to any New Securities acquired by the Purchaser pursuant to Section 9.3 or Section 9.6(c).

9.5 Tax Representations.

For so long as the Purchaser and its affiliates own Shares representing at least 9.9% of the outstanding Common Stock, the Purchaser will use its reasonable commercial efforts to perform a periodic review to ascertain that the tax representations contained in Section 3.9 of this Agreement remain accurate and immediately will notify the Company upon determining that any of such tax representations is no longer true and correct or if it becomes aware of facts that it reasonably expects will cause any of the tax representations contained in Section 3.9 to no longer be true and correct.

9.6 Restrictions on Certain Actions.

- (a) For a period ending on the third anniversary of the Closing Date, except as provided in Section 9.3, Section 9.6(b) or Section (c), the Purchaser, without the prior consent of the Board of Directors of the Company (excluding the director designated by the Purchaser), will not, nor will it permit any affiliate of the Purchaser to:
 - acquire (other than through a stock split, stock dividend, reorganization, recapitalization, merger, consolidation or other distribution made to holders of Common Stock of the Company), directly or indirectly or in conjunction with or through any other person, by purchase or otherwise, beneficial ownership of any additional Shares or any other securities of the Company (the "Securities");

- (ii) directly or indirectly or through any other person, solicit proxies with respect to Securities under any circumstances or become a "participant" in any "election contest" relating to the election of directors of the Company (as such terms are used in Rule 14a-11 of Regulation 14A under the Exchange Act); provided that the foregoing shall not be deemed to prohibit (a) Purchaser from voting (or casting a written consent solicited by the Company) regarding its shares of Common Stock in the manner it deems appropriate or (b) Purchaser's designee on the Board of Directors of the Company from participating in board deliberations, subject to compliance with the Company's governing documents;
- deposit any Securities in a voting trust, or subject any Securities to a voting or similar agreement;
- (iv) directly or indirectly or through or in conjunction with any other person, engage in a tender or exchange offer for Securities made by any other person or entity without the prior approval of the Company, or engage in any proxy solicitation or any other activity with any other person or entity relating to the Company without the prior approval of the Company; or
- (v) take any action alone or in concert with any other person to acquire or change the control of the Company or, directly or indirectly, participate in, or encourage the formation of, any group seeking to obtain or take control of the Company.
- (b) Notwithstanding the provisions of Section 9.6(a), the Purchaser shall be permitted to acquire additional shares of Common Stock in the open market in an amount sufficient so as to maintain its Percentage Interest if the Shares owned by it and its affiliates come to represent less than its Percentage Interest as a result of (i) the issuance, grant or sale of Common Stock, options to purchase Common Stock or Common Stock issuable upon the exercise of options or other equity awards pursuant to any stock option, stock bonus or other stock plan or arrangement adopted by the Company, (ii) the issuance of securities by the Company in connection with an acquisition, merger, joint venture, or sale or purchase of assets, (iii) any Common Stock issuable upon the redemption of outstanding Units in the Operating Partnership, or (iv) a Non Eligible Public Offering; provided, however, that notwithstanding anything to the contrary contained in this Agreement, if at any time or from time to time the Purchaser does not elect to purchase its Percentage Interest of New Securities as provided in Section 9.3, then the Percentage Interest shall automatically be reduced each such time to be calculated on a fully diluted basis at the time of each closing of the sale of New Securities.
- (c) Notwithstanding the provisions of subsection (a) of this Section 9.6 in the event that a public tender offer is made for the Company and not withdrawn or in the event that the Board of Directors of the Company announces (i) a recommended sale or merger transaction or (ii) a process to solicit proposals to acquire or merge with the Company, then the restrictions contained in Section 9.6(a) shall no longer apply and the Purchaser will not be contractually restricted in any way from acquiring additional securities of the Company or making an offer to acquire the Company or taking any other actions that would otherwise be prohibited under Section 9.6(a). Notwithstanding anything to the contrary in this Section 9.6, nothing shall prevent the Purchaser from making a confidential proposal to the Board of Directors of the Company relating to an offer for the Company's Securities or assets or another business combination or strategic transaction involving the Purchaser (or an affiliate of the Purchaser) and the Company.

9.7 NYSE Matters.

The Company will use its commercially reasonable efforts to (i) cause the Common Stock, including the Shares and the Warrant Shares, to continue to be listed on the NYSE, subject to, in the case of Shares and Warrant Shares, official notice of issuance and (ii) cause the Common Stock (other than the Shares and the Warrant Shares) to be quoted on the NYSE.

9.8 Filings.

The Company shall file with the SEC on a timely basis a report on Form 8-K or such other appropriate form as determined by counsel to the Company, relating to the transactions contemplated by this Agreement. The Company shall timely file with the NYSE all other documents and notices required by the NYSE of companies that have or will have issued securities that are traded on the NYSE. The Purchaser shall file with the SEC on a timely basis a Schedule 13D and Form 3 and such amendments or other appropriate forms as determined by counsel to the Purchaser, relating to the transactions contemplated by this Agreement.

9.9 Investment Company Act.

The Company shall use its commercially reasonable efforts to conduct its affairs, and shall use its commercially reasonable efforts to cause the affairs of its subsidiaries to be conducted, in such a manner so as to reasonably ensure that neither the Company nor any of its subsidiaries will be or become an "investment company," as such term is defined in the Investment Company Act.

9.10 Securities Act and Exchange Act.

Each of the Company and the Purchaser shall use its commercially reasonable efforts to comply with all requirements imposed upon it by the Securities Act and the Exchange Act as from time to time in force.

9.11 Sarbanes-Oxley Act.

The Company shall use its commercially reasonable efforts to comply with all effective applicable

provisions of the Sarbanes-Oxley Act of 2002.

9.12 Credit Facility.

The Company shall use its commercially reasonable efforts to obtain commitments to renew its Loan Agreement dated January 30, 2004, as amended, in an amount that would not be less than \$196,000,000.

9.13 Contribution of Net Proceeds to Operating Partnership

Immediately following the Closing, the Company shall contribute to the Operating Partnership the net proceeds received by the Company from the issuance of the Shares and the Warrant in consideration for the issuance of the OP Units and the OP Warrant to the Company pursuant to Section 4.5(g) of the OP Partnership Agreement.

ARTICLE X MISCELLANEOUS

10.1 Entire Understanding.

This Agreement, the exhibits attached hereto and the confidentiality agreement dated September 25, 2009 between the Company and RioCan (the "Confidentiality Agreement") state the entire understanding between the parties with respect to the subject matter hereof, and supersede all prior oral and written communications and agreements, and all contemporaneous oral communications and agreements, with respect to the subject matter hereof. The Confidentiality Agreement shall remain in full force and effect in accordance with its terms notwithstanding the execution of this Agreement; provided that the provisions of Section 9.6 of this Agreement shall govern the parties from and after the Closing Date and the 24--month standstill described in the Confidentiality Agreement shall terminate and be of no further force or effect from and after such time. This Agreement may not be amended, modified or waived except by an instrument in writing signed by each of the parties hereto.

10.2 Parties in Interest.

This Agreement shall bind, benefit, and be enforceable by and against each party hereto and its successors and permitted assigns; provided, however, that neither party may assign this Agreement without the prior consent of the other party. Nothing in this Agreement shall confer any rights or liabilities upon any person or entity that is not a party to this Agreement, except as expressly provided herein.

10.3 Severability.

If any provision of this Agreement is construed to be invalid, illegal or unenforceable, then the remaining provisions hereof shall not be affected thereby and shall be enforceable without regard thereto.

10.4 Section Headings.

Article and Section headings in this Agreement are for convenience of reference only, do not constitute a part of this Agreement, and shall not affect its interpretation.

10.5 Inclusion.

Where the words "including" or "includes" is used in this Agreement, it means "including (or includes) without limitation".

10.6 Brokers.

The Company represents and warrants to the Purchaser that no brokerage commission or finder's fees have been incurred by the Company in connection with the sale of the Shares and Warrant to the Purchaser, except for the engagement by the Company of Goldman, Sachs & Co.

10.7 Counterparts.

This Agreement may be executed simultaneously in several counterparts and by facsimile, each of which will be deemed to be an original, but all of which together will constitute one and the same instrument.

10.8 Notices.

All notices hereunder shall be in writing and shall be deemed to have been given at the time when hand delivered, when received if sent by telecopier or by same day or overnight recognized commercial courier service, or three days after mailed by registered or certified mail, addressed to the address below stated of the party to which notice is given, or to such changed address as such party may have fixed by notice:

To the Company or the Operating Partnership: Cedar Shopping Centers, Inc. 44 South Bayles Avenue Port Washington, New York 11050

Attention: Leo S. Ullman Fax No. 516-767-6497

-with a copy to-

Stroock & Stroock & Lavan LLP 180 Maiden Lane New York, NY 10038

Attention: Martin H. Neidell Fax No. 212-806-7836

To Purchaser or RioCan:

RioCan Real Estate Investment Trust RioCan Yonge Eglinton Centre 2300 Yonge Street, Suite 500, PO Box 2386 Toronto, Ontario M4P 1E4

Attention: Rags Davloor Fax No. 416 866-3020

-with a copy to-

(before December 19, 2009) Goodmans LLP 250 Yonge Street, Suite 2400 Toronto, Ontario M5B 2M6

(on and after December 19, 2009) Goodmans LLP Bay Adelaide Centre 333 Bay Street, Suite 3400 Toronto, Ontario M5H 2S7

Attention: David Matlow Fax No. 416 979-1234

provided, however, that any notice of change of address shall be effective only upon receipt.

10.9 Survival of Representations.

All representations, warranties and covenants contained in this Agreement shall survive the execution and delivery of this Agreement and the Closing Date, but, in the case of such representations and warranties, shall terminate and be of no further force or effect 18 months after the Closing Date.

10.10 Officers and Trustees of RioCan.

The Company and the Operating Partnership each acknowledge that this Agreement shall be conclusively taken to have been executed by, or by officers of RioCan on behalf of, the trustees of RioCan only in their capacity as trustees of RioCan. The Company and the Operating Partnership each hereby disavow any liability upon and waive any claim against holders of units of RioCan and annuitants under plans of which holders of units of RioCan act as trustee or carrier and the obligations created hereunder are not personally binding upon, nor shall resort be had to, nor shall recourse or satisfaction be sought from, the private property of any trustee or officer of RioCan or any holder of units of RioCan or annuitant, but the property of RioCan from time to time or a specific portion thereof only shall be bound. It is agreed that the benefit of this provision is restricted to the trustees and officers of RioCan, each holder of units issued by RioCan and annuitants and, solely for that purpose, the undersigned signing officers of RioCan have entered into this provision as agent and trustee for and on behalf of the trustees of RioCan, each holder of units of RioCan and each annuitant.

10.11 Specific Performance.

Without limiting or waiving any rights or remedies of either of the parties hereto, the parties hereto agree that irreparable damage would occur in the event any provisions of this Agreement were not performed by the parties in accordance with the terms hereof and that each of the parties hereto shall be entitled to an injunction or injunctions to prevent breaches of this Agreement by the other party and to seek specific performance of the obligations of the other party under this Agreement in any court of the United States or the State of New York having jurisdiction as agreed to in Section 10.15, this being in addition to any other remedy to which they are entitled at law or in equity.

10.12 Waiver of Jury Trial.

Each of the parties hereto hereby waives to the fullest extent permitted by applicable law any right it may have to a trial by jury with respect to any litigation directly or indirectly arising out of, under or in connection with this Agreement or the transactions contemplated hereby.

10.13 Mutual Drafting; Expenses; Press Release.

Each party hereto has participated in the drafting of this Agreement, which each party acknowledges is the result of negotiations among the parties. Each party to this Agreement shall pay its own expenses in connection with the preparation of this Agreement and the consummation of the transactions contemplated hereunder, including the fees of any attorneys, accountants or others engaged by such party. Each of the Company and the Purchaser shall consult with each other before issuing, and shall provide each other a reasonable opportunity to review and comment on, any press release with respect to this Agreement and the transactions contemplated hereby, and shall not issue any press release without the prior approval of the other party, except as may be required by applicable law, in which case the party required to make the release shall use its reasonable best efforts to allow the other party a reasonable time to approve such release in advance of issuing such release.

10.14 Dollar Amounts.

All references to \$ or dollars in this Agreement shall be to U.S. dollars.

10.15 Controlling Law.

THIS AGREEMENT IS MADE UNDER, AND SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK APPLICABLE TO AGREEMENTS MADE AND TO BE PERFORMED SOLELY THEREIN, WITHOUT GIVING EFFECT TO PRINCIPLES OF CONFLICTS OF LAW. ANY DISPUTES OR CLAIMS ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT AND THE TRANSACTIONS CONTEMPLATED OR DOCUMENTS REQUIRED HEREBY SHALL BE SUBMITTED TO THE EXCLUSIVE JURISDICTION OF THE COURTS OF THE STATE AND CITY OF NEW YORK SITTING IN THE CITY OF NEW YORK, BOROUGH OF MANHATTAN, AND THE FEDERAL COURTS OF THE UNITED STATES OF AMERICA LOCATED IN THE SOUTHERN DISTRICT OF NEW YORK, AND APPROPRIATE APPELLATE COURTS THEREFROM. THE PARTIES HEREBY IRREVOCABLY WAIVE, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY OBJECTION WHICH THEY MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY DISPUTE ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OF THE TRANSACTIONS CONTEMPLATED HEREBY BROUGHT IN SUCH COURT OR ANY DEFENSE OF INCONVENIENT FORUM FOR THE MAINTENANCE OF SUCH DISPUTE. EACH OF THE PARTIES HERETO AGREES THAT A JUDGMENT IN ANY SUCH DISPUTE MAY BE ENFORCED IN OTHER JURISDICTIONS BY SUIT ON THE JUDGMENT OR IN ANY OTHER MANNER PROVIDED BY LAW. THIS CONSENT TO JURISDICTION IS BEING GIVEN SOLELY FOR PURPOSES OF THIS AGREEMENT AND IS NOT INTENDED TO, AND SHALL NOT, CONFER CONSENT TO JURISDICTION WITH RESPECT TO ANY OTHER DISPUTE IN WHICH A PARTY TO THIS AGREEMENT MAY BECOME INVOLVED. EACH OF THE PARTIES HERETO HEREBY CONSENTS TO PROCESS BEING SERVED BY ANY PARTY TO THIS AGREEMENT IN ANY SUIT, ACTION, OR PROCEEDING OF THE NATURE SPECIFIED IN THIS SECTION 10.15 BY PROVIDING A COPY THEREOF IN THE MANNER SPECIFIED BY THE PROVISIONS OF SECTION 10.8.

IN WITNESS WHEREOF, the Purchaser, RioCan, the Company and the Operating Partnership have executed this Agreement as of the date first written above.

KIO	an Holdings USA Inc.
By:	lal
Dy.	/s/ Name:
	Title:
RioC	Can Real Estate Investment Trust
By:	/s/
	Name:
	Title:
Ced	ar Shopping Centers, Inc.
_	
By:	/s/ Name:
	Name:
	Title:
Ceda	ar Shopping Centers Partnership L.P.
By:	Cedar Shopping Centers Inc., its general partner
2,,	count shopping conters they no general parener
By:	<u>/s/</u>
	Name:
	Title:

Exhibit A

GUARANTY

This Guaranty is given this _____ day of October, 2009, by RioCan Real Estate Investment Trust, an unincorporated closed end trust constituted in accordance with the laws of the Province of Ontario (the "Guarantor"), in favor of Cedar Shopping Centers, Inc., a Maryland corporation, and Cedar Shopping Centers Partnership, L.P., a Delaware limited partnership (collectively "Cedar").

WHEREAS, pursuant to a securities purchase agreement dated the date hereof (the "Agreement"), a direct or indirect wholly-owned subsidiary of the Guarantor (the "Purchaser") will be acquiring shares of

common stock and warrants to purchase common stock of Cedar Shopping Centers, Inc.;

WHEREAS, the Guarantor is the direct or indirect sole stockholder of the Purchaser and expects to derive a financial advantage from this Guaranty; and

WHEREAS, in order to induce Cedar to enter into the Agreement with Purchaser, the Guarantor has agreed to give this Guaranty in favor of Cedar.

NOW, THEREFORE, FOR VALUE RECEIVED, THIS GUARANTY WITNESSETH as follows:

- 1. The Guarantor hereby absolutely, unconditionally and irrevocably guarantees the due performance and prompt payment in full of any and all existing and future obligations of any and all kinds of the Purchaser to Cedar arising out of or in connection with the Agreement. If any term, condition or provision of the Agreement or the application thereof to any party or circumstance shall be ruled invalid, illegal or unenforceable in any respect, the validity, legality or enforceability of any term, condition or provision of this Guaranty shall not in any manner be affected or impaired.
- 2. The liability of the Guarantor under this Guaranty is an absolute, primary, direct, present and unconditional guaranty of payment and performance and may be enforced without Cedar first attempting to collect from or without Cedar first resorting to any rights or remedies Cedar may have against the Purchaser or any other person or entity.
- 3. The Guarantor hereby acknowledges that no release or extinguishment of the Purchaser's obligations or liabilities (other than in accordance with the terms of the Agreement or otherwise with the written agreement of Cedar), whether by decree in any bankruptcy proceeding or otherwise, shall affect the continuing validity and enforceability of this Guaranty.
- 4. The Guarantor hereby consents to all of the terms and provisions of the Agreement as the same may be from time to time hereafter amended or changed, and waives and agrees not to assert as a defense in any action upon this Guaranty (a) notice of any amendment, modification or change in any term, condition or provision of the Agreement; (b) notice of any default under the Agreement; (c) demand for performance or observance of, and any enforcement of any term, condition or provision of, or any pursuit or exhaustion of any rights or remedies against Purchaser; and (d) to the extent it may lawfully do so, any and all demands and notices of every kind and description with respect to the foregoing or which may be required to be given by any statute or rule of law and any defense of any kind which it may now or hereafter have with respect to this Guaranty or the Agreement, except to the extent any such defense is available to Purchaser.
- 5. No failure on the part of Cedar to exercise, and no delay in exercising, any right, remedy or power hereunder shall operate as a waiver under this Guaranty, nor shall any single or partial exercise by Cedar of any right, remedy or power hereunder preclude any other or future exercise of any other right, remedy or power.
- 6. The Guaranty herein contained shall be a continuing guaranty and shall remain in full force until all of the obligations of the Purchaser under the Agreement shall have been satisfied in full.
- 7. This Guaranty shall be binding upon the Guarantor, its successors and assigns, and shall inure to the benefit of Cedar, its successors and assigns.
- 8. Cedar acknowledges that this letter agreement shall be conclusively taken to have been executed by, or by officers of the Guarantor on behalf of, the trustees of the Guarantor only in their capacity as trustees of the Guarantor. Cedar hereby disavows any liability upon and waive any claim against holders of units of the Guarantor and annuitants under plans of which holders of units of the Guarantor act as trustee or carrier and the obligations created hereunder are not personally binding upon, nor shall resort be had to, nor shall recourse or satisfaction be sought from, the private property of any trustee or officer of the Guarantor or any holder of units of the Guarantor or annuitant, but the property of the Guarantor from time to time or a specific portion thereof only shall be bound. It is agreed that the benefit of this provision is restricted to the trustees and officers of the Guarantor, each holder of units issued by the Guarantor and annuitants and, solely for that purpose, the undersigned signing officers of the Guarantor have entered into this provision as agent and trustee for and on behalf of the trustees of the Guarantor, each holder of units of the Guarantor and each annuitant.
- 9. Any demand or notice to be made or given by Cedar to the Guarantor under this Guaranty shall be given by notice addressed to the Guarantor at the address provided in Section 10.8 of the Agreement.
- 10. This Guaranty cannot be changed or terminated orally and shall be governed by and construed in accordance with the laws of the State of New York without giving effect to the principles of conflict of laws.

IN WITNESS WHEREOF, the Guarantor has executed this Guaranty on the day and year first abovementioned.

RioCan Real Estate Investment Trust

Ву:	
-----	--

THE WARRANT AND ANY SHARES OF COMMON STOCK ISSUED UPON EXERCISE HEREOF HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR THE SECURITIES LAWS OF ANY STATE AND MAY NOT BE OFFERED, SOLD, TRANSFERRED, PLEDGED OR OTHERWISE DISPOSED OF WITHOUT AN EFFECTIVE REGISTRATION STATEMENT UNDER THE ACT AND UNDER ANY APPLICABLE STATE SECURITIES LAWS, RECEIPT OF A NO-ACTION LETTER ISSUED BY THE SECURITIES AND EXCHANGE COMMISSION (TOGETHER WITH EITHER REGISTRATION OR AN EXEMPTION UNDER APPLICABLE STATE SECURITIES LAWS) OR AN OPINION OF COUNSEL ACCEPTABLE TO THE COMPANY THAT THE PROPOSED TRANSACTION WILL BE EXEMPT FROM REGISTRATION UNDER THE ACT AND APPLICABLE STATE SECURITIES LAWS.

WARRANT TO PURCHASE

SHARES OF COMMON STOCK

OF

CEDAR SHOPPING CENTERS, INC.

NO. W 1

1,428,570 Shares November __, 2009

FOR VALUE RECEIVED, subject to the provisions hereinafter set forth, the undersigned, CEDAR SHOPPING CENTERS, INC., a Maryland corporation (together with its successors and permitted assigns, the "Company"), hereby certifies that

RIOCAN HOLDINGS USA INC.

or its registered assigns is entitled to subscribe for and purchase, during the period specified in this Warrant (this "Warrant"), 1,428,570 shares (subject to adjustment as hereinafter provided) of duly authorized, validly issued, fully paid and non-assessable shares of common stock of the Company, \$0.06 par value ("Common Stock"), at an initial exercise price of \$7.00 per share ("Exercise Price"), subject, however, to the provisions and upon the terms and conditions hereinafter set forth. This Warrant is issued pursuant to the Securities Purchase Agreement dated October 26, 2009 among the Company, Cedar Shopping Centers Partnership, L.P., RioCan Real Estate Investment Trust and RioCan Holdings USA Inc. (the "Purchase Agreement"), and, by execution of the Purchase Agreement, the holder hereof has agreed (and any transferee of this Warrant will agree) to comply with the obligations set forth herein and therein.

Exercise of This Warrant. The rights represented by this Warrant may be exercised by the holder hereof, in whole or in part (but not as to a fractional share of Common Stock), by the surrender of this Warrant (properly endorsed if required) at the office of the Company, 44 South Bayles Avenue, Port Washington, New York 11050, or such other office or agency of the Company as the Company may designate by notice in writing to the holder hereof at the address of such holder appearing on the books of the Company, at any time and from time to time after the date hereof and before 5:00 P.M., New York City time, two years from the date hereof (the "Exercise Period") upon payment to the Company of the Exercise Price for such shares.

The exercise price for shares of common stock issuable upon exercise of this warrant shall be payable as follows: (a) by payment to the company of the exercise price in cash, by check or by wire transfer of funds or (b) by surrender to the company for cancellation of notes or debt securities of the company having a principal balance plus accrued interest on the date of exercise equal to the exercise price for such shares or (c) by a combination of the methods described in clauses (a) and (b) above. The company agrees that the shares so purchased shall be deemed to be issued to the holder hereof as the record owner of such shares as of the close of business on the date on which this warrant shall have been surrendered and payment made for such shares as aforesaid. Certificates for the shares of common stock so purchased shall be delivered to the holder hereof as promptly as practicable after the rights represented by this warrant shall have been so exercised, and, unless this warrant has expired, a new warrant representing the number of shares, if any, with respect to which this warrant shall not then have been exercised shall also be delivered to the holder hereof within such time.

2. Reservation, Issuance and Listing of Stock.

- A. The Company will at all times reserve and keep available, free from preemptive rights, out of its authorized but unissued Common Stock, solely for the purpose of issue upon exercise of the Warrant, such number of shares of Common Stock as shall then be issuable upon exercise of the Warrant. The Company will, at its expense, use its commercially reasonable efforts to cause such shares to be listed (subject to issuance or notice of issuance) on all stock exchanges, if any, on which the Company's Common Stock may become listed during the Exercise Period.
- B. The Company covenants that all shares which may be issued upon the exercise of the rights represented by this Warrant will, upon issuance, be duly authorized and validly issued, fully paid and nonassessable, not subject to preemptive rights and free from all taxes, liens, charges and security interests with respect to the issue thereof.
- 3. Exercise Price. The above provisions are, however, subject to the following:

- A. Initial Exercise Price. The initial Exercise Price of \$7.00 per share shall be subject to adjustment from time to time as hereinafter provided.
- B. Adjustment of Exercise Price and Number of Shares of Common Stock Issuable. The Exercise Price and number of shares of Common Stock issuable upon exercise hereof shall be subject to adjustment from time to time in accordance with the following provisions, such adjustments shall take effect at the close of business on the date of any such split, subdivision, dividend or combination, as the case may be:
- (1) In case the Company shall at any time split or subdivide the outstanding shares of Common Stock or shall issue a stock dividend with respect to the Common Stock resulting in the issuance of additional shares of Common Stock, the number of shares of Common Stock for which this Warrant is exercisable immediately prior to such event shall be proportionally increased and the Exercise Price in effect immediately prior to such event shall be proportionately decreased to reflect such split, subdivision or stock dividend, as the case may be.
- (2) In case the Company shall at any time combine the outstanding shares of Common Stock, the number of shares of Common Stock for which this Warrant is exercisable immediately prior to such event shall be proportionally decreased and the Exercise Price in effect immediately prior to such event shall be proportionately increased to reflect such combination.
- (3) In case the Company shall declare a cash dividend on its Common Stock (excluding Dividends Paid in the Ordinary Course), the Exercise Price in effect immediately prior to the record date for such dividend shall be proportionately reduced immediately thereafter to reflect such dividend; such adjustment shall be made successively whenever such a record date is fixed. In the event that such dividend is not so paid, the Exercise Price then in effect shall be readjusted, effective as of the date when the Board of Directors of the Company determines not to pay such dividend, to the Exercise Price that would then be in effect if such record date had not been fixed. For purposes of this Warrant, "Dividends Paid in the Ordinary Course" shall mean dividends on Common Stock not to exceed \$.225 per share per quarter.
- C. Reorganization, Consolidation, Merger or Sale. If the Company shall effect a reorganization, shall merge with or consolidate into another corporation, or shall sell, transfer or otherwise dispose of all or substantially all of its assets and pursuant to the terms of such reorganization, merger, consolidation or disposition of assets, securities, property or assets of the Company, successor or transferee or an affiliate thereof or cash are to be received by or distributed to the holders of Common Stock, then the holder of this Warrant shall have the right thereafter to receive, upon the exercise of this Warrant, the number of shares of stock or other securities, property or assets of the Company, successor, transferee or affiliate thereof or cash receivable upon or as a result of such reorganization, merger, consolidation or disposition of assets by a holder of the number of shares of Common Stock equal to that to which the holder of this Warrant upon the exercise thereof immediately prior to such event would have been entitled. The provisions of this subsection 3.C shall similarly apply to successive reorganizations, mergers, consolidations or dispositions of assets. Upon any reorganization, consolidation, merger or transfer hereinabove referred to, this Warrant shall continue in full force and effect and the terms hereof shall be applicable to the shares of stock and other securities, property, assets and cash receivable upon the exercise of this Warrant after the consummation of such reorganization, consolidation, merger or transfer, as the case may be. The Company shall not effect any such reorganization, consolidation, merger or transfer unless prior to the consummation thereof the successor corporation (if other than the Company) resulting therefrom or the corporation purchasing such assets shall, by written instrument executed and mailed to the registered holder hereof at the last address of such holder appearing on the books of the Company, (i) assume the obligation to deliver to such holder such shares of stock, securities, property or assets as, in accordance with the foregoing provisions, such holder may be entitled to purchase, and (ii) agree to be bound by all the terms of this Warrant.
- D. Notice of Adjustment. Upon any adjustment of the Exercise Price and any increase or decrease in the number of shares of Common Stock purchasable upon the exercise of this Warrant, then, and in each such case, the Company, as promptly as practicable thereafter, shall give written notice thereof to the holder hereof at the address of such holder as shown on the books of the Company which notice shall state the Exercise Price as adjusted and the increased or decreased number of shares purchasable upon the exercise of this Warrant, setting forth in reasonable detail the method of calculation of each.
- E. Other Notices. In case at any time the Company shall propose to (a) reorganize, or reclassify the capital stock of the Company, or consolidate, merge or otherwise combine with, or sell all or substantially all of its assets to, another person or (b) voluntarily or involuntarily dissolve, liquidate or wind up of the affairs of the Company; then, in any one or more of said cases, the Company shall give to the holder hereof at least ten (10) days' prior written notice of the date on which the books of the Company shall close or a record shall be taken for determining rights to vote in respect of any such reorganization, reclassification, consolidation, merger, sale, dissolution, liquidation or winding up.
- No Voting Rights. This Warrant shall not entitle the holder hereof to any voting rights or other rights as a stockholder of the Company.

- 5. Transfer and Registration of Warrants. This Warrant and all rights hereunder are transferable, in whole or in part, at the office or agency of the Company referred to in Section 1 hereof by the holder hereof in person or his duly authorized attorney, upon surrender of this Warrant properly endorsed; provided, that this Warrant may not be transferred or assigned in whole or in part without compliance with (i) the provisions of the Purchase Agreement, and (ii) all applicable federal and state securities laws by the transferor and the transferee (including, to the extent required by law or reasonably requested by the Company, the delivery of appropriate investment representations in customary form). Each taker and holder of this Warrant, by taking or holding the same, consents and agrees that this Warrant, when endorsed in blank, shall be deemed negotiable, and that the holder hereof, when this Warrant shall have been so endorsed may be treated by the Company and all other persons dealing with this Warrant as the absolute owner thereof for any purpose and as the person entitled to exercise the rights represented by this Warrant, or to the transfer hereof on the books of the Company, any notice to the contrary notwithstanding; but until such transfer on such books, the Company may treat the registered holder hereof as the owner for all purposes.
- 6. Exchanges of Warrants. This Warrant is exchangeable, upon the surrender hereof by the holder hereof at such office or agency of the Company, for new Warrants of like tenor representing in the aggregate the right to subscribe for and purchase the number of shares that may be subscribed for and purchased hereunder, each of such new Warrants to represent the right to subscribe for and purchase such number of shares as shall be designated by said holder hereof at the time of such surrender.
- 7. No Fractional Shares. No fractional shares of Common Stock or scrip representing fractional shares of Common Stock shall be issued upon the exercise of the Warrant, but, in lieu thereof, there shall be paid an amount in cash equal to the same fraction of the closing price of a whole share of Common Stock on the business day preceding the day of exercise.
- 8. Replacement of Warrants. Upon receipt of evidence reasonably satisfactory to the Company of the loss, theft, destruction or mutilation of the Warrant and, in the case of any such loss, theft or destruction, upon delivery of an indemnity agreement reasonably satisfactory in form and amount to the Company or, in the case of any such mutilation, upon surrender and cancellation of such Warrant, the Company at its expense will execute and deliver, in lieu thereof, a new Warrant of like tenor.
- 9. Remedies. The Company stipulates that the remedies at law of the holder of this Warrant in the event of any default or threatened default by the Company in the performance of or compliance with any of the terms of this Warrant are not and will not be adequate, and that such terms may be specifically enforced by a decree for the specific performance of any agreement contained herein or by an injunction against a violation of any of the terms hereof or otherwise.
- Notices. Except as otherwise provided herein, any notices hereunder shall be deemed to have been given five (5) days after having been mailed in the United States by registered or certified mail, addressed if given to the Company to the principal office of the Company, Attention: President, or if given to a holder of this Warrant addressed to such holder at his address as the same shall appear on the books of the Company.
- 11. Miscellaneous. This Warrant and any term hereof may be changed, waived, discharged or terminated only by an instrument in writing signed by both the Company and the holder hereof. This Warrant is being delivered in the State of New York and shall be construed and enforced in accordance with and governed by the laws of such State. The headings in this Warrant are for purposes of reference only, and shall not limit or otherwise affect any of the terms hereof.
- 12. **Expiration of Warrants.** At 5:00 P.M. New York time on the last day of its Exercise Period, the Warrant, if not exercised prior thereto, shall be and become wholly void and of no value.
- 13. **No Impairment.** The Company shall not by any action, including, without limitation, amending its charter documents or through any reorganization, reclassification, transfer of assets, consolidation, merger, dissolution, issue or sale of securities or any other similar voluntary action, avoid or seek to avoid the observance or performance of any of the terms of this Warrant, but will at all times in good faith assist in the carrying out of all such terms and in the taking of all such actions as may be necessary or appropriate to protect the rights of the holder hereof against impairment. Without limiting the generality of the foregoing, the Company shall take all commercially reasonable action as may be necessary or appropriate in order that the Company may validly and legally issue fully paid and nonassessable shares of Common Stock upon the exercise of this Warrant, free and clear of all liens, and shall use its commercially reasonable efforts to obtain all such authorizations, exemptions or consents from any public regulatory body having jurisdiction over it as may be necessary to enable the Company to perform its obligations under this Warrant.
- 14. Supplying Information; Rule 144. The Company shall cooperate with the holder hereof in supplying such information as may be reasonably necessary for such holder to complete and file any information reporting forms presently or hereafter required by the Securities and Exchange Commission (the "Commission") as a condition to the availability of an exemption from the Securities Act of 1933, as amended (the "Act") for the transfer hereof. The Company shall use its commercially reasonable efforts to at all times make public information available so as to afford the holder hereof the benefits of Rule 144 of the Commission in connection with resales.

IN WITNESS WHEREOF, Ceda	r Shopping C	Centers,	Inc. has	caused this	Warrant to	be signed	by on	e
of its duly authorized officers.								

Dated: November _____, 2009

IN HUMBNESS WHIEDESE S 1 SI ' S

CEDAR SHOPPING CENTERS, INC.

	by:	
	Name:	
	Title:	
	ELECTION TO PURCHASE signed only upon exercise of War	rrant)
To: Cedar Shopping Centers, Inc 44 South Bayles Avenue Port Washington, New York		
Attn: Leo S. Ullman		
right represented by such Warrant for, a Stock of Cedar Shopping Centers, Inc. a surrenders for cancellation notes or debt requests that the certificates for	and to purchase thereunder, and herewith (a) makes payment securities of the Company equal	of \$ therefor or (b) to the Exercise Price. The undersigned the name of, and delivered to,
Dated: , 20		
		st conform in all respects to name of ified on the face of the Warrant)
	(Address)	
	(Address)	
	(Address)	
Insert here the number of shares on the portion thereof as to which the Warran additional Common Stock or any other adjustment provisions of the Warrant, many	nt is being exercised), in either carstock or other securities or pro-	
(To be	ASSIGNMENT signed only upon transfer of War	rant)
FOR VALUE RECEIVED, the undersigned under the within Warrant, Shopping Centers, Inc. covered thereby s	with respect to the number of	ns and transfers all of the rights of the shares of Common Stock of Cedar
Name of Assignee	Address	No. of Shares
Name of Assignee	Addicss	No. of Shares
Dated:,		
	(Sionature mu	st conform in all respects to name of
		ified on the face of the Warrant)

	(Address)	
	(Address)	
Signed in the presence of:		
Signature)		
Print)		

Exhibit C

REGISTRATION RIGHTS AGREEMENT

Registration Rights Agreement (this "Agreement") dated November __, 2009, by and between Cedar Shopping Centers, Inc., a Maryland corporation (the "Company"), and RioCan Holdings USA Inc., a Delaware corporation (the "Investor").

$\underline{W}\underline{I}\underline{T}\underline{N}\underline{E}\underline{S}\underline{S}\underline{E}\underline{T}\underline{H}$:

WHEREAS, the Company and the Investor entered into a Securities Purchase Agreement dated October 26, 2009 (the "**Purchase Agreement**") pursuant to which the Investor will acquire shares of Common Stock of the Company and a Warrant to acquire shares of Common Stock of the Company;

WHEREAS, the Company desires to grant to the Investor certain registration rights with respect to Registrable Securities (as hereinafter defined) acquired by the Investor pursuant to the Purchase Agreement;

NOW, THEREFORE, in consideration of the mutual covenants contained herein and for other good and valuable consideration set forth herein, the parties hereto agree as follows:

1 . <u>Certain Definitions.</u> For purposes of this Agreement, the following terms shall have the respective meanings set forth below:

"Commission" means the Securities and Exchange Commission.

"Common Stock" means shares of common stock, par value \$.06 per share, of the Company.

"Costs and Expenses" means all of the costs and expenses relating to the Registration Statement described in Section 2.1 including, but not limited to, blue-sky expenses, printing expenses, Commission filing fees, Financial Industry Regulatory Authority fees and fees and disbursements of counsel to the Company; provided, however, that Costs and Expenses shall not include (x) fees and disbursements for counsel for any Holder, which shall be borne by such Holder, or (y) underwriting discounts and commissions and reimbursable underwriters' expenses attributable to the Registrable Securities being sold by the Selling Holders thereof, which shall be borne by such Selling Holders pro rata on the basis of the relative number of Registrable Securities included in the applicable registration statement. With respect to Piggy Back Registrations pursuant to Section 2.2(b), the Holders shall bear all Costs and Expenses relating to inclusion of their Registrable Securities pro rata on the basis of the relative number of Registrable Securities included in the Piggy Back Registration, except for fees and disbursements of counsel to the Company.

"Holder" means (i) the Investor as a holder of Registrable Securities and (ii) any affiliate of the Investor that is a direct or indirect transferee of such Registrable Securities from the Investor that has agreed to be bound by the terms of this Agreement as a Holder.

"Registrable Securities" means (a) any shares of Common Stock issuable to the Investor pursuant to the provisions of the Purchase Agreement, (b) any shares of Common Stock issuable upon exercise of the Warrant granted to the Investor pursuant to the Purchase Agreement and (c) any securities issued as a dividend on or other distribution with respect to, or in exchange for or in replacement of, the shares of Common Stock referred to in subsections (a) and (b) above including, without limitation, stock splits, reclassifications, mergers, consolidations, recapitalizations or similar events; provided, however, that a security shall cease to be a Registrable Security only upon it having been effectively registered under the Securities Act and transferred pursuant to such registration or otherwise transferred to holders who may trade such security without restriction under the Securities Act, and the Company has delivered a new certificate or other evidence of ownership for such security not bearing a restrictive legend.

"Securities Act" means the Securities Act of 1933, as amended, or any similar Federal law then in force.

"Selling Holder" means any Holder who is selling Registrable Securities pursuant to a public offering registered hereunder.

- Registration.
 - Registration Procedures and Expenses.
 - (a) The Company shall prepare and file with the Commission a registration statement on

Form S-3 meeting the requirements of the Securities Act relating to the sale of the Registrable Securities by the Holders (the "Registration Statement") from time to time. Such Registration Statement (i) shall be a shelf registration statement providing for the registration and the sale of Registrable Securities by the Holders on a continuous or delayed basis pursuant to Rule 415 of the Securities Act, (ii) shall comply as to form in all material respects with the requirements of Form S-3 and include all financial statements required by the Commission to be filed therewith or be incorporated therein, (iii) shall be reasonably acceptable to the Investor's counsel and (iv) shall provide for the resale of the Registrable Securities from time to time pursuant to any method or combination of methods legally available by the Holders, and the Registration Statement and any form of prospectus included or incorporated by reference therein (or any prospectus supplement relating thereto) shall reflect such plan of distribution or method of sale.

- (b) The Company shall, subject to receipt of necessary information from the Holders, use its reasonable best efforts to cause the Commission to declare the Registration Statement effective on or prior to the date that is one year from the date hereof (such date, the "Lockup Termination Date").
- The Company shall (i) prepare and file with the Commission such amendments and supplements to the Registration Statement and the prospectus used in connection therewith as may be (A) reasonably necessary to keep the Registration Statement continuously effective until the earliest of (x) twenty four months after the later of the effective date of the Registration Statement and the Lockup Termination Date, (y) the date on which all Registrable Securities have been disposed of under the Registration Statement and (z) such time as all Registrable Securities have been otherwise transferred to holders who may trade such securities without restriction under the Securities Act, and the Company has delivered a new certificate or other evidence of ownership for such securities not bearing a restrictive legend (the earliest of such dates being referred to herein as the "Expiration Date"), or (B) reasonably requested by the Holders (whether or not required by the form on which the securities are being registered), and shall use its reasonable best efforts to cause each such amendment to be declared effective by the Commission, if required, as soon as practicable after the filing thereof, (ii) cause any related prospectus to be supplemented by any required supplement, and as so supplemented to be filed with the Commission pursuant to Rule 424 under the Securities Act (or any similar provisions then in force under the Securities Act), to the extent required, and (iii) comply in all material respects with the provisions of the Securities Act with respect to the disposition of all securities covered by the Registration Statement during the applicable period in accordance with the intended methods of disposition as may be reasonably requested from time to time by the Holders and set forth in such Registration Statement as so amended or such prospectus as so supplemented.
 - (d) The Company shall bear the Costs and Expenses of the Registration Statement.

2.2 Other Holders; Piggyback Rights.

- (a) The Company shall have the right to include in the Registration Statement the shares of Common Stock of any shareholder of the Company who has registration rights enabling it to include its shares in such Registration Statement, provided that such shares of Common Stock of any other shareholder of the Company may only be included in the Registration Statement to the extent such inclusion (i) will not reduce the amount of Registrable Securities of the Holders included in the Registration Statement and (ii) in the case of an underwritten offering, based on the written advice of the underwriter for the Registrable Securities, will not materially adversely affect the proposed offering price, the timing, the distribution method, or the probability of success of the offering.
- For a period commencing on the Expiration Date and ending three years after the Expiration Date, if the Company (i) proposes to file a registration statement under the Securities Act with respect to an offering of equity securities by the Company for its own account or for stockholders of the Company for their account other than a registration statement on Forms S-8 or S-4 or any equivalent form then in effect and the registration form to be used may be used for the registration of Registrable Securities, then the Company shall (A) give written notice of such proposed filing to the Holders as soon as practicable but in no event less than ten business days before the anticipated filing date, which notice shall describe the amount and type of securities to be included in such offering, the intended method(s) of distribution, and the name of the proposed underwriter(s), if any, of the offering, and (B) offer to the Holders in such notice the opportunity to register the sale of such number of Registrable Securities as such Holders may request in writing within five business days following receipt of such notice or (ii) has then in effect a registration statement under the Securities Act with respect to equity securities by the Company (other than a registration statement on Forms S-8 or S-4 or any equivalent form then in effect or the registration statement on Form S-3 filed by the Company with the Commission on November 17, 2008) and such registration statement may be used for the registration of Registrable Securities, then the Company shall register the sale of such number of Registrable Securities as any Holder may request in writing (any such registration described in clause (i)(B) or clause (ii) of this Section 2.2(b) is referred to as a "Piggy-Back Registration"); provided, however, that no request for a Piggy-Back Registration may be made prior to the Expiration Date. If at any time after giving written notice of its intention to register any securities and prior to the effective date of the registration statement filed in connection with such registration, the Company shall determine for any reason not to register or to delay registration of such securities, the Company may, at its election, give written notice of such determination to each Holder and, (x) in the case of a determination not to register, shall be relieved of its obligation to register any Registrable Securities in connection with such registration, and (y) in the case of a determination to delay registering, shall be permitted to delay registering any Registrable Securities for the same period as the delay in registering such other securities. The Company shall use its reasonable best efforts to cause such Registrable Securities to be included in such Piggy-Back Registration and shall use reasonable best efforts to cause the underwriter(s) of a proposed underwritten offering to permit the Registrable Securities requested to be included in a Piggy-Back Registration on the same terms and conditions as any similar securities of the Company and to permit the sale or other disposition of such Registrable Securities in accordance with the intended method(s) of distribution thereof. All Holders proposing to distribute their securities through a Piggy-Back Registration that involves an underwriter(s) shall enter into an underwriting agreement in customary form with the underwriter(s) selected for such Piggy-Back Registration and (ii) complete and execute all questionnaires, powers-of-attorney, indemnities, opinions and other documents required under the terms of such underwriting agreement. The Holder shall be responsible for all Costs and Expenses with respect to each Piggy Back Registration pro rata on the basis of the relative number of Registrable Securities included in the Piggy Back Registration.

- (c) If the underwriter for a Piggy-Back Registration advises the Company and the Holders in writing that in their opinion the inclusion of Registrable Securities by a Holder would materially adversely affect the proposed offering price, the timing, the distribution method, or the probability of success of the offering, then the Company shall include in such offering, as to each Holder exercising piggyback rights pursuant to Section 2.2(b) and any other person or persons having a written contractual right to request their shares of Common Stock or other securities of the Company be included in such offering, that number of shares of Common Stock or other securities of the Company that the Company is so advised can be sold in such offering without materially adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of the offering, determined as follows:
 - (i) if the registration is undertaken for the Company's account: (a) first, the shares of Common Stock or other securities of the Company that the Company desires to sell and (b) second, any shares of Common Stock or other securities of the Company held by holders of written contractual piggy-back registration rights (including the Holders) which can be included therein without, in the opinion of the underwriters, materially adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of the offering, pro rata among such holders on the basis of the relative number of shares of Common Stock or other securities of the Company requested to be included in such registration by such holders, and
 - (ii) if the registration is a "demand" registration undertaken at the demand of persons other than the Holders: (a) first, the shares of Common Stock or other securities of the Company to be sold for the account of such demanding persons that can be sold without, in the opinion of the underwriters, materially adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of the offering and (b) second, any shares of Common Stock or other securities of the Company held by other holders of written contractual piggy-back registration rights (including the Holders) which can be included therein without, in the opinion of the underwriters, materially adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of the offering, pro rata among such holders on the basis of the relative number of shares of Common Stock or other securities of the Company requested to be included in such registration by such holders.
- 2 . 3 Other Registration Procedures. With respect to any registration of Registrable Securities effected pursuant to Section 2.1 or Section 2.2(b), the Company shall:
- (a) furnish to the Selling Holders with respect to such Registrable Securities (and to each underwriter, if any, of such Registrable Securities), without charge, such reasonable number of copies of any registration statement in respect of such Registrable Securities or any amendment or supplement thereto (in each case including all exhibits thereto and documents incorporated by reference therein) and the prospectus included in such registration statement (including each preliminary prospectus) and such other documents as the Selling Holders or such underwriter may reasonably request, in order to facilitate the public sale or other disposition of all or any of the Registrable Securities by such Selling Holders; provided, however, that the obligation of the Company to deliver copies of prospectuses to the Selling Holders shall be subject to the receipt by the Company of reasonable assurances from such Selling Holders that such Selling Holders will comply with the applicable provisions of the Securities Act and of such other securities or blue sky laws as may be applicable in connection with any use of such prospectuses;
- (b) use its reasonable best efforts to (x) register and qualify the Registrable Securities under such other securities or blue sky laws of such states or jurisdictions specified in writing by the Selling Holders, keep each such registration or qualification (or exemption therefrom) effective during the period in which the registration statement for such Registrable Securities is required to be kept effective, and (y) do any and all other acts and things which may be necessary or advisable to enable each Selling Holder to consummate the disposition of the Registrable Securities in such jurisdictions; provided, however, that the Company shall not be required to qualify to do business or consent to service of process in any jurisdiction in which it is not now so qualified or has not so consented;
- (c) enter into and perform customary agreements (including an underwriting agreement in customary form) and take such other actions (including, without limitation, participation in road shows and investor conference calls) as are required in order to expedite or facilitate the sale of Registrable Securities;
- (d) after the filing of any registration statement in respect of such Registrable Securities, promptly notify such Selling Holders of any stop order issued or, to the Company's knowledge, threatened to be issued by the Commission and use its reasonable best efforts to prevent the entry of such stop order or to remove it if entered;
- (e) at the request of any underwriter of the Registrable Securities in connection with an underwritten offering, furnish (i) an opinion of counsel, addressed to such underwriters and such Selling Holders, covering such customary matters as the underwriter and the Selling Holders may reasonably request and (ii) a comfort letter or comfort letters from the Company's independent public accountants addressed to the underwriters and such Selling Holders covering such customary matters as the underwriter or the Selling Holders may reasonably request;
- (f) provide a CUSIP number for the Registrable Securities included in any registration statement in respect of such Registrable Securities not later than the effective date of such registration statement;
- (g) cooperate with such Selling Holders and any underwriter participating in the disposition of the Registrable Securities and their respective counsel in connection with any filings required to be made with the Financial Industry Regulatory Authority ("FINRA");
- (h) during the period when the prospectus is required to be delivered under the Securities Act, use its reasonable best efforts promptly to file all documents required to be filed with the Commission pursuant to Section 13(a) of the Exchange Act; and
- (i) take such other actions as are reasonably required in order to expedite or facilitate the disposition of Registrable Securities included in any registration statement in respect of such Registrable

Securities.

<u>Indemnification by the Company</u>. The Company will indemnify and hold harmless each Holder, any underwriter (as defined in the Securities Act) for the Holders, each officer, director, trustee, employee and agent of each Holder, and each person, if any, who controls such Holder or such underwriter within the meaning of the Securities Act (but, in the case of an underwriter or a controlling person, only if such underwriter or controlling person agrees to indemnify the persons mentioned in subdivision (b) of Section 2.5 hereof in substantially the manner set forth therein), from and against any losses, claims, damages or liabilities, joint or several, to which such Holder or any such underwriter, officer, director, trustee, employee, agent or controlling person becomes subject, under the Securities Act or otherwise, insofar as such losses, claims, damages or liabilities (or actions in respect thereof) (i) are caused by, arise out of, result from or relate to any untrue statement or alleged untrue statement of any material fact contained or incorporated by reference in the Registration Statement under which Registrable Securities were registered under the Securities Act, the prospectus contained therein, or any amendment or supplement thereto, including all documents attached thereto or incorporated by reference therein or (ii) are caused by, arise out of, result from or relate to the omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein not misleading or (iii) are caused by, arise out of, result from or relate to any violation by the Company of any rule or regulation promulgated under the Securities Act applicable to the Company and relating to any action or inaction required of the Company in connection with such registration; and the Company will reimburse such Holder and any such underwriter, officer, director, trustee, employee, agent or controlling person for any legal or other expenses reasonably incurred by such Holder, or any such underwriter, officer, director, trustee, employee, agent or controlling person in connection with investigating or defending any such loss, claim, damage, liability or action; provided, however, that the Company will not be liable to any such persons in any such case to the extent that any such loss, claim, damage, liability or action arises out of or is based upon any untrue statement or alleged untrue statement or omission or alleged omission made in reliance upon and in conformity with information furnished to the Company in writing by such person expressly for inclusion in any of the foregoing documents. The remedies provided in this Section 2.4 are not exclusive and shall not limit any rights or remedies which may otherwise be available to the Holders at law or in equity.

2.5 <u>Indemnification by the Holders</u>. Each Holder shall:

- (a) Furnish in writing all information to the Company concerning itself and its holdings of securities of the Company as shall be required in connection with the preparation and filing of any registration statement covering any Registrable Securities; and
- Indemnify and hold harmless the Company, each of its directors, each of its officers who has signed any registration statement covering any Registrable Securities, each person, if any, who controls the Company within the meaning of the Securities Act and any underwriter (as defined in the Securities Act) for the Company, against any losses, claims, damages or liabilities to which the Company or any such director, officer, controlling person or underwriter may become subject under the Securities Act or otherwise, insofar as such losses, claims, damages or liabilities (or actions in respect thereof) are caused by any untrue or alleged untrue statement of any material fact contained in any registration statement under which Registrable Securities were registered under the Securities Act, the prospectus contained therein, or any amendment or supplement thereto, or arising out of or based upon the omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein not misleading; in each case to the extent, but only to the extent, that such untrue statement or alleged untrue statement or omission or alleged omission was made in reliance upon and in conformity with information furnished in writing to the Company by such Holder expressly for inclusion in any of the foregoing documents, and each Holder shall reimburse the Company and any such underwriter, officer, director or controlling person for any legal or other expenses reasonably incurred by the Company or any such director, officer or controlling person in connection with investigating or defending any such loss, claim, damage, liability or action. Each Holder's liability under this Section 2.5 shall be limited to an amount equal to the net proceeds (after deducting any applicable underwriting discount and expenses associated with the Registrable Securities sold thereunder) received by such Holder from the sale of Registrable Securities by such Holder. The remedies provided in this Section 2.5 are not exclusive and shall not limit any rights or remedies which may otherwise be available to the Company at law or in equity.
- 2.6 <u>Contribution</u>. In order to provide for just and equitable contribution to joint liability under the Securities Act in any case in which either (a) the indemnification required by Section 2.4 is unavailable to an indemnified party in respect of any losses, claims, damages or liabilities, or legal or other expenses, with respect to which indemnity is to be provided thereunder or (b) contribution under the Securities Act may be required on the part of the indemnified party in circumstances for which indemnification is provided under Section 2.4; then, and in each such case, the Company, in lieu of indemnifying such indemnified party, shall contribute to the amount paid or payable by such indemnified party as a result of such losses, claims, damages, liabilities or expenses in such proportion as is appropriate to reflect the relative fault of the Company and indemnified parties in connection with the actions which resulted in such losses, claims, damages or liabilities, or legal or other expenses, as well as any other relevant equitable considerations. The relative fault of the Company and the indemnified party shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by such party and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission. The parties hereto agree that it would not be just and equitable if contribution pursuant to this Section 2.6 were determined by pro rata allocation or by any other method of allocation which does not take into account the equitable considerations referred to in this Section 2.6. No Holder shall be required to contribute any amount in excess of the amount by which the net proceeds of the offering (before deducting expenses) received by such Holder exceeds the amount of any damages that such Holder has otherwise been required to pay under Section 2.5 and Section 2.6 or otherwise by reason of untrue or alleged untrue statements or omissions or alleged omissions. No person guilty of fraudulent misrepresentation (within the meaning of Section 12 of the Securities Act) will be entitled to contribution from any person who was not guilty of such fraudulent misrepresentation.
- 2 . 7 <u>Conduct of Indemnification Proceedings.</u> Each Holder, the Company and each other person indemnified pursuant to Section 2.4 or 2.5 hereof will, in the event it receives notice of the commencement of any action against it in respect of which indemnity may be sought pursuant to Section 2.4 or 2.5, promptly notify the indemnifying party, in writing, of the commencement of such action and permit the indemnifying party, if the

indemnifying party so notifies the indemnified party within 10 days after receipt by the indemnifying party of notice of the commencement of the action, to participate in and to assume the defense of such action with counsel selected by the indemnifying party; provided, however, that an indemnified party shall have the right to retain its own counsel, with the fees and disbursements and expenses to be paid by the indemnifying party, if representation of such indemnified party by the counsel retained by the indemnifying party would be inappropriate due to actual or potential differing interests between such indemnified party and any other party represented by such counsel in such proceeding. The omission to notify the indemnifying party promptly of the commencement of any such action shall not relieve the indemnifying party of any liability to indemnify the indemnified party, under Section 2.4 or Section 2.5 hereof, as applicable, except to the extent the indemnifying party shall suffer any loss by reason of such failure to give notice and shall not relieve the indemnifying party of any other liabilities which it may have under this or any other agreement. Any fees and expenses incurred by the indemnified party (including any fees and expenses incurred in connection with investigating or preparing to defend such action or proceeding) shall be advanced to the indemnified party, as incurred, within thirty (30) days of written notice thereof to the indemnifying party. Any such indemnified party shall have the right to employ separate counsel in any such action, claim or proceeding and to participate in the defense thereof, but the fees and expenses of such counsel shall be the expenses of such indemnified party unless (i) the indemnifying party has agreed to pay such fees and expenses or (ii) the indemnifying party shall have failed to promptly assume the defense of such action, claim or proceeding or (iii) the named parties to any such action, claim or proceeding (including any impleaded parties) include both such indemnified party and the indemnifying party, and such indemnified party shall have been advised by counsel that there may be one or more legal defenses available to it which are different from or in addition to those available to the indemnifying party and that the assertion of such defenses would create a conflict of interest such that counsel employed by the indemnifying party could not faithfully represent the indemnified party (in which case, if such indemnified party notifies the indemnifying party in writing that it elects to employ separate counsel at the expense of the indemnifying party, the indemnifying party shall not have the right to assume the defense of such action, claim or proceeding on behalf of such indemnified party, it being understood, however, that the indemnifying party shall not, in connection with any one such action, claim or proceeding or separate but substantially similar or related actions, claims or proceedings in the same jurisdiction arising out of the same general allegations or circumstances, be liable for the reasonable fees and expenses of more than one separate firm of attorneys (together with appropriate local counsel) at any time for all such indemnified parties, unless in the reasonable judgment of such indemnified party a conflict of interest may exist between such indemnified party and any other of such indemnified parties with respect to such action, claim or proceeding, in which event the indemnifying party shall be obligated to pay the fees and expenses of such additional counsel or counsels).

2.8 Rule 144. The Company shall use reasonable best efforts to file any and all reports required to be filed by it under the Securities Act and the Exchange Act and shall take such further action as the Holders may reasonably request to the extent required from time to time to enable the Holders to sell Registrable Securities without registration under the Securities Act within the limitation of the exemptions provided by Rule 144 under the Securities Act, as such Rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission. Upon the request of any Holder, the Company will deliver to such Holder a written statement as to whether it has complied with such reporting requirements.

Miscellaneous.

- 3 . 1 Notices. All notices, requests, demands and other communications provided for by this Agreement shall be in writing (including telecopier or similar writing) and shall be deemed to have been given at the time five days after being mailed in any general or branch office of the United States Postal Service, enclosed in a registered or certified postpaid envelope, or received if sent by Federal Express or other similar overnight courier service, addressed to the address of the parties provided in the Purchase Agreement, or, with respect to any permitted transferee of the Investor that agrees to be bound by the terms of the Agreement as a Holder, to such address as shall be specified in writing by such Holder to the other parties hereto.
- 3.2 <u>Successors and Assigns</u>. This Agreement is solely for the benefit of and shall be binding upon the parties and their respective successors and permitted assigns, including, without limitation, any successor of the Company by merger, acquisition, reorganization, recapitalization or otherwise. Neither the Company nor the Investor may assign this Agreement or any of its rights, duties or obligations hereunder without the prior written consent of the other party; provided, however, that the Investor may assign its rights, duties or obligations hereunder to any affiliate of the Investor, provided that such affiliate agrees to be bound by the terms of this Agreement as a Holder. Except as expressly set forth herein, nothing herein shall be construed to provide any rights to any other entity or individual.
- 3.3 <u>Counterparts</u>. This Agreement may be executed in several counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same document.
- 3.4 <u>Headings</u>. Section headings are for convenience only and do not control or affect the meaning or interpretation of any terms or provisions of this Agreement.
- 3 . 5 <u>Governing Law.</u> This Agreement shall be governed by and construed in accordance with the laws of the State of New York governing contracts to be made and performed therein without giving effect to principles of conflicts of law, and, with respect to any dispute arising out of this Agreement, each party hereby consents to the exclusive jurisdiction of the courts sitting in the City of New York as provided in Section 10.15 of the Purchase Agreement.
- 3.6 <u>Severability.</u> Should any part, term, condition or provision hereof or the application thereof be declared illegal, invalid or otherwise unenforceable or in conflict with any other law by a court of competent jurisdiction, the validity of the remaining parts, terms, conditions or provisions of this Agreement shall not be affected thereby, and the illegal, invalid or unenforceable portions of this Agreement shall be and hereby are redrafted to conform with applicable law, while leaving the remaining portions of this Agreement intact, except to the extent necessary to conform to the redrafted portions hereof.
- 3.7 <u>Entire Agreement</u>. This Agreement sets forth the entire agreement and understanding between the parties and supersedes all proposals, commitments, writings, negotiations, discussions, agreements and understandings, oral or written, of every kind and nature between them concerning the subject matter hereof. This Agreement may not be amended or otherwise modified and no provision hereof may be waived, without the

consent of the Company and the Investor. No discharge of the terms hereof shall be deemed valid unless by full performance by the parties or by a writing signed by the parties. A waiver by any party of any breach or violation of any provision of this Agreement shall not be deemed or construed as a waiver of any other breach or violation hereof.

3.8 <u>Injunctive Relief.</u> Each of the parties hereto acknowledges that in the event of a breach by any of them of any material provision of this Agreement, the aggrieved party may be without an adequate remedy at law. Each of the parties therefore agrees that in the event of such a breach hereof the aggrieved party may elect to institute and prosecute proceedings in any court having jurisdiction as agreed to in Section 3.5 to seek to enforce specific performance or to enjoin the continuing breach hereof. By seeking or obtaining any such relief, the aggrieved party shall not be precluded from seeking or obtaining any other relief to which it may be entitled.

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first written above.

	Name:		
	Title:		
RIO	OCAN HOLDINGS USA I	NC.	
Per:			
Per:	Name:		

CEDAR SHOPPING CENTERS, INC.

Included as Exhibit 10.2

BY AND BETWEEN

CEDAR SHOPPING CENTERS PARTNERSHIP, L.P., as seller

AND

RIOCAN HOLDINGS USA INC., as purchaser

Dated as of October 26, 2009

Premises:

Columbus Crossing Shopping Center	Franklin Village Plaza
Philadelphia, PA	Franklin, MA
Loyal Plaza Shopping Center	Stop & Shop Plaza
Williamsport, PA	Bridgeport, CT
Blue Mountain Commons	Sunset Crossing Shopping Center
Susquehanna Township, PA	Dickson, PA
Charrie Dlama	

Shaw's Plaza Raynham, MA

TABLE OF CONTENTS

		Page
1.	Certain Definitions.	1
2.	Reorganizations and Consideration.	1 13
3.	Closing.	15
3. 4.	Earn-Out	16
4. 5.	Substitution.	18
5. 6.		19
7.	Closing Costs.	20
7. 8.	Blue Mountain Development Parcel.	20 22
	Investigations.	
9.	Indemnification.	24
10.	Confidentiality.	24
11.	Undertaking.	25
12.	Lender Approval.	25
13.	Representations and Warranties of Cedar.	27
14.	Representations and Warranties of RioCan.	38
15.	Investment Representations, Etc.	39
16.	Interim Covenants of Cedar.	40
17.	Deliveries to be made on the Closing Date.	44
18.	Conditions to the Closings.	47
19.	Apportionments.	50
20.	Condemnation or Destruction of the Properties.	54
21.	Release.	55
22.	Brokers.	56
23.	Limitation of Liability.	56
24.	Remedies For Default.	57
25.	Title Reviews.	59
26.	Notices.	62
27.	Amendments.	64
28.	Governing Law; Jurisdiction; Construction.	64
29.	Partial Invalidity.	64
30.	Counterparts.	64
31.	No Third Party Beneficiaries.	65
32.	Waiver.	65
33.	Assignment.	65
34.	Binding Effect.	66
35.	Entire Agreement.	66
36.	Further Assurances.	66
37.	Paragraph Headings/Schedules.	66
38.	Waiver of Trial by Jury.	66
39.	Litigation Costs.	66
40.	Currency.	67
41.	Time of the Essence	67
42.	Press Releases.	67

EXHIBITS

EXHIBIT B FORM OF LIMITED PARTNERSHIP AGREEMENT

EXHIBIT C PRE-RIOCAN PROPERTY OWNER AGREEMENTS

EXHIBIT D FORM OF MANAGEMENT AGREEMENT

EXHIBIT E ALLOTTED CONSIDERATION

EXHIBIT F FORM OF ESCROW AGREEMENT

EXHIBIT G FORM OF ASSIGNMENT AND ASSUMPTION AGREEMENT

EXHIBIT H ADDITIONAL MATERIALS

EXHIBIT I BLUE MOUNTAIN DEVELOPMENT PARCEL

EXHIBIT J FRANKLIN VILLAGE EARN-OUT SPACE

SCHEDULES

SCHEDULE 1 EXISTING OWNERSHIP CHARTS

SCHEDULE 2 REORGANIZATION STEPS

SCHEDULE 3 POST CLOSING CHARTS

SCHEDULE 4 SERVICE CONTRACTS

SCHEDULE 5 LEASES

SCHEDULE 6 EXISTING TITLE POLICIES

SCHEDULE 7 LITIGATION

SCHEDULE 8 LOAN DOCUMENTS

SCHEDULE 9 EARN-OUT PROCEEDS

SCHEDULE 10 MATERIAL LEASE DEFAULTS

SCHEDULE 11 BLUE MOUNTAIN LEASES

SCHEDULE 12 MAJOR TENANTS

SCHEDULE 13 EXISTING SURVEYS

SCHEDULE 14 EXCLUDED COMPETITORS

SCHEDULE 15 LEASING PROFORMA

AGREEMENT REGARDING PURCHASE OF PARTNERSHIP INTERESTS

AGREEMENT REGARDING PURCHASE OF PARTNERSHIP INTERESTS (this "Agreement"), made as of the 26th day of October, 2009, by and between CEDAR SHOPPING CENTERS PARTNERSHIP, L.P., a Delaware limited partnership ("Cedar") and RIOCAN HOLDINGS USA INC., a Delaware corporation ("RioCan").

WITNESSETH:

WHEREAS, subject to the terms and provisions of this Agreement, Cedar and RioCan have agreed to form a Delaware limited partnership (the "Partnership") for the purpose of directly or indirectly acquiring all of Cedar's interests in the Properties (as hereinafter defined); and

WHEREAS, the Partnership will be comprised of (x) Cedar LP (as hereinafter defined) having a nineteen percent (19%) limited partnership interest, (y) Cedar GP (as hereinafter defined) having a one percent (1%) general partnership interest, and (z) RioCan, having an eighty percent (80%) limited partnership interest.

NOW THEREFORE, in consideration of the mutual covenants contained herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Cedar and RioCan hereby agree as follows:

1. Certain Definitions.

For purposes of this Agreement, the following terms shall have the respective meanings set forth below:

Additional Title Objections: As defined in Section 25(b).

Adjustment Date: As defined in Section 19(a).

Agreement: As defined in the Preamble.

Allotted Consideration: As defined in Section 2(b).

Balance Sheets: As defined in Section 13(a).

Blue Mountain: The Property located in Susquehanna Township, Pennsylvania and known as "Blue Mountain Commons".

Blue Mountain Closing: The Closing with respect to Blue Mountain.

Blue Mountain Closing Date: The Closing Date with respect to Blue Mountain.

Blue Mountain Closing Earn-Out Proceeds: The Earn-Out Proceeds with respect to the Blue Mountain Leases for which the conditions provided in Section 4(a) shall have been satisfied as of the Blue Mountain Closing Date.

Blue Mountain Condo Conversion: As defined in Section 7(b).

Blue Mountain Development Parcel: That certain parcel of land located in Susquehanna Township, Pennsylvania, more particularly described on Exhibit I.

Blue Mountain Ground Lease: As defined in Section 7(b).

Blue Mountain Leases:

- (a) Those Leases as set out in Schedule 11;
- Leases of space in Blue Mountain executed in accordance with the terms of this Agreement;
 and
- (c) Leases of space in Blue Mountain that are approved after the Blue Mountain Closing, in accordance with the Partnership Agreement and relevant Property Management Agreement.

Blue Mountain Line of Credit The revolving line of credit from KeyBank, National Association, as administrative agent, encumbering, inter alia, Blue Mountain as of the date hereof.

Blue Mountain Loan: As defined in Section 16(b).

Blue Mountain Loan Application: As defined in Section 16(b)

Blue Mountain Loan Documents: As defined in Section 16(b).

Blue Mountain REA: As defined in Section 7(a).

Blue Mountain Separation: As defined in Section 7(a).

Blue Mountain Shopping Center Parcel: That certain parcel of land located in Susquehanna Township, Pennsylvania, more particularly described on Exhibit A-5, excluding the Blue Mountain Development Parcel

Bridgeport Ground Lease: Ground Lease dated December 8, 2004, by and between The Housing Authority of the City of Bridgeport, LLC, as landlord, and Cedar Bridgeport, LLC (as successor-in-interest to Fairfield Avenue Investors, LLC), as tenant.

Bridgeport Loan: As defined in Section 13(a).

Bridgeport Loan Documents: As defined in Section 13(a).

Buildings: With respect to each parcel of Land, all buildings, structures (surface and subsurface), installations and other improvements located thereon.

Business Day: Any day, other than a Saturday or Sunday, on which commercial banks in the State of New York are not required or authorized to be closed for business.

Cedar: As defined in the Preamble.

Cedar GP: The wholly-owned subsidiary of Cedar that will be the general partner of the Partnership.

Cedar LP: The wholly-owned subsidiary of Cedar that will be the limited partner of the Partnership.

Cedar Partners: Individually and collectively, as applicable, Cedar GP and Cedar LP.

Cedar Related Parties: Cedar and any agent, advisor, representative, affiliate, employee, director, partner, member, beneficiary, investor, servant, shareholder, trustee or other person or entity acting on Cedar's behalf or otherwise related to or affiliated with Cedar, including, without limitation, the Cedar Partners.

Closing: The closing of a Transaction contemplated hereby.

Closing Date: As defined in Section 3.

Closing Date Representations: As defined in Section 17(a).

Closing Documents: The agreements, instruments and other documents to be delivered by Cedar to RioCan pursuant to Section 17(a) or otherwise pursuant to this Agreement and the agreements, instruments and other documents to be delivered by RioCan to Cedar pursuant to Section 17(b) or otherwise pursuant to this Agreement.

Columbus Crossing: The Property located in Philadelphia, Pennsylvania and known as "Columbus Crossing Shopping Center."

Columbus Crossing Loan: As defined in Section 13(a).

Columbus Crossing Loan Documents: As defined in Section 13(a).

Columbus Crossing Loan Guaranty: Surety Agreement, dated as of June 9, 2009, by Cedar in favor of Susquehanna Bank, a Pennsylvania banking corporation.

Columbus Crossing Preferred Interests: The partnership interests in the Columbus Crossing Property Owner owned directly or indirectly by Welsh-Square, Inc., The Indenture of Trust of Bart Blatstein dated as of June 9, 1998 and/or The Irrevocable Indenture of Trust of Barton Blatstein dated July 13, 1999.

Columbus Crossing Preferred Partner Loan: The loan evidenced by the Columbus Crossing Preferred Partner Loan Documents.

Columbus Crossing Preferred Partner Loan Documents: As defined in Section 13(a).

Columbus Crossing Property Owner: Delaware 1851 Associates, L.P., a Pennsylvania limited partnership that is the owner of Columbus Crossing.

Columbus Crossing Reimbursement Agreement: As defined in Section 17(a).

Commission: The United States Securities and Exchange Commission.

Confidentiality Agreement: The Confidentiality Agreement between CSCI and RioCan REIT dated September 25, 2009.

Consideration: As defined in Section 2(b).

Control: means, when used with respect to any specified Person, the power to direct the management and policies of such Person, directly or indirectly, whether through the ownership of voting securities or other beneficial interest, by contract or otherwise; and the terms "Controlling" and "Controlled" have the meanings correlative to the foregoing.

CSCI: Cedar Shopping Centers, Inc., a Maryland corporation, and any successors thereto.

Default Notice: As defined in Section 24(c).

Defaulting Party: As defined in Section 24(c).

Deposit: As defined in Section 2(c).

Direct Interests: Individually and collectively, as applicable, the partnership interests in the applicable Property Owners (other than the Columbus Crossing Preferred Interests).

Due Diligence Site: The internet based "virtual vault" created by Cedar where Cedar has assembled and has made available the Information to RioCan prior to the date hereof.

Earn-Out Proceeds: An amount equal to the NOI payable under the applicable lease (or renewal thereof), divided by a capitalization rate of 8.5%. A sample calculation of the Earn-Out Proceeds with respect to each Blue Mountain Lease executed prior to the date hereof and each portion of the Franklin Village Earn-Out Space is attached hereto and made a part hereof as Schedule 9.

Environmental Claim: With respect to any Property, any action, cause of action, suit, order, decree, award, proceeding, judgment, penalty, assessment, claim, or fine imposing or alleging potential liability for any violation of Environmental Laws or otherwise relating to Hazardous Substances.

Environmental Laws: All applicable federal, state, municipal and local laws (including without limitation all statutes, by-laws and regulations and all orders, directives and decisions rendered by, and policies, instructions, guidelines and similar guidance of, any ministry, department or administrative or regulatory agency), relating to the protection of the environment, or the manufacture, processing, distribution, use, treatment, storage, disposal, packaging, transport, handling, containment, clean-up or other remediation or corrective action of any Hazardous Substances.

Escrow Agent: Commonwealth Land Title Insurance Company, Two Grand Central Tower 140 East 45th Street, 22nd Floor, New York, NY 10017, Attention: Robert Fitzgerald.

Executive Order 13224: Executive Order 13224–Blocking Property and Prohibiting Transactions with Persons Who Commit, Threaten to Commit, or Support Terrorism, issued by OFAC.

Existing Columbus Crossing Preferred Partner Lender. Cedar Lender LLC, a Delaware limited liability company that is a wholly-owned subsidiary of Cedar and is the holder of the Columbus Crossing Preferred Partner Loan on the date hereof.

Existing Surveys: The surveys listed on Schedule 13 attached hereto and made a part hereof.

Existing Title Policies: The title policies listed on Schedule 6 attached hereto and made a part hereof.

Extension Period: As defined in Section 12(b).

Financial Statements: As defined in Section 13(a).

First Scheduled Closing Date: As defined in Section 3.

Franklin Village: The Property located in Franklin, Massachusetts and known as "Franklin Village Plaza".

Franklin Village Applebee's Space: All or any portion of the space located at Franklin Village that is designated as the "Applebee's Space" on Exhibit J.

Franklin Village Closing: The Closing with respect to Franklin Village.

Franklin Village Closing Date: The Closing Date with respect to Franklin Village.

Franklin Village Earn-Out Space: Individually and collectively, as applicable, the Franklin Village Applebee's Space, the Franklin Village New Lease Space and the Franklin Village Renewal Lease Space.

Franklin Village Loan: As defined in Section 13(a).

Franklin Village Loan Documents: As defined in Section 13(a).

Franklin Village New Lease Space: All or any portion of the space located at Franklin Village that is designated as the "Franklin Village New Lease Space" on Exhibit J.

Franklin Village Renewal Lease Space: All or any portion of the space located at Franklin Village that is designated as the "Franklin Village Renewal Lease Space" on Exhibit J.

GAAP: Generally accepted accounting principles of the United States, consistently applied.

Governmental Authority: Any agency, instrumentality, department, commission, court, tribunal or board of any government, whether foreign or domestic and whether national, federal, state, provincial, local or any quasi-governmental entity or any Person acting under the authority of any of the foregoing.

Ground Leases:The Bridgeport Ground Lease and the Loyal Plaza Ground Lease, in each case as amended, renewed or otherwise varied.

Ground Lessor Estoppels: As defined in Section 16(d).

Ground Lessors: The lessors under the Bridgeport Ground Lease and the Loyal Plaza Ground Lease.

Hazardous Substances: Any pollutants, contaminants, chemicals, deleterious substances, waste (including without limitation industrial, toxic or hazardous wastes), petroleum or petroleum products, asbestos, PCBs, underground storage tanks and the contents thereof, flammable materials or radioactive materials.

Indirect Interests: Individually and collectively, as applicable, the equity interests in the Indirect Owners.

Indirect Owner(s): Individually and collectively, as applicable, the subsidiaries of Cedar that are direct or indirect owners of equity interests in the Property Owners, which (a) exist on the date hereof, will survive the Reorganizations (as more particularly described in Schedule 2) and will be wholly owned directly or indirectly by REIT Property Subsidiary following the Closings (as more particularly described in Schedule 3) or (b) will be formed in connection with the Reorganizations (as more particularly described in Schedule 2) and will be wholly owned directly or indirectly by REIT Property Subsidiary following the Closings (as more particularly described in Schedule 3).

Information: Any of the following: (i) all information and documents in any way relating to the Properties, the Owners, the Interests, the Columbus Crossing Preferred Interests, the operation thereof or the sale thereof, all leases and contracts furnished to RioCan or the RioCan Representatives by Cedar or any Cedar Related Party or their agents or representatives, including, without limitation, their contractors, engineers, attorneys, accountants, consultants, brokers or advisors, and all information and documents posted on the Due Diligence Site, whether prior to or after the date hereof, and (ii) all analyses, compilations, data, studies, reports or other information or documents prepared or obtained by RioCan or the RioCan Representatives containing or based on, in whole or in part, the information or documents described in the preceding subparagraph (i), the Investigations, or otherwise reflecting their review or investigation of the Properties, the Owners, the Interests and/or the Columbus Crossing Preferred Interests.

Institutional Investor:Any of the following types of entities (or any entity that is directly or indirectly wholly-owned and Controlled by any of the following types of entities), whether domestic or Canadian: (a) a commercial bank, trust company (whether acting individually or in a fiduciary capacity for another entity that constitutes an Institutional Investor), savings and loan association, savings bank, financing company or similar institution; (b) an insurance company; (c) an investment bank; or (d) an employee's welfare, benefit, profit-sharing, pension or retirement trust, fund or system (whether federal, state, municipal, private or otherwise); in each case on the condition that such Institutional Investor (i) is regularly engaged in the business of owning or operating commercial real estate properties, (ii) is recognized as a reputable investor, (iii) has net assets (in name or under management) in excess of One Billion Dollars (\$1,000,000,000), (iv) is not one of Persons described in Section 14(a)(iv) hereof, (v) is neither one of the Persons listed on Schedule 14 nor Controlled by any such Persons, and (vi) is otherwise reasonably acceptable to Cedar, it being acknowledged that CPP Investment Board, a Canadian corporation, is acceptable to Cedar.

Intellectual Property: The interest, if any, of the applicable Property Owner in any trademarks, trade names, logos, names, coined words, abbreviations, designs, styles, certification marks, copyrights, industrial designs and other similar property relating solely to any Property.

Interests: Individually and collectively, as applicable, the Direct Interests and the Indirect Interests.

Investigations: As defined in Section 8.

Land: As applicable, that certain parcel of land located in (i) Philadelphia, Pennsylvania, (ii) Franklin, Massachusetts, (iii) Williamsport, Pennsylvania, (iv) Bridgeport, Connecticut, (v) Susquehanna Township, Pennsylvania, (vi) Dickson, Pennsylvania, and (vii) Raynham, Massachusetts, all as more particularly described in Exhibit A-1 through A-7 hereof, respectively.

Lease Exhibit: As defined in the definition of "Leases" set forth in this Agreement.

Leases: With respect to each Property, (i) the leases described on Schedule 5 attached hereto and made a part hereof (collectively, the "**Lease Exhibit**") with respect to such Property, and (ii) the leases entered into by any Property Owner in accordance with Section 4 and Section 16(a) hereof.

Leasing Costs: As defined in Section 19(b).

Lenders: Collectively, the mortgage lenders under each of the Loans.

Loan Approval Deadline: As defined in Section 12(b).

Loan Approvals: As defined in Section 12(a).

Loan Documents: Collectively, the Columbus Crossing Loan Documents, the Franklin Village Loan Documents, the Loyal Plaza Loan Documents, the Bridgeport Loan Documents and the Shaw's Plaza Loan Documents and, if applicable, the Blue Mountain Loan Documents and the Sunset Crossing Loan Documents, as each of the foregoing may be amended in accordance with this Agreement.

Loan Estoppel Statement: As defined in Section 12(a).

Loans: Collectively, the Columbus Crossing Loan, the Franklin Village Loan, the Loyal Plaza Loan, the Bridgeport Loan and the Shaw's Plaza Loan and, if applicable, the Blue Mountain Loan and the Sunset Crossing Loan.

Loyal Plaza Ground Lease: Agreement of Lease dated January 15, 1963, by and between Robert M. Zaner and Ruth S. Zaner, his wife, as landlord, and Loyal Plaza Associates, L.P. (as ultimate successor-in-interest to Murray H. Goodman), as tenant, as amended by that certain Amendatory Agreement, dated March 26, 1964.

Loyal Plaza Loan: As defined in Section 13(a).

Loyal Plaza Loan Documents: As defined in Section 13(a).

Major Tenants: Those Tenants listed on Schedule 12 attached hereto.

Management Agreements: With respect to each Property, the Property Management Agreement to be entered into at the applicable Closing between the applicable Property Owner and Manager for the management and leasing of such Property, the form of which is attached hereto as Exhibit D.

Manager: Cedar or an affiliate of Cedar, as determined by Cedar (provided such affiliate is directly or indirectly wholly-owned by Cedar or CSCI and generally manages the other properties directly or indirectly owned by Cedar).

Mandatory Cure Item: As defined in Section 25(c).

Material Title Contracts: Any common use agreements and easement agreements of record the termination of which would materially and adversely affect or interfere with the ordinary use or operation of the applicable Property.

Net Consideration: As defined in Section 2(b).

New Columbus Crossing Preferred Partner Lender. A wholly-owned subsidiary of REIT Property Subsidiary that shall be a Delaware limited liability company, to be formed to acquire, at the Closing of Columbus Crossing, and thereafter own the Columbus Crossing Preferred Partner Loan and the Columbus Crossing Preferred Partner Loan Documents.

NOI: Means the annualized amounts payable by the applicable Tenant pursuant to a Lease (or renewal thereof), less the sum of (i) the annualized operating costs and realty tax recoveries included in such amounts, (ii) the Permanent Shortfall, (iii) a vacancy allowance equal to 3% of annualized "gross receipts" (as defined in the applicable Management Agreement) to be generated from the applicable Tenant (other than, in the case of Blue Mountain only, Giant Food Stores LLC, for which no such vacancy allowance shall be deducted); and (iv) 3.5% of "gross receipts" to be generated from the applicable Tenant, on account of management fees.

Non-Defaulting Party: As defined in Section 24(c).

OFAC: The Office of Foreign Assets Control of the United States Department of the Treasury.

OFAC Lists: As defined in Section 14(a).

Outside Adjustment Date: As defined in Section 19(f)

Outside Closing Date: As defined in Section 3.

Outside RioCan Adjournment Date: As defined in Section 3.

Owners: Individually and collectively, as applicable, the Property Owners and the Indirect Owners.

Partnership: As defined in the Recitals.

Partnership Agreement: The Limited Partnership Agreement to be entered into by and among the Cedar Partners and RioCan on or prior to the first Closing hereunder in respect of their relationship as partners of the Partnership, in substantially the form attached hereto as Exhibit B.

Partnership Interests: As defined in Section 15(a).

Partnership Subsidiary GP: Means a limited liability company wholly-owned by the Partnership and formed to own the general partnership interest in the REIT.

Percentage Interest: The respective partnership interests of the Cedar Partners and RioCan in the Partnership from and after the first Closing hereunder as follows: (x) the percentage interest of Cedar GP shall be one percent (1%), (y) the percentage interest of Cedar LP shall be nineteen percent (19%) and (z) the percentage interest of RioCan shall be eighty percent (80%), as the same may be adjusted pursuant to the provisions of the Partnership Agreement.

Permanent Shortfall: The annualized amount, if any, by which the Tenant's annualized proportionate share of operating cost and realty tax recoveries (as determined by Cedar, acting reasonably, based on the recoveries provided for in the standard form lease used by Cedar) exceeds the annualized amount actually payable by such Tenant under such Lease on account of operating costs and realty taxes.

Permitted Exceptions: With respect to each Property (unless otherwise provided herein): (i) the state of facts shown on the Existing Survey of such Property; (ii) as applicable, subject to the rights of RioCan pursuant to Section 16(b) and (c) hereof, respectively, any Loan Documents; (iii) with respect to the Properties, those matters specifically set forth on Schedule B of the Existing Title Policy of the applicable Property and any matters omitted or affirmatively insured over pursuant to or in connection with such Existing Title Policy; (iv) all laws, ordinances, rules and regulations of the United States, the Commonwealth or State in which the Property is located, or any Governmental Authority, as the same may now exist or may be hereafter modified, supplemented or promulgated; (v) all presently existing and future liens of real estate taxes or assessments and water rates, water meter charges, water frontage charges and sewer taxes, rents and charges, if any, provided that such items are not yet due and payable and are apportioned as provided in this Agreement; (vi) any other matter or thing affecting title to such Property that RioCan shall have agreed in writing or be deemed to have agreed pursuant to the express terms of this Agreement to waive as a Title Objection or Additional Title Objection; (vii) all violations of laws, ordinances, orders, requirements or regulations of any Governmental Authority known by RioCan as of the date of this Agreement; (viii) all utility easements; and (ix) all other matters of record which do not, individually or in the aggregate, prohibit or materially and adversely interfere with the present use or operation of the applicable Property, or materially and adversely affect the value or marketability of the applicable Property.

Person: An individual, partnership, joint venture, corporation, trust or other entity.

Personal Property: With respect to each Property, all right, title and interest of the applicable Property Owner, if any, in and to the fixtures, equipment and other personal property owned by such Property Owner and attached or appurtenant to the applicable Property.

Pre-RioCan Owner Agreements: Prior to the respective Reorganizations, the organizational documents of each Owner identified on Exhibit C attached hereto and following the Reorganizations but prior to the applicable Closing Date, the limited partnership or limited liability company agreement of each Owner in a form consistent with the terms of the Partnership Agreement and the applicable Loan Documents (as modified by any applicable Loan Approval) in all material respects and otherwise reasonably acceptable to the parties.

Property or Properties: Collectively or individually, as applicable, those certain real properties commonly known as: (i) Columbus Crossing Shopping Center, located in Philadelphia, Pennsylvania; (ii) Franklin Village Plaza, located in Franklin, Massachusetts; (iii) Loyal Plaza Shopping Center, located in Williamsport, Pennsylvania; (iv) Stop & Shop Plaza, located in Bridgeport, Connecticut; (v) Blue Mountain Commons, located in Susquehanna Township, Pennsylvania; (vi) Sunset Crossing Shopping Center, located in Dickson, Pennsylvania; and (vii) Shaw's Plaza Shopping Center, located in Raynham, Massachusetts, as more particularly described in Exhibit A-1 through A-7 attached hereto, respectively, together with all of the Buildings located or to be developed thereon, and also together with all rights, interests, entitlements, benefits, and privileges of any nature or kind whatsoever related thereto including, without limitation, the Land, the Personal Property, Service Contracts, Leases, Intellectual Property, Warranties, and all easements for ingress, egress, parking, utility service and other appurtenances thereto.

Property Owner(s): Individually and collectively, as applicable, the entities identified in Schedule 1 attached hereto, each as owner or ground lessee of each Property indicated opposite its name.

REIT: Means a limited partnership owned by the Partnership, as limited partner, the Partnership Subsidiary GP, as general partner, and certain outside investors, as preferred interest holders, and formed to own all of the limited partnership interests in the REIT Property Subsidiary. The REIT shall be treated as a corporation for U. S. tax purposes.

REIT Property Subsidiary: Means a limited partnership owned by the REIT, as limited partner, and the REIT Subsidiary GP, as general partner, and formed to own directly or indirectly through one or more wholly-owned subsidiaries (a) all of the Interests and (b) all of the membership interests in the New Columbus Crossing Preferred Partner Lender.

REIT Subsidiary GP: Means a limited liability company wholly-owned by the REIT and formed to own all of the general partnership interests in the REIT Property Subsidiary.

Remaining Scheduled Closing Date(s): As defined in Section 3.

Reorganizations: As defined in Section 2(a).

Required Operating Tenant: Individually and collectively, as applicable, any Required Tenant that is a Giant Food Store, Super Fresh Supermarket, Stop & Shop, Marshall's or Shaw's Supermarket.

Required Tenants: As defined in Section 16(d)

RioCan: As defined in the Preamble.

RioCan REIT: RioCan Real Estate Investment Trust, an Ontario trust.

RioCan Related Party: As defined in Section 21(a).

RioCan Representatives: The directors, officers, employees, affiliates, partners, members, brokers, agents or other representatives, including, without limitation, attorneys, accountants, contractors, consultants, engineers and financial advisors of RioCan.

Scheduled Closing Date(s): As defined in Section 3.

SEC: The Securities and Exchange Commission.

Securities Act. The Securities Act of 1933, as amended.

Service Contracts: With respect to each Property, (i) the contracts described on Schedule 4 attached hereto and made a part hereof, (ii) Terminable Service Contracts, and (iii) contracts entered into by any Property Owner in accordance with Section 16 hereof.

Settlement Statement: As defined in Section 17(a).

Shaw's Plaza: The Property located in Raynham, Massachusetts and known as "Shaw's Plaza."

Shaw's Plaza Loan: As defined in Section 13(a).

Shaw's Plaza Loan Documents: As defined in Section 13(a).

Subject Interests: As defined in Section 12(b).

Subject Property: As defined in Section 12(b).

Subject Transaction(s): As defined in Section 12(b).

Substitute Property: As defined in Section 5.

Substitution Date: As defined in Section 5.

Sunset Crossing: The Property located in Dickson, Pennsylvania and known as "Sunset Crossing Shopping Center".

Sunset Crossing Line of Credit: The revolving line of credit from Bank of America, N.A., as administrative agent, encumbering, inter alia, Sunset Crossing as of the date hereof.

Sunset Crossing Loan: As defined in Section 16(c).

Sunset Crossing Loan Application: As defined in Section 16(c).

Sunset Crossing Loan Documents: As defined in Section 16(c).

Supplemental Due Diligence Period: As defined in Section 8.

 $\textbf{Supplemental Due Diligence Termination Notice:} \ As \ defined \ in \ Section \ 8.$

Supplemental Testing: Individually or collectively, as applicable, (a) in the case of Shaw's Plaza, a Phase 1 environmental assessment and an engineering review of Shaw's Plaza, and (b) in the case of Sunset Crossing, a Phase 2 environmental assessment of Sunset Crossing pursuant to the scope of work submitted by RioCan and approved by Cedar prior to the date of this Agreement.

Tenant Estoppels: As defined in Section 16(d).

Tenants: Means all Persons having a right to occupy any rentable area of any Property pursuant to a Lease.

Terminable Service Contracts: With respect to any Property, contracts entered into in the ordinary course of business that are cancellable on sixty (60) days notice or less without premium or penalty.

Title Company: A nationally recognized title company agreed upon by RioCan and Cedar acting reasonably and in good faith.

Title Objection Deadline: As defined in Section 25(a).

Title Objection Letter: As defined in Section 25(a).

Title Objection Response: As defined in Section 25(a).

Title Objections: As defined in Section 25(a).

Title Reports: As defined in Section 25(a).

Transaction(s): Individually or collectively, as applicable, the Closing of the transfer of the Interests related to one or more of the Properties to the REIT Property Subsidiary and the transfer of the Columbus

Crossing Preferred Partner Loan to the New Columbus Crossing Preferred Partner Lender, all in accordance with the terms of this Agreement.

Transfer Taxes: As defined in Section 6(a).

Update Certificate: As defined in Section 17(a).

Warranties: The existing warranties, guarantees and indemnities for the construction and/or the existing operation of the Buildings.

2. Reorganizations and Consideration.

- (a) Schedule 1 attached hereto and made a part hereof depicts the ownership structure of each of the Properties as of the date hereof. Prior to or contemporaneously with the Closing of a Transaction, Cedar shall cause to be effectuated the applicable assignments, transfers and conversions described on Schedule 2 attached hereto (the "Reorganizations"). Schedule 3 attached hereto and made a part hereof depicts the ownership structure following the Closing of the Transactions (assuming all of the Transactions close in accordance with the terms of this Agreement). From and after the applicable Closing Date, no Cedar Related Party shall have any continuing obligations to RioCan with respect to the Properties as transferor or seller thereof other than as expressly provided in this Agreement.
- (b) The aggregate consideration payable by RioCan to Cedar on the Closing Dates for the Interests (other than the Interests of the Indirect Owners of Blue Mountain and the Earn-Out Proceeds in respect of the Franklin Village Applebee's Space and the Franklin Village New Lease Space) shall be One Hundred Sixteen Million Five Hundred Twenty Thousand Dollars (\$116,520,000) (the "Consideration") as allocated to each Property (other than Blue Mountain) as set forth in the applicable pro forma price schedule attached hereto as Exhibit E (the "Allotted Consideration"). For purposes of this Agreement, (a) the "Consideration" for the Interests of the Indirect Owners of Blue Mountain shall be the sum of (x) the Blue Mountain Closing Earn-Out Proceeds and (y) any amounts paid to Cedar as additional Earn-Out Proceeds pursuant to Section 4(a), (b) the Blue Mountain Closing Earn-Out Proceeds shall constitute the "Allotted Consideration" payable by RioCan to Cedar on the Blue Mountain Closing Date for the Interests of the Indirect Owners of Blue Mountain, and (c) the "Consideration" for the Interests of the Indirect Owners of Franklin Village shall be the sum of (x) the Allotted Consideration for Franklin Village, less any portion of the Franklin Village Earn-Out Escrow that is not disbursed to Cedar and (y) any amounts paid to Cedar as additional Earn-Out Proceeds pursuant to Section 4(b) in respect of the Franklin Village Applebee's Space and the Franklin Village New Lease Space. Accordingly, for the avoidance of doubt, wherever this Agreement shall provide that, following the failure to close a particular Transaction that "the Consideration shall be reduced by the amount of the applicable Allotted Consideration", the parties acknowledge and agree that (A) if such Transaction is the Blue Mountain Closing, no deduction of Consideration shall be made with respect to the other Transactions since the Consideration for Blue Mountain consists entirely of Earn-Out Proceeds and (B) if such Transaction is the Franklin Village Closing, only the portion of the Consideration representing the Franklin Village Earn-Out Escrow and the Allotted Consideration for the portion of Franklin Village that is not included in the Franklin Village Earn-Out Space, shall be deducted. The Allotted Consideration shall be (i) reduced for each Transaction by eighty percent (80%) of the outstanding principal amount as of the Closing Date of the Loan applicable thereto and (ii) adjusted pursuant to the express terms of this Agreement (the Allotted Consideration, as so reduced and adjusted, the "Net Consideration"). Each of Cedar and RioCan (and their respective direct and indirect partners, members, owners, beneficiaries, investors, and shareholders) agree to allocate the Consideration as determined for U.S. federal income tax purposes (which shall include all capitalizable costs incurred in connection with the transactions hereunder) among the Properties for all purposes (including, without limitation, accounting, financial reporting and federal and applicable state and local income tax purposes) on the basis of Section 1060 of the Internal Revenue Code, as amended, and in a manner consistent with Exhibit E, as such allocation may be amended from time to time pursuant to the next sentence. The allocation of the Consideration shall be amended to reflect any adjustment to the Consideration. The Net Consideration shall be payable as follows:
 - RioCan shall pay the applicable Net Consideration to Cedar, and in consideration therefor, Cedar shall transfer or cause to be transferred the subject Interests to the REIT Property Subsidiary; and
 - (ii) RioCan shall pay the Net Consideration to Cedar by wire transfer of immediately available federal funds to an account or accounts designated by Cedar.
- (c) Within two (2) Business Days after the date this Agreement is executed and delivered by Cedar and RioCan, RioCan shall deposit with the Escrow Agent, as escrowee, by wire transfer of immediately available federal funds to an account designated by the Escrow Agent, the sum of Five Hundred Thousand Dollars (\$500,000) (together with all interest thereon, the "Deposit"). The Deposit shall be held by the Escrow Agent pursuant to the escrow agreement attached hereto as Exhibit F. If RioCan shall fail to deposit the Deposit with the Escrow Agent within two (2) Business Days after the date this Agreement shall be executed and delivered by Cedar and RioCan, at Cedar's election exercised by delivery of written notice to RioCan following such two (2) Business Day period but prior to receipt of the Deposit, this Agreement shall be null, void ab initio and of no force or effect. The Deposit shall be applied in partial payment of the applicable Allotted Consideration required to be made by RioCan at the Closing of the Transaction with respect to the last remaining Property.

Closing.

The closing (each a "Closing") of the Transactions shall occur in stages. The first Closing shall occur at 10:00 a.m. (Eastern time) on the date that is ten (10) days after the satisfaction (or waiver) of the last of all conditions precedent for one or more Transactions (the "First Scheduled Closing Date"). Each of the remaining Transactions with respect to which all conditions precedent thereto have been satisfied or waived by the party entitled to do so, shall occur on the date that is ten (10) days after the satisfaction (or waiver) of the last of all such conditions precedent for the applicable Transaction (each, a "Remaining Scheduled Closing Date"; together with the First Scheduled Closing Date, the "Scheduled Closing Date(s)"); provided, however, that RioCan shall have the right to adjourn a particular Scheduled Closing Date (other than the First Scheduled Closing Date) not more than two (2) times to a Business Day that is not later sixty (60) days following the applicable Scheduled Closing Date (the "Outside RioCan Adjournment Date") by delivery of written notice to Cedar on or prior to the original Scheduled Closing Date of the adjourned Scheduled Closing Date. Without limitation to the foregoing, the parties agree to use commercially reasonable efforts to close as many of the Transactions on the same date as practicable. Notwithstanding the foregoing but subject to the right of Cedar to adjourn the Closing of one or more Transactions pursuant to Section 12(b) or Section 25 hereof, in the event that all of the conditions precedent with respect to any Transaction shall not have been satisfied or waived by the party entitled to do so by July 26, 2010(the "Outside Closing Date"), then, subject to Section 5, this Agreement shall automatically terminate on such Outside Closing Date as to such Transaction, the Deposit shall be refunded to RioCan (if no other Closing with respect to another Property remains outstanding) and the Consideration shall be reduced by the amount of the applicable Allotted Consideration, whereupon the parties hereto shall be relieved of all further liability and responsibility under this Agreement with respect to such Transaction (except for any obligation expressly provided to survive a termination of this Agreement). The Closings shall occur at the offices of the Title Company through an escrow and pursuant to escrow instructions consistent with the terms of this Agreement and otherwise mutually satisfactory to Cedar and RioCan (the date on which any Closing shall occur being herein referred to as a "Closing Date"). Each Closing shall constitute approval by each of Cedar and RioCan of all matters to which such party has a right of approval and a waiver of all conditions precedent related to the applicable Transaction.

4. Earn-Out

- (a) <u>Blue Mountain</u>. Notwithstanding any other provision of this Agreement, for each Blue Mountain Lease for which the following conditions have not been satisfied on or prior to the Blue Mountain Closing, additional Earn-Out Proceeds shall be paid by RioCan to Cedar on or prior to the tenth (10th) Business Day immediately following the date that all of the following conditions shall have been satisfied in respect of such lease:
 - (i) The tenant thereunder shall have commenced paying regularly scheduled rent in accordance with such lease;
 - (ii) RioCan shall have received either:
 - (A) a Tenant Estoppel from the applicable tenant, which shall be in a form consistent with the form required pursuant to Section 16(d); or
 - (B) where such Tenant Estoppel is not available and the applicable tenant is not a Major Tenant, a certificate of Cedar confirming substantially the same information as would have been in a Tenant Estoppel certificate from the applicable tenant; and
 - (iii) RioCan shall have received a reasonably detailed calculation prepared by Cedar of the Earn-Out Proceeds due to Cedar, which shall be conclusive and binding absent manifest error, it being acknowledged and agreed that the calculations set forth on Schedule 9 satisfy the foregoing requirement with respect to the Blue Mountain Leases executed prior to the date hereof and are hereby deemed to have been delivered to RioCan in accordance with this subparagraph (iii);

provided, however, that no Earn-Out Proceeds shall be payable with respect to any Blue Mountain Lease unless each of the conditions set forth in subparagraph (a) above shall have been satisfied (x) on or prior to the second (2nd) anniversary of the Blue Mountain Closing Date, with respect to any Blue Mountain Lease executed prior to the Blue Mountain Closing Date or (y) on or prior to the third (3rd) anniversary of the Blue Mountain Closing Date, with respect to any Blue Mountain Lease executed during the period commencing on the Blue Mountain Closing Date and ending on the second (2nd) anniversary of the Blue Mountain Closing Date.

- (b) Franklin Village. Notwithstanding any other provision of this Agreement, at the Franklin Village Closing, RioCan shall deposit with the Escrow Agent, as escrowee, by wire transfer of immediately available federal funds to an account designated by the Escrow Agent, a portion of the Net Consideration for Franklin Village in the amount of Four Million Three Hundred Eighty Thousand Dollars (\$4,380,000) (together with all interest thereon, the "Franklin Village Earn-Out Escrow") for the purposes of funding the Earn-Out Proceeds payable to Cedar pursuant to this Section 4(b) with respect to all of the Franklin Village Renewal Lease Space. The Franklin Village Earn-Out Escrow shall be held by the Escrow Agent pursuant to an escrow agreement reasonably satisfactory to the parties. For each lease (or renewal thereof) of all or any portion of the Franklin Village Earn-Out Space for which the following conditions are satisfied, Earn-Out Proceeds shall be paid by Escrow Agent to Cedar from the Franklin Village Earn-Out Escrow (or, in the case of the Franklin Village Applebee's Space and the Franklin Village New Lease Space, by RioCan directly) (x) at the Franklin Village Closing, to the extent such conditions shall have been satisfied prior to the Franklin Village Closing or (y) to the extent such conditions shall not have been satisfied on or prior to the Franklin Village Closing with respect to any such lease (or renewal thereof), on or prior to the tenth (10th) Business Day immediately following the date that all of the following conditions shall have been satisfied in respect of such lease (or renewal thereof):
 - The tenant thereunder shall have commenced paying regularly scheduled rent in accordance with such lease in respect to the primary or renewal term thereof, as applicable;
 - (ii) RioCan shall have received either:
 - (A) a Tenant Estoppel from the applicable tenant, which shall be in a form consistent with the form required pursuant to Section 16(d); or
 - (B) where such Tenant Estoppel is not available within the applicable timeframe provided in this Section 4(b), a certificate of Cedar confirming substantially the same information as would have been in a Tenant Estoppel certificate from the applicable tenant;
 - (iii) RioCan shall have received a reasonably detailed calculation prepared by Cedar of the Earn-Out Proceeds due to Cedar, which shall be conclusive and binding absent manifest error, it being acknowledged and agreed that the calculations set forth on Schedule 9 satisfy the foregoing requirement with respect to the leases described thereon and are hereby deemed to have been delivered to RioCan in accordance with this subparagraph (iii);
 - (iv) with respect to the Franklin Village Renewal Lease Space only, such renewal is executed with a tenant identified on Exhibit J;
 - (v) with respect to the Franklin Village New Lease Space only, to the extent not executed and delivered prior to the date hereof, such lease is executed with a tenant identified on Exhibit J in substantial accordance with the terms of the applicable executed letter of intent furnished to RioCan on or prior to the date of this Agreement (or on terms that are more favorable to the applicable Property Owner);
 - (vi) with respect to the Franklin Village Applebee's Space only, such lease (A) is for the entire Franklin Village Applebee's Space, (B) is executed with a tenant that is unaffiliated with Cedar, (C) provides for a term of not less than five (5) years, (D) includes a market rental and other material terms that are market (or on terms that are more favorable to the applicable Property Owner) and (E) is with a tenant reasonably approved by RioCan (which approval shall not be unreasonably withheld, conditioned or delayed),

provided, however, that (x) no Earn-Out Proceeds shall be payable with respect to any such lease (or renewal thereof) unless the conditions set forth in subparagraph (b) above shall have been satisfied (A) within three (3) months following the expiration of the applicable lease, with respect to the existing leases demising the Franklin Village Renewal Lease Space and described on Exhibit J, (B) March 26, 2010, with respect to any new lease demising any portion of the Franklin Village New Lease Space, and (C) February 28, 2011, with respect to the new lease demising the Franklin Village Applebee's Space; and (y) in no event shall Cedar be entitled to Earn-Out Proceeds for Franklin Village in excess of the amount of the Franklin Village Earn-Out Escrow with respect to any portion of the Franklin Village Renewal Lease Space (it being understood and agreed that the remainder of the Franklin Village Earn-Out Space shall not be subject to such limitation). Any amounts remaining in the Franklin Village Earn-Out Escrow after payment to Cedar of all Earn-Out Proceeds to which it is entitled pursuant to this Section 4(b) shall be promptly returned to RioCan.

- (c) Payment. The Earn-Out Proceeds shall be paid by the Escrow Agent or RioCan, as applicable, by wire transfer of immediately available federal funds to an account designated by Cedar.
- (d) <u>Survival</u>. The provisions of this Section 4 shall survive the Closings.

If RioCan terminates this Agreement with respect to the acquisition of the Interests applicable to Shaw's Plaza pursuant to Section 8, or any of the conditions to RioCan's obligation to consummate a Transaction contained in Section 18(a) are not satisfied or waived on or before the Outside Closing Date, provided that no default by RioCan hereunder or under the Partnership Agreement shall have occurred, then Cedar will co-operate with RioCan for a period of six (6) months following the Outside Closing Date to identify another property or properties owned by it that has a value which is similar to the value of the Property that was subject to such Transaction to be substituted for such Property (each a "Substitute Property"); provided, however, that in no event shall anything contained in this Section 5 obligate either party to consummate any such substitution, it being acknowledged and agreed that any such election shall be made by each party in its sole and absolute discretion and in writing (the date that the parties shall have acknowledged in writing the inclusion of each Substitute Property under this Agreement shall be referred to herein as a "Substitution Date"). If a Substitute Property or Substitute Properties are selected by the parties as aforesaid, the parties shall enter into an amendment to this Agreement on or prior to the Substitution Date acceptable to the parties in their sole and absolute discretion that will provide for all the terms and provisions applicable to such substitution, including, without limitation, modification of defined terms (e.g., "Property"), applicable deadlines (e.g., "Outside Closing Date", "Outside RioCan Adjournment Date", "Loan Approval Deadline", "Title Objection Deadline", etc.) and the Allotted Consideration. The provisions of this Section 5 shall survive until April 26, 2011.

6. Closing Costs.

Costs in connection with each of the Transactions shall be allocated as follows:

- The Cedar Partners and RioCan shall pay their respective Percentage Interests of the following costs and expenses due and payable in connection with the Reorganizations and/or the Transactions: (A) any and all state and local recording charges and fees, if any; (B) all of the costs, expenses and charges in connection with the Loan Approvals, including, without limitation, all application fees, processing fees, assumption fees, attorneys' fees, consultants' fees and costs and expenses associated with survey updates, record searches, title examinations and updated mortgagee title insurance policies (including endorsements thereto), if any, required by any Lender; (C) any escrow fees charged by the Escrow Agent; (D) any and all state and local deed taxes, real property transfer taxes, controlling-interest taxes and similar taxes (collectively, "Transfer Taxes"); (E) all costs and expenses associated with the formation of additional Indirect Owners and the New Columbus Crossing Preferred Partner Lender and, including, without limitation, legal and filing fees and disbursements; (F) with respect to the Transaction involving Blue Mountain, all of the reasonable costs, expenses and charges incurred in connection with the release of Blue Mountain from the Blue Mountain Line of Credit; and (G) with respect to the Transaction involving Sunset Crossing, all of the reasonable costs, expenses and charges incurred in connection with the release of Sunset Crossing from the Sunset Crossing Line of Credit.
- (b) RioCan shall pay all costs and expenses associated with (A) record searches, title examinations and updated owner title insurance policies (including endorsements thereto), if any, desired by RioCan and not by any Lender; (B) any title insurance policy and/or endorsements insuring or otherwise providing coverage to, RioCan as a partner of the Partnership; (C) obtaining updates to the surveys of the Properties as and to the extent desired by RioCan and not by any Lender; (D) as applicable, all costs and expenses associated with the formation of the REIT, including, without limitation, legal and filing fees and disbursements; (E) as applicable, all costs and expenses associated with the contemplated conversion of the Owners that are limited liability companies, including, without limitation, legal and filing fees and disbursement; and (F) all costs and expenses associated with its Investigations, including, without limitation, legal and filing fees and disbursements.
- (c) In addition, RioCan hereby agrees to pay to Cedar, in its capacity as Manager, at the applicable Closing and as more particularly set forth in the Management Agreement to be executed at such Closing, its Percentage Interest of any Leasing Commission (as defined in such Management Agreement) payable to Cedar with respect to any leases or renewals thereof entered into by and between a Tenant and the applicable Property Owner at any time during the period between the date hereof and the applicable Closing Date (as if such Management Agreement had been effective during such period), provided that such Tenant has paid its first month's rent on or prior to the applicable Closing Date and provided further that this subparagraph (c) shall not apply to any leases or renewals thereof for which Earn-Out Proceeds shall be earned by Cedar pursuant to Section 4, it being understood and agreed that this subparagraph (c) shall apply to any leases or renewals thereof for which Earn-Out Proceeds shall not be earned by Cedar pursuant to Section 4.
- (d) Except as set forth in Section 39 below, each party shall pay the cost of the fees and disbursements of its attorneys in connection with this Agreement.

The provisions of this Section 6 shall survive the Closings.

7. Blue Mountain Development Parcel.

The parties acknowledge and agree that the Blue Mountain Development Parcel is not intended to be included in the Blue Mountain Closing and no portion of the Consideration for Blue Mountain is attributable to the Blue Mountain Development Parcel. Accordingly, notwithstanding anything to the contrary contained in this Agreement, the parties covenant and agree as follows:

- If Cedar shall be able to legally separate the Blue Mountain Shopping Center Parcel and the (a) Blue Mountain Development Parcel and deed fee title to the Blue Mountain Development Parcel (the "Blue Mountain Separation") to its affiliate at or prior to the Blue Mountain Closing, the parties shall cause the Blue Mountain Property Owner and the owner of the Blue Mountain Development Parcel at the Blue Mountain Closing or, at Cedar's election, at any time thereafter, to enter into a development declaration and reciprocal easement agreement in a form reasonably acceptable to the parties that shall provide for, inter alia, (i) such exclusive and nonexclusive easements as shall be reasonably necessary for the siting, designing, constructing, installing, repairing, restoring, maintaining, improving, demolishing, adding to or replacing of, all or any portion of the improvements now existing or thereinafter located on each property, and (ii) such other mutual rights and obligations as shall be reasonably necessary for the ordinary maintenance and operation of both properties, including, without limitation, customary insurance requirements and mutual indemnities from credit-worthy entities (it being acknowledged that the Partnership is a credit-worthy entity for this purpose) that are reasonably satisfactory to the parties (the "Blue Mountain REA").
- (b) If, for any reason, the Blue Mountain Separation shall not have occurred as of the time that Cedar satisfies all other conditions set forth herein for the Blue Mountain Closing, then (i) the same shall not constitute a default by Cedar under this Agreement or a failure of a condition precedent to either party's obligation to proceed with the Blue Mountain Closing under this Agreement, (ii) the parties shall proceed to close the Transaction for Blue Mountain as and when provided in this Agreement, and the Blue Mountain Development Parcel shall be included in such Transaction, (iii) at the Blue Mountain Closing, the parties shall cause the Blue Mountain Property Owner to enter into a ground lease (the "Blue Mountain Ground Lease") demising the Blue Mountain Development Parcel to an affiliate of Cedar that is not an Owner in a form reasonably acceptable to the parties that shall provide for, inter alia (A) a term of ninety-nine (99) years (or, at Cedar's election, any lesser term); (B) an annual fixed rent of One Hundred Dollars (\$100) for the entirety of the term thereof; (C) the lessee to pay all costs, expenses and charges of every kind and nature relating solely to the Blue Mountain Development Parcel, including, without limitation, its to be agreed upon proportionate share of real estate taxes and other impositions attributable thereto; (D) the lessor to pay all amounts payable under any financing of the Blue Mountain Shopping Center Parcel (and the improvements thereon) that may also encumber the Blue Mountain Development Parcel (except as provided in clause (iv) below); (E) the right of the lessee thereunder to cause the Blue Mountain Separation to occur or, alternatively, at the lessee's sole election, to cause the Blue Mountain Development Parcel and the Blue Mountain Shopping Center Parcel (and the improvements thereon) to be submitted to a condominium regime whereby each property shall constitute a separate condominium unit and the documents governing such a condominium regime shall provide, inter alia, for substantially the same terms and conditions as are provided in the Blue Mountain REA, in forms otherwise reasonably acceptable to the parties (the "Blue Mountain Condo Conversion"), in each case, pursuant to and in accordance with documentation reasonably acceptable to the parties; (F) an option in favor the lessee thereunder to purchase fee title to the Blue Mountain Development Parcel or the condominium unit comprised of the Blue Mountain Development Parcel, as the case may be, for One Hundred Dollars (\$100) at any time following the occurrence of the Blue Mountain Separation or the Blue Mountain Condo Conversion, as applicable; (G) without restriction as to use or subleasing; and (H) the right of the lessee thereunder to record a memorandum of ground lease, and (iv) the parties shall not permit the Blue Mountain Property Owner to enter into any financing unless the same shall permit the Blue Mountain Separation, the Blue Mountain Condo Conversion and the release of the Blue Mountain Development Parcel from the lien of such financing, in each case, without the payment of any fee or expense (other than reimbursement of the applicable lender's out-of-pocket expenses). Simultaneously with the execution and delivery of the Blue Mountain Ground Lease or, at Cedar's election, at any time thereafter, the parties thereto shall execute and deliver the Blue Mountain REA.
- (c) Cedar shall pay the following costs and expenses associated with the Blue Mountain Separation, the Blue Mountain REA, the Blue Mountain Ground Lease and/or the Blue Mountain Condo Conversion: (i) any and all state and local recording charges and fees, if any; (ii) the reasonable out-of-pocket expenses of RioCan and any lender under a financing of Blue Mountain in connection with the Blue Mountain Separation, the Blue Mountain Condo Conversion, the Blue Mountain Ground Lease and/or the release of the Blue Mountain Development Parcel from the lien of such financing, including all attorneys' fees, consultants' fees and costs and expenses associated with survey updates, record searches, title examinations and updated mortgagee title insurance policies (including endorsements thereto), if any; and (iii) any and all Transfer Taxes.
- (d) The parties shall, and shall cause the Blue Mountain Property Owner and the owner or lessee of the Blue Mountain Development Parcel (as applicable) to cooperate with all reasonable requests of any party hereto in order to effectuate or otherwise accomplish the purposes of this Section 7, including, without limitation, executing and delivering such further documents and instruments as shall be reasonably required in connection therewith.
- (e) The terms and provisions of this Section 7 shall survive the Closings.

8. Investigations.

RioCan hereby acknowledges that it has been afforded full access to the Due Diligence Site and the Properties, and that it has performed and completed its due diligence examinations, reviews and inspections of all matters pertaining to the Transactions, the Owners, the Interests, the Columbus Crossing Preferred Interests, the Properties, the Loans, and the Columbus Crossing Preferred Partner Loan, including, without limitation, the Information (its "Investigations") prior to the date of this Agreement except for the completion of the

Supplemental Testing.

RioCan shall have until 5:00 p.m. (Eastern time) on (i) November 10, 2009, with respect to the Supplemental Testing for Shaw's Plaza and (ii) November 24, 2009, with respect to the Supplemental Testing for Sunset Crossing, in each case, TIME BEING OF THE ESSENCE (the period of time commencing upon the date hereof and continuing through and including such time on such date being herein called the "Supplemental Due Diligence Period"), within which to complete the Supplemental Testing, which shall at all times be subject to RioCan's compliance with the provisions of this Section 8 and Section 9 hereof. Any entry upon the applicable Properties shall be made or performed during Cedar's normal business hours and at the sole risk and expense of RioCan, and shall not materially interfere with the activities on or about the applicable Properties, their respective Tenants and their employees and invitees. During the Supplemental Due Diligence Period, Cedar shall provide RioCan with reasonable access to the applicable Properties upon reasonable advance notice for the sole purpose of performing the Supplemental Testing. In connection with the foregoing, RioCan shall:

- (a) promptly repair any damage to the applicable Properties resulting from any such Supplemental Testing and replace, refill and regrade any holes made in, or excavations of, any portion of the applicable Properties used for the Supplemental Testing so that each of the applicable Properties shall be substantially in the same condition that they existed in prior to the Supplemental Testing;
- (b) fully comply with all laws applicable to the Supplemental Testing;
- (c) permit Cedar to have a representative present during the Supplemental Testing;
- (d) take all actions and implement all protections reasonably necessary to ensure that the Supplemental Testing and the equipment, materials, and substances generated, used or brought onto the applicable Properties in connection with the Supplemental Testing, pose no threat to the safety or health of persons or the environment, and cause no damage to the applicable Properties or other property of Cedar or other persons;
- (e) furnish to Cedar, at no cost or expense to Cedar, copies of all studies and reports relating to the Supplemental Testing, which RioCan shall obtain promptly after RioCan's receipt of same:
- (f) maintain or cause to be maintained, at RioCan's expense, a policy of commercial general liability insurance, with a broad form contractual liability endorsement and with a combined single limit of not less than \$2,000,000 per occurrence for bodily injury and property damage, automobile liability coverage including owned and hired vehicles with a combined single limit of \$2,000,000 per occurrence for bodily injury and property damage, and an excess umbrella liability policy for bodily injury and property damage in the amount of \$5,000,000, insuring RioCan, Cedar, CSCI and Cedar-Raynham, LLC, as additional insureds, against any injuries or damages to persons or property that may result from or are related to Supplemental Testing, and/or any and all other activities undertaken by RioCan and/or the RioCan Representatives, all of which insurance shall be on an "occurrence form" and otherwise in such forms acceptable to Cedar and with an insurance company acceptable to Cedar, and deliver a copy of such insurance policy to Cedar prior to the first entry on the applicable Properties; and
- (g) not permit the Supplemental Testing or any other activities undertaken by RioCan or the RioCan Representatives to result in any liens, judgments or other encumbrances being filed or recorded against any of the applicable Properties, and RioCan shall, at its sole cost and expense, promptly discharge of record any such liens or encumbrances that are so filed or recorded (including, without limitation, liens for services, labor or materials furnished).

If, on or before the expiration of the Supplemental Due Diligence Period, the Supplemental Testing for either of the applicable Properties shall evidence a physical or environmental condition not previously known to RioCan that, in its reasonable opinion, can reasonably be expected to have a material adverse impact on the value or operation of the Property that is the subject of such Supplemental Testing and RioCan shall determine that it no longer intends to acquire such Property as a result thereof, then RioCan shall promptly notify Cedar of such determination in writing on or before 5:00 p.m. (Eastern time) on the date that the Supplemental Due Diligence Period shall expire (each such notice being herein called the "Supplemental Due Diligence Termination Notice"), which notice shall contain a reasonably detailed description of such condition and reasonable evidence supporting RioCan's determination of material adverse impact as aforesaid, whereupon the Consideration shall be reduced by the amount of the applicable Allotted Consideration, and the parties hereto shall be relieved of all further liability and responsibility under this Agreement with respect to the applicable Transaction, except for any obligation expressly provided to survive a termination of this Agreement. In the event that RioCan shall fail to deliver the applicable Supplemental Due Diligence Termination Notice to Cedar on or before 5:00 p.m. (Eastern time) on the date that the Supplemental Due Diligence Period shall expire, TIME BEING OF THE ESSENCE, RioCan shall be deemed to have agreed that the foregoing matters are acceptable to RioCan and that it intends to proceed with the acquisition of the applicable Properties without a reduction in, or an abatement of or credit against, the applicable Allocated Consideration (and, thereafter, RioCan shall have no further right to terminate this Agreement pursuant to this Section 8).

9. Indemnification.

RioCan shall indemnify, defend and hold harmless the Cedar Related Parties from and against any and all claims, demands, causes of action, losses, damages, liabilities, costs and expenses (including, without limitation, reasonable attorneys' fees and disbursements and costs of enforcement of the indemnification obligation hereunder), suffered or incurred by Cedar or any Cedar Related Party, and arising out of or in connection with (i) the entry by RioCan and/or the RioCan Representatives upon any of the Properties (whether conducted prior to or after the date hereof), (ii) any of its Investigations or other activities conducted thereon by RioCan or the

RioCan Representatives, (iii) any liens or encumbrances filed or recorded against any Property as a consequence of its Investigations and/or (iv) any and all other activities undertaken by RioCan or the RioCan Representatives with respect to the Properties. The foregoing obligation to indemnify, defend and hold harmless shall not include any claims, demands, causes of action, losses, damages, liabilities, costs or expenses (including, without limitation, attorneys' fees and disbursements) that result solely from the mere discovery, by RioCan or the RioCan Representatives, of existing conditions on any Property during its Investigations.

The provisions of this Section 9 shall survive the Closings and/or any termination of this Agreement.

10. Confidentiality.

The Confidentiality Agreement remains in full force and effect and is incorporated herein by reference as if fully set forth herein, provided that (a) RioCan shall be deemed to have all the rights and obligations of RioCan REIT set forth therein, (b) Cedar shall be deemed to have all the rights and obligations of CSCI set forth therein, (c) a default by or breach by RioCan REIT, RioCan, any other RioCan Related Party and/or any RioCan Representative under the Confidentiality Agreement (including, but not limited to, as incorporated herein by reference) shall constitute a default by RioCan under this Agreement (including, but not limited to, as incorporated herein by reference) shall constitute a default by Codar under this Agreement (including, but not limited to, as incorporated herein by reference) shall constitute a default by Cedar under this Agreement.

11. Undertaking.

Cedar shall use commercially reasonable efforts to locate and deliver to RioCan the materials listed on Exhibit H attached hereto and made a part hereof.

12. Lender Approval.

With respect to each of the Transactions (other than the Transactions involving Blue Mountain and Sunset Crossing), Cedar shall use commercially reasonable efforts commencing promptly after the date hereof to obtain from the Lenders their respective written approval or agreement, in a form reasonably acceptable to RioCan of (i) the Reorganizations, if applicable, and the Transactions, (ii) the applicable Management Agreement, (iii) a one time transfer on or after the applicable Closing of either (A) forty-nine percent (49%) or a lesser amount of the direct or indirect interests in RioCan to a single Institutional Investor or (B) RioCan's entire partnership interest in the Partnership to a U.S. entity wholly owned and Controlled by RioCan REIT and a single Institutional Investor and at least fifty-one percent (51%) owned, directly or indirectly, by RioCan REIT and not more than forty-nine percent (49%) owned, directly or indirectly, by such Institutional Investor, and (iv) a transfer pursuant to the buy/sell provisions of the Partnership Agreement (collectively, with any other related approvals required pursuant to the applicable Loan Documents the "Loan Approvals"). Notwithstanding the foregoing, the refusal of a Lender to pre-approve or otherwise permit any of the following without the consent of such Lender shall not be grounds for RioCan to claim that a Loan Approval is not reasonably acceptable to RioCan (or that a condition precedent to RioCan's obligation to close the applicable Transaction has not been satisfied): (w) any transfer of a partnership interest in the Partnership from Cedar to RioCan (or any affiliate of either of the foregoing) after Closing, (x) any transfer of up to forty-nine percent (49%) of the stock in RioCan to a third party Institutional Investor on or after Closing, (y) any transfer of RioCan's entire partnership interest in the Partnership to a U.S. entity wholly owned and Controlled by RioCan REIT and a single Institutional Investor on or after Closing and/or (z) any transfer pursuant to the buy/sell and/or right of first refusal provisions of the Partnership Agreement. In addition, if any Lender shall condition its Loan Approval upon modifying the applicable organizational documents of the Owners, any other agreements to be delivered in connection herewith and/or the terms of the contemplated Reorganizations in order for the same to comply with the single purpose entity and/or bankruptcy remoteness requirements of the applicable Loan Documents and/or other reasonable requirements of such Lender, in each case, which modifications do not materially increase the liabilities (including, without limitation, potential tax liabilities) or materially limit the rights or economic benefits of Cedar or RioCan under this Agreement or any other agreements to be delivered in connection herewith, the same shall constitute neither a default by Cedar under this Agreement nor the failure of a condition precedent to the obligation of any party to close hereunder, and the parties shall use commercially reasonable efforts to satisfy any such requirements to the satisfaction of such Lender. Cedar shall request that the documents evidencing a Loan Approval contain a statement from the Lender identifying, in writing, the outstanding principal balance and interest rate of the applicable Loan and whether, to Lender's knowledge, any default exists under the applicable Loan Documents (the "Loan Estoppel Statement"). Cedar and RioCan agree to use commercially reasonable efforts to cooperate with each other in connection with the foregoing (including, without limitation, promptly furnishing to the Lenders all information and documents (financial and otherwise) which may be required under the Loan Documents or otherwise reasonably requested by the Lenders). For avoidance of doubt, failure by Cedar to obtain (x) any Loan Approval in the manner provided herein shall not constitute a default by Cedar under this Agreement, but shall constitute the mere failure of a condition precedent as more particularly set forth in Section 18 below and/or (y) any Loan Estoppel Statement in the manner provided herein shall constitute neither a default by Cedar under this Agreement nor the failure of a condition precedent to the obligation of any party to close hereunder.

- (b) If, with respect to one (1) or more of the applicable Properties (each, a 'Subject Property'), necessary Loan Approvals shall not have been obtained by Cedar and RioCan prior to 5:00 P.M. (Eastern time) on July 26, 2010 (the "Loan Approval Deadline"), then Cedar shall have the right, in its sole and absolute discretion, exercisable by delivery of written notice to RioCan to either (x) extend the Loan Approval Deadline with respect to the Subject Property(ies) by a period not to exceed, in the aggregate, thirty (30) days (the "Extension Period") and, if necessary, extend the Closing of the related Transaction(s) (the "Subject Transaction(s)") in connection therewith, or (y) remove the Interests associated with the Subject Property(ies) (the "Subject Interests") from the Interests being conveyed pursuant to this Agreement, in which case this Agreement shall terminate as to the Subject Transaction, the Deposit shall be refunded to RioCan (if no other Closing with respect to a Property that is not a Subject Property remains outstanding) and the Consideration shall be reduced by the amount of the applicable Allotted Consideration, whereupon the parties hereto shall be relieved of all further liability and responsibility under this Agreement with respect to the Subject Interests, the Subject Property and the Subject Transaction (except for any obligation expressly provided to survive a termination of this Agreement). If Cedar shall make an election under clause (x) of this Section 12(b), then the following shall apply:
 - (i) The parties shall proceed with the Closing of any other Transaction that is not a Subject Transaction in accordance with the terms of this Agreement.
 - (ii) If Cedar does not obtain any or all outstanding Loan Approval(s) by the expiration of the Extension Period, then this Agreement shall automatically terminate with respect to the Subject Transaction only, in which case the Consideration shall be reduced by the amount of the applicable Allotted Consideration, the Deposit shall be refunded to RioCan (if no other Closing with respect to a Property that is not a Subject Property remains outstanding), and the parties hereto shall be relieved of all further liability and responsibility under this Agreement with respect to the Subject Interests, the Subject Property and the Subject Transaction, except for any obligation expressly provided to survive a termination of this Agreement.

13. Representations and Warranties of Cedar.

- (a) Cedar hereby makes the following representations and warranties to RioCan:
 - (i) Due Authority. This Agreement and all agreements, instruments and documents herein provided to be executed by Cedar will be duly authorized, executed and delivered by and binding upon Cedar as of each Closing Date. As of each Closing Date, this Agreement will constitute the legal, valid and binding obligations of Cedar and shall be enforceable against Cedar in accordance with its terms, except as such enforceability may be limited by (i) bankruptcy, insolvency or other similar laws affecting creditor's rights generally and (ii) general principles of equity. Cedar is a limited partnership, duly organized and validly existing and in good standing under the laws of the State of Delaware and, as of the Closing Date, will be duly authorized and qualified to do all things required of it under this Agreement and all agreements, instruments and documents herein provided to be executed by Cedar. Each of the Owners is, on the date of this Agreement, a limited liability company or limited partnership, duly formed and validly existing and in good standing under the laws of the State or Commonwealth of its formation and, with respect to each Property Owner, is in good standing under the laws of the State or Commonwealth in which its Property is located. On the applicable Closing Date, each of the Owners (other than Columbus Crossing Property Owner) will be a limited partnership or a limited liability company, duly formed and validly existing and in good standing under the laws of the State of Delaware and, with respect to each Property Owner, in good standing in the State or Commonwealth in which its Property is located. On the applicable Closing Date, Columbus Crossing Property Owner will be a limited partnership, duly formed and validly existing and in good standing under the laws of the Commonwealth of Pennsylvania.
 - (ii) No Options: Except as set forth in the Leases, as of the date of this Agreement, neither Cedar nor the Property Owners have entered into any agreement, option, understanding or commitment, or granted any right or privilege (whether by law, pre-emptive or contractual) capable of becoming an agreement, option or commitment with any Person (other than RioCan), for the purchase or ground lease from Cedar and/or any Property Owner of any of the Properties or any rights or interest therein, which remains outstanding. For avoidance of doubt, the foregoing representation shall in no event be interpreted to cover any agreement, option, understanding, commitment right or privilege with respect to all or any portion of the direct or indirect interests in Cedar or CSCI.

- (iii) Pre-RioCan Owner Agreements; Assets. Annexed hereto as Exhibit C and made a part hereof is a true and complete list (in all material respects) of the Pre-RioCan Owner Agreements of each Owner as modified and/or amended through the date of this Agreement, true and correct copies (in all material respects) of which have been delivered or made available to RioCan. As of the date of this Agreement, the Pre-RioCan Owner Agreements of each Owner, as listed in Exhibit C, are in full force and effect and have not been modified, supplemented or amended. Since its inception, no Owner has owned, as applicable, assets other than the applicable Property or Owner or engaged in any business other than the ownership and operation of the applicable Property or other Owner, and the applicable Owners have no liabilities (contingent or otherwise) other than liabilities incurred in connection with the ownership and operation of the applicable Property or other Owner.
- (iv) Interests. Immediately prior to each Closing, Cedar (or its wholly-owned direct or indirect subsidiary) shall own, legally and beneficially, all of the Interests included in such Closing free of all security interests, liens, encumbrances and pledges. There are no options, subscriptions, warrants, calls, preemptive rights, rights of first refusal or other rights, commitments or arrangements, written or oral, outstanding with respect to the Interests or any unissued equity interests in the Owners or any security convertible into or exchangeable or exercisable for any equity interests in the Owners, in each case, other than with respect to the Columbus Crossing Preferred Interests and the terms and conditions of the Loan Documents, the Blue Mountain Line of Credit and the Sunset Crossing Line of Credit. Except for the Interests and the Columbus Crossing Preferred Interests, there are no other equity interests in any Owner held by any Person. Upon delivery to the REIT Property Subsidiary in accordance with Schedule 2 attached hereto, good and valid title to the Interests will pass to the REIT Property Subsidiary, free and clear of any and all security interests, liens, encumbrances and pledges.
- (vi) Taxes. All tax returns that have been required to be filed with respect to the business, operations and assets of each Owner have been filed. All taxes, charges, fees, levies or other assessments, including, without limitation, income, real and personal property taxes, imposed by any Governmental Authority having jurisdiction that are due and payable as of the applicable Closing Date with respect to the business, operations and assets of the applicable Owner, have been paid or shall be paid as of the applicable Closing Date. As of the date of this Agreement, there are no pending audits with respect to taxes payable by the Owners. Immediately following the applicable Reorganization and as of the applicable Closing Date, each applicable Owner (other than the Columbus Crossing Property Owner) shall be classified as a disregarded entity for federal income tax purposes.
- Leases. Cedar has no knowledge of any leases to which any Property Owner is a (vii) party affecting any portion of the applicable Property that will be in force on the applicable Closing Date other than the Leases. As of the date of this Agreement, to the knowledge of Cedar, (x) the Leases are in full force and effect and have not been amended except as set forth in the Lease Exhibit, and (y) the Lease Exhibit is true and correct in all material respects. True and complete (in all material respects) copies of the Leases have been provided to RioCan on the Due Diligence Site. As of the date of this Agreement, except as noted on Schedule 10, Cedar has no knowledge of any material default by any party to any Lease that remains uncured (including, without limitation, violations of (A) representations that would give rise to a termination right under the applicable Lease and (B) radius restrictions). To the knowledge of Cedar, the rent rolls provided in the Due Diligence Site are true and correct in all material respects as of the date of such rent rolls. To Cedar's knowledge as of the date of this Agreement, no Major Tenant has requested in writing Cedar's consent to the assignment or surrender of such Major Tenant's lease, which consent request remains outstanding. With respect to each Blue Mountain Lease executed by a Tenant prior to the date of this Agreement, to the knowledge of Cedar, no such Tenant has an outstanding right (if any) to terminate its Lease or receive a rent reduction by reason of the landlord's failure to complete the initial construction of Blue Mountain as and when required thereunder.
- (viii) Ground Leases: There are no ground leases pursuant to which a Property Owner, as lessee, leases all or any portion of any Property from a third party, as lessor, other than the Ground Leases. As of the date of this Agreement, to the knowledge of Cedar, the Ground Leases are in full force and effect and have not been amended. True and complete (in all material respects) copies of the Ground Leases have been provided to RioCan on the Due Diligence Site. As of the date of this Agreement, Cedar has no knowledge of any material default by any party to any Ground Lease that remains uncured.
- (ix) Environmental Claims: Except as disclosed in the environmental reports provided to RioCan on the Due Diligence Site, to the knowledge of Cedar, as of the date of this Agreement, neither Cedar nor any Property Owner has received written notice of any material Environmental Claim attributable to the period of Cedar's or such Property Owner's ownership of the applicable Property that remains uncured or unsatisfied in any material respect.

- (x) Zoning: To the knowledge of Cedar, as of the date of this Agreement, no written notice has been received by Cedar or any Property Owner of any pending or threatened change to, any zoning by-law materially affecting all or any portion of any Property, or any local improvements made by any authority and chargeable (and not paid) to all or any portion of any Property, in any event, that would have a material adverse effect on the value, use or operation of such Property.
- (xi) Leasing Agents and Commissions: To the knowledge of Cedar as of the date of this Agreement, there are no outstanding agreements with leasing agents in respect of leasing space in the Property nor are there any outstanding commissions payable to any brokers with respect to any Leases, except (x) to Cedar or its affiliate at the applicable Closing pursuant to Section 6(c) hereof, (y) such outstanding commissions that will remain the sole obligation of Cedar pursuant to Section 19(b) hereof and (z) such outstanding commissions that shall be the responsibility of the Partnership pursuant to Section 19 hereof.
- (xii) Intellectual Property: To the knowledge of Cedar as of the date of this Agreement, neither Cedar nor any Property Owner has granted any licenses, rights or interests in the Intellectual Property, none of the Intellectual Property is subject to any licenses, rights or interests, and no payments are made by or to Cedar in respect of the use of the Intellectual Property.
- (xiii) Service Contracts. Cedar has no knowledge of any service or equipment leasing contracts to which any Property Owner is a party affecting any portion of the applicable Property which will be in force on the applicable Closing Date other than the Service Contracts. As of the date of this Agreement, to the knowledge of Cedar, (x) all of the material Service Contracts are in full force and effect and (y) true and complete (in all material respects) copies of the Service Contracts listed on Schedule 4 have been delivered to RioCan on the Due Diligence Site. As of the date of this Agreement, Cedar has no knowledge of any material default by any party to any Service Contract that remains uncured.
- (xiv) Employees. As of the date of this Agreement and the applicable Closing Date, the Owners have no, and shall not have any, employees or former employees.
- (xv) <u>Litigation</u>. As of the date of this Agreement, except as set forth in Schedule 7 attached hereto, there is no material pending (for which any Owner has been served) or, to Cedar's knowledge, material threatened litigation, claim or proceeding against any Property or against any Owner other than claims made in the ordinary course of the business of owning and operating the Properties and the Property Owners, as applicable, which are covered by insurance maintained by Cedar and/or the applicable Owner. To Cedar's knowledge as of the date of this Agreement, there is not outstanding against Cedar, any Owner or any Property any judgment, decree, injunction, rule or order of any court, governmental department, commission, agency or arbitrator which materially and adversely affects the Properties.
- (xvi) <u>No Insolvency</u>. Neither Cedar nor any Owner is or shall be on the applicable Closing Date, a debtor in any state or federal insolvency, bankruptcy or receivership proceeding.
- (xvii) Non-Foreign Person. Neither Cedar nor any Owner is or shall be as of the applicable Closing Date, a "foreign person" as defined in Section 1445 of the Internal Revenue Code, as amended.
- Columbus Crossing Loan. The Property commonly known as Columbus Crossing, located in Philadelphia, Pennsylvania is currently encumbered by a mortgage loan in the original principal amount of \$17,000,000 made by Susquehanna Bank, a Pennsylvania banking corporation (the "Columbus Crossing Loan") to the applicable Property Owner. As of the date of this Agreement, (x) to Cedar's knowledge, the documents and instruments identified on Schedule 8 attached hereto constitute all of the material documents and instruments delivered in connection with the Columbus Crossing Loan (the "Columbus Crossing Loan Documents"), true and complete (in all material respects) copies of which have been provided to RioCan on the Due Diligence Site; (y) to Cedar's knowledge, the Columbus Crossing Loan Documents are in full force and effect and have not been amended except as set forth on Schedule 8 attached hereto, and (z) to Cedar's knowledge, the applicable Property Owner is not in material default of, and has not received written notice from the applicable Lender of any uncured default under, any of such Property Owner's material obligations under the Columbus Crossing Loan Documents. To the knowledge of Cedar, as of the applicable Closing Date, the outstanding principal amount of the Columbus Crossing Loan set forth on the applicable Settlement Statement shall be the true and correct outstanding principal amount of the Columbus Crossing Loan as of the applicable Closing Date.

- (xix) Franklin Village Loan. The Property commonly known as Franklin Village, located in Franklin, Massachusetts is currently encumbered by a mortgage loan in the original principal amount of \$43,500,000 made by Eurohypo AG, New York Branch, a New York branch of a German banking corporation (as subsequently assigned, the "Franklin Village Loan") to the applicable Property Owner. As of the date of this Agreement, (x) to Cedar's knowledge, the documents and instruments identified on Schedule 8 attached hereto constitute all of the material documents and instruments delivered in connection with the Franklin Village Loan (the "Franklin Village Loan Documents"), true and complete (in all material respects) copies of which have been provided to RioCan on the Due Diligence Site; (y) to Cedar's knowledge, the Franklin Village Loan Documents are in full force and effect and have not been amended except as set forth on Schedule 8 attached hereto, and (z) to Cedar's knowledge, the applicable Property Owner is not in material default of, and has not received written notice from the applicable Lender of any uncured default under, any of such Property Owner's material obligations under the Franklin Village Loan Documents. To the knowledge of Cedar, as of the applicable Closing Date, the outstanding principal amount of the Franklin Village Loan set forth on the applicable Settlement Statement shall be the true and correct outstanding principal amount of the Franklin Village Loan as of the applicable Closing Date.
- (xx) Loyal Plaza Loan. The Property commonly known as Loyal Plaza, located in Williamsport, Pennsylvania is currently encumbered by a mortgage loan in the original principal amount of \$14,000,000 made by Lehman Brothers Bank, FSB and subsequently assigned to LaSalle Bank National Association, as Trustee for the Registered Holders of LB-UBS Commercial Mortgage Trust 2001-C3, Commercial Mortgage Pass-Through Certificates, Series 2001-C3 (the "Loyal Plaza Loan") to the applicable Property Owner. As of the date of this Agreement, (x) to Cedar's knowledge, the documents and instruments identified on Schedule 8 attached hereto constitute all of the material documents and instruments delivered in connection with the Loyal PlazaLoan (the "Loyal Plaza Loan Documents"), true and complete (in all material respects) copies of which have been provided to RioCan on the Due Diligence Site; (y) to Cedar's knowledge, the Loyal Plaza Loan Documents are in full force and effect and have not been amended except as set forth on Schedule 8 attached hereto, and (z) to Cedar's knowledge, the applicable Property Owner is not in material default of, and has not received written notice from the applicable Lender of any uncured default under, any of such Property Owner's material obligations under the Loyal Plaza Loan Documents. To the knowledge of Cedar, as of the applicable Closing Date, the outstanding principal amount of the Loyal Plaza Loan set forth on the applicable Settlement Statement shall be the true and correct outstanding principal amount of the Loyal Plaza Loan as of the applicable Closing
- (xxi) Bridgeport Loan. The Property commonly known as Shop N Shop Plaza, located in Bridgeport, Connecticut is currently encumbered by a mortgage loan in the original principal amount of \$7,000,000 made by Morgan Stanley Mortgage Capital Inc., a New York corporation, as subsequently assigned to LaSalle Bank National Association, as Trustee and Custodian for Bear Stearns Commercial Mortgage Securities Inc., Commercial Mortgage Pass-Through Certificates, Series 2007-Top 26 (the "Bridgeport Loan") to the applicable Property Owner. As of the date of this Agreement, (x) to Cedar's knowledge, the documents and instruments identified on Schedule 8 attached hereto constitute all of the material documents and instruments delivered in connection with the Bridgeport Loan (the "Bridgeport Loan Documents"), true and complete (in all material respects) copies of which have been provided to RioCan on the Due Diligence Site; (y) to Cedar's knowledge, the Bridgeport Loan Documents are in full force and effect and have not been amended except as set forth on Schedule 8 attached hereto, and (z) to Cedar's knowledge, the applicable Property Owner is not in material default of, and has not received written notice from the applicable Lender of any uncured default under, any of such Property Owner's material obligations under the Bridgeport Loan Documents. To the knowledge of Cedar, as of the applicable Closing Date, the outstanding principal amount of the Bridgeport Loan set forth on the applicable Settlement Statement shall be the true and correct outstanding principal amount of the Bridgeport Loan as of the applicable Closing Date.

- (xxii) Shaw's Plaza Loan. The Property commonly known as Shaw's Plaza, located in Raynham, Massachusetts is currently encumbered by a mortgage loan in the original principal amount of \$14,200,000 made by Bear Stearns Commercial Mortgage, Inc., a New York corporation, as subsequently assigned to LaSalle Bank National Association, as Trustee for Bear Stearns Commercial Mortgage Securities Inc., Commercial Mortgage Pass-Through Certificates, Series 2004-Top 14 (the "Shaw's Plaza Loan") to the applicable Property Owner. As of the date of this Agreement, (x) to Cedar's knowledge, the documents and instruments identified on Schedule 8 attached hereto constitute all of the material documents and instruments delivered in connection with the Shaw's Plaza Loan (the "Shaw's Plaza Loan Documents"), true and complete (in all material respects) copies of which have been provided to RioCan on the Due Diligence Site; (y) to Cedar's knowledge, the Shaw's Plaza Loan Documents are in full force and effect and have not been amended except as set forth on Schedule 8 attached hereto, and (z) to Cedar's knowledge, the applicable Property Owner is not in material default of, and has not received written notice from the applicable Lender of any uncured default under, any of such Property Owner's material obligations under the Shaw's Plaza Loan Documents. To the knowledge of Cedar, as of the applicable Closing Date, the outstanding principal amount of the Shaw's Plaza Loan set forth on the applicable Settlement Statement shall be the true and correct outstanding principal amount of the Shaw's Plaza Loan as of the applicable Closing Date.
- (xxiii) Columbus Crossing Preferred Partner Loan. The original principal amount of the Columbus Crossing Preferred Partner Loan is \$6,367,000, which has not been repaid. As of the date of this Agreement (x) to the knowledge of Cedar, the documents and instruments identified on Schedule 8 attached hereto constitute all of the material documents and instruments entered into in connection with the Columbus Crossing Preferred Partner Loan (the "Columbus Crossing Preferred Partner Loan Documents"), true and complete (in all material respects) copies of which have been provided to RioCan on the Due Diligence Site; (y) to the knowledge of Cedar, the Columbus Crossing Preferred Partner Loan Documents are in full force and effect and have not been amended except as set forth on Schedule 8 attached hereto, and (z) the Existing Columbus Crossing Preferred Partner Leaner is the holder of the Columbus Crossing Preferred Partner Loan Documents. To Cedar's knowledge, as of the date of this Agreement, neither the Existing Columbus Crossing Preferred Partner Lender nor the borrower is in material default under the Columbus Crossing Preferred Partner Loan Documents.
- (xxiv) <u>Loan Documents</u>. No Owner has entered into any loan documents secured in whole or in part by the applicable Property or Property Owner that will be binding on such Owner after the applicable Closing Date other than the Loan Documents.
- (xxv) Notices of Condemnation, Violations. To the knowledge of Cedar, neither Cedar nor any Property Owner has received written notice from any Governmental Authority having jurisdiction of (a) any condemnation of all or any part of the Properties as of the date of this Agreement or (b) any violations by any Property Owner of any zoning ordinance, fire codes, law or other legal requirement relating to the ownership of the Properties, which have not been corrected in all material respects, which are not the responsibility of Tenants and which have a material adverse effect on the value, use or operation of such Property.
- (xxvi) Reports: Cedar has provided to RioCan complete copies of all environmental and physical reports it has commissioned or in its possession or control in respect of the Properties.
- (xxvii) Exempt Assets: The Properties constitute exempt assets for purposes of the Hart-Scott-Rodino Antitrust Improvements Act of 1976 and Rules 802.2(h) and 802.5 thereunder.
- (xxviii) Material Title Contracts. As of the date of this Agreement, to Cedar's knowledge, (x) the Material Title Contracts are in full force and effect and (y) Cedar is not in default in any material respect with respect to any material obligation under any Material Title Contract.
- (xxix) Surveys. As of the date of this Agreement, to Cedar's knowledge, there has been no material change with respect to any material physical improvements disclosed on the Existing Surveys furnished to RioCan prior to the date of this Agreement (other than with respect to Blue Mountain), except for such matters that would not materially and adversely affect the ordinary operation of the Properties as currently operated or the value or marketability of the Properties.

- Financial Statements. Prior to the date hereof, Cedar has delivered to RioCan: (i) the (xxx) balance sheets of each applicable Property Owner as of December 31, 2007 and 2008 (collectively, the "Balance Sheets"); (ii) statements of operations, cash flows and owners' equity for the year ended December 31, 2008; and (iii) a balance sheet of each Property Owner as of June 30, 2009 (the foregoing financial statements, including any notes thereto and any related compilations, reviews and other reports issued by the Property Owners' accountants with respect thereto, the "Financial Statements"). The Financial Statements have been prepared from the books and records of each respective Property Owner in accordance with GAAP during the periods covered thereby (except as otherwise disclosed therein). To the knowledge of Cedar, the books and records of each Property Owner, all of which have been made available to RioCan before the date hereof, are true and complete in all material respects, have been maintained in accordance with sound business practices and accurately present and reflect in all material respects all of the transactions and actions therein described. To the knowledge of Cedar, no Property Owner has any material liabilities or obligations of a nature required by GAAP to be reflected on a balance sheet of such Property Owner, except (i) as disclosed, reflected or reserved against in the Balance Sheets and (ii) for liabilities and obligations incurred in the ordinary course of business since the date of the applicable Balance Sheet.
- (xxxi) Withholding Information. Cedar has not knowingly withheld any factual information or documentation regarding the Properties or the Interests which would make any of the representations and warranties contained herein untrue in any material respect
- (b) Knowledge of Cedar; References to "Owner". References to the "knowledge" of Cedar or words of similar import shall refer only to (i) the knowledge of Cedar of information actually and specifically set forth in written materials physically located in the files and property records maintained by Cedar at its offices and (ii) the current actual (as opposed to implied or constructive) knowledge of Leo S. Ullman and Brenda Walker and shall not be construed, by imputation or otherwise, to refer to the knowledge of Cedar or any parent, subsidiary or affiliate of Cedar or to any other officer, agent, manager, representative or employee of Cedar or to impose upon Leo S. Ullman or Brenda Walker any duty to investigate the matter to which such actual knowledge, or the absence thereof, pertains other than the duty of either Leo S. Ullman or Brenda Walker to cause inquiry, verbal or in writing, of the regional asset managers for the applicable Properties, with respect to Property specific representations contained in Section 13(a). Notwithstanding anything to the contrary contained in this Agreement, neither Leo S. Ullman nor Brenda Walker shall have any personal liability hereunder. Notwithstanding anything to the contrary contained in this Agreement, references to "Owner" or "Owners" in the representations and warranties made in this Section 13(a) as of the date of this Agreement shall refer to the Owner or Owners as and to the extent the same has or have been formed as of the date of this Agreement.
- Knowledge of RioCan. Notwithstanding anything to the contrary contained in this (c) Agreement, with respect to each Transaction, (i) if any of the representations or warranties of Cedar contained in this Agreement or in any document or instrument delivered in connection herewith are materially false or inaccurate, or Cedar is in material breach or default of any of its obligations under this Agreement that survive a Closing, and RioCan nonetheless closes such Transaction hereunder, then none of the Cedar Partners shall have any liability or obligation respecting such false or inaccurate representations or warranties or other breach or default (and any cause of action resulting therefrom shall terminate upon such Closing) in the event that either (x) on or prior to the applicable Closing, RioCan shall have had actual knowledge of the false or inaccurate representations or warranties or other breach or default, or (y) the accurate state of facts pertinent to such false or inaccurate representations or warranties or other breach or default was contained in any of the Information and (ii) to the extent the copies of the Leases, the Service Contracts, any estoppel certificates or any other such Information furnished to or otherwise obtained by RioCan prior to the applicable Closing contain provisions or information that are inconsistent with the foregoing representations and warranties, none of the Cedar Partners shall have any liability or obligation respecting such inconsistent representations or warranties (and RioCan shall have no cause of action with respect thereto), and such representations and warranties shall be deemed modified to the extent necessary to eliminate such inconsistency and to conform such representations and warranties to such Leases, Service Contracts and other Information.

- DISCLAIMER OF REPRESENTATIONS. EXCEPT AS SPECIFICALLY SET FORTH IN (d) THIS AGREEMENT, THE TRANSFER OF THE INTERESTS AND THE PROPERTIES HEREUNDER IS AND WILL BE MADE ON AN "AS IS", "WHERE IS," AND "WITH ALL FAULTS" BASIS, WITHOUT REPRESENTATIONS AND WARRANTIES OF ANY KIND OR NATURE, EXPRESS, IMPLIED OR OTHERWISE, INCLUDING ANY REPRESENTATION OR WARRANTY CONCERNING TITLE TO THE INTERESTS, PROPERTIES, THE PHYSICAL CONDITION OF THE PROPERTIES (INCLUDING THE CONDITION OF THE SOIL OR THE IMPROVEMENTS), ENVIRONMENTAL CONDITION OF THE PROPERTIES (INCLUDING THE PRESENCE OR ABSENCE OF HAZARDOUS SUBSTANCES ON OR AFFECTING THE PROPERTY), THE COMPLIANCE OF THE PROPERTIES OR THE OWNERS WITH APPLICABLE LAWS AND REGULATIONS (INCLUDING ZONING AND BUILDING CODES OR THE STATUS OF DEVELOPMENT OR USE RIGHTS RESPECTING THE PROPERTIES), THE FINANCIAL CONDITION OF THE PROPERTIES, THE OWNERS OR ANY OTHER REPRESENTATION OR WARRANTY RESPECTING ANY INCOME, EXPENSES, CHARGES, LIENS OR ENCUMBRANCES, RIGHTS OR CLAIMS ON, AFFECTING OR PERTAINING TO THE PROPERTIES, THE OWNERS, THE INTERESTS OR ANY PART THEREOF. EXCEPT AS EXPRESSLY PROVIDED IN SECTION 25 OF THIS AGREEMENT, RIOCAN ACKNOWLEDGES THAT PRIOR TO THE DATE OF THIS AGREEMENT RIOCAN HAS EXAMINED, REVIEWED AND INSPECTED ALL MATTERS WHICH IN THE JUDGMENT OF RIOCAN BEAR UPON THE PROPERTIES, THE INTERESTS AND THEIR VALUE AND SUITABILITY EXCEPT FOR THE COMPLETION OF THE SUPPLEMENTAL TESTING (WHICH RIOCAN ACKNOWLEDGES SHALL BE COMPLETED BY THE EXPIRATION OF THE SUPPLEMENTAL DUE DILIGENCE PERIOD). EXCEPT AS TO MATTERS SPECIFICALLY SET FORTH IN THIS AGREEMENT: (A) RIOCAN WILL ACQUIRE THE INTERESTS (INCLUDING AN INDIRECT INTEREST IN THE PROPERTIES) SOLELY ON THE BASIS OF ITS OWN PHYSICAL AND FINANCIAL EXAMINATIONS, REVIEWS AND INSPECTIONS AND (B) WITHOUT LIMITING THE FOREGOING, RIOCAN WAIVES ANY RIGHT IT OTHERWISE MAY HAVE AT LAW OR IN EQUITY, INCLUDING, WITHOUT LIMITATION, THE RIGHT TO SEEK DAMAGES FROM CEDAR IN CONNECTION WITH THE CONDITION OF THE PROPERTIES AND THE INTERESTS, INCLUDING ANY RIGHT OF CONTRIBUTION UNDER THE COMPREHENSIVE ENVIRONMENTAL RESPONSE COMPENSATION AND LIABILITY ACT. THE PROVISIONS OF THIS SECTION 13(d) SHALL SURVIVE THE CLOSINGS.
- (e) <u>Survival of Representations and Warranties of Cedar</u>. Notwithstanding anything to the contrary contained in this Agreement, all representations and warranties of Cedar contained in this Section 13 with respect to each Transaction and the related Property, Owner(s), and Interests shall survive the Closing of such Transaction for a period of one (1) year (except that the representations and warranties of Cedar contained in Section 13(a)(i), (iii), (iv) and (v) shall survive the Closing of the applicable Transaction for a period of two (2) years and the representations and warranties of Cedar contained in Section 13(a)(vi) shall survive the Closing until the expiration of the applicable statute of limitations). This Section 13(e) shall survive the Closings.

14. Representations and Warranties of RioCan.

- (a) RioCan does hereby make the following representations and warranties to Cedar:
 - (i) <u>Due Authority</u>. This Agreement and all agreements, instruments and documents herein provided to be executed by RioCan have been or by Closing will be, duly authorized, executed and delivered by and are binding upon RioCan. As of the Closing Date, this Agreement will constitute the legal, valid and binding obligations of RioCan and shall be enforceable against RioCan in accordance with its terms, except as such enforceability may be limited by (i) bankruptcy, insolvency or other similar laws affecting creditor's rights generally and (ii) general principles of equity. RioCan is a corporation validly existing and in good standing under the laws of the Delaware, and is duly authorized and qualified to do all things required of it under this Agreement and all agreements, instruments and documents herein provided to be executed by RioCan.
 - (ii) <u>Litigation</u>. To the knowledge of RioCan, there is no material pending or threatened litigation, claim or proceeding against RioCan.
 - (iii) No Insolvency. RioCan is not and as of the applicable Closing Date, RioCan will not be, a debtor in any state, federal or foreign insolvency, bankruptcy, receivership proceeding.

- (iv) OFAC. Neither RioCan nor any member, partner or shareholder of RioCan, nor to the knowledge of RioCan, any Person with actual authority to direct the actions of RioCan nor, to the knowledge of RioCan any other Persons holding any legal or beneficial interest whatsoever in RioCan (A) are named on any list of Persons and governments issued by OFAC pursuant to Executive Order 13224, as in effect on the date hereof, or any similar list known to RioCan or publicly issued by OFAC or any other department or agency of the United States of America (collectively, the "OFAC Lists"), (B) are included in, owned by, controlled by, knowingly acting for or on behalf of, knowingly providing assistance, support, sponsorship, or services of any kind to, or otherwise knowingly associated with any of the Persons referred to or described in the OFAC Lists, or (C) has knowingly conducted business with or knowingly engaged in any transaction with any Person named on any of the OFAC Lists or any Person included in, owned by, controlled by, acting for or on behalf of, providing assistance, support, sponsorship, or services of any kind to, or, to the knowledge of RioCan, otherwise associated with any of the Persons referred to or described in the OFAC Lists.
- (v) Conflicts. Neither the entry into nor the performance of this Agreement by RioCan will (i) violate or result in a breach under, or constitute a default under, any corporate charter, certificate of incorporation, by-law, partnership agreement, indenture, contract, permit, judgment, decree or order to which RioCan is a party or by which RioCan is bound, or (ii) except with respect to the Loan Approvals, require the consent of any third party other than as has already been obtained or is otherwise specifically set forth herein.
- (b) Notwithstanding anything to the contrary contained in this Agreement, all representations and warranties of RioCan contained in this Section 14 shall survive the Closings for a period of one (1) year (except that the representations and warranties of RioCan contained in Section 14(a)(i) and (v) shall survive the Closings for a period of two (2) years and the representations and warranties of Cedar contained in Section 14(a)(iv) shall survive the Closing until the expiration of the applicable statute of limitations, or if there is no applicable statute of limitations, then forever). This Section 14(b) shall survive the Closings.

15. Investment Representations, Etc.

- Cedar, for itself and for each Cedar Partner, and RioCan, each represents and warrants to the (a) other and each Owner, that (i) it is an "accredited investor" as that term is defined in the Securities Act and was not formed solely for the purpose of purchasing partnership interests in the Partnership (the "Partnership Interests"); (ii) as applicable, the Partnership Interests have been or are being acquired by it, directly or indirectly, pursuant to the Partnership Agreement as an investment for its own account with no intention of distributing or reselling such Partnership Interests in any transaction that would be in violation of the securities laws of the United States or of any state, subject however, to the rights of such purchasers at all times to sell or otherwise dispose of all or any part of the Partnership Interests under an effective registration statement under the Securities Act, or under an exemption from such registration available under the Securities Act and, subject, nevertheless, to the disposition of such purchaser's property being at all times within its control; (iii) it (A) has such knowledge and experience in financial and business matters that it is capable of evaluating the merits and risks of the investment in the Partnership Interests, (B) has had the opportunity to ask questions of and receive answers concerning such Owner and its investment in the Partnership Interests and to obtain any information necessary to verify the information obtained by it, and (C) is able to bear the economic risks of such investment; and (iv) it has full power and authority to own or acquire the Partnership Interests to be acquired by it directly or indirectly as set forth herein or in the Partnership Agreement.
- (b) Cedar, for itself and for each Cedar Partner, and RioCan each acknowledges that: (i) the offering of the Partnership Interests has not been, and will not be, registered with the Commission under and pursuant to the Securities Act; (ii) the Partnership Interests have not been qualified for sale in any state under applicable state securities or Blue Sky Laws; (iii) in purchasing the Partnership Interests directly or indirectly it must bear the economic risks of the investment for an indefinite period of time because the Partnership Interests cannot be sold unless the offering of such Partnership Interests is subsequently registered under that Securities Act or an exemption from such registration is available; (iv) with respect to the tax and other legal consequences of an investment in the Partnership Interests, it is relying solely upon advice of its own tax and legal advisors; and (v) the Partnership Agreement and any other evidence of ownership of Partnership Interests will bear a legend reflecting the unregistered and restricted nature of the Partnership Interests; provided, however the foregoing Sections 15(a) and 15(b) are subject to and do not derogate from the reliance by each of RioCan and Cedar on the truth and accuracy of the express representations, warranties and covenants of the other in this Agreement or any of the closing documents executed and delivered by the other in connection with a Closing.
- (c) Cedar and RioCan each agrees that: (i) it will not, directly or indirectly, dispose of any of the Partnership Interests without registration under the Securities Act unless and until the proposed sale or transfer of the Partnership Interests is exempt from the registration requirements of the Securities Act, as evidenced (if desired by such Owner) by a written opinion of counsel of recognized standing in Securities Law.
- (d) The provisions of this Section 15 shall survive the Closings.

- (a) With respect to each of the Properties, Cedar shall cause each of the Property Owners to operate its Property in substantially the same manner as prior hereto pursuant to its normal course of business, including the maintenance of insurance and the making of claims thereunder, until the applicable Closing Date or prior termination of this Agreement with respect to such Property as provided herein; provided, however, that, without the prior consent of RioCan, Cedar shall not (except to the extent expressly provided herein) prior to the applicable Closing Date or prior termination of this Agreement with respect to such Property as provided herein:
 - refinance any of the Loans or, except as contemplated by Section 12(a) hereof, amend, modify or terminate in any material respect any of the Loan Documents;
 - (ii) except as required pursuant to the terms of a Lease, enter into, terminate (including evict), accept a surrender, modify or amend, in any material respect, or waive in writing or otherwise any material covenant or obligation of a Tenant under, any Lease for an area in excess of 10,000 rentable square feet;
 - (iii) enter into any lease of less than 10,000 rentable square feet if (y) the net effective rent is less than ninety-seven percent (97%) of the proforma net effective rent shown for the applicable portion of a Property shown on Schedule 15 or (z) if such lease is neither substantially in the standard form of lease for the applicable Property (with commercially reasonable changes thereto) nor is otherwise on commercially reasonable terms;
 - (iv) terminate, modify or amend, in any material respect, or waive in writing or otherwise any material covenant or obligation of a Ground Lessor under, any Ground Lease: or
 - (v) except for leases (or terminations, surrenders, modifications or amendments thereof) which would not require RioCan's consent pursuant to the foregoing clauses of this Section 16(a), grant any person or entity the right to acquire any fee or leasehold interest in any Property (or any portion thereof).

The provisions of this Section 16(a) shall survive the Closings.

(b) Cedar shall use commercially reasonable efforts as soon as practicable following the date of this Agreement to cause the release of Blue Mountain from the Blue Mountain Line of Credit. Prior to the Closing of the Transaction involving Blue Mountain, following the release of Blue Mountain from the Blue Mountain Line of Credit, Cedar may, subject to the terms of this Section 16(b), cause the applicable Property Owner to finance Blue Mountain with a mortgage loan secured by such Property on such commercially reasonable terms as Cedar shall determine (the "Blue Mountain Loan"). Promptly upon receipt thereof, Cedar agrees to deliver a copy of either the loan application or commitment received from the applicable lender in connection with the Blue Mountain Loan (the "Blue Mountain Loan Application") to RioCan for its review and approval, not to be unreasonably withheld or conditioned. Likewise, prior to entering into the loan documents and instruments evidencing the Blue Mountain Loan (the "Blue Mountain Loan Documents"), Cedar agrees to deliver copies of the same to RioCan for its review and approval, not to be unreasonably withheld or conditioned; provided, however, that RioCan shall have no right to disapprove the Blue Mountain Loan Documents unless the same materially and adversely conflict with the terms of the Blue Mountain Loan Application. In the event that RioCan shall fail to deliver written approval or disapproval of the terms of either the Blue Mountain Loan Application or the Blue Mountain Loan Documents within five (5) Business Days after receipt thereof, RioCan shall be deemed to have approved the same. If, in accordance with the terms of this Section 16(b), RioCan shall disapprove of the terms of either the Blue Mountain Loan Application or the Blue Mountain Loan Documents, Cedar shall have the option, in its sole direction, to either (i) cause the Blue Mountain Loan Application or the Blue Mountain Loan Documents, as applicable, to be modified until RioCan shall approve the same (which approval shall not be unreasonably withheld, conditioned or delayed) or (ii) cease pursuit of such Blue Mountain Loan. If the Blue Mountain Loan shall close prior to the Closing of the Transaction involving Blue Mountain as contemplated herein, the Allotted Consideration payable by RioCan at such Closing shall be adjusted (x) to account for the outstanding principal amount of the Blue Mountain Loan and (y) such that RioCan shall be responsible for its respective Percentage Interests of all third party transaction costs and closing costs incurred in obtaining the Blue Mountain Loan. In addition, if the Blue Mountain Loan shall have closed prior to the Closing of the Transaction involving Blue Mountain, RioCan shall pay to Cedar at the Closing of such Transaction, its Percentage Interest of a financing fee equal to one-quarter of one percent (0.25%) of the original principal amount of the Blue Mountain Loan; provided, however, that fifty percent (50%) of such financing fee shall be paid to RioCan if RioCan or RioCan REIT was the sole procuring party with respect to such financing or refinancing, and provided further that any such financing fee payable hereunder shall not exceed \$50,000. In the event the Blue Mountain Loan shall close on or after the Closing of the Transaction involving Blue Mountain, the financing fee payable to Cedar in connection therewith shall be governed by the terms of the applicable Management Agreement. The provisions of this Section 16(b) shall survive the Closings.

- (c) Cedar shall use commercially reasonable efforts as soon as practicable following the date of this Agreement to cause the release of Sunset Crossing from the Sunset Crossing Line of Credit. Prior to the Closing of the Transaction involving Sunset Crossing, following the release of Sunset Crossing from the Sunset Crossing Line of Credit, Cedar may, subject to the terms of this Section 16(b), cause the applicable Property Owner to finance Sunset Crossing with a mortgage loan secured by such Property on such commercially reasonable terms as Cedar shall determine (the "Sunset Crossing Loan"). Promptly upon receipt thereof, Cedar agrees to deliver a copy of either the loan application or commitment received from the applicable lender in connection with the Sunset Crossing Loan (the "Sunset Crossing Loan Application") to RioCan for its review and approval, not to be unreasonably withheld or conditioned. Likewise, prior to entering into the loan documents and instruments evidencing the Sunset Crossing Loan (the "Sunset Crossing Loan Documents"), Cedar agrees to deliver copies of the same to RioCan for its review and approval, not to be unreasonably withheld or conditioned; provided, however, that RioCan shall have no right to disapprove the Sunset Crossing Loan Documents unless the same materially and adversely conflict with the terms of the Sunset Crossing Loan Application. In the event that RioCan shall fail to deliver written approval or disapproval of the terms of either the Sunset Crossing Loan Application or the Sunset Crossing Loan Documents within five (5) Business Days after receipt thereof, RioCan shall be deemed to have approved the same. If, in accordance with the terms of this Section 16(c), RioCan shall disapprove of the terms of either the Sunset Crossing Loan Application or the Sunset Crossing Loan Documents, Cedar shall have the option, in its sole direction, to either (i) cause the Sunset Crossing Loan Application or the Sunset Crossing Loan Documents, as applicable, to be modified until RioCan shall approve the same (which approval shall not be unreasonably withheld, conditioned or delayed) or (ii) cease pursuit of such Sunset Crossing Loan. If the Sunset Crossing Loan shall close prior to the Closing of the Transaction involving Sunset Crossing as contemplated herein, the Allotted Consideration payable by RioCan at such Closing shall be adjusted (x) to account for the outstanding principal amount of the Sunset Crossing Loan and (y) such that RioCan shall be responsible for its respective Percentage Interests of all third party transaction costs and closing costs incurred in obtaining the Sunset Crossing Loan. In addition, if the Sunset Crossing Loan shall have closed prior to the Closing of the Transaction involving Sunset Crossing, RioCan shall pay to Cedar at the Closing of such Transaction, its Percentage Interest of a financing fee equal to one-quarter of one percent (0.25%) of the original principal amount of the Sunset Crossing Loan; provided, however, that fifty percent (50%) of such financing fee shall be paid to RioCan if RioCan or RioCan REIT was the sole procuring party with respect to such financing or refinancing, and provided further that any such financing fee payable hereunder shall not exceed \$50,000. In the event the Sunset Crossing Loan shall close on or after the Closing of the Transaction involving Sunset Crossing, the financing fee payable to Cedar in connection therewith shall be governed by the terms of the applicable Management Agreement. The provisions of this Section 16(c) shall survive the Closings.
- (d) Cedar shall use commercially reasonable efforts to deliver to RioCan before the applicable Closing Date, (i) tenant estoppel certificates ("Tenant Estoppels") from each Major Tenant for the applicable Property and a sufficient number of other Tenants occupying space at the Property such that estoppel certificates have been received with respect to not less than seventy-five percent (75%) of the aggregate occupied rentable square footage of each Property (collectively, "Required Tenants"), in each case dated not more than forty-five (45) days prior to the applicable Closing Date and on the agreed to form prescribed by its Lease (or the substantive equivalent), or on the form previously executed by such Tenant (or the substantive equivalent), a copy of which was furnished to and accepted by RioCan on or prior to the date hereof, or on any other form approved by RioCan in writing (such approval not to be unreasonably withheld), in each case, disclosing no materially adverse matters, and (ii) an estoppel certificate (the "Ground Lessor Estoppel") from each Ground Lessor, dated not more than forty-five (45) days prior to the applicable Closing Date and on the agreed to form prescribed by the applicable Ground Lease (or the substantive equivalent), or on the form previously obtained by Cedar with respect to such Ground Lessor (or the substantive equivalent), a copy of which was furnished to and accepted by RioCan on or prior to the date hereof, or on any other form approved by RioCan in writing (such approval not to be unreasonably withheld), and disclosing no materially adverse matters. Cedar shall deliver to RioCan (or RioCan's lawyers) the initial draft of each Tenant Estoppel and Ground Lessor Estoppel at least two (2) Business Days prior to delivery of such estoppels to the recipients. Cedar shall deliver to RioCan copies of all executed Tenant Estoppels and the Ground Lessor Estoppel promptly following receipt thereof by Cedar. For avoidance of doubt, failure by Cedar to obtain any Tenant Estoppel or Ground Lessor Estoppel in the manner provided herein shall not constitute a default by Cedar under this Agreement, but shall constitute the mere failure of a condition precedent as more particularly set forth in Section 18 below.
- (e) Except for any matters disclosed in the documents included in the Due Diligence Site or otherwise made available to RioCan and/or RioCan's Representatives on or prior to the date of this Agreement, Cedar covenants that, if after the date of this Agreement it obtains knowledge or information prior to Closing of matters then existing which make any representation or warranty of Cedar hereunder materially and adversely inaccurate, Cedar will promptly communicate such information to RioCan.
- (f) Cedar will provide to RioCan draft budgets and leasing plans for the Properties for the 2010 calendar year on or prior to December 8, 2009 and the parties shall use commercially reasonable efforts to approve same prior to the applicable Closings.

17. Deliveries to be made on the Closing Date.

(a) <u>Cedar Deliveries</u>: Cedar shall deliver or cause to be delivered to the Owners, RioCan or the Title Company, as the case may be, on the applicable Closing Date the following documents:

- with respect to the first Closing to occur pursuant to the terms of this Agreement, the Partnership Agreement and any formation or similar certificates required by the laws of the State of Delaware, executed by the Cedar Partners;
- (ii) assignment and assumption agreements in the form attached hereto as Exhibit G required in accordance with the steps outlined on Schedule 2 attached hereto, between the applicable Cedar Partner, as assignor, and REIT Property Subsidiary, as assignee, of the applicable Interests, executed by all parties thereto;
- (iii) with respect to Transaction involving Columbus Crossing only, an operating agreement for the New Columbus Crossing Preferred Partner Lender in a form consistent with the terms of the Partnership Agreement in all material respects and otherwise reasonably acceptable to the parties, executed by REIT Property Subsidiary, as the sole member, together with a copy of the certificate of formation of the New Columbus Crossing Preferred Partner Lender;
- (iv) with respect to Transaction involving Columbus Crossing only, conveyance documents mutually acceptable to the parties by which all right, title and interest of Existing Columbus Crossing Preferred Partner Lender in, to and under the Columbus Crossing Preferred Partner Loan and the Columbus Crossing Preferred Partner Loan Documents shall be transferred to New Columbus Crossing Preferred Partner Lender, executed by Existing Columbus Crossing Preferred Partner Lender and New Columbus Crossing Preferred Partner Lender;
- (v) with respect to Transaction involving Columbus Crossing only, a reimbursement agreement in a form reasonably acceptable to the parties whereby Cedar and RioCan REIT shall share liability under the Columbus Crossing Loan Guaranty in accordance with the Percentage Interests of the Cedar Partners and RioCan, respectively (the "Columbus Crossing Reimbursement Agreement"), executed by Cedar;
- (vi) with respect to each Property, the Management Agreement, executed by the applicable Property Owner and the Manager;
- (vii) all applicable transfer tax forms, if any;
- (viii) the affidavit referred to in Section 1445 of the Internal Revenue Code, as amended, with all pertinent information confirming that Cedar is not a foreign person, trust, estate, corporation or partnership;
- (ix) evidence reasonably satisfactory to the Title Company respecting the due organization of the Cedar Partners and the due authorization and execution by the applicable Cedar Partners of this Agreement and the documents required to be delivered hereunder;
- to the extent reasonably required by the Title Company, an affidavit of title in form and substance reasonably acceptable to Cedar and the Title Company;
- (xi) a certificate (the "Update Certificate") of Cedar dated as of the applicable Closing Date certifying that the representations and warranties of Cedar set forth in Section 13(a) of this Agreement, other than the representations and warranties set forth in Section 13(a) of this Agreement which are made as of the date of this Agreement (such representations and warranties of Cedar set forth in Section 13(a) of this Agreement (other than as aforesaid) are hereafter referred to as "Closing Date Representations") with respect to the applicable Closing remain true and correct in all material respects as of the applicable Closing Date, it being agreed that if any Closing Date Representation with respect to a particular Closing shall no longer be true and correct in any material respect due to a change in the facts or circumstances which do not otherwise constitute a default of Cedar pursuant to the express terms of this Agreement and Cedar is unable to deliver the Update Certificate, the failure of Cedar to deliver the Update Certificate shall constitute a failure of a condition to such Closing and shall not constitute a default by Cedar under this Agreement (or the failure of a condition to any other Closing), and the sole remedy of RioCan in connection therewith shall be to terminate this Agreement with respect to the applicable Transaction by written notice to Cedar in which event the Deposit shall be refunded to RioCan (if no other Closing with respect to another Property remains outstanding) and no party hereto shall have any further obligations under this Agreement with respect to such Transaction, except under those provisions of this Agreement that expressly survive a termination of this Agreement);
- (xii) a settlement statement prepared by the Title Company and approved by Cedar and RioCan (the "Settlement Statement");
- (xiii) evidence of the Lender Approval (if applicable);
- (xiv) copies of the Tenant Estoppels;

- (xv) a copy of the Ground Lessor Estoppel (if applicable);
- (xvi) documents with respect to the Blue Mountain Development Parcel in accordance with the terms of Section 7; and
- (xvii) all such other documents which are required or reasonably necessary to give effect to the Agreement and which have been reasonably requested by RioCan on or before the applicable Closing Date.
- (b) <u>RioCan Deliveries</u>: RioCan shall deliver or cause to be delivered to Cedar, the Owners or the Title Company, as the case may be, on the applicable Closing Date the following:
 - (i) the applicable Net Consideration required to be paid by RioCan to Cedar pursuant to Section 2 hereof;
 - with respect to the first Closing to occur pursuant to the terms of this Agreement, the Partnership Agreement and any formation or similar certificates required by the laws of the State of Delaware, executed by RioCan;
 - with respect to Transaction involving Columbus Crossing only, the Columbus Crossing Reimbursement Agreement, executed by RioCan REIT;
 - (iv) all applicable transfer tax forms, if any;
 - evidence reasonably satisfactory to the Title Company respecting the due organization of RioCan and the due authorization and execution by RioCan of this Agreement and the documents required to be delivered hereunder;
 - (vi) the Settlement Statement;
 - (vii) documents with respect to the Blue Mountain Development Parcel in accordance with the terms of Section 7; and
 - (viii) all such other documents which are required or reasonably necessary to give effect to the Agreement and which have been reasonably requested by Cedar on or before the applicable Closing Date.
- (c) All Closing Documents, the forms of which are not attached to this Agreement, shall be in form and content acceptable to the parties acting reasonably and in good faith.

18. Conditions to the Closings.

- (a) <u>Conditions Precedent to Obligations of RioCan</u>. The obligation of RioCan to consummate each Transaction contemplated by this Agreement shall be subject to the following, as applicable:
 - performance and observance in all material respects, by Cedar of all covenants, warranties and agreements of this Agreement to be performed or observed by Cedar with respect to such Transaction prior to or on the applicable Closing Date;
 - (ii) receipt of any Loan Approval applicable to such Transaction;
 - (iii) the Reorganizations applicable to such Transaction shall have occurred;
 - (iv) with respect to the Transaction involving Blue Mountain only, such Property shall have been released from the lien of the Blue Mountain Line of Credit;
 - (v) with respect to the Transaction involving Sunset Crossing only, such Property shall have been released from the lien of the Sunset Crossing Line of Credit;
 - (vi) the representations and warranties of Cedar set forth in Section 13(a) (other than the Closing Date Representations) being true and correct in all material respects as of the date of this Agreement;
 - (vii) the Closing Date Representations and the representations and warranties of Cedar set forth in Section 15 hereof being true and correct in all material respects as of the applicable Closing Date;
 - (viii) RioCan shall have received Tenant Estoppels from all the Required Tenants of the applicable Property(ies) (other than Blue Mountain) in the forms consistent with the forms required pursuant to Section 16(d);

- (ix) no Required Operating Tenant of the applicable Property(ies) shall have ceased conducting its business and paying rent, as provided in its Lease, or shall be a debtor in any state or federal insolvency or bankruptcy proceeding;
- (x) as of the Closing Date, the applicable Property Owner shall not be in material default of, and shall not have received written notice from the applicable Lender of any uncured material default under, any of such Property Owner's material obligations under the applicable Loan Documents;
- (xi) as of the Closing Date, the applicable Property Owner shall not be in material default of, and shall not have received written notice from the applicable Major Tenant or Ground Lessor of any uncured material default under, any of such Property Owner's material obligations under a Lease with a Major Tenant or a Ground Lease;
- (xii) with respect to the Transaction involving the Property located in Bridgeport, Connecticut only, RioCan shall have received the Ground Lessor Estoppel and the Bridgeport Ground Lease shall be in full force and effect;
- (xiii) with respect to the Transaction involving the Property located in Williamsport, Pennsylvania only, the Loyal Plaza Ground Lease shall be in full force and effect; and
- (xiv) the fulfillment on or before the applicable Closing Date of all other conditions precedent to Closing benefiting RioCan specifically enumerated in this Agreement respecting the subject Transaction.

The conditions set forth in this Section 18(a) are for the sole benefit of RioCan and may be waived in writing in whole or in part by notice to Cedar on or before Closing of the applicable Transaction, without prejudice to the right to terminate this Agreement in the event of the non-fulfillment of any other condition or conditions with respect to such Transaction not so waived.

- (b) <u>Conditions Precedent to Obligations of Cedar</u>. The obligation of Cedar to consummate each Transaction contemplated by this Agreement shall be subject to the following, as applicable:
 - performance and observance by RioCan in all material respects, of all covenants and agreements of this Agreement to be performed or observed by RioCan with respect to such Transaction prior to or on the applicable Closing Date;
 - (ii) receipt of any Loan Approval applicable to such Transaction;
 - (iii) with respect to the Transaction involving Blue Mountain only, such Property shall have been released from the lien of the Blue Mountain Line of Credit and all of the conditions to the payment of Earn-Out Proceeds set forth in Section 4(a) shall have been satisfied with respect to the requisite number of Blue Mountain Leases that would entitle Cedar to receive Blue Mountain Closing Earn-Out Proceeds of not less than Twenty-Four Million Dollars (\$24,000,000);
 - (iv) with respect to the Transaction involving Sunset Crossing only, such Property shall have been released from the lien of the Sunset Crossing Line of Credit;
 - (v) the representations and warranties of RioCan set forth in Section 14(a) and Section 15 hereof being true and correct in all material respects; and
 - (vi) the fulfillment on or before the Closing Date of all other conditions precedent to Closing benefiting Cedar specifically set forth in this Agreement respecting the subject Transaction.

The conditions set forth in this Section 18(b) are for the sole benefit of Cedar and may be waived in writing in whole or in part by notice to RioCan on or before Closing of the applicable Transaction, without prejudice to the right to terminate this Agreement in the event of the non-fulfillment of any other condition or conditions with respect to such Transaction not so waived.

(c) Reasonable Commercial Efforts to Satisfy Conditions Each of the parties shall act in good faith and use reasonable commercial efforts in the circumstances to satisfy or cause to be satisfied (at its sole cost, except as expressly provided in Section 6 hereof) the conditions set forth in subparagraphs (a) and (b), respectively, provided that no party shall be required to spend money or incur additional obligations to obtain the necessary assistance or cooperation of any third party to satisfy any condition, other than expenditure of reasonable legal fees and provided further that the foregoing shall not limit or prejudice the rights of any party hereto (or its board of directors) to be satisfied in its sole and unfettered discretion as to the fulfillment of a condition in its respective favor if such right is provided pursuant to the terms of such condition. Each party will cooperate in support of all things necessary to give effect to this Agreement.

- (d) If one or more of the conditions set forth in subparagraph (a) is not satisfied or waived as therein provided on or before the applicable Scheduled Closing Date (as such date may be extended by written agreement of the parties or as otherwise expressly provided herein) with respect to a particular Property, this Agreement shall be automatically terminated with respect to such Property, the Deposit shall be refunded to RioCan (if no other Closing with respect to another Property remains outstanding), the Consideration shall be reduced by the amount of the applicable Allotted Consideration and, in such event, Cedar shall be released from all obligations hereunder with respect to such Property (except for any obligation expressly provided to survive a termination of this Agreement) and, subject to claims for breach of obligations pursuant to subparagraph (c) hereof, RioCan shall also be released from all obligations hereunder with respect to such Property (except for any obligation expressly provided to survive a termination of this Agreement); it being acknowledged and agreed that the consummation of the Closing with respect to any Property shall constitute the waiver of any such conditions that were not waived or satisfied with respect to such Property.
- (e) If one or more of the conditions set forth in subparagraph (b) is not satisfied or waived as therein provided on or before the applicable Scheduled Closing Date (as such date may be extended by written agreement of the parties or as otherwise expressly provided herein) with respect to a particular Property, this Agreement shall be automatically terminated with respect to such Property, the Deposit shall be refunded to RioCan (if no other Closing with respect to another Property remains outstanding), the Consideration shall be reduced by the amount of the applicable Allotted Consideration and, in such event, RioCan shall be released from all obligations hereunder with respect to such Property (except for any obligation expressly provided to survive a termination of this Agreement) and, subject to claims for breach of obligations pursuant to subparagraph (c) hereof, Cedar shall also be released from all obligations hereunder with respect to such Property (except for any obligation expressly provided to survive a termination of this Agreement); it being acknowledged and agreed that the consummation of the Closing with respect to any Property shall constitute the waiver of any such conditions that were not waived or satisfied with respect to such Property.
- (f) The parties agree that all conditions herein are conditions of the obligations of the party named in the relevant section to complete the applicable Transaction and are not conditions precedent to the existence or enforceability of this Agreement.

19. Apportionments.

- (a) With respect to each Property, the following shall be prorated between the applicable Property Owner as constituted immediately prior to the Closing (assuming that Cedar owned 100% of the Interests in such Property Owner), and the applicable Property Owner as constituted immediately following the Closing (assuming that Cedar owned 20% of the Interests in such Property Owner and the remaining 80% of such Interests were owned by RioCan), as of 11:59 p.m. on the day preceding the Closing Date (the "Adjustment Date") (on the basis of the actual number of days elapsed over the applicable period):
 - (i) Fixed rents, additional rents, percentage rent and all other sums and credits due or payable under the applicable Leases and any other items of income, as and when collected (it being acknowledged that all such amounts received after the applicable Closing shall be applied as provided in Section 19(i));
 - (ii) All real estate taxes, water charges, sewer rents, vault charges and assessments on the Property on the basis of the fiscal year for which assessed (except to the extent required to be paid by Tenants in good standing pursuant to Leases);
 - (iii) All operating expenses, including, without limitation, all amounts payable by the applicable Property Owners pursuant to the Ground Leases (except to the extent required to be paid by Tenants in good standing pursuant to Leases);
 - (iv) Any prepaid items, including, without limitation, fees for licenses and annual permit and inspection fees;
 - (v) Utilities, including, without limitation, telephone, steam, electricity and gas, on the basis of the most recently issued bills therefor (except to the extent required to be paid by Tenants pursuant to Leases);
 - (vi) Deposits with telephone and other utility companies;
 - (vii) Payments of principal and interest and other costs payable under any Loan Documents;
 - (viii) Cash and cash equivalents (e.g., U.S. Treasuries) held by or for the account of any of the Owners;
 - (ix) Deposits, reserves or escrows made by or on behalf of any of the Owners with respect to loans and other obligations that will remain in effect on and after the applicable Closing; and

- (x) Such other items as are customarily apportioned between sellers and purchasers of real properties (and interests therein) of a type similar to the Properties and located in the State or Commonwealth in which each such Property is located, including, without limitation, any items not expressly included in the items that are the responsibility of Cedar pursuant to Section 19(b) through (d) below).
- (b) Items to be Paid by Cedar. Subject to the provisions of Section 6(c) hereof, the parties hereby acknowledge and agree that Cedar shall be responsible and liable to pay and shall pay when due the following (collectively, "Leasing Costs") for each Property (other than Blue Mountain and the Franklin Village Earn-Out Space, which is provided for in clause (c) below) and the Partnership shall be responsible for payment for all leasing costs not the responsibility of Cedar pursuant to this clause 19(b):
 - any real estate or leasing commission in respect of the Leases (or any modification or amendment thereof) executed prior to the date of this Agreement;
 - (ii) any tenant inducements or tenant allowances payable under the Leases (or any modification or amendment thereof) executed prior to the date of this Agreement;
 - (iii) any costs and expenses of any lease take-over, assignment, assumption or other commitments required pursuant to the Leases (or any modification or amendment thereof) executed prior to the date of this Agreement; and
 - (iv) the costs and expenses of any initial tenant build-out work or improvements to rentable or rental space in the Building required to be performed by the landlord thereunder and arising pursuant to the Leases (or any modification or amendment thereof) executed prior to the date of this Agreement.
- (c) <u>Blue Mountain and Franklin Village</u>. With respect to Blue Mountain only, Cedar shall be responsible and liable to pay, and shall pay when due, (i) the cost of completion of the initial construction of the Blue Mountain Property (including, without limitation, penalties incurred in connection therewith) and (ii) all Leasing Costs for which Earn-Out Proceeds shall be earned by Cedar pursuant to Section 4(a) of this Agreement, but excluding all related tenant build-out work or improvements and other Leasing Costs with respect to Leases for which Earn-Out Proceeds shall not be earned by Cedar pursuant to Section 4(a) of this Agreement. With respect to Franklin Village only, Cedar shall be responsible and liable to pay, and shall pay when due, all Leasing Costs for which Earn-Out Proceeds shall be earned by Cedar pursuant to Section 4(b) of this Agreement, but excluding all related tenant build-out work or improvements and other Leasing Costs with respect to leases for the Franklin Village Earn-Out Space for which Earn-Out Proceeds shall not be earned by Cedar pursuant to Section 4(b) of this Agreement.
- (d) <u>Deferred Costs.</u> There will be no adjustment (other than current year adjustments) for deferred amortized common area costs, if any, that are recoverable from Tenants after any Closing nor will any adjustments (other than current year adjustments) be made with respect thereto after any Closing regardless of any amounts received from Tenants relating thereto.
- (e) <u>Statement of Adjustments</u>. A statement of adjustments shall be prepared by Cedar for approval of RioCan, acting reasonably, at least five (5) Business Days before each Closing Date.
- (f) Readjustments. If, on the Closing Date, any items of additional rent or percentage rent under the Leases or other income or expense of the Properties shall not have been ascertained, then such items shall be adjusted retroactively as and when the same are ascertained. If the final cost or amount of any item which is to be adjusted cannot be determined at Closing, then an initial adjustment for such item shall be made at Closing, such amount to be estimated by Cedar, acting reasonably, as of the Adjustment Date on the basis of the best evidence available at the Closing as to what the final cost or amount of such item will be. A final adjustment shall be made no later than the date being one (1) year after the Closing Date for such Property (each, an "Outside Adjustment Date") other than Blue Mountain. For the case at Blue Mountain the Outside Adjustment Date shall be the third (3rd) anniversary of the Blue Mountain Closing Date. Except as may otherwise be permitted pursuant to this Agreement, no re-adjustment may be claimed by any party with respect to any Property later than the Outside Adjustment Date.
- (g) If, with respect to any Property, the Closing shall occur before the applicable real estate tax rate is fixed, the apportionment of real estate taxes for such Property at the Closing shall be based upon the tax rate for the next preceding year applied to the latest assessed valuation. Promptly after the new tax rate or assessment is fixed, the apportionment of taxes or assessments shall be recomputed and any discrepancy resulting from such recomputation and any errors or omissions in computing apportionments at Closing shall be promptly corrected and the proper party reimbursed.
- (h) All apportionments made under this Agreement shall be calculated (1) as between the Property Owners, as constituted prior to the applicable Closing as the prior owners of the Properties (assuming that Cedar owned 100% of the Interests in such Property Owner), and such Property Owners, as constituted following the applicable Closing as the new owners of the Properties (assuming that Cedar owned 20% of the Interests in such Property Owner and the remaining 80% of such Interests were owned by RioCan) and then (2) the applicable Allotted Consideration shall be adjusted at the applicable Closing such that Cedar and RioCan shall share in the credits and debits of the Property Owners in proportion to their respective interests in such Property Owners immediately following the Closing.

- (i) If any tenant at a Property is in arrears in the payment of rent or any other receivables on the Closing Date, any and all rents and receivables received from such tenant after the Closing shall be applied in the following order of priority: (i) first to the month in which the Closing occurred; (ii) then to any month or months following the month in which the Closing occurred; and (iii) then to the months preceding the month in which the Closing occurred. If rents or other receivables or any portion thereof received after the Closing are payable to the other party by reason of this allocation, the appropriate sum, less a proportionate share of any reasonable attorneys' fees, costs and expenses of collection thereof, shall be promptly paid to the other party.
- (j) Notwithstanding anything to the contrary contained in this Agreement, with respect to the Properties, Cedar shall remain liable for actual damages (including out-of-pocket expenses actually incurred by the Owners) resulting from (x) any uninsured third party tort claims arising and accruing prior to the applicable Closing Date and which are both unrelated to the environmental condition of any Property or any physical condition known by or disclosed to RioCan or any RioCan Representatives and based solely on the actions or omissions of any Owner prior to the applicable Closing Date (the parties acknowledge that third party tort claims shall not be deemed "uninsured" if the applicable insurance policy provides for a deductible), (y) any breach by any Property Owner of its obligations under any of the Service Contracts and Leases arising prior to the applicable Closing Date, except (A) with respect to any Lease, if the Tenant shall have delivered a Tenant Estoppel prior to the applicable Closing confirming that no such breach exists, (B) if such breach shall have been disclosed to or known by RioCan prior to the applicable Closing Date or (C) if the applicable Allotted Consideration shall have been adjusted to reflect such monetary obligation or breach, or (z) any tax liability of any Owner allocable to periods prior to the applicable Closing Date. RioCan acknowledges and agrees that its sole and exclusive remedy against Cedar in connection with the foregoing responsibilities shall be either an action for specific performance or a claim for direct damages (excluding special, indirect, consequential and punitive damages), RioCan hereby waiving any other right or remedy it may otherwise have at law or equity. The provisions of this Section 19 notwithstanding, nothing contained herein shall limit or in any way be deemed to modify the "as is, where is" nature of the Transactions as more particularly set forth in Section 13(d) of this Agreement and RioCan hereby confirms its agreement to waive any right it may have at law or in equity, including, without limitation, the right to seek damages or contribution from Cedar in connection with the physical (including, without limitation, environmental) condition of the Properties (except in connection with any breach of applicable representations and warranties of Cedar contained in Section 13(a) in accordance with the terms and conditions of this Agreement).

The provisions of this Section 19 shall survive the Closings.

20. Condemnation or Destruction of the Properties.

- In the event that, after the date hereof but prior to the applicable Closing Date with respect to any Property then owned by Cedar or any affiliate, either any portion of such Property is taken (or so threatened by written notice delivered to the applicable Property Owner by a Governmental Authority having jurisdiction) pursuant to eminent domain proceedings or condemnation or any of the improvements on such Property are damaged or destroyed by fire or other casualty, Cedar shall, promptly upon becoming aware of the same, deliver or cause to be delivered to RioCan, notice of any such eminent domain proceedings or casualty. Except as otherwise expressly provided herein, neither Cedar nor any Owner shall have the obligation to restore, repair or replace any portion of any Property or any such damage or destruction. If, with respect to any Property, the amount of the damage (as determined by an independent third party contractor or engineer selected by Cedar and reasonably approved by RioCan) or the amount of condemnation award shall exceed an amount equal to ten percent (10%) of the applicable Allotted Consideration, RioCan shall have the right to elect to terminate this Agreement as to the applicable Transaction only by written notice to Cedar given within ten (10) days after notification to RioCan of the estimated amount of damages or the determination of the amount of any condemnation award, whereupon the Deposit shall be refunded to RioCan (if no other Closing with respect to another Property remains outstanding), the Consideration shall be reduced by an amount equal to the applicable Allotted Consideration and the parties hereto shall be relieved of all further liability and responsibility under this Agreement with respect to such Transaction (except for any obligation expressly provided to survive a termination of this Agreement).
- (b) In the event of any condemnation or casualty as aforesaid, the applicable Closing Date shall be extended as and to the extent necessary to permit the determination of the damage amount or condemnation award in the manner herein provided, to a Business Day selected by Cedar and reasonably approved by RioCan. The parties hereby waive the provisions of any statute which provides for a different outcome or treatment in the event of a casually or a condemnation or eminent domain proceeding.

(c) RioCan shall not have any right to terminate this Agreement with respect to any Transaction on account of any condemnation or casualty except as expressly provided in this Section 20. In the event that the parties shall proceed to close a Transaction notwithstanding the occurrence of any condemnation or casualty as aforesaid, the corresponding condemnation awards or proceeds of insurance shall be adjusted, settled, collected and/or applied by the applicable Property Owner or Lender in accordance with the applicable Loan Documents provided that in no event shall any condemnation awards or proceeds of insurance received prior to the applicable Closing Date be distributed to Cedar and the same shall be held by or for the benefit of the applicable Property Owner or applied to the costs and expenses of the applicable Property and provided further that Cedar shall be solely responsible for the payment of the deductible under any insurance policy with respect to any casualty occurring prior to the applicable Closing Date.

21. Release.

- EFFECTIVE AS OF THE CLOSING OF EACH TRANSACTION, RIOCAN SHALL BE (a) DEEMED TO HAVE RELEASED CEDAR AND ALL CEDAR RELATED PARTIES FROM ALL CLAIMS WHICH RIOCAN OR ANY AGENT, REPRESENTATIVE, AFFILIATE, EMPLOYEE, DIRECTOR, OFFICER, PARTNER, MEMBER, SERVANT, SHAREHOLDER OR OTHER PERSON OR ENTITY ACTING ON BEHALF OF OR OTHERWISE RELATED TO OR AFFILIATED WITH, RIOCAN (EACH, A "RIOCAN RELATED PARTY") HAS OR MAY HAVE ARISING FROM OR RELATED TO ANY MATTER OR THING RELATED TO OR IN CONNECTION WITH THE APPLICABLE PROPERTY AND THE APPLICABLE INTERESTS (INCLUDING, AT THE CLOSING OF COLUMBUS CROSSING, THE COLUMBUS CROSSING PREFERRED INTERESTS AND THE COLUMBUS CROSSING PREFERRED PARTNER LOAN) INCLUDING THE DOCUMENTS AND INFORMATION REFERRED TO HEREIN, THE LEASES AND THE TENANTS THEREUNDER, ANY CONSTRUCTION DEFECTS, ERRORS OR OMISSIONS IN THE DESIGN OR CONSTRUCTION OF ALL OR ANY PORTION OF THE APPLICABLE PROPERTY AND ANY ENVIRONMENTAL CONDITIONS, AND RIOCAN SHALL NOT LOOK TO CEDAR OR ANY CEDAR RELATED PARTIES IN CONNECTION WITH THE FOREGOING FOR ANY REDRESS OR RELIEF. THIS RELEASE SHALL BE GIVEN FULL FORCE AND EFFECT ACCORDING TO EACH OF ITS EXPRESSED TERMS AND PROVISIONS, INCLUDING THOSE RELATING TO UNKNOWN AND UNSUSPECTED CLAIMS, DAMAGES AND CAUSES OF ACTION; PROVIDED, HOWEVER, THAT THIS RELEASE SHALL NOT BE APPLICABLE TO ANY CLAIMS ARISING OUT OF THE EXPRESS COVENANTS, REPRESENTATIONS, OR WARRANTIES SET FORTH IN THIS AGREEMENT OR ANY CLOSING DELIVERY THAT SHALL EXPRESSLY SURVIVE THE CLOSING OF A TRANSACTION.
- (b) The provisions of this Section 21 shall survive the Closings or a termination of this Agreement.

22. Brokers.

- (a) Cedar represents and warrants to RioCan, and RioCan represents and warrants to Cedar, that no broker or finder except Goldman, Sachs & Co. and RBC Dominion Securities Inc. (whose fees and commissions will be paid in accordance with subparagraphs (b) and (c) below) has been engaged by it, respectively, in connection with the Transactions contemplated under this Agreement. In the event of a claim for broker's or finder's fee or commissions in connection with the sale contemplated by this Agreement (other than as provided in subparagraphs (b) and (c) below), then Cedar shall indemnify, defend and hold harmless RioCan from the same if it shall be based upon any statement or agreement alleged to have been made by Cedar, and RioCan shall indemnify, defend and hold harmless Cedar from the same if it shall be based upon any statement or agreement alleged to have been made by RioCan.
- (b) Cedar will pay whatever commission is payable to Goldman, Sachs & Co. by reason of the Transactions and will indemnify and save RioCan harmless in respect of any claim or action against RioCan on account thereof.
- (c) RioCan will pay whatever commission is payable to RBC Dominion Securities Inc. by reason of the Transactions and will indemnify and save Cedar harmless in respect of any claim or action against Cedar on account thereof.
- (d) The provisions of this Section 22 shall survive the Closings and/or a termination of this Agreement.

23. Limitation of Liability.

- (a) Notwithstanding anything to the contrary contained in this Agreement or any documents executed in connection herewith, if one or more of the Transactions shall have closed hereunder, Cedar shall have not have any liability arising pursuant to or in connection with the representations, warranties, indemnifications, covenants or other obligations (whether express or implied) of Cedar under this Agreement (or any document or certificate executed or delivered in connection herewith) unless claims made by RioCan shall collectively equal or exceed One Hundred Thousand Dollars (\$100,000) in the aggregate for all the Properties; provided, however, in no event shall the aggregate liability of Cedar exceed five percent (5%) of the aggregate amount of the Consideration received from all of the Transactions that have closed pursuant to and in accordance with this Agreement (whether paid at Closing or as Earn-Out Proceeds). Notwithstanding the foregoing, the obligations of Cedar contained in Section 19 hereof shall not be subject to the limitations on liability contained in this Section 23(a).
- (b) No partner, member, shareholder, officer, director, employee or agent of Cedar, nor any Cedar Related Parties, shall have any personal liability, directly or indirectly, under or in connection with this Agreement or any agreement made or entered into under or pursuant to the provisions of this Agreement, or any amendment or amendments to any of the foregoing made at any time or times, heretofore or hereafter, and RioCan, on behalf of itself and its successors and assigns, hereby waives any and all such personal liability.
- (c) This Agreement and any agreement made or entered into under or pursuant to the provisions of this Agreement, any amendment or amendments to any of the foregoing made at any time or times, shall be conclusively taken to have been executed by, or by officers of RioCan on behalf of, the trustees of RioCan only in their capacity as trustees of RioCan. Cedar hereby disavows any liability upon and waives any claim against holders of units of RioCan and annuitants under plans of which holders of units of RioCan act as trustee or carrier and the obligations created hereunder are not personally binding upon, nor shall resort be had to, nor shall recourse or satisfaction be sought from, the private property of any trustee or officer of RioCan or any holder of units of RioCan or annuitant, but the property of RioCan from time to time or a specific portion thereof only shall be bound. It is agreed that the benefit of this provision is restricted to the trustees and officers of RioCan, each holder of units issued by RioCan and annuitants and, solely for that purpose, the undersigned signing officers of RioCan have entered into this Agreement and any agreement made or entered into under or pursuant to the provisions of this Agreement, any amendment or amendments to any of the foregoing made at any time or times, as agent and trustee for and on behalf of the trustees of RioCan, each holder of units of RioCan and each annuitant.
- (d) The provisions of this Section 23 shall survive the Closings and/or a termination of this Agreement.

24. Remedies For Default.

CEDAR DEFAULTS. IF ANY TRANSACTION SHALL NOT BE CLOSED SOLELY BY REASON OF CEDAR'S BREACH OR DEFAULT UNDER THIS AGREEMENT, THEN RIOCAN SHALL HAVE AS ITS EXCLUSIVE REMEDY THE RIGHT TO (A) SUBJECT TO THE LAST SENTENCE OF THIS CLAUSE 24(a), TERMINATE THIS AGREEMENT WITH RESPECT TO ALL TRANSACTIONS NOT YET CLOSED (IN WHICH EVENT THE DEPOSIT SHALL BE RETURNED TO RIOCAN AND NO PARTY HERETO SHALL HAVE ANY FURTHER OBLIGATION OR LIABILITY TO THE OTHER EXCEPT WITH RESPECT TO THOSE PROVISIONS OF THIS AGREEMENT WHICH EXPRESSLY SURVIVE A CLOSING OR TERMINATION OF THIS AGREEMENT), RIOCAN HEREBY WAIVING ANY RIGHT OR CLAIM TO DAMAGES FOR CEDAR'S BREACH; OR (B) SPECIFICALLY ENFORCE THIS AGREEMENT (BUT NO OTHER ACTION, FOR DAMAGES OR OTHERWISE, SHALL BE PERMITTED); PROVIDED THAT ANY ACTION BY RIOCAN FOR SPECIFIC PERFORMANCE MUST BE FILED, IF AT ALL, WITHIN FORTY-FIVE (45) DAYS OF CEDAR'S BREACH OR DEFAULT, AND THE FAILURE TO FILE WITHIN SUCH PERIOD SHALL CONSTITUTE A WAIVER BY RIOCAN OF SUCH RIGHT AND REMEDY. NOTWITHSTANDING THE FOREGOING, IF A BREACH OF A REPRESENTATION, WARRANTY OR COVENANT IS SPECIFIC TO A PARTICULAR PROPERTY, RIOCAN'S EXCLUSIVE REMEDY SHALL BE TO TERMINATE THIS AGREEMENT AS TO SUCH PROPERTY IN WHICH EVENT THE CONSIDERATION SHALL BE REDUCED BY THE AMOUNT OF THE APPLICABLE ALLOTTED CONSIDERATION, THE DEPOSIT SHALL BE REFUNDED TO RIOCAN (IF NO OTHER CLOSING WITH RESPECT TO ANOTHER PROPERTY REMAINS OUTSTANDING) AND THE PARTIES HERETO SHALL BE RELIEVED OF ALL FURTHER LIABILITY AND RESPONSIBILITY UNDER THIS AGREEMENT WITH RESPECT TO THE APPLICABLE TRANSACTION, EXCEPT FOR ANY OBLIGATION EXPRESSLY PROVIDED TO SURVIVE A TERMINATION OF THIS AGREEMENT OR, IF APPLICABLE, ANY CLOSING THAT MAY HAVE ALREADY OCCURRED HEREUNDER. THE PARTIES ACKNOWLEDGE AND AGREE THAT THE TERMS AND PROVISIONS OF THIS SECTION 24 ARE NOT INTENDED TO LIMIT RIOCAN'S RIGHTS AND REMEDIES IN THE EVENT OF A BREACH OF CEDAR'S POST-CLOSING OBLIGATIONS UNDER THIS AGREEMENT NOTWITHSTANDING THE FACT THAT ONE OR MORE TRANSACTIONS MAY NOT HAVE CLOSED.

- RIOCAN DEFAULTS. IN THE EVENT ANY TRANSACTION SHALL NOT CLOSE (b) SOLELY ON ACCOUNT OF RIOCAN'S BREACH OR DEFAULT, THEN, AT CEDAR'S ELECTION, THIS AGREEMENT SHALL TERMINATE WITH RESPECT TO ALL TRANSACTIONS NOT YET CLOSED, AND THE RETENTION OF THE DEPOSIT SHALL BE CEDAR'S SOLE AND EXCLUSIVE REMEDY UNDER THIS AGREEMENT, SUBJECT TO THE PROVISIONS OF THIS AGREEMENT THAT EXPRESSLY SURVIVE SUCH TERMINATION. IN CONNECTION WITH THE FOREGOING, THE PARTIES RECOGNIZE THAT CEDAR WILL INCUR EXPENSE IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT AND THAT THE INTERESTS (AND RELATED PROPERTIES) WILL BE REMOVED FROM THE MARKET; FURTHER, THAT IT IS EXTREMELY DIFFICULT AND IMPRACTICABLE TO ASCERTAIN THE EXTENT OF DETRIMENT TO CEDAR CAUSED BY THE BREACH BY RIOCAN UNDER THIS AGREEMENT AND THE FAILURE OF THE CONSUMMATION OF ANY TRANSACTION CONTEMPLATED BY THIS AGREEMENT OR THE AMOUNT OF COMPENSATION CEDAR SHOULD RECEIVE AS A RESULT OF RIOCAN'S BREACH OR DEFAULT. THE PARTIES ACKNOWLEDGE AND AGREE THAT THE TERMS AND PROVISIONS OF THIS SECTION 24 ARE NOT INTENDED TO LIMIT CEDAR'S RIGHTS AND REMEDIES IN THE EVENT OF A BREACH OF RIOCAN'S POST-CLOSING OBLIGATIONS UNDER THIS AGREEMENT (INCLUDING, WITHOUT LIMITATION, THE OBLIGATION TO PAY EARN-OUT PROCEEDS AS AND WHEN DUE HEREUNDER) NOTWITHSTANDING THE FACT THAT ONE OR MORE TRANSACTIONS MAY NOT HAVE CLOSED.
- (c) Prior to the exercise by Cedar or RioCan of any right or remedy afforded to it pursuant to Section 24(a) or Section 24(b) herein, as applicable, such party (the "Non-Defaulting Party") shall deliver written notice (a "Default Notice") to the other party hereunder (the "Defaulting Party") identifying the applicable breach or default and the Defaulting Party shall have ten (10) days after delivery such Default Notice to cure such breach or default. If a Defaulting Party fails to cure any default or breach that is the subject of a Default Notice within such ten (10) day period, the Non-Defaulting Party may exercise all rights and remedies afforded to it pursuant to Section 24(a) or Section 24(b) above, as applicable.
- (d) The provisions of this Section 24 shall survive the Closings and/or a termination of this Agreement.

25. Title Reviews.

RioCan has obtained, or shall promptly after the date hereof obtain, current title reports for each of the Properties from the Title Company or Stewart Title Insurance Company (the "Title Reports"). If any exceptions(s) to title to any Property should appear in the Title Reports that are not Permitted Exceptions, then, no later than November 5, 2009 (the "Title Objection Deadline"), RioCan shall promptly deliver copies thereof to Cedar, together with copies of the applicable exception documentation and written notice of disapproval of said exceptions (a "Title Objection Letter"). Any such title exceptions so objected to by RioCan pursuant to this Section 25(a) shall be deemed to be "Title Objections." Subject to Section 25(c) below, within ten (10) days following receipt of the Title Objection Letter, Cedar shall deliver written notice to RioCan of any Title Objections with respect to which Cedar, in its sole and absolute discretion, elects to undertake the removal prior to or at the applicable Closing (the "Title Objection Response"); provided, however, that if Cedar shall fail to deliver any Title Objection Response by the expiration of such ten (10) day period, Cedar shall be deemed to have elected not to undertake the removal of the subject Title Objections. Subject to Section 25(c) below, if Cedar elects or is deemed to have elected not to cure any Title Objection, RioCan's only options in response thereto shall be waive the Title Objections or to terminate this Agreement within ten (10) days following its receipt or deemed receipt of the Title Objection Response as it relates to the applicable Property or Properties (it being acknowledged and agreed that RioCan's failure to so terminate with the aforementioned ten (10) day period shall constitute an election to waive such Title Objections), whereupon the Deposit shall be refunded to RioCan (if no other Closing with respect to another Property remains outstanding), the Consideration shall be reduced by the amount of the applicable Allotted Consideration, and the parties hereto shall be relieved of all further liability and responsibility under this Agreement with respect to the applicable Transaction, except for any obligation expressly provided to survive a termination of this Agreement or, if applicable, any Closing that may have already occurred hereunder. Subject to Section 25(c) below, if Cedar shall have elected to undertake the removal of a Title Objection but does not cause the removal thereof by the applicable Scheduled Closing Date, RioCan shall have the option, to be exercised by RioCan by written notice to Cedar on or before the applicable Scheduled Closing Date, to either (A) accept the Property "as is" with respect to such Title Objections and consummate the Closings in accordance with the terms of this Agreement or (B) terminate this Agreement as it relates to the applicable Property or Properties by written notice thereof to Cedar, whereupon the Deposit shall be refunded to RioCan (if no other Closing with respect to another Property remains outstanding), the Consideration shall be reduced by the amount of the applicable Allotted Consideration, and the parties hereto shall be relieved of all further liability and responsibility under this Agreement with respect to the applicable Transaction, except for any obligation expressly provided to survive a termination of this Agreement or, if applicable, any Closing that may have already occurred hereunder. Should RioCan fail to elect an option in writing by the applicable Scheduled Closing Date, RioCan shall be deemed to have elected option (A) above. For avoidance of doubt, Cedar shall not under any circumstance be required or obligated to cause the cure or removal of any Title Objection (other than Mandatory Cure Items) including, without limitation, to bring any action or proceeding, to make any payments or otherwise to incur any expense in order to eliminate any Title Objection or to arrange for title insurance insuring against enforcement of such Title Objection against, or collection of the same out of, the applicable Property, notwithstanding that Cedar may have attempted to

(a)

Exceptions, of which RioCan is first made aware in any update made to any Title Report or in any updated survey received after the earlier of the date of the Title Objection Letter delivered pursuant to Section 25(a) above or the Title Objection Deadline, RioCan shall deliver copies thereof to Cedar, together with copies of the applicable exception documentation(s) or updated survey, if applicable, and written notice of disapproval of said exceptions no later than the earlier of (i) the applicable Scheduled Closing Date and (ii) ten (10) days after receipt by RioCan of the applicable updated Title Report. Any such title exceptions so objected to by RioCan pursuant to this Section 25(b) shall be deemed to be "Additional Title Objections." Subject to Section 25(c) below, no later than the earlier of (i) the applicable Scheduled Closing Date and (ii) ten (10) days after receipt by Cedar of written notice from RioCan of any Additional Title Objections, Cedar shall deliver a Title Objection Response to RioCan of any Additional Title Objections with respect to which Cedar, in its sole and absolute discretion, elects to undertake the removal prior to or at the applicable Closing; provided, however, that if Cedar shall fail to deliver any Title Objection Response by the applicable Scheduled Closing Date, Cedar shall be deemed to have elected not to cause the removal of the subject Additional Title Objections. Notwithstanding the foregoing, in the event Cedar shall elect to undertake the removal of any Additional Title Objections hereunder, Cedar shall have the right in its sole and absolute discretion upon delivery of prior written notice to RioCan, to extend the applicable Scheduled Closing Date by up to thirty (30) days in the aggregate, to cause the removal thereof. Subject to Section 25(c) below, if Cedar indicates or is deemed to have indicated that it will not cure any Additional Title Objection or, if Cedar shall have elected to undertake the removal of an Additional Title Objection but does not cause the removal thereof by the applicable Scheduled Closing Date, RioCan shall have the option, by (I) if Cedar shall have elected (or is deemed to have elected) not to cause the removal of the Additional Title Objection, the earlier of the Scheduled Closing Date and third (3rd) Business Day after receipt of the Title Objection Response (or the date such Title Objection Response shall have been due, as applicable) or (II) if Cedar shall have elected to undertake the removal of an Additional Title Objection but does not cause the removal thereof by the applicable Scheduled Closing Date, the Scheduled Closing Date, to either (A) accept the applicable Property "as is" with respect to such Additional Title Objections and consummate applicable the Closings in accordance with the terms of this Agreement or (B) terminate this Agreement as it relates to the applicable Property or Properties by written notice thereof to Cedar, whereupon the Deposit shall be refunded to RioCan (if no other Closing with respect to another Property remains outstanding), the Consideration shall be reduced by the amount of the applicable Allotted Consideration, and the parties hereto shall be relieved of all further liability and responsibility under this Agreement with respect to the applicable Transaction, except for any obligation expressly provided to survive a termination of this Agreement or, if applicable, any Closing that may have already occurred hereunder. Should RioCan fail to elect an option in writing within said three (3) Business Day period, RioCan shall be deemed to have elected option (A) above. For avoidance of doubt, Cedar shall not under any circumstance be required or obligated to cause the cure or removal of any Additional Title Objection (other than Mandatory Cure Items) including, without limitation, to bring any action or proceeding, to make any payments or otherwise to incur any expense in order to eliminate any Additional Title Objection or to arrange for title insurance insuring against enforcement of such Additional Title Objection against, or collection of the same out of, the applicable Property, notwithstanding that Cedar may have attempted to do so.

If RioCan shall object to any exceptions(s) to title to the Property, other than the Permitted

(c) Notwithstanding anything to the contrary contained herein, Cedar shall cause the removal (by bonding or otherwise) prior to the applicable Scheduled Closing Date of any monetary liens encumbering any Property (that is not a Permitted Exception hereunder) objected to by RioCan in accordance with Section 25(a) or Section 25(b) above, if the placing of such lien was solely the direct result of the actions of Cedar (including, without limitation, resulting from the initial construction of Blue Mountain) and not otherwise caused by any tenant at, or prior owner of, the Property or any other third party (each, a "Mandatory Cure Item").

26. Notices.

(b)

All notices, demands, consents, reports and other communications provided for in this Agreement shall be in writing, shall be given by a method prescribed in this Section and shall be given to the party to whom it is addressed at the address set forth below.

To Cedar:

c/o Cedar Shopping Centers, Inc. 44 South Bayles Avenue Port Washington, New York 11050 Attention: Leo S. Ullman Facsimile: (516) 767-6497

With a copy to:

Stroock & Stroock & Lavan LLP 180 Maiden Lane New York, New York 10038-4982 Attention: Steven P. Moskowitz, Esq. Facsimile: (212) 806-6006

To RioCan (prior to December 20, 2009):

c/o RioCan Holdings USA Inc. Yonge Eglinton Centre 2300 Yonge Street Suite 500, P.O. Box 2386 Toronto, Ontario M4P 1E4 Attention: Rags Davloor

Facsimile: (416) 866-3020

With a copy to:

c/o RioCan Holdings USA Inc. Yonge Eglinton Centre 2300 Yonge Street Suite 500, P.O. Box 2386 Toronto, Ontario M4P 1E4 Attention: Jonathan Gitlin

Facsimile: (416) 866-3020

With a copy to (if prior to December 20, 2009):

Goodmans LLP 250 Yonge Street, Suite 2400 Toronto, Ontario M5B 2M6 Attention: Juli Morrow Facsimile: (416) 979-1234

With a copy to (if on or after December 20, 2009):

Goodmans LLP 333 Bay Street, Suite 3400 Bay Adelaide Centre, West Tower Toronto, Ontario M5H 2S7 Attention: Juli Morrow Facsimile: (416) 979-1234

Any party hereto may change the address to which notice may be delivered hereunder by the giving of written notice thereof to the other Parties as provided herein below. Any notice or other communication delivered pursuant to this Section may be mailed by United States or Canadian certified air mail, return receipt requested, postage prepaid, deposited in a United States or Canadian Post Office or a depository for the receipt of mail regularly maintained by the United States Post Office or the Canadian Post Office, as applicable. Such notices, demands, consents and reports may also be delivered (i) by hand or reputable international courier service which maintains evidence of receipt or (ii) by facsimile with a confirmation copy delivery by hand or reputable international courier service which maintains evidence of receipt. Any notices, demands, consents or other communications shall be deemed given and effective when delivered by hand or courier or facsimile if sent before 5:00 p.m. on a Business Day, and otherwise, the Business Day next following the day of transmittal, or if mailed only, five (5) Business Days after mailing. Notwithstanding the foregoing, no notice or other communication shall be deemed ineffective because of refusal of delivery to the address specified for the giving of such notice in accordance herewith. The provisions of this Section 26 shall survive the Closings and/or a termination of this Agreement.

27. Amendments.

Except as otherwise expressly set forth in this Agreement, this Agreement may not be modified or terminated orally or in any manner other than by an agreement in writing signed by all the parties hereto or their respective successors in interest. The provisions of this Section 27 shall survive the Closings and/or a termination of this Agreement.

28. Governing Law; Jurisdiction; Construction.

- (a) This Agreement (i) shall be governed by and construed in accordance with the laws of the State of New York, without giving effect to principles of conflicts of law and (ii) shall be given a fair and reasonable construction in accordance with the intentions of the parties hereto and without regard to, or aid of, any rules of construction requiring construction against any party drafting this Agreement.
- (b) The parties agree that this Agreement has been made in New York, New York and that exclusive jurisdiction for matters arising under this Agreement shall be in the State courts in New York County, New York. Each party by signing this Agreement irrevocably consents to and shall submit to such jurisdiction.
- (c) Each party hereto acknowledges that it has participated in the drafting of this Agreement, and any applicable rule of construction to the effect that ambiguities are to be resolved against the drafting party shall not be applied in connection with the construction or interpretation hereof. Each party has been represented by independent counsel in connection with this Agreement.
- (d) Words importing the singular include the plural and vice versa. Words importing gender include all genders. If anything herein is to be done or held on a day that is not a Business Day, the same will be done or held either on the next succeeding Business Day or as otherwise expressly provided in this Agreement.

(e) The provisions of this Section 28 shall survive the Closings and/or a termination of this Agreement.

29. Partial Invalidity.

If any provision of this Agreement is held to be invalid or unenforceable as against any Person or under certain circumstances, the remainder of this Agreement and the applicability of such provision to other Persons or circumstances shall not be affected thereby. Each provision of this Agreement shall be valid and enforceable to the fullest extent permitted by law. The provisions of this Section 29 shall survive the Closings and/or a termination of this Agreement.

30. Counterparts.

This Agreement may be executed in any number of counterparts, each of which shall constitute an original, but all of which, taken together, shall constitute but one and the same instrument. This Agreement may be executed by facsimile, which shall be deemed an original for all purposes. In the event this Agreement is executed by the exchange of facsimile copies, the parties agree to exchange ink-signed counterparts promptly after the execution and delivery of this Agreement. The provisions of this Section 30 shall survive the Closings and/or a termination of this Agreement.

31. No Third Party Beneficiaries.

The warranties, representations, agreements and undertakings contained herein shall not be deemed to have been made for the benefit of any Person or entity other than the parties hereto and the Cedar Related Parties, except as otherwise expressly set forth in Section 33 hereof. The provisions of this Section 31 shall survive the Closings and/or a termination of this Agreement.

32. Waiver.

No failure or delay of either party in the exercise of any right given to such party hereunder or the waiver by any party of any condition hereunder for its benefit (unless the time specified herein for exercise of such right, or satisfaction of such condition, has expired) shall constitute a waiver of any other or further right nor shall any single or partial exercise of any right preclude other or further exercise thereof or any other right. The waiver of any breach hereunder shall not be deemed to be waiver of any other or any subsequent breach hereof. The provisions of this Section 32 shall survive the Closings and/or a termination of this Agreement.

33. Assignment.

Without the prior written consent of the other parties hereunder, no party hereto may assign this Agreement or any of its rights or obligations hereunder, and any purported unpermitted assignment shall be null and void. Notwithstanding the foregoing, (a) Cedar shall be permitted to assign this Agreement without the consent of any other party to any entity Controlled, directly or indirectly, by Cedar, provided that any such assignment by Cedar shall not release Cedar of its obligations under this Agreement and (b) RioCan shall have the right to designate either (i) a wholly-owned subsidiary of RioCan or (ii) an entity wholly-owned and Controlled by RioCan and an Institutional Investor and at least fifty-one percent (51%) owned, directly or indirectly, by RioCan and not more than forty-nine percent (49%) owned, directly or indirectly, by such Institutional Investor, to be the RioCan affiliated limited partner in the Partnership, in which case such subsidiary or other entity shall be entitled to the rights and benefits of RioCan under this Agreement and shall be deemed a third party beneficiary of the representations, warranties, certifications, covenants, agreements and indemnities made by Cedar for the benefit of RioCan pursuant to this Agreement, in any event under this clause (b), however, subject to any and all restrictions set forth in the Partnership Agreement and provided that (i) all of the Lenders shall have delivered Loan Approvals expressly permitting such a subsidiary or entity to be a limited partner in the Partnership, (ii) any such designation by RioCan shall not release RioCan of its obligations under this Agreement, and (iii) such designee shall assume all of the obligations of RioCan under this Agreement (including, without limitation, the obligation to pay the Earn-Out Proceeds as and when required hereunder) pursuant to an assumption agreement reasonably satisfactory to Cedar. The provisions of this Section 33 shall survive the Closings and/or a termination of this Agreement.

34. Binding Effect.

All of the covenants and agreements in this Agreement shall be binding upon the parties hereto and their respective successors and assigns and shall inure to the benefit of and be enforceable by the parties hereto and their respective successors and permitted assigns pursuant to the terms and conditions of this Agreement. The provisions of this Section 34 shall survive the Closings and/or a termination of this Agreement.

35. Entire Agreement.

This Agreement and the documents herein contemplated to be entered into by the parties sets forth the entire agreement between the parties pertaining to the subject matter hereof and there are no other terms, obligations, covenants, representations, statements or conditions, oral or otherwise, of any kind whatsoever. Any agreement hereafter made shall be ineffective to change, modify, discharge or effect an abandonment of this Agreement in whole or in part unless such agreement is in writing and signed by the party against whom enforcement of the change, modification, discharge or abandonment is sought. The provisions of this Section 35 shall survive the Closings and/or a termination of this Agreement.

36. Further Assurances.

After the Closings, the parties hereunder shall from time to time execute and deliver each to the other such documents and instruments and take such further actions as may be reasonably necessary or required to consummate the Transactions contemplated by this Agreement or more effectually implement and carry out the true intent and meaning of this Agreement. The provisions of this Section 36 shall survive the Closings and/or a

termination of this Agreement.

37. Paragraph Headings/Schedules.

The headings of the various sections of this Agreement have been inserted only for the purpose of convenience and are not part of this Agreement and shall not be deemed in any manner to modify, expand, explain or restrict any of the provisions of this Agreement. The provisions of this Section 37 shall survive the Closings and/or a termination of this Agreement. The Schedules referred to in this Agreement are attached to and form part of it.

38. Waiver of Trial by Jury.

THE PARTIES HERETO WAIVE TRIAL BY JURY IN ANY ACTION OR PROCEEDING ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT. THE PROVISIONS OF THIS SECTION 38 SHALL SURVIVE THE CLOSINGS AND/OR A TERMINATION OF THIS AGREEMENT.

39. Litigation Costs.

Notwithstanding anything to the contrary contained in this Agreement (including, without limitation, the terms of Section 5), in the event of any litigation arising in connection with this Agreement, the substantially prevailing party shall be entitled to recover from the substantially non-prevailing party its reasonable legal fees and expenses at trial and all appellate levels. The provisions of this Section 39 shall survive the Closings and/or a termination of this Agreement.

40. Currency.

Any and all amounts owing by any party hereto pursuant to this Agreement, shall be paid in lawful currency of the United States of America (i.e. U.S. Dollars). The provisions of this Section 40 shall survive the Closings and/or a termination of this Agreement.

41. Time of the Essence

Time shall be of the essence of this Agreement, provided that the time for doing or completing any matter may be extended or abridged by an agreement in writing between the parties hereto or their respective solicitors.

42. Press Releases.

Cedar and RioCan agree to consult with each other before issuing any press releases with respect to this Agreement or the Transactions and shall endeavor to agree as to the content of such press releases (which agreement shall not be unreasonably withheld, conditioned or delayed); provided, however, that nothing herein shall be deemed to prevent either party, or their respective affiliates, from issuing any press release if such parties shall believe, in the exercise of its reasonable judgment, that such press release is required to be made by applicable law. The provisions of this Section 42 shall survive the Closings.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the parties hereto have duly executed this Agreement on the day and year first above written.

CEDAR:

CEDAR SHOPPING CENTERS PARTNERSHIP, L.P., a Delaware limited partnership

By: Cedar Shopping Centers, Inc., a Maryland corporation, its general partner

By: /s/ Leo S. Ullman Leo S. Ullman President

RIOCAN:

RIOCAN HOLDINGS USA INC., a Delaware corporation

By: /s/ Rags Davloor Rags Davloor Chief Financial Officer

EXHIBIT A

LAND

Property Exhibit #

Columbus Crossing	A-1
Franklin Village	A-2
Loyal Plaza	A-3
Stop & Shop – Bridgeport	A-4
Blue Mountain Commons	A-5
Sunset Crossing	A-6
Shaw's Plaza	A-7

EXHBIIT B

AGREEMENT OF LIMITED PARTNERSHIP

OF

[CR] L.P.

Dated as of _____, 2009

THE PARTNERSHIP INTERESTS REPRESENTED BY THIS LIMITED PARTNERSHIP AGREEMENT HAVE BEEN ACOUIRED FOR INVESTMENT AND HAVE NOT BEEN REGISTERED WITH THE SECURITIES AND EXCHANGE COMMISSION UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR UNDER THE DELAWARE SECURITIES ACT, OR OTHER SIMILAR FEDERAL OR STATE STATUTES OR AGENCIES IN RELIANCE UPON EXEMPTIONS FROM REGISTRATION AS PROVIDED IN THOSE STATUTES. THE SALE, ACQUISITION, ASSIGNMENT, TRANSFER, EXCHANGE, MORTGAGE, PLEDGE OR OTHER DISPOSITION OF ANY PARTNERSHIP INTEREST IS RESTRICTED IN ACCORDANCE WITH THE PROVISIONS OF THIS LIMITED PARTNERSHIP AGREEMENT, AND THE EFFECTIVENESS OF ANY SUCH SALE, ACQUISITION, ASSIGNMENT, TRANSFER, EXCHANGE, MORTGAGE, PLEDGE OR OTHER DISPOSITION MAY BE CONDITIONED UPON, AMONG OTHER THINGS, RECEIPT BY THE GENERAL PARTNER OF THE PARTNERSHIP OF AN OPINION OF COUNSEL SATISFACTORY TO IT AND ITS COUNSEL THAT SUCH SALE, ACQUISITION, ASSIGNMENT, TRANSFER, EXCHANGE, MORTGAGE, PLEDGE OR OTHER DISPOSITION CAN BE MADE WITHOUT REGISTRATION UNDER THE SECURITIES ACT OF 1933, AS AMENDED, THE DELAWARE SECURITIES ACT AND OTHER APPLICABLE FEDERAL OR STATE STATUTES. BY ACQUIRING THE PARTNERSHIP INTERESTS REPRESENTED BY THIS LIMITED PARTNERSHIP AGREEMENT, EACH PARTNER REPRESENTS THAT IT WILL NOT SELL, ACQUIRE, ASSIGN, TRANSFER, EXCHANGE, MORTGAGE, PLEDGE OR OTHERWISE DISPOSE OF A PARTNERSHIP INTEREST WITHOUT REGISTRATION OR OTHER COMPLIANCE WITH THE AFORESAID STATUTES AND RULES AND REGULATIONS THEREUNDER AND THE TERMS AND PROVISIONS OF THIS LIMITED PARTNERSHIP AGREEMENT.

Table of Contents

		<u>Page</u>
ARTICLE	I GENERAL PROVISIONS	1
1.1	Formation.	1
1.2	Name.	2
1.3	Principal Office.	2
1.4	Registered Office and Registered Agent.	2
1.5	Qualification.	2
1.6	Purpose.	2
1.7	Term.	3
1.8	Definitions.	3
ARTICLE ALLOCAT	II CAPITAL CONTRIBUTIONS; CAPITAL ACCOUNTS; DISTRIBUTIONS;	19
2.1	Initial Capital Contributions.	19
2.2	Additional Capital Contributions.	20
	III PARTNERSHIP INTERESTS	25
3.1	Percentage Interests of General Partner and Limited Partners.	25
3.2	Capital Accounts.	25
3.3	Return of Capital.	27

ARTICLE	IV DISTRIBUTIONS	27
4.1	General.	27
4.2	Net Cash Flow.	27
4.3	Net Proceeds of a Capital Transaction.	27
4.4	Tax Payments.	27
4.5	Limitation on Distributions.	28
	VALVA O GARNON ON PROPERTY AND A COORD	••
	V ALLOCATION OF PROFITS AND LOSSES	28
5.1	Allocations for Accounting Purposes.	28
5.2	Special Allocations.	28
5.3	Other Allocation Rules.	31
5.4	Tax Allocations. Code Section 704(c).	32
	VI MANAGEMENT; LIABILITY OF PARTNERS; EXPENSES	32
6.1	Management.	32
6.2	Advisory Board.	36
6.3	Partnership Decisions.	38
6.4	Duties and Conflicts.	42
6.5	Partnership Counsel.	43
6.6	Exculpation/Indemnification.	43
6.7	Cedar LP Obligations	46
6.8	Transactions with Partners or Affiliates.	46
6.9	Rights of the Limited Partners.	47
6.10	Expenses.	48
6.11	Certain Tax Matters.	48
ADTICI E	VII INVESTMENT OPPORTUNITIES; NONCOMPETITION AND	49
	CITATION	49
7.1	Investment Opportunities.	49
7.2	Noncompetition.	52
7.3	Nonsolicitation.	53
ARTICLE	VIII BOOKS AND RECORDS, REPORTS TO PARTNERS	54
8.1	Bank Accounts.	54
8.2	Books of Account.	54
8.3	Audit and Reports.	54
8.4	Accountants.	57
8.5	Annual Budget.	58
8.6	Accounting Fee.	58
ADTICLE	IV TO ANCEED, WITHIND AWAL, DEMOVAL OF CENEDAL DADTNED	50
	IX TRANSFER; WITHDRAWAL; REMOVAL OF GENERAL PARTNER	58
9.1	Transfers Generally.	58
9.2	Succession by Operation of Law/Prorations/Cooperation.	59
9.3	General Conditions Applicable to Transfers.	59
9.4	Buy Sell Rights.	61
9.5	Right of First Refusal.	65
9.6	Bankruptcy or Withdrawal of a Partner.	67
9.7	Death or Incompetency of an Individual Partner.	68
9.8	General Partner's Withdrawal Rights.	68
9.9	Intentionally Omitted.	68
9.10	Removal of General Partner.	68
ARTICLE X TERMINATION 69		
10.1	Dissolution.	69
10.1	Termination.	70
10.2	Liquidating Partner.	71
10.3	Partnership Assets Reserved and Pending Claims.	71
10.5	No Redemption.	72
10.5	Governance.	72
10.7	Return of Capital.	73
10.7	Totali of Capital.	75
ARTICLE		73
ARTICI E	XI INTENTIONALLY OMITTED	
12.1		73
14.1	XII CONFIDENTIALITY	
12.2	XII CONFIDENTIALITY Disclosure.	73 73
12.2 12.3	XII CONFIDENTIALITY Disclosure. Confidentiality.	73 73
12.2 12.3	XII CONFIDENTIALITY Disclosure.	73 73
12.3	XII CONFIDENTIALITY Disclosure. Confidentiality.	73
12.3 ARTICLE	XII CONFIDENTIALITY Disclosure. Confidentiality. Additional Information/Deliveries. XIII POWER OF ATTORNEY	73 73 74 75
12.3 ARTICLE	XII CONFIDENTIALITY Disclosure. Confidentiality. Additional Information/Deliveries.	73 73 74
12.3 ARTICLE ARTICLE 14.1	XII CONFIDENTIALITY Disclosure. Confidentiality. Additional Information/Deliveries. XIII POWER OF ATTORNEY XIV AMENDMENTS; WAIVER Amendments; Waiver.	73 73 74 75 76 76
ARTICLE ARTICLE 14.1 ARTICLE	XII CONFIDENTIALITY Disclosure. Confidentiality. Additional Information/Deliveries. XIII POWER OF ATTORNEY XIV AMENDMENTS; WAIVER Amendments; Waiver. XV MISCELLANEOUS	73 73 74 75 76 76
ARTICLE 14.1 ARTICLE 15.1	XII CONFIDENTIALITY Disclosure. Confidentiality. Additional Information/Deliveries. XIII POWER OF ATTORNEY XIV AMENDMENTS; WAIVER Amendments; Waiver. XV MISCELLANEOUS Further Assurances.	73 73 74 75 76 76 76
ARTICLE 14.1 ARTICLE 15.1 15.2	XII CONFIDENTIALITY Disclosure. Confidentiality. Additional Information/Deliveries. XIII POWER OF ATTORNEY XIV AMENDMENTS; WAIVER Amendments; Waiver. XV MISCELLANEOUS Further Assurances. Notices.	73 73 74 75 76 76 76 76
12.3 ARTICLE 14.1 ARTICLE 15.1 15.2 15.3	XII CONFIDENTIALITY Disclosure. Confidentiality. Additional Information/Deliveries. XIII POWER OF ATTORNEY XIV AMENDMENTS; WAIVER Amendments; Waiver. XV MISCELLANEOUS Further Assurances. Notices. Applicable Law.	73 73 74 75 76 76 76 76 76
12.3 ARTICLE 14.1 ARTICLE 15.1 15.2 15.3 15.4	XII CONFIDENTIALITY Disclosure. Confidentiality. Additional Information/Deliveries. XIII POWER OF ATTORNEY XIV AMENDMENTS; WAIVER Amendments; Waiver. XV MISCELLANEOUS Further Assurances. Notices. Applicable Law. Headings, Etc.	73 73 74 75 76 76 76 76 76 76 78
12.3 ARTICLE 14.1 ARTICLE 15.1 15.2 15.3	XII CONFIDENTIALITY Disclosure. Confidentiality. Additional Information/Deliveries. XIII POWER OF ATTORNEY XIV AMENDMENTS; WAIVER Amendments; Waiver. XV MISCELLANEOUS Further Assurances. Notices. Applicable Law.	73 73 74 75 76 76 76 76 76

15.7	No Waiver.	79
15.8	Rule of Construction.	79
15.9	Severability.	79
15.10	Consents.	79
15.11	Entire Agreement.	79
15.12	Consent to Jurisdiction.	80
15.13	Counterparts.	80
15.14	Representations and Warranties.	80
15.15	Partnership Name.	82
15.16	Ownership of Partnership Property.	83
15.17	Time of the Essence.	83
15.18	Status Reports.	83
15.19	Waiver of Partition.	83
15.20	Calculation of Days	84
15.21	Dollar Amounts.	84
15.22	No Third-Party Rights.	84

Exhibits

EXHIBIT A PROPERTY MANAGEMENT AGREEMENT

EXHIBIT B PROPERTY INFORMATION

EXHIBIT C EXCLUDED PARTIES

AGREEMENT OF LIMITED PARTNERSHIP OF [CR] L.P., dated as of __, 2009, by and among __ LLC, a Delaware limited liability company having an office at 44 South Bayles Avenue, Port Washington, New York, 11050, as the initial General Partner (as hereinafter defined), and ___ LLC, a Delaware limited liability company having an office at 44 South Bayles Avenue, Port Washington, New York 11050, and RIOCAN HOLDINGS USA INC., a Delaware corporation, having an office at c/o RioCan Real Estate Investment Trust, RioCan Yonge Eglinton Centre, 2300 Yonge St., Suite 500, P.O. Box 2386, Toronto, Ontario, M4P 1E4, as the initial Limited Partners (as hereinafter defined), pursuant to the provisions of the Delaware Revised Uniform Limited Partnership Act, Title 6 of the Delaware Code, Section 17-101 et. seq. (as amended from time to time, including any successor thereto, the "Delaware Act"). Capitalized terms used herein are defined in Section 1.8 below or as elsewhere provided herein.

WHEREAS, the Partners desire to form a limited partnership pursuant to the provisions of the Delaware Act and to constitute themselves a limited partnership for the purposes set forth in Section 1.6 of this Agreement; and

WHEREAS, each Partner desires to make its respective capital contributions to the Partnership as described in this Agreement and to be a Partner of the Partnership.

NOW, THEREFORE, in consideration of the mutual promises, obligations and agreements contained herein, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties, intending to be legally bound, hereby agree as follows:

ARTICLE I GENERAL PROVISIONS

1.1 Formation.

- (a) The General Partner has caused the Partnership to be formed pursuant to the provisions of the Delaware Act and on the terms and conditions set forth in the Certificate. The rights and liabilities of all Partners shall be as provided under the Delaware Act, the Certificate and this Agreement. To the extent permitted by applicable law, the provisions of this Agreement shall override the provisions of the Delaware Act in the event of any inconsistency or contradiction between them.
- (b) In order to maintain the Partnership as a limited partnership under the laws of the State of Delaware, the Partnership shall, from time to time, take appropriate action, including the preparation and filing of such amendments to the Certificate and such other assumed name certificates, documents, instruments and publications as may be required by or desirable under law, including, without limitation, action to reflect:
 - (i) any change in the Partnership name; or
 - (ii) any correction of false or erroneous statements in the Certificate or the desire of the Partners to make a change in any statement therein in order that it shall accurately represent the agreement among the Partners.
- (c) Each necessary Partner shall further execute, and the Partnership shall file and record (or cause to be filed and recorded) and shall publish, if required by law, such other and further certificates, statements or other instruments as may be necessary or desirable under the laws of the State of Delaware or the state in which any of the Portfolio Investments is located in connection with the Partnership carrying on of its business. The General Partner shall be an authorized person of the Partnership for purposes of any filings under the Delaware Act and shall be authorized to execute and deliver on behalf of the Partnership any of the foregoing certificates.

1.2 Name.

The name of the Partnership is "[CR] L.P." All business of the Partnership shall be conducted under the name of the Partnership and title to all property, real, personal, or mixed, owned by or leased to the Partnership shall be held in such name.

1.3 Principal Office.

The principal place of business and office of the Partnership shall be located at 44 South Bayles Avenue, Port Washington, New York 11050 or at such other place or places in the United States as the General Partner may from time to time designate. The Partnership may have such additional offices and places of business as may be established at such other locations as may be determined from time to time by the Partners. The Partnership shall keep its books and records at its principal office. The General Partner shall give prompt notice to each Limited Partner of any change in the location of the Partnership's principal office.

1.4 Registered Office and Registered Agent.

The street address of the registered office of the Partnership in the State of Delaware is at 1209 Orange Street, Wilmington, Delaware 19801, or such other place in the State of Delaware as may from time to time be designated by the General Partner in accordance with the Delaware Act, and the Partnership's registered agent at such address is Corporation Trust Company. The General Partner shall promptly notify the Limited Partners of any change in the registered office or registered agent of the Partnership.

1.5 Qualification.

The General Partner shall qualify the Partnership to do business or become licensed in each jurisdiction where the activities of the Partnership make such qualification or licensing necessary or where failure to so qualify or become licensed would have an adverse effect on the limited liability of the Limited Partners.

1.6 Purpose.

- (a) The purpose and business of the Partnership shall be to (i) directly or indirectly, through one or more Property Owners, acquire, own and hold the Initial Properties and other Portfolio Investments, and in connection therewith, finance, own, operate, lease, develop, construct, redevelop, manage, dispose of (in whole or in part) and otherwise deal with the Initial Properties and any Partnership Assets acquired, directly or indirectly, by the Partnership in accordance with the terms hereof, (ii) engage in activities incidental or ancillary thereto; and (iii) engage in any other lawful acts or activities consistent with the terms of this Agreement and the foregoing for which limited partnerships may be organized under the Delaware Act.
- (b) The Partnership shall not engage in any other business or activity without the prior written consent of all the Partners.

1.7 Term.

The term of the Partnership (the "**Term**") commenced on the filing of the Certificate with the Secretary of State of the State of Delaware and shall continue until December 31, 2059, unless sooner terminated pursuant to the provisions hereof. The existence of the Partnership as a separate legal entity shall continue until the cancellation of the Certificate in the manner required by the Delaware Act.

1.8 Definitions.

For the purpose of this Agreement, the following terms shall have the following meanings:

"\$" or "U.S. Dollars" shall mean the official currency of the United States of America.

"Accountant" shall have the meaning ascribed thereto in Section 8.4.

"Adjusted Capital Account" means, with respect to any Partner, the balance, if any, in such Partner's Capital Account as of the end of the relevant Fiscal Year, after giving effect to the adjustments set forth herein and the following adjustments:

- (a) Credit to such Capital Account any amounts which such Partner is obligated to restore pursuant to the terms of this Agreement or is deemed to be obligated to restore pursuant to Treasury Regulations Section 1.704-1(b)(2)(ii)(c) or pursuant to the penultimate sentences of Treasury Regulations Sections 1.704-2(g)(1) and 1.704-2(i)(5); and
- (b) Debit to such Capital Account the items described in paragraphs (4), (5) and (6) of Treasury Regulations Section 1.704-1(b)(2)(ii)(d).

The foregoing definition of Adjusted Capital Account is intended to comply with the provisions of Section 1.704-1(b)(2)(ii)(d) of the Treasury Regulations to the extent relevant thereto and shall be interpreted consistently therewith.

"Advisory Board" shall mean the advisory board of the Partnership established pursuant to Section 6.2(a).

"Advisory Board Member" shall have the meaning ascribed thereto in Section 6.2(a).

"Affiliate" shall mean, with respect to any Person, any other Person that directly, or indirectly through one or more intermediaries, Controls, or is Controlled by, or is under common Control with, such Person. No Limited Partner shall be deemed to be an "Affiliate" of the Partnership or the General Partner solely by reason of

being a Limited Partner of the Partnership.

- "Agreement" shall mean this Agreement of Limited Partnership, as amended from time to time in accordance with the terms hereof.
 - "Applicable Contributions" shall have the meaning ascribed thereto in Section 2.2(d)(vii).
- "Approved Budget" shall mean the annual budget for each Portfolio Investment for the applicable Fiscal Year prepared by the applicable Property Manager pursuant to and in incordance with its Property Management Agreement that has been approved by the Advisory Board.
- "Approved Investment Structure" shall mean the acquisition of Portfolio Invesments, directly or indirectly, through REIT Property Subsidiary pursuant to the terms of this Agreement and/or the Purchase and Sale Agreement, as applicable.
- "Approved Leasing Plan" shall mean the leasing plan for each Portfolio Investment for the applicable Fiscal Year prepared by the applicable Property Manager pursuant to and in incordance with its Property Management Agreement that has been approved by the Advisory Board.
- "Bankruptcy Event' means, with respect to any Person, the occurrence of any of the following events: (i) the filing by it of a voluntary petition in bankruptcy, (ii) an adjudication that it is bankrupt or insolvent unless such adjudication is stayed or dismissed within sixty (60) days, or the entry against it of an order for relief of debtors in any bankruptcy or insolvency proceeding unless such order is stayed or dismissed within ninety (90) days, (iii) the filing by it of a petition or an answer seeking for itself any reorganization, arrangement, composition, readjustment, liquidation, dissolution or similar relief under any statute, law or regulation, (iv) the filing by it of an answer or other pleading admitting or failing to contest the material allegations of the petition filed against it in any proceeding of the nature described in the preceding clause (iii), (v) its seeking, consenting to or acquiescing in the appointment of a trustee, receiver or liquidator of it or of all or any substantial part of its assets, or (vi) the failure within ninety (90) days after the commencement of any proceeding against it seeking reorganization, arrangement, composition, readjustment, liquidation, dissolution or similar relief under any statute, law or regulation, to have the proceeding stayed or dismissed, or the failure within one hundred twenty (120) days after the appointment without its consent or acquiescence of a trustee, receiver or liquidator of it or of all or any substantial part of its assets to have such the appointment vacated or stayed, or the failure within ninety (90) days after the expiration of any such stay to have the proceeding dismissed or the appointment vacated, or (vii) such party generally shall be unable to pay, or shall admit in writing its inability to pay, its debts as they become due.
 - "Bonus Percentage" shall have the meaning ascribed thereto in Section 2.2(d)(vii).
- "Business Day" means any day other than Saturday, Sunday or any other day on which banks or savings and loan associations in New York, New York are not open for business.
 - "Buy Sell Applicable Purchase Price" shall have the meaning ascribed thereto in Section 9.4(a).
 - "Buy Sell Deposit" shall have the meaning ascribed thereto in Section 9.4(c).
 - "Buy Sell Election Date" shall have the meaning ascribed thereto in Section 9.4(b).
- "Buy Sell Exercise Period" means (a) with respect to the Cedar Partners, any time on or after (i) the third (3rd) anniversary of the date of this Agreement, or (ii) a Change of Control with respect to RioCan, or (iii) the removal of Cedar GP as the General Partner pursuant to Section 9.10 hereof or (iv) the removal of Cedar Operating Partnership (or its Affiliate) as a Property Manager by reason of a Cause Event (as defined in the applicable Property Management Agreement), and (b) with respect to RioCan, any time on or after (i) the third (3rd) anniversary of the date of this Agreement, or (ii) a Change of Control with respect to the Cedar Partners, or (iii) the removal of Cedar GP as the General Partner pursuant to Section 9.10 hereof or (iv) the removal of Cedar Operating Partnership (or its Affiliate) as a Property Manager by reason of a Cause Event (as defined in the applicable Property Management Agreement).
 - "Buy Sell Interests" shall have the meaning ascribed thereto in Section 9.4(a).
 - "Buy Sell Notice" shall have the meaning ascribed thereto in Section 9.4(a).
 - "Buy Sell Offeree" shall have the meaning ascribed thereto in Section 9.4(a).
 - "Buy Sell Offeror" shall have the meaning ascribed thereto in Section 9.4(a).
 - "Call Amounts" shall have the meaning ascribed thereto in Section 2.2(c)(i).
 - "Call Notice" shall have the meaning ascribed thereto in Section 2.2(c)(i).
- "Capital Account" shall mean, with respect to each Partner, an account to be maintained by the Partnership in accordance with the provisions of Treasury Regulations Section 1.704-1(b)(2)(iv) and Section 3.2 of this Agreement.
- "Capital Contribution" shall mean, with respect to any Partner, the amount of money and the Gross Asset Value of any property (other than money) contributed or deemed contributed by such Partner to the capital of the Partnership (for Portfolio Investments or otherwise, in each case, in accordance with this Agreement) (net of any liabilities secured by such property or to which such property is otherwise subject, or otherwise assumed by the Partnership in connection with the acquisition of such property).
 - "Capital Expenditures" means for any period, the amount expended for items capitalized under GAAP.
 - "Capital Transaction" means any of the following: (a) a sale, transfer or other disposition of all or a

portion of any Partnership Asset (other than tangible personal property that (i) is not sold, transferred or otherwise disposed in connection with the sale, transfer or other disposition of a fee interest or leasehold interest in real property and (ii) is otherwise sold, transferred or disposed in the ordinary course of business); (b) any condemnation or deeding in lieu of condemnation of all or a portion of any Partnership Asset; (c) any financing or refinancing of any Partnership Asset or other financing obtained by the Partnership or any of its Subsidiaries; (d) the receipt of proceeds due to any fire or other casualty to any Portfolio Investment or any other Partnership Asset; and (e) any other transaction involving Partnership Assets, in each case the proceeds of which, in accordance with GAAP, are considered to be capital in nature.

"Cause Event" means, with respect to any Partner, the occurrence of any of the following events: (a) such Partner committed fraud, willful misconduct or gross negligence in the performance of its duties and obligations under this Agreement; (b) such Partner is in material default in the performance or observance of any of its covenants or obligations under this Agreement, which default continues uncured for a period of sixty (60) days after written notice to such Partner, provided, that if such default is not reasonably susceptible of being cured with such sixty (60) day period and such Partner shall have commenced a cure of such default within such sixty (60) day period and is diligently pursuing a cure of such default, such Partner shall have such additional time as is reasonably necessary to cure such default; (c) in the case of either Cedar Partner only, the Cedar Partners shall not be under common Control; (d) in the case of either Cedar Partner only, if the Property Manager is under common Control with the Cedar Partners, the removal of the Property Manager by reason of a "Cause Event" pursuant to the terms of a Property Management Agreement; or (e) any Bankruptcy Event with respect to such Partner. Notwithstanding the foregoing, in no event shall any default under Section 2.2 (including, without limitation, a Default), 9.4 or 9.5 of this Agreement constitute or give rise to a Cause Event.

"Cedar Direct Contribution" shall have the meaning ascribed thereto in Section 2.2(e).

"Cedar GP" means _____ LLC, a Delaware limited liability company, and its permitted successors and assigns hereunder.

"Cedar LP" means LLC, a Delaware limited liability company, and its permitted successors and

"Cedar Operating Partnership" means Cedar Shopping Centers Partnership, L.P., a Delaware limited partnership.

"Cedar Partners" means Cedar GP and Cedar LP.

assigns hereunder.

"Certificate" means the Certificate of Limited Partnership for the Partnership that complies with Section 17-201 of the Delaware Act dated [____] filed with the Secretary of State of the State of Delaware pursuant to Section 17-206 of the Delaware Act, as the same may be amended and restated.

"Change of Control" shall mean, (a) with respect to the Cedar Partners, if (i) either of the Cedar Partners shall no longer be Controlled or at least fifty-one percent (51%) owned, directly or indirectly, by CSCI; (ii) there shall be consummated (x) any consolidation or merger of CSCI in which CSCI is not the continuing or surviving corporation or pursuant to which the stock of CSCI would be converted into cash, securities or other property, other than a merger or consolidation of CSCI in which the holders of CSCI's stock immediately prior to the merger or consolidation hold more than fifty percent (50%) of the stock or other forms of equity of the surviving corporation immediately after the merger or (y) any sale, lease, exchange or other transfer (in one transaction or series of related transactions) of all, or substantially all, the assets of CSCI; (iii) the board of directors of CSCI approves any plan or proposal for liquidation or dissolution of CSCI; or (iv) any Person acquires more than twenty-five percent (25%) of the issued and outstanding common stock of CSCI; and (b) with respect to RioCan, if (i) RioCan shall no longer be Controlled and at least fifty-one percent owned, directly or indirectly, by RioCan REIT; (ii) any direct or indirect interests in RioCan (other than direct or indirect interests in RioCan REIT) are held by any Person that is not an Institutional Investor; (iii) there shall be consummated (x) any consolidation or merger of RioCan REIT in which RioCan REIT is not the continuing or surviving corporation or pursuant to which the units of RioCan REIT would be converted into cash, securities or other property, other than a merger or consolidation of RioCan REIT in which the holders of RioCan REIT's units immediately prior to the merger or consolidation hold more than fifty percent (50%) of the units or other forms of equity of the surviving corporation immediately after the merger or (y) any sale, lease, exchange or other transfer (in one transaction or series of related transactions) of all, or substantially all, the assets of RioCan REIT; (iv) the board of directors of RioCan REIT approves any plan or proposal for liquidation or dissolution of RioCan REIT; or (v) any Person acquires more than twenty-five percent (25%) of the issued and outstanding units of RioCan REIT.

"Closing" shall mean the execution and delivery of this Agreement by the General Partner and the Limited Partners.

"Code" shall mean the United States Internal Revenue Code of 1986, as amended from time to time.

"Confidential Information" shall have the meaning ascribed thereto in Section 12.2(b).

"Consent Notice" shall have the meaning ascribed thereto as defined in Section 6.3(c).

"Control" means with respect to any specified Person, the power to direct the management and policies of such Person, directly or indirectly, whether through the ownership of voting securities or other beneficial interest, by contract or otherwise; and the terms "Controlling" and "Controlled" have the meanings correlative to the foregoing.

"CSCI" means Cedar Shopping Centers, Inc., a Maryland corporation.

"Default" shall have the meaning ascribed thereto in Section 2.2(d)(i).

"Default Contribution" shall have the meaning ascribed thereto in Section 2.2(d)(v).

- "Default Contribution Notice" shall have the meaning ascribed thereto in Section 2.2(d)(v).
- "Default Loan" shall have the meaning ascribed thereto in Section 2.2(d)(ii).
- "Default Loan Maturity Date" shall have the meaning ascribed thereto in Section 2.2(d)(iii).
- "Default Loan Rate" means eighteen percent (18%) per annum, compounded monthly.
- "Defaulting Partner" shall have the meaning ascribed thereto in Section 2.2(d)(i).
- "Defaulting Partner Contribution" shall have the meaning ascribed thereto in Section 2.2(d)(ii).
- "Delaware Act" shall have the meaning ascribed thereto in the Preamble.
- "Depreciation" shall mean, for each Fiscal Year or other period, an amount equal to the depreciation, amortization, or other cost recovery deduction allowable with respect to an asset for such Fiscal Year or other period, except that if the Gross Asset Value of an asset differs from its adjusted basis for federal income tax purposes at the beginning of such Fiscal Year or other period, Depreciation shall be an amount which bears the same ratio to such beginning Gross Asset Value as the federal income tax depreciation, amortization, or other cost recovery deduction for such Fiscal Year or other period bears to such beginning adjusted tax basis. If any asset shall have a zero adjusted basis for federal income tax purposes, Depreciation shall be determined utilizing any reasonable method selected by the Partners.
 - "Dissolution Event" shall have the meaning ascribed thereto in 10.1.
 - "Economic Risk of Loss" shall have the meaning ascribed thereto in Regulations Section 1.752-2.
- "Escrow Agent" shall mean any reputable, nationally recognized and financially solvent title insurance company designated by the Partner purchasing an Interest, an Indirect Owner, a Property Owner or a Portfolio Investment.
- **"Executive Order 13224"** shall mean Executive Order 13224 Blocking Property and Prohibiting Transactions with Persons Who Commit, Threaten to Commit, or Support Terrorism, issued by OFAC.
- "Fair Market Value" shall mean the value of the particular asset or interest in question determined on the basis of an arm's length transaction for cash between an informed and willing seller (under no compulsion to sell) and an informed and willing purchaser (under no compulsion to purchase), taking into account, among other things, the anticipated cash flow, taxable income and taxable loss attributable to the asset or interest in question. Except as otherwise expressly set forth herein, in the case of any asset other than a marketable security, the Fair Market Value shall be determined in good faith by the General Partner provided, that if the General Partner determines Fair Market Value other than by engaging an independent third-party appraiser, the Advisory Board shall have the right to object in its reasonable discretion to any determination of Fair Market Value made by the General Partner, in which case the Fair Market Value of the asset(s) in question shall be re-determined at the expense of the Partners (based on their respective Percentage Interests in the applicable Portfolio Investment) by an independent third-party appraiser selected mutually by the General Partner and the Advisory Board. In determining the value of any asset other than a marketable security, the General Partner may, but shall not be under any obligation to, engage an independent appraiser having recognized qualifications necessary in order to make such determination and the fees and expenses of such appraiser shall be borne by the Partnership. Except as otherwise expressly set forth herein, in the case of any marketable security at any date, the Fair Market Value of such security shall equal the closing sale price of such security on the Business Day (on which any national securities exchange is open for the normal transaction of business) next preceding such date, as appearing in any published list of any national securities exchange (other than NASDAQ Stock Market, Inc.) or in the Global Market List of NASDAQ Stock Market, Inc., or, if there is no such closing sale price of such security, the final price of such security at face value quoted on such Business Day by a financial institution of recognized standing which regularly deals in securities of such type.
- "Financing" means any indebtedness, financing or refinancing by debt, bonds, sale and leaseback, derivatives (e.g., hedging instruments) or other form of financing with respect to any Portfolio Investment or any of the direct or indirect interests in the Partnership's Subsidiaries or any debt or other similar monetary obligation of the Partnership or any of its Subsidiaries (but excluding trade payables incurred in the ordinary course of business).
- "Financing Document" shall mean any loan agreement, security agreement, mortgage, deed of trust, indenture, bond, note, debenture or other instrument or agreement relating to a Financing.
- "Fiscal Year" shall mean, except as otherwise required by law, the calendar year, except that the first Fiscal Year of the Partnership shall have commenced on the date of commencement of the Partnership and end on the next succeeding December 31, and the last Fiscal Year of the Partnership shall end on the date on which the Partnership shall terminate and commence on the January 1 immediately preceding such date of termination.
 - "GAAP" shall mean generally accepted accounting principles of the United States, consistently applied.
- "General Partner" shall mean the general partner or general partners, from time to time, of the Partnership authorized to carry out the management of the business and affairs of the Partnership pursuant to Article VI hereof. As of the date of this Agreement, the General Partner is Cedar GP.
- "Governing Agreements" means, with respect to each Subsidiary of the Partnership or any joint venture in which the Partnership owns a direct or indirect interest, its operating agreement, partnership agreement, shareholder agreement or similar governing agreement.
- "Governmental Authority" shall mean: (i) any government or political subdivision thereof, whether foreign or domestic, national, state, county, municipal or regional; (ii) any agency or instrumentality of any such government, political subdivision or other government entity (including any central bank or comparable agency);

"Gross Asset Value" shall mean, with respect to any asset, the asset's adjusted basis for federal income tax purposes, except as follows:

- (a) The initial Gross Asset Value of any asset contributed by a Partner to the Partnership shall be the gross Fair Market Value of such asset, as determined by the Partners (as evidenced by this Agreement or an amendment hereto);
- (b) The Gross Asset Values of all Partnership Assets shall be adjusted to equal their respective gross Fair Market Values, (taking Section 7701(g) of the Code into account), as of the following times: (i) the acquisition of an interest or an additional interest in the Partnership by any new or existing Partner in exchange for more than a de minimis Capital Contribution or other consideration; (ii) the distribution by the Partnership to a Partner of more than a de minimis amount of property or money as consideration for an Interest in the Partnership; (iii) a more than de minimis change in the Interests of the Partners; and (iv) the liquidation of the Partnership within the meaning of Treasury Regulations Section 1.704-1(b)(2)(ii)(g); provided, however, that adjustments pursuant to clauses (i) and (ii) above shall be made only if the General Partner, acting reasonably and in good faith, determines that such adjustments are necessary or appropriate to reflect the relative economic interests of the Partners;
- (c) The Gross Asset Value of any Partnership Asset distributed to a Partner shall be the gross Fair Market Value (taking Section 7701(g) of the Code into account) of such asset on the date of distribution;
- (d) The Gross Asset Values of Partnership Assets shall be increased (or decreased) to reflect any adjustments to the adjusted basis of such assets pursuant to Code Section 734(b) or Code Section 743(b), but only to the extent that such adjustments are taken into account in determining Capital Accounts pursuant to Treasury Regulations Section 1.704-1(b)(2)(iv)(m), clause (f) of the definition of Profits and Losses and Section 5.2(a)(vii); provided, however, that Gross Asset Values shall not be adjusted pursuant to this paragraph (d) to the extent the General Partner determines that an adjustment pursuant to paragraph (b) hereof is necessary or appropriate in connection with a transaction that would otherwise result in an adjustment pursuant to this paragraph (d); and
- (e) If the Gross Asset Value of an asset has been determined or adjusted pursuant to paragraphs (a), (b) or (d), such Gross Asset Value shall thereafter be adjusted by the Depreciation taken into account with respect to such asset for purposes of computing Profits and Losses;

"IFRS" shall mean International Financial Reporting Standards.

"Impositions" shall mean all taxes (including sales and use taxes), assessments (including all assessments for public improvements or benefits, whether or not commenced or completed prior to the date hereof), water, sewer or other rents, rates and charges, excises, levies, license fees, permit fees, inspection fees and other authorization fees and other charges, in each case whether general or special, ordinary or extraordinary, of every character (including all interest and penalties thereon), which at any time may be assessed, levied, confirmed or imposed by any Governmental Authority having jurisdiction over the Partnership, any of its Subsidiaries or any Portfolio Investment, as applicable, on or in respect of or be a lien upon (i) such Portfolio Investment or any estate or interest therein, (ii) any occupancy, use or possession of, or activity conducted on, such Portfolio Investment, or (iii) the rents from such Portfolio Investment or the use or occupancy thereof.

"Indemnified Losses" shall have the meaning ascribed thereto in Section 6.6(a)(iii).

"Indirect Owner" shall mean any entity that is a Subsidiary of REIT Property Subsidiary and is an indirect owner of assets constituting a Portfolio Investment.

"Initial Properties" means the Portfolio Investments acquired pursuant to the Purchase and Sale Agreement.

"Initial Real Property Costs" means with respect to any Portfolio Investment, as of the date of determination thereof, without duplication, the aggregate of:

- (a) the aggregate acquisition cost of such Portfolio Investment including, without any limitations or duplications, deposits paid on account of options for the purchase or leasing of such real property, net of interest earned, whether or not credited to the purchase price; cash payments made on account of the purchase price of such real property, whether paid before, at, or after acquisition of such real property; principal payments made as prepaid rentals or in lieu of rentals to acquire a leasehold interest in real property; and principal payments made upon mortgages, charges and encumbrances upon real property, paid at or in connection with the acquisition of such real property; and
- (b) the aggregate of all third party expenses ancillary to the acquisition of such real property, including, without any limitation or duplication, due diligence expenses, and subject to, clause (a) above, (y) any travel, lodging or meal expenses of the Partners or their Affiliates, real estate commissions, fees and expenses relating to obtaining financing, legal fees, consultants fees, land transfer taxes, survey expenses, registration fees, inspection fees, title premiums, insurance premiums, and all other acquisition expenses relating to such acquisition;

Notwithstanding the foregoing, Initial Real Property Costs shall not include the Net Consideration payable for any Initial Property.

"Institutional Investor" means any of the following types of entities (or any entity that is directly or indirectly wholly-owned (except de minimis interests) and Controlled by any of the following types of entities),

whether domestic or Canadian: (a) a commercial bank, trust company (whether acting individually or in a fiduciary capacity for another entity that constitutes an Institutional Investor), savings and loan association, savings bank, financing company or similar institution; (b) an insurance company; (c) an investment bank; or (d) an employee's welfare, benefit, profit-sharing, pension or retirement trust, fund or system (whether federal, state, municipal, private or otherwise); in each case on the condition that such Institutional Investor (i) is regularly engaged in the business of owning or operating commercial real estate properties, (ii) is recognized as a reputable investor, (iii) has net assets (in name or under management) in excess of One Billion Dollars (\$1,000,000,000), (iv) is not one of Persons described in Section 9.3(a) hereof, (v) is neither one of the Persons listed on Exhibit C nor Controlled by any such Persons, and (vi) is otherwise reasonably acceptable to the Cedar Partners, it being acknowledged that CPP Investment Board, a Canadian corporation, is acceptable to the Cedar Partners.

"Interest" means, with respect to each Partner at a given time, the interest of such Partner in the Partnership at such time, including the rights and obligations of such Partner as provided in the Agreement and under applicable law and any economic interest of such Partner in the Partnership.

"Investment Period" shall mean the period ending on the second anniversary of the Closing.

"Joint Venture Investment" means a Portfolio Investment that is not 100% owned, directly or indirectly, by the Partnership or any of its Subsidiaries.

"Lending Eligible Partner" shall have the meaning ascribed thereto in Section 2.2(d)(i).

"Lending Partner" shall have the meaning ascribed thereto in Section 2.2(d)(iii).

"Limited Partner" means, at any time, any Person admitted and remaining as a limited partner of the Partnership pursuant to the terms of this Agreement, including any substituted limited partners admitted to the Partnership pursuant to Article IX. As of the date of this Agreement, the Limited Partners of the Partnership are Cedar LP and RioCan.

"Liquidating Partner" shall have the meaning ascribed thereto in Section 10.3(a).

"Litigation" shall have the meaning ascribed thereto in Section 6.3(b)(vii).

"Majority Decision" shall have the meaning ascribed thereto in Section 6.3(b).

"Maximum Rate" means the highest lawful rate of interest allowable under applicable law.

"Necessary Expenses" shall mean expenses incurred or required to be incurred (without duplication) for (a) compliance with legal requirements, (b) Impositions, (c) amounts payable by the Partnership under Section 6.6, (d) obligations under any (i) ground lease existing as of the date hereof or entered into in accordance with the Purchase and Sale Agreement or this Agreement, pursuant to which the Partnership or any of its Subsidiaries is the lessee, to the extent the expenses thereunder (or the amounts thereof) are non-discretionary, (ii) other leases and contracts with third parties existing as of the date hereof or entered into in accordance with the Purchase and Sale Agreement or this Agreement, to the extent the expenses thereunder (or the amounts thereof) are nondiscretionary, (iii) Financing Documents existing as of the date hereof or entered into in accordance with the Purchase and Sale Agreement or this Agreement, other than amounts due upon maturity of the applicable Financing unless the payment of such amounts has been unanimously approved by the Advisory Board, and (iv) Governing Agreements with respect to Joint Venture Investments existing as of the date hereof or entered into in accordance with the Purchase and Sale Agreement or this Agreement, to the extent the expenses thereunder (or the amounts thereof) are non-discretionary, (e) utility charges, (f) amounts payable to or reimbursable to a Property Manager under its Property Management Agreement, (g) amounts payable to or reimbursable to the General Partner under this Agreement, (h) insurance, and (i) protecting against (or deemed necessary or prudent in the good faith judgment of the General Partner to protect against) injury to persons or damage to property, including, without limitation, in respect of security and life safety.

"Net Cash Flow" means, with respect to the Partnership, with respect to any period, the sum of all money available to the Partnership at the end of that period for distribution to its Partners after (1) payment of all debt service and other expenses (including, without limitation, payments due on or with respect to operating and maintenance expenses, general and administrative expenses, insurance costs, Impositions and other expenses paid or required to be paid) by the Partnership or any of its Subsidiaries; (2) satisfaction of the Partnership's and each of its Subsidiaries' liabilities as they come due; and (3) establishment of (and contributions to) such reserves as are required under any Financing Documents or additional reasonable reserves required to operate the Partnership and/or any of its Subsidiaries (including the Portfolio Investments) established in accordance with Section 6.3; provided, however, that Net Cash Flow shall not include Net Proceeds of a Capital Transaction, Capital Contributions, loans, tenant security deposits or earnest money deposits or any interest thereon so long as the Partnership or one of its Subsidiaries has a contingent obligation to return the same.

"Net Proceeds of a Capital Transaction" means the net cash proceeds (other than insurance proceeds for lost rental incomes) from a Capital Transaction less any portion thereof used to (i) establish (and contribute to) such reserves as are required under any Financing Documents or additional reasonable reserves required to operate the applicable Portfolio Investment and established in accordance with Section 6.3, (ii) repay any debts or other obligations of the Partnership and/or its applicable Subsidiaries in connection with such Capital Transaction, (iii) restore a Portfolio Investment following a casualty or condemnation, (iv) pay costs reasonably and actually incurred in connection with the Capital Transaction, or (v) pay creditors in the event of a liquidation. "Net Proceeds of a Capital Transaction" shall include all principal, interest and other payments as and when received with respect to any note or other obligation received by the Partnership and/or any of its Subsidiaries in connection with a Capital Transaction.

"Nonrecourse Deductions" shall have the meaning set forth in Treasury Regulations Section 1.704-2(b) (1). The amount of Nonrecourse Deductions for a Fiscal Year equals the excess, if any, of the net increase, if any, in the amount of Partnership Minimum Gain during that Fiscal Year, over the aggregate amount of any distributions during that Fiscal Year of proceeds of a Nonrecourse Liability that are allocable to an increase in Partnership Minimum Gain, determined according to the provisions of Treasury Regulations Section 1.704-2(c).

- "Nonrecourse Liability" shall have the meaning ascribed thereto in Treasury Regulations Section 1.704-2(b)(3).
 - "Notices" shall have the meaning ascribed thereto in Section 15.2.
 - "OFAC" means the Office of Foreign Assets Control of the United States Department of the Treasury.
 - "OFAC Lists" shall have the meaning ascribed thereto in Section 9.3(a).
 - "Open Call Amount" shall have the meaning ascribed thereto in Section 2.2(d)(i).
- "Partner" means, at any time, any person or entity admitted and remaining as a partner of the Partnership pursuant to the terms of this Agreement. As of the date of this Agreement, the Partners of the Partnership are Cedar GP, Cedar LP and RioCan.
- "Partner Nonrecourse Debt" means "partner non-recourse debt" as set forth in Treasury Regulations Section 1.704-2(b)(4).
- "Partner Nonrecourse Debt Minimum Gain" means an amount, with respect to each Partner Nonrecourse Debt, equal to the Partnership Minimum Gain that would result if such Partner Nonrecourse Debt were treated as a Nonrecourse Liability, determined in accordance with Treasury Regulations Section 1.704-2(i) (2) and (3).
- "Partner Nonrecourse Deductions" means "partner nonrecourse deductions" as set forth in Treasury Regulations Section 1.704-2(i)(2). For any Fiscal Year, the amount of Partner Nonrecourse Deductions with respect to a Partner Nonrecourse Debt equals the excess, if any, of the net increase, if any, in the amount of the Partner Nonrecourse Debt Minimum Gain attributable to such Partner Nonrecourse Debt over the aggregate amount of any distributions during such Year to the Partner that bears the Economic Risk of Loss for such Partner Nonrecourse Debt to the extent such distributions are from proceeds of such Partner Nonrecourse Debt and are allocable to an increase in Partner Nonrecourse Debt Minimum Gain, determined according to the provisions of Treasury Regulations Section 1.704-2(i)(2).
 - "Partnership" shall mean [CR] L.P., a Delaware limited partnership.
- "Partnership Assets" means the assets and property, whether tangible or intangible and whether real, personal, or mixed, at any time owned by or held for the benefit of the Partnership and all direct or indirect interests in the Partnership's Subsidiaries and the Portfolio Investments.
 - "Partnership Asset Price" shall have the meaning ascribed thereto in Section 9.4(a)
 - "Partnership Counsel" shall have the meaning ascribed thereto in Section 6.5.
 - "Partnership Decision" means a Majority Decision or a Unanimous Decision, as applicable.
- "Partnership Minimum Gain" shall mean the amount of "partnership minimum gain" determined in accordance with the principles of Treasury Regulations Section 1.704-2(d).
- "Partnership Subsidiary GP" shall mean _____, a Delaware limited liability company wholly owned by the Partnership that is the general partner of REIT.
- "Percentage Interest" means, as to any Partner, the percentage interest of such Partner specified in Section 3.1, as the same may be increased or decreased pursuant to Section 2.2(d)(vii) or 2.2(e).
- "Person" shall mean an individual, a corporation, a company, a voluntary association, a partnership, a joint venture, a limited liability company, a trust, an estate, an unincorporated organization, a Governmental Authority or other entity.
- "Portfolio Investment" shall mean all real property (including, without limitation, leasehold interests) now owned or hereafter acquired by the Partnership or any of its Subsidiaries pursuant to and in accordance with this Agreement, and all improvements, fixtures, tangible and intangible personal property, appurtenances, rights and interests in connection therewith, including, without limitation, the Initial Properties, and any equity interest now owned or hereafter acquired by the Partnership or any Subsidiary thereof in or relating to such real property.
- "**Profits**" and "Losses" means for each Fiscal Year or other period, an amount equal to the Partnership's taxable income or loss for such Fiscal Year or period, determined in accordance with Code Section 703(a) (for this purpose, all items of income, gain, loss, or deduction required to be stated separately pursuant to Code Section 703(a)(1) shall be included in taxable income or loss), with the following adjustments:
 - (a) Any income of the Partnership that is exempt from federal income tax, and not otherwise taken into account in computing Profits or Losses pursuant to this definition, shall be added to such taxable income or loss:
 - (b) Any expenditures of the Partnership described in Code Section 705(a)(2)(B) or treated as Code Section 705(a)(2)(B) expenditures pursuant to Treasury Regulations Section 1.704-1(b)(2)(iv) (i), and not otherwise taken into account in computing Profits or Losses pursuant to this definition, shall be subtracted from such taxable income or loss;
 - (c) In the event the Gross Asset Value of any Partnership Asset is adjusted pursuant to paragraph (b) or (c) under the definition of "Gross Asset Value," the amount of such adjustment shall be taken into account as gain or loss from the disposition of such Partnership Asset for purposes of computing Profits or Losses;

- (d) Gain or loss resulting from any disposition of Partnership property with respect to which gain or loss is recognized for federal income tax purposes shall be computed by reference to the Gross Asset Value of the property disposed of, notwithstanding that the adjusted tax basis of such property differs from its Gross Asset Value;
- (e) In lieu of the depreciation, amortization and other cost recovery deductions taken into account in computing such taxable income or loss, there shall be taken into account Depreciation for such Fiscal Year or other period, computed in accordance with the definition thereof;
- (f) In the event of an adjustment of the Gross Asset Value of any Partnership Asset which requires that the Capital Accounts of the Partnership be adjusted pursuant to Treasury Regulation Section 1.704-1(b)(2)(iv)(e),(f) and (m), the amount of such adjustment shall be taken into account as additional Profit or Loss; and
- (g) Notwithstanding any other provision of this definitional Section, any items which are specially allocated under this Agreement shall not be taken into account in computing Profits or Losses.
- "Property Management Agreement" means any property management agreement between a Property Owner and a Property Manager respecting the management of a Portfolio Investment entered into in accordance with this Agreement, as the same may be amended, restated, replaced, supplemented or modified from time to time in accordance with this Agreement.
- "Property Manager" means any property manager engaged by a Property Owner to manage a Portfolio Investment in accordance with this Agreement.
- "Property Owner" shall means any entity that is the direct owner of real property assets (including, without limitation, leasehold interests) constituting a Portfolio Investment.
 - "Proposed Transfer" shall have the meaning ascribed thereto in Section 12.2(a)(ii).
 - "Protected Person" shall have the meaning ascribed thereto in Section 6.6(d).
- "Purchase and Sale Agreement" means that certain Agreement Regarding Purchase of Partnership Interests, dated October 26, 2009, between Cedar Operating Partnership and RioCan.
 - "Purchasing Partner" shall have the meaning ascribed thereto in Section 9.4(c).
 - "Receiving Party" shall have the meaning ascribed thereto in Section 7.1(c).
 - "Referring Party" shall have the meaning ascribed thereto in Section 7.1(b).
- "Regulations" or "Treasury Regulations" means the Income Tax Regulations promulgated under the Code as such regulations may be amended from time to time (including Temporary Regulations).
 - "Regulatory Allocations" shall have the meaning ascribed thereto in Section 5.2(a)(viii).
- "REIT" shall mean _____, a Delaware limited partnership that has elected to be treated as a real estate investment trust for United States income tax purposes.
- "REIT Property Subsidiary" shall mean _____, a Delaware limited partnership that is owned 99.9% by REIT, as the limited partner, and 0.1% by REIT Subsidiary GP, as the general partner.
- "REIT Subsidiary GP" shall mean _____, a Delaware limited liability company wholly owned by REIT that is the general partner of REIT Property Subsidiary.
 - "Related Party" shall have the meaning ascribed thereto in Section 6.8(b).
 - "Related Party Transaction" shall have the meaning ascribed thereto in Section 6.8(a).
 - "Representatives" shall have the meaning ascribed thereto in Section 12.2(a).
 - "Restricted Party" shall have the meaning ascribed thereto in Section 7.1(a).
- "RioCan" means RioCan Holdings USA Inc., a Delaware corporation, and its permitted successors and assigns hereunder.
 - "RioCan REIT" mean RioCan Real Estate Investment Trust, an Ontario trust.
 - "ROFR Interest" shall have the meaning ascribed thereto in Section 9.5(a).
 - "ROFR Offer Notice" shall have the meaning ascribed thereto in Section 9.5(a).
 - "ROFR Offeree" shall have the meaning ascribed thereto in Section 9.5(a).
 - "ROFR Offeror" shall have the meaning ascribed thereto in Section 9.5(a).
 - "ROFR Response Notice" shall have the meaning ascribed thereto in Section 9.5(b).
 - "ROFR Sale" shall have the meaning ascribed thereto in Section 9.5(a).
 - "ROFR Third Party Offer" shall have the meaning ascribed thereto in Section 9.5(c).

"Selling Partner" shall have the meaning ascribed thereto in Section 9.4(c).

"Shortfall" means that the gross receipts of the Partnership, its applicable Subsidiary or the applicable Portfolio Investment (including, without limitation, proceeds under any applicable Financings) together with the proceeds of any accessible reserve account maintained by or on behalf of the Partnership or its applicable Subsidiary, are reasonably anticipated to be insufficient to pay when due (a) all expenses incurred or reasonably anticipated to be incurred by the Partnership or any of its Subsidiaries to own, operate, lease, develop, construct, redevelop, manage, dispose of (in whole or in part) or otherwise deal with such Portfolio Investment pursuant to the applicable Approved Budget, and (b) all Necessary Expenses, irrespective of whether the same shall have been included in the applicable Approved Budget.

"Subsidiary" of any Person shall mean (i) a corporation all or any portion of the outstanding voting stock of which is owned, directly or indirectly, by such Person or by one or more other Subsidiaries of such Person or by such Person and one or more Subsidiaries thereof, or (ii) any other Person (other than a corporation) in which such Person, or one or more other Subsidiaries of such Person or such Person and one or more other Subsidiaries thereof, directly or indirectly, has an ownership interest and/or the power to Control such other Person. The Subsidiaries of the Partnership include, without limitation, REIT, Partnership Subsidiary GP, REIT Property Subsidiary, REIT Subsidiary GP, the Indirect Owners and the Property Owners.

"Target Investment" shall have the meaning ascribed thereto in Section 7.1(a).

"Tax Matters Partner" shall have the meaning ascribed thereto in Section 6.11(a).

"Tax Payments" shall have the meaning ascribed thereto in Section 4.4.

"Taxed Partner" shall have the meaning ascribed thereto in Section 4.4.

"Term" shall have the meaning ascribed thereto in Section 1.7.

"**Termination**" shall mean the date of the cancellation or withdrawal of the Certificate by the filing of a Certificate of Cancellation of the Partnership in the Office of the Secretary of State of the State of Delaware pursuant to Section 17-203 of the Delaware Act.

"Transaction Documents" shall have the meaning ascribed thereto in Section 15.14(a)(ii).

"Transfer" shall have the meaning ascribed thereto in Section 9.1.

"Unanimous Decision" shall have the meaning ascribed thereto in Section 6.3(a).

"United States" or "U.S." shall mean the United States of America, its territories and possessions, any State of the United States and the District of Columbia, as the context requires.

"Withdrawal Event" shall have the meaning ascribed thereto in Section 9.6.

"Withdrawn Partner" shall have the meaning ascribed thereto in Section 9.6.

ARTICLE II CAPITAL CONTRIBUTIONS; CAPITAL ACCOUNTS; DISTRIBUTIONS; ALLOCATIONS

2.1 Initial Capital Contributions.

(a) As of the date of this Agreement, the Partners shall be deemed to have made Capital Contributions, and the Capital Accounts of the Partners shall be, as follows:

Capital Contribution

 Cedar GP:
 \$1

 Cedar LP:
 \$19

 RioCan:
 \$80

(b) Except with the prior written consent of all of the Partners or as otherwise provided in this Agreement, no Partner shall be required or permitted to make any further Capital Contribution to the Partnership.

2.2 Additional Capital Contributions.

(a) Additional Capital Contribution Obligations. Each Partner shall be required to make additional Capital Contributions from time to time in accordance with this Section 2.2 to fund its Percentage Interest of (i) any Initial Real Property Costs for a Portfolio Invesment approved for acquisition in accordance with Section 7.1 and (ii) any Shortfall; provided, that if any capital call to the Cedar Partners for Capital Expenditures in any single Fiscal Year, when aggregated with all prior capital calls for Capital Expenditures funded by the Cedar Partners in such Fiscal Year with respect to all of the Portfolio Investments, exceeds Five Million Dollars (\$5,000,000), such capital call shall require the unanimous approval of the Advisory Board. No Partner shall be entitled to interest on its Capital Contributions. All payments by the Partners to the Partnership pursuant to this Section 2.2 shall be made in U.S. Dollars and in immediately available funds.

(b) Net Consideration for the Initial Properties. On each Closing Date (as defined in the Purchase and Sale Agreement) RioCan shall contribute to the Partnership as a Capital Contribution, the Net Consideration (as defined in the Purchase and Sale Agreement) related to the applicable Closing (as defined in the Purchase and Sale Agreement) in exchange for the transfer by Cedar Operating Partnership (or its Subsidiaries) of all of the ownership interests in the related Indirect Owners and/or Property Owners, as applicable, to REIT Property Subsidiary, all in accordance with the Purchase and Sale Agreement. The amounts contributed by RioCan shall be distributed to the Cedar Partners.

(c) <u>Call Amounts and Call Notices</u>.

- (i) Each Partner's Capital Contribution required to be made pursuant to subsections (a) and (b) above shall be paid to the Partnership as calls are made by the General Partner upon the Partners, in such amounts (the "Call Amounts") and on such dates as shall be specified by the General Partner upon at least ten (10) Business Days' prior written notice (the "Call Notice") by the General Partner. With respect to each Call Notice, the Call Amount therein shall not exceed the applicable Partner's Percentage Interest of the total amount required to be funded by all Partners. The General Partner may withdraw a Call Notice at any time. Notwithstanding the foregoing, the parties acknowledge and agree that the Advisory Board (at the election of a majority of the Advisory Board Members) shall be entitled to send a Call Notice to the Partners in the event that the General Partner shall have failed to make a call in accordance with Section 2.2(a) if, in the reasonable judgment of such Advisory Board Members, any further delay in making such a capital call would have an imminent material adverse effect on a Portfolio Investment, the Partnership and/or any of its Subsidiaries, in which event the provisions of this Section 2.2 shall apply to such Call Notice
- (ii) In addition to the Call Amount, each Call Notice shall set forth (A) whether it relates to a Portfolio Investment (in each case including a detailed itemized listing of such expenses) and other fees payable by the Partnership to the General Partner (or its Affiliate) pursuant to this Agreement, and (B) the aggregate amount of Capital Contributions to be made by the Partners on such date.

(d) <u>Defaults</u>.

- (i) If any Partner shall fail to contribute all or a portion of any Call Amount on or before the date set forth on the applicable Call Notice (each of the foregoing, a "Default", such defaulting Partner being herein referred to as a "Defaulting Partner," and the unpaid portion of such Defaulting Partner's share of any Call Amount, the "Open Call Amount"), such Defaulting Partner shall remain liable in respect of its obligation to fund its Call Amount and any other amounts due and payable by such Partner hereunder, and the General Partner or its Affiliate that is a Limited Partner (or if the Defaulting Partner is the General Partner or such Affiliate, the other Limited Partner) may elect, in its sole discretion, to take any one or more of the remedial actions set forth in the remaining provisions of this subsection (d), provided that the General Partner or such Affiliate (or if the Defaulting Partner is the General Partner or such Affiliate, the other Limited Partner) shall have funded its entire Call Amount under the applicable Call Notice and does not have an outstanding Default Loan made to it hereunder (such non-defaulting Partner being referred to herein as the "Lending Eligible Partner").
- Default Loans. The Lending Eligible Partner shall have the right, but not the (ii) obligation, to make a loan (a "Default Loan") to such Defaulting Partner in an amount equal to the Open Call Amount at any time within forty-five (45) days after the funding date set forth on the applicable Call Notice, provided that such Defaulting Partner has not funded the entire Open Call Amount prior to the making of such Default Loan. If a Default Loan shall be made in accordance with this Section 2.2(d)(ii), the Lending Partner shall notify the Defaulting Partner of the amount and date of the Default Loan, which, subject to Section 2.2(d)(vi) below, shall be deemed to be a Capital Contribution (a "Defaulting Partner Contribution") made by the Defaulting Partner, and the Capital Account of the Defaulting Partner shall be credited to reflect the payment of the proceeds of the Default Loan to the Partnership. Each Default Loan shall be deemed to be made to the Defaulting Partner, with the proceeds of each Default Loan being delivered to the Partnership by the Lending Eligible Partner making same in immediately available funds on such Defaulting Partner's behalf. A Default Loan shall be deemed to have been advanced on the date actually advanced. Default Loans shall earn interest on the outstanding principal amount thereof at a rate equal to the lesser of (i) the Default Loan Rate or (ii) the Maximum Rate, from the date actually advanced until the same is repaid in full.

- (iii) Default Loans shall be secured as provided in Section 2.2(d)(iv) and shall have a term of one hundred eighty (180) days (the "Default Loan Maturity Date"). A Lending Eligible Partner making a Default Loan (a "Lending Partner") may, in the exercise of such Partner's sole and absolute discretion, extend the term of a Default Loan for a period(s) to be determined by such Partner. If a Default Loan has been made, the Defaulting Partner shall not receive any distributions of Net Cash Flow or Net Proceeds of a Capital Transaction or any proceeds from the transfer of all or any part of its Interest while the Default Loan, including all interest thereon, if applicable, remains unpaid. Instead, the Defaulting Partner's share of Net Cash Flow and Net Proceeds of a Capital Transaction or such other proceeds shall first be paid to the Lending Partner until all Default Loans to such Defaulting Partner, including interest thereon, shall have been repaid in full. Such payments shall be applied first to accrued interest on such Default Loans and then to the repayment of the principal amounts thereof, but shall be considered, for all other purposes of this Agreement, to have been distributed to the Defaulting Partner. Distributions of Net Cash Flow and Net Proceeds of a Capital Transaction to such Defaulting Partner shall be immediately reinstated prospectively upon the full repayment of a Default Loan, including all accrued and unpaid interest thereon, to the Lending Partner. If a Default Loan, including all accrued and unpaid interest thereon, has not been repaid in full on or before the date the same is due, in addition to any other rights or remedies provided in this Agreement, the Lending Partner shall have all rights and remedies available at law or in equity against the Defaulting Partner. The Defaulting Partner shall be liable for the reasonable fees and expenses incurred by the Lending Partner (including, without limitation, reasonable attorneys' fees and disbursements) in connection with any enforcement or foreclosure upon any Default Loan and such costs, to the extent enforceable under applicable law, shall be added to the principal amount of the applicable Default Loan. In addition, at any time during the term of such Default Loan, the Defaulting Partner shall have the right to repay, in full, the Default Loan (including interest and other charges).
- (iv) The Defaulting Partner shall be deemed to have pledged to the Lending Partner, and granted to such Lending Partner, a continuing first priority security interest in all of the Defaulting Partner's Interest to secure the payment of the principal of, and interest on, any Default Loans made in accordance with the provisions hereof, and for such purpose this Agreement shall constitute a security agreement. The Defaulting Partner shall promptly execute, acknowledge and deliver such financing statements, continuation statements or other documents and take such other actions as the Lending Partner shall request in order to perfect or continue the perfection of such security interest; and, if the Defaulting Partner shall fail to do so within seven (7) days after demand therefor, the Lending Partner is hereby appointed the attorney-infact of, and is hereby authorized on behalf of, the Defaulting Partner, to execute, acknowledge and deliver all such documents and take all such other actions as may be required to perfect such security interest. Such appointment and authorization are coupled with an interest and shall be irrevocable.
- Default Contributions. Except as provided in Section 2.2(d)(vi) below, in lieu of (v) making a Default Loan, a Lending Eligible Partner may elect to make a Capital Contribution to the Partnership (a "Default Contribution") in the amount of the Open Call Amount by delivering a notice (a "Default Contribution Notice") to the Defaulting Partner which shall include the following statement set forth in all capital letters "NOTE: YOU HAVE FAILED TO MAKE A REQUIRED CAPITAL CONTRIBUTION TO [CR], L.P. IN THE AMOUNT OF \$_____, AND THE UNDERSIGNED CAN ELECT TO FUND THE SAME AS A "DEFAULT CONTRIBUTION" AS DEFINED IN SECTION 2.2(D)(V) OF THE AGREEMENT OF LIMITED PARTNERSHIP OF [CR], L.P. IF SUCH REQUIRED CAPITAL CONTRIBUTION IS NOT MADE BY YOU ON OR BEFORE SEVEN (7) DAYS FOLLOWING THE DATE HEREOF." A Lending Eligible Partner shall have the right, but not the obligation, to make a Default Contribution to the Partnership in an amount equal to the Open Call Amount at any time within forty-five (45) days after the seventh (7th) day following the delivery of a Default Contribution Notice provided that such Defaulting Partner has not funded the entire Open Call Amount prior to the making of such Default Contribution. If a Default Contribution(s) shall be made in accordance with this Section 2.2(d)(v), the Lending Eligible Partner shall notify the Defaulting Partner of the amount and date of the Default Contribution(s), and the Capital Account of the Lending Eligible Partner shall be credited to reflect the contribution of the Default Contribution to the Partnership. A Default Contribution shall be deemed to have been made on the date actually delivered to the Partnership.
- (vi) A Lending Partner may elect at any time to convert all of a Default Loan into a Default Contribution on the terms hereinafter provided at any time after the seventh (7th) day following the delivery of a Default Contribution Notice. If a Default Loan is converted into a Default Contribution, then as of the date of such conversion, (1) the Lending Partner will be deemed to have made a Default Contribution in the amount of the outstanding balance of principal and accrued and unpaid interest under such Default Loan, (2) the Defaulting Partner shall be treated as receiving a distribution in the amount of such Default Contribution which distribution shall be deemed as having repaid the outstanding balance of principal and interest under such Default Loan, and (3) the Defaulting Partner Contribution shall be deemed refunded and shall be null and void *ab initio*.

- (vii) At the time the Lending Eligible Partner makes a Default Contribution or converts all of a Default Loan into a Default Contribution, (x) the Percentage Interest of each Partner shall be recalculated to equal the percentage equivalent of a fraction the numerator of which is the amount by which (A) the sum of (1) all Applicable Contributions made or deemed made by such Partner, and (2) an amount equal to ten percent (10%)(the "Bonus Percentage") of all Default Contributions made or deemed made by such Partner, exceeds (B) the Bonus Percentage of all Default Contributions made or deemed made by the other unaffiliated Partners (i.e., for purposes of this subparagraph (vii), the Cedar Partners shall be "affiliates" of each other, but shall be "unaffiliated" with RioCan); and the denominator of which is the total amount of all Applicable Contributions by all Partners, and (y) the Capital Accounts and Capital Contributions of each Partner shall be adjusted so that the ratio of such Partner's Capital Account and Capital Contributions to the aggregate Capital Accounts and aggregate Capital Contributions, respectively, of all Partners is equal to such Partner's Percentage Interest (as adjusted hereunder). As used herein, "Applicable Contributions" means all Capital Contributions to the Partnership by all Partners, including Default Contributions and Defaulting Partner Contributions, but excluding all Defaulting Partner Contributions that are deemed to have been refunded pursuant to clause (3) of Section 2.2(d)(vi).
- (viii) Withdrawal. If a Partner is not a Lending Eligible Partner but nevertheless funded all of its applicable Call Amount, or such Partner elects not to make a Default Loan or Default Contribution (or fails to make a Default Loan or Default Contribution in the time periods provided in Section 2.2(d)(ii) and Section 2.2(d)(v), as applicable), at the election of such Partner, the Capital Contribution advanced by such Partner in respect of such Call Amount shall promptly be returned by the Partnership to such Partner
- (ix) No Third Party Beneficiaries. The right of a Partner to send a Call Notice or to make a Default Loan or Default Contribution shall not confer upon any creditor or other third party having dealings with the Partnership or any of its Subsidiaries any right, claim or other benefit, including the right to require any such Call Notice, Default Loan or Default Contribution.
- If Cedar LP is required to make additional Capital Contributions pursuant to Section 2.2(a) to (e) fund Initial Real Property Costs or Shortfalls with respect to any Portfolio Investment (other than any of the Initial Properties), and Cedar LP, in good faith, has determined that making such additional Capital Contributions directly to the Partnership may adversely impact CSCI's qualification as a "real estate investment trust" in accordance with Section 856 of the Code or otherwise may have material adverse tax consequences to the Cedar Partners, CSCI or Cedar Operating Partnership, Cedar LP shall have the right to make a contribution (such contribution, a "Cedar Direct Contribution") to REIT Property Subsidiary, in an amount equal to the applicable Call Amount in lieu of making a Capital Contribution of such amount to the Partnership. Cedar LP shall notify RioCan of its intention to make a Cedar Direct Contribution and the Partners shall work in good faith to amend REIT Property Subsidiary's partnership agreement to reflect the Cedar Direct Contribution and to amend this Agreement to reflect the dilution of the interests of Cedar LP in the Partnership that results from RioCan and Cedar GP making contributions to the Partnership with respect to the applicable Call Notice pursuant to Section 2.2(a) and as further provided in this Section 2.2(e). The obligation of Cedar LP to make additional Capital Contributions under Section 2.2(a) shall be reduced by the amount of any Cedar Direct Contribution. It is the intention of the Partners that following such amendments the aggregate interests, directly and indirectly, of the Cedar Partners in the assets owned by REIT Property Subsidiary shall equal twenty percent (20%) and the aggregate interests, directly and indirectly, of RioCan in the assets owned by REIT Property Subsidiary shall equal eighty percent (80%), subject, in each case, to any applicable adjustment that may have occurred pursuant to Section 2.2(d)(vii). Prior to or simultaneously with the funding of the first such Cedar Direct Contribution, REIT Property Subsidiary's partnership agreement shall also be amended to provide for the admission of Cedar LP as an additional limited partner of REIT Property Subsidiary and the inclusion of such other provisions that are analogous to the provisions of this Agreement as may be necessary to preserve the substantive rights and obligations of the Partners provided herein in all material respects (e.g., with respect to transfer rights, in order to provide that all of the interests of Cedar LP in the Partnership and REIT Property Subsidiary will be included in a sale pursuant to Section 9.4 or 9.5 and the analogous provisions of REIT Property Subsidiary's partnership agreement).

ARTICLE III PARTNERSHIP INTERESTS

3.1 Percentage Interests of General Partner and Limited Partners.

The Percentage Interest of Cedar GP as a general partner in the Partnership shall be one percent (1%), the Percentage Interest of Cedar LP as a limited partner in the Partnership shall be nineteen percent (19%) and the Percentage Interest of RioCan as a limited partner in the Partnership shall be eighty percent (80%). The Percentage Interests shall not be changed without the prior written consent of all of the Partners, except as expressly provided in Section 2.2(d)(vii) or (e).

3.2 Capital Accounts.

(a) <u>Maintenance of Capital Accounts</u>. The Partnership shall establish and maintain a separate "Capital Account" for each Partner on the books of the Partnership in accordance with the following provisions for accounting purposes:

- (i) To each Partner's Capital Account there shall be credited such Partner's Capital Contributions, such Partner's allocable share of Profits, and any items in the nature of income or gain that are specially allocated to such Partner under this Agreement, and the amount of any Partnership liabilities that are assumed by such Partner in accordance with the terms hereof (other than liabilities that are secured by any Partnership Asset distributed to such Partner).
- (ii) To each Partner's Capital Account there shall be debited the amount of cash and the Gross Asset Value of any Partnership property distributed to such Partner pursuant to any provision of this Agreement (net of liabilities secured by such distributed property that such Partner is considered to assume or take subject to under Code Section 752), such Partner's allocable share of Losses, and any items in the nature of expenses or losses that are specially allocated to such Partner under this Agreement, and the amount of any liabilities of such Partner that are assumed by the Partnership (other than liabilities that are secured by any property contributed by such Partner to the Partnership).
- (iii) In the event an Interest is transferred in accordance with the terms of this Agreement, the transferee shall succeed to the Capital Account of the transferor to the extent it relates to the transferred Interest. In the case of Transfer of an Interest at a time when an election under Code Section 754 is in effect, the Capital Account of the transferee Partner shall not be adjusted to reflect the adjustments to the adjusted tax bases of Partnership property required under Code Sections 754 and 743, except as otherwise permitted by Treasury Regulations Section 1.704-1(b)(2)(iv)(m).
- (iv) In determining the amount of any liability for purposes of paragraphs (a) and (b) above, there shall be taken into account Code Section 752(c) and the Treasury Regulations promulgated thereunder, and any other applicable provisions of the Code and Regulations.
- (v) The foregoing provisions and the other provisions of this Agreement relating to the maintenance of Capital Accounts are intended to comply with Treasury Regulations Section 1.704-1(b) and 1.704-2, and shall be interpreted and applied in a manner consistent with such Regulations.
- (b) Adjustments of Capital Accounts. The Partnership shall revalue the Capital Accounts of the Partners in accordance with Regulations Section 1.704-1(b)(2)(iv)(f) at the following times: (i) immediately prior to the contribution of more than a *de minimis* amount of money or other property to the Partnership by a new or existing Partner as consideration for an interest in the Partnership; (ii) the distribution by the Partnership to a Partner of more than a *de minimis* amount of property as consideration for an Interest; and (iii) the liquidation of the Partnership within the meaning of Treasury Regulations Section 1.704-1(b)(2)(ii)(g); provided, that adjustments pursuant to clauses (i) and (ii) above need not be made if the General Partner reasonably determines that such adjustments are not necessary or appropriate to reflect the relative economic interest of the Partners.

3.3 Return of Capital.

No Partner shall be liable for the return of the Capital Contributions (or any portion thereof) of any other Partner, it being expressly understood that any such return shall be made solely from the Partnership Assets. No Partner shall be required to pay to the Partnership or to any other Partner any deficit in its Capital Account upon dissolution of the Partnership or otherwise, and no Partner shall be entitled to withdraw any part of its Capital Contributions or Capital Account, to receive interest on its Capital Contributions or Capital Account or to receive any distributions from the Partnership, except as expressly provided for in this Agreement or under the Delaware Act as then in effect.

ARTICLE IV DISTRIBUTIONS

4.1 General.

Net Cash Flow and/or Net Proceeds of a Capital Transaction shall be distributed to the Partners as set forth in Section 4.2 and 4.3 below.

4.2 Net Cash Flow.

Subject to Section 10.2, Net Cash Flow shall be distributed monthly (if available) by the General Partner to the Partners in accordance with their respective Percentage Interests.

4.3 Net Proceeds of a Capital Transaction.

Subject to Section 10.2, Net Proceeds of a Capital Transaction shall be distributed by the General Partner as soon as practicable after the receipt thereof to the Partners in accordance with their respective Percentage Interests.

4.4 Tax Payments.

To the extent that any taxes or withholding taxes are due on behalf of or with respect to any Partner and the Partnership is required by law to withhold or to make such tax payments ("Tax Payments"), the Partnership shall withhold such amounts and make such Tax Payments as so required. The withholdings referred to in this Section 4.4 shall be made at the required applicable statutory rate under the applicable tax law. Each Tax Payment made on behalf of or with respect to a Partner shall be deemed a distribution of Net Cash Flow in such amount to such Partner to the extent such Tax Payment was not attributable to a Capital Transaction, and to the

extent such Tax Payment is attributable to a Capital Transaction, it shall be deemed a distribution of Net Proceeds of a Capital Transaction to such Partner, and any such deemed distribution shall be deemed to have been paid to the Partner on the earlier of the date when the corresponding Tax Payment is made by the Partnership or the date that the distributions, if any, giving rise to the obligation to make such Tax Payment were made. If the Partnership is required to make a Tax Payment on behalf of or with respect to any Partner (the "Taxed Partner") and the amount of such payment exceeds the cash that would otherwise be distributed to such Taxed Partner, the Taxed Partner shall pay to the Partnership by wire transfer the amount of such Tax Payment within ten (10) days of receipt by the Taxed Partner of a notice from the General Partner that it is required to make such Tax Payment. Any amounts paid by the Taxed Partner to the Partnership pursuant to the preceding sentence shall not be treated as a Capital Contribution and the remittance of such Tax Payment to the appropriate taxing authority shall not be treated as a deemed distribution to the Taxed Partner. Each Partner for which the Partnership is required to make a Tax Payment shall indemnify, defend and hold the Partnership and the other Partners harmless of, from and against Indemnified Losses incurred by the Partnership or any other Partner arising out of or in connection with the Tax Payments or obligations attendant thereto.

4.5 Limitation on Distributions.

Notwithstanding anything to the contrary contained herein, without the prior consent of the Partners, no distribution of Net Cash Flow or Net Proceeds of a Capital Transaction shall be made hereunder if such distribution would cause the Partnership to violate Section 17-607 of the Delaware Act or any other applicable law

ARTICLE V ALLOCATION OF PROFITS AND LOSSES

5.1 Allocations for Accounting Purposes.

- (a) Profits and Losses. Except as otherwise provided in this Agreement, Profits and Losses (and, to the extent necessary in the year of liquidation and to the extent permitted by Section 761 of the Code, the prior year, individual items of income, gain, loss, deduction or credit) of the Partnership shall be allocated among the Partners for accounting purposes in a manner such that the Capital Account of each Partner, immediately after making such allocation, is, as nearly as possible, equal (proportionately) to the amount each Partner's Percentage Interests.
- (b) Tax Allocations. For United States federal, state and local income tax purposes, items of income, gain, loss, deduction and credit shall be allocated to the Partners in accordance with the allocations of the corresponding items for Capital Account purposes under Section 5.1(a), except that items with respect to which there is a difference between tax basis and Gross Asset Value will be allocated in accordance with Section 704(c) of the Code, the Regulations thereunder including Regulation Section 1.704-1(b)(4)(i).

5.2 Special Allocations.

- (a) The following special allocations shall be made in the following order:
 - (i) Minimum Gain Chargeback. Notwithstanding any other provision of this Article V, subject to the exceptions set forth in Treasury Regulations Section 1.704-2(f), if there is a net decrease in Partnership Minimum Gain during any Fiscal Year, the Partners shall be specially allocated items of Partnership income and gain for such Fiscal Year (and, if necessary, subsequent Fiscal Years) in an amount equal to such Partner's share of the net decrease in Partnership Minimum Gain, determined in accordance with Treasury Regulations Section 1.704-2(g)(2). Allocations pursuant to the previous sentence shall be made in proportion to the respective amounts required to be allocated to each Partner pursuant thereto. The items so allocated shall be determined in accordance with Treasury Regulations Section 1.704-2(f). This Section 5.2(a)(i) is intended to comply with the minimum gain chargeback requirement in Section 1.704-2(f) of the Treasury Regulations and shall be interpreted consistently therewith.
 - (ii) Partner Nonrecourse Debt Minimum Gain Chargeback. Notwithstanding any other provision of this Article V, except Section 5.2(a), subject to the exceptions contained in Treasury Regulations Section 1.704-2(i)(4), if there is a net decrease in Partner Nonrecourse Debt Minimum Gain attributable to a Partner Nonrecourse Debt during any Fiscal Year, each Partner who has a share of the Partner Nonrecourse Debt Minimum Gain attributable to such Partner Nonrecourse Debt, determined in accordance with Treasury Regulations Section 1.704-2(i)(5) as at the beginning of such Fiscal Year, shall be specially allocated items of Partnership income and gain for such Fiscal Year (and, if necessary, subsequent Fiscal Years) in an amount equal to such Partner's share of the net decrease in Partner Nonrecourse Debt Minimum Gain attributable to such Partner Nonrecourse Debt, determined in accordance with Treasury Regulations Section 1.704-2(i)(4) and 1.704-2(g)(2). Allocations pursuant to the previous sentence shall be made in proportion to the respective amounts required to be allocated to each Partner pursuant thereto. The items so allocated shall be determined in accordance with Treasury Regulations Section 1.704-2(i)(4). This Section 5.2(a)(ii) is intended to comply with the partner nonrecourse debt minimum gain chargeback requirement in such Section of the Treasury Regulations and shall be interpreted consistently therewith.

- (iii) Qualified Income Offset. In the event any Partner unexpectedly receives any adjustments, allocations, or distributions described in paragraphs (4), (5) and (6) of Treasury Regulations Section 1.704-1(b)(2)(ii)(d), modified, as appropriate, by Treasury Regulations Sections 1.704-2(g)(1) and 1.704-2(i)(5), items of Partnership income and gain for such Fiscal Year shall be specially allocated to such Partners in an amount and manner sufficient to eliminate, to the extent required by such Regulations, the Adjusted Capital Account deficit of such Partners as quickly as possible, provided that an allocation pursuant to this Section 5.2(a)(iii) shall be made only if and to the extent that such Partners would have an Adjusted Capital Account deficit after all other allocations provided for in this Section 5.2 have been tentatively made as if this Section 5.2(a)(iii) were not in this Agreement.
- (iv) <u>Nonrecourse Deductions</u>. Nonrecourse Deductions shall be allocated to the Partners in accordance with their respective Percentage Interests.
- (v) Partner Nonrecourse Deductions. Any Partner Nonrecourse Deductions for any Fiscal Year or other period shall be specially allocated to the Partner who bears the Economic Risk of Loss with respect to the Partner Nonrecourse Debt to which such Partner Nonrecourse Deductions are attributable in accordance with Treasury Regulations Section 1.704-2(i)(1). If more than one Partner bears the Economic Risk of Loss, such deduction shall be allocated between or among such Partners in accordance with the ratios in which such Partners share such Economic Risk of Loss.
- (vi) <u>Limitation on Allocation of Losses</u>. Notwithstanding any provision of this Agreement, in no event shall Losses be allocated to a Partner to the extent such allocation would result in such Partner having an Adjusted Capital Account deficit at the end of any Fiscal Year. All such Losses shall be allocated to the other Partners in proportion to their respective Percentage Interests, provided, however, that appropriate adjustments shall be made to the allocation of future Profits in order to offset such specially allocated Losses hereunder.
- (vii) Section 754 Adjustments. To the extent an adjustment to the adjusted tax basis of any Partnership Asset pursuant to Code Section 734(b) or Code Section 743(b) is required, pursuant to Treasury Regulations Section 1.704-1(b)(2)(iv)(m), to be taken into account in determining Capital Accounts as the result of distributions to a Partner, the amount of such adjustment to the Capital Accounts shall be treated as an item of gain (if the adjustment increases the basis of the asset) or an item of loss (if the adjustment decreases such basis) and such gain or loss shall be specially allocated to the Partners in a manner consistent with the manner in which their Capital Accounts are required to be adjusted pursuant to such Section of the Treasury Regulations as the result of distributions to a Partner in accordance with their interests in the Partnership as determined under Regulations Section 1.704-1(b)(3) in the event Regulations Section 1.704-1(b)(2)(iv)(m)(2) applies, or to the Partner to whom such distribution was made in the event Regulations Section 1.704-1(b)(2)(iv) (m)(4) applies.
- Curative Allocations. The allocations contained in Sections 5.2(a)(i) through 5.2(a) (viii) (vii) (the "Regulatory Allocations") are intended to comply with certain requirements of the Code and Treasury Regulations. The Partners intend that, to the extent possible, all Regulatory Allocations shall be offset either by other Regulatory Allocations or with special allocations of other items of Partnership income, gain, loss or deduction pursuant to this Section 5.2(a)(viii). Therefore, notwithstanding any other provisions of this Agreement (other than the Regulatory Allocations), the Partnership shall make such offsetting special allocations of Partnership income, gain, loss or deduction in whatever manner they reasonably determine to be appropriate so that, after such offsetting allocations are made, each Partner's Capital Account balance is, to the extent possible, equal to the Capital Account balance such Partner would have had if the Regulatory Allocations were not part of this Agreement and all items were allocated pursuant to Section 5.1. In exercising its discretion under this Section 5.2(a)(viii), the Partnership shall take into account future Regulatory Allocations under Section 5.2(a)(i) through Section 5.2(a)(viii) that are likely to offset other Regulatory Allocations previously made.

5.3 Other Allocation Rules.

- (a) For purposes of determining the Profits, Losses, or any other items allocable to any period, Profits, Losses, and any such other items shall be determined on a daily, monthly, or other basis, as reasonably determined by the Partners using any permissible method under Code Section 706 and the Treasury Regulations thereunder.
- (b) Except as otherwise provided in this Agreement, all items of Partnership income, gain, loss, deduction, and any other allocations not otherwise provided for shall be divided among the Partners for tax purposes in the same proportions as they share Profits or Losses, as the case may be, for the Fiscal Year.
- (c) The Partners are aware of the income tax consequences of the allocations made by this Article V and hereby agree to be bound by the provisions of this Article V in reporting their shares of Partnership income and loss for income tax purposes.
- (d) Solely for purposes of determining a Partner's proportionate share of the "excess nonrecourse liabilities" of the Partnership within the meaning of Treasury Regulations Section 1.752-3(a)
 (3), the interest of the Partners in Partnership Profits equals one hundred percent (100%), in proportion to their Percentage Interests.

(e) To the extent permitted by Treasury Regulations Section 1.704-2(h)(3), the Partners shall treat distributions of Net Proceeds of a Capital Transaction as not allocable to an increase in Partnership Minimum Gain to the extent the distribution does not cause or increase a deficit balance in the Adjusted Capital Account of any Partner.

5.4 Tax Allocations. Code Section 704(c).

- (a) In accordance with Code Section 704(c) and the Treasury Regulations thereunder, income, gain, loss, and deduction with respect to any property contributed to the capital of the Partnership shall, solely for tax purposes, be allocated among the Partners so as to take account of any variation between the adjusted basis of such property to the Partnership for federal income tax purposes and its initial Gross Asset Value using the "traditional method" with curative allocations upon disposition.
- (b) In the event the Gross Asset Value of any Partnership property is adjusted pursuant to paragraph (b) of the definition of Gross Asset Value, subsequent allocations of income, gain, loss, and deduction with respect to such asset shall take account of any variation between the adjusted basis of such asset for federal income tax purposes and its Gross Asset Value in the same manner as under Code Section 704(c) and the Treasury Regulations thereunder using the traditional method.
- (c) Any elections or other decisions relating to such allocations shall be made by the General Partner, in any manner that reasonably reflects the purpose and intention of this Agreement. Allocations pursuant to this Section 5.4 are solely for purposes of federal, state, and local taxes and shall not affect, or in any way be taken into account in computing, any Partner's Capital Account or share of Profits, Losses, other items, or distributions pursuant to any provision of this Agreement.

ARTICLE VI MANAGEMENT: LIABILITY OF PARTNERS: EXPENSES

6.1 Management.

- Except as otherwise expressly provided in this Agreement, the business and affairs of the (a) Partnership shall be exclusively vested in the General Partner. The General Partner shall carry out and implement the day to day affairs of the Partnership within the scope of the authority granted pursuant to this Agreement. The General Partner shall keep the other Partners reasonably informed as to all matters of concern to the Partnership and the Partners. The General Partner shall devote to the Partnership's business such time as reasonably shall be necessary in connection with its duties and responsibilities hereunder. Except to the extent limited by the provisions of Section 6.3 or otherwise in this Agreement, the General Partner shall have the full, exclusive and complete discretion in the management and control of the affairs of the Partnership and no Limited Partner shall participate in the management of the Partnership or have any control over the Partnership business or have any right or authority to act for or by the Partnership, including, without limitation, the authority provided by the Delaware Act and, in addition, the General Partner shall have the power on behalf of the Partnership, without the consent of the other Partners except as expressly provided in this Agreement, including without limitation, Section 6.3, to (or cause any of its Subsidiaries to):
 - (i) acquire, hold, manage, own, operate, repair, maintain, remediate, improve, develop, redevelop, construct, reconstruct, grant options with respect to, sell, transfer, convey, assign, exchange or otherwise dispose of, grant easements with respect to, or otherwise restrict the use of, all or any part of any Portfolio Investment or the Partnership's or any of its Subsidiaries' interests therein, and to execute and deliver in the Partnership's or any such Subsidiary's name any and all instruments necessary to effectuate such transactions;
 - execute, in furtherance of any or all of the purposes of the Partnership or any of its Subsidiaries, any deed, assignment, lease, easement, covenant, restriction, bill of sale, contract or other document or instrument;
 - (iii) vote at any election or meeting of any Person, or by proxy, and appoint agents to do so in its place and stead;
 - (iv) enter into sale and leaseback financing arrangements with respect to all or part of any Portfolio Investment and, in connection therewith, execute for and on behalf of the Partnership or any of its Subsidiaries any documents relating thereto;
 - (v) lease or sublease, in whole or in part, any Portfolio Investment, real and personal, as lessor, sublessor, lessee or sublessee, and, in connection therewith, execute for and on behalf of the Partnership or any of its Subsidiaries any leases or subleases or agreements terminating, amending or modifying leases or subleases;

- (vi) borrow money on behalf of the Partnership or any of its Subsidiaries, and, in connection therewith, execute for and on behalf of the Partnership or any of its Subsidiaries, bonds, notes, mortgages, security agreements, financing statements, assignments, guarantees and other agreements and documents creating liens on or otherwise affecting any Portfolio Investment, and extensions, renewals, and modifications thereof, and to repay in whole or in part, refinance, recast, increase, modify or extend any indebtedness of the Partnership or any of its Subsidiaries;
- (vii) engage, on behalf of the Partnership or any of its Subsidiaries, such Persons as it shall reasonably deem advisable for the operation and management of the business of the Partnership or any of its Subsidiaries, in each case as independent contractors (and not as employees of the Partnership or any of its Subsidiaries), including, without limitation, agents, managers, accountants, attorneys, consultants, and brokers, all on such terms and for such compensation as the General Partner shall reasonably determine to be proper;
- make and implement all decisions for the Partnership and each of its Subsidiaries, other than Partnership Decisions;
- (ix) implement all Partnership Decisions made in accordance with Section 6.3;
- (x) deposit, withdraw, invest, pay, retain and distribute the Partnership's and each of its Subsidiaries' funds, and open and maintain bank accounts for such funds in the name of the Partnership or its applicable Subsidiary and designate the persons authorized on behalf of the Partnership or any of its Subsidiaries to make deposits therein and withdrawals therefrom;
- (xi) pay, extend, renew, modify, adjust, submit to arbitration, prosecute, defend or compromise any obligation, suit, liability, cause of action or claim, either in favor of or against the Partnership or any of its Subsidiaries, and execute all documents and make all representations, admissions and waivers in connection therewith;
- (xii) enter into, execute, acknowledge and deliver any and all contracts, agreements or other instruments the General Partner deems necessary or appropriate in connection with the business or affairs of the Partnership or any of its Subsidiaries;
- (xiii) apply for, file, prosecute, obtain, appeal and challenge any permit, approval, authorization, filing or consent with respect to the Partnership, any of its Subsidiaries or any Portfolio Investment issued by any Governmental Authority;
- (xiv) either by itself or by contract with others (including with a Person whose shareholders, partners, officers or employees are also shareholders, partners, officers or employees of the General Partner or its Affiliates), establish, have, maintain or close one or more offices, and in connection therewith to maintain office space, facilities and equipment and to engage and pay personnel and to do such other acts and things, it being acknowledged and agreed that the Partnership and its Subsidiaries will enjoy the benefit of the existing facilities and personnel of the General Partner and its Affiliates without payment of any fees or expenses, except as expressly set out herein or in the applicable Property Management Agreement;
- (xv) possess, transfer, or otherwise deal in, and to exercise all rights, powers, privileges and other incidents of ownership or possession with respect to, any Portfolio Investment or other property held or owned by the Partnership or any of its Subsidiaries;
- (xvi) set aside funds for reasonable reserves, anticipated contingencies and working capital;
- (xvii) to distribute the Partnership's funds in accordance with the provisions of this Agreement;
- (xviii) take all actions that may be reasonably necessary or appropriate for the continuation of the Partnership's valid existence as a limited partnership under the Delaware Act (and each of its Subsidiaries' valid existence in accordance with the applicable laws of its state of formation) and under the laws of each other jurisdiction in which such action is necessary to protect the limited liability of the Limited Partners or to enable the Partnership and each of its Subsidiaries, consistent with such limited liability, to conduct the business in which it is engaged;
- (xix) register or qualify the Partnership or any of its Subsidiaries under any applicable United States federal or state laws, or to obtain exemptions under such laws, if such registration, qualification or exemption is reasonably deemed necessary or advisable by the General Partner;
- (xx) enter into, make and perform all contracts, agreements, instruments and other undertakings and pay all expenses as the General Partner may reasonably determine to be necessary, advisable or incidental to the carrying out of the purposes of the Partnership or any of its Subsidiaries;

- (xxi) create special purpose entities to make or pursue Portfolio Investments in accordance with the Approved Investment Structure;
- (xxii) engage in any kind of activity and execute, perform and carry out contracts of any kind necessary, or in connection with or convenient or incidental to any of the foregoing or the Partnership's or any of its Subsidiaries' purposes as set forth herein; execute any and all other documents to carry out the intention and purpose hereof;
- (xxiii) otherwise take any other action in furtherance of the Partnership's or any of its Subsidiaries' stated purposes hereunder.

No Person dealing with the Partnership or any of its Subsidiaries or their respective assets (other than the Property Manager), whether as lender, assignee, purchaser, lessee, grantee, or otherwise, shall be required to investigate the authority of the General Partner in dealing with the Partnership or any of its Subsidiaries or any of their assets, nor shall any Person entering into a contract with the Partnership or any of its Subsidiaries or relying on any such contract or agreement be required to inquire as to whether such contract or agreement was properly approved by the General Partner. Any such Person may conclusively rely on a certificate of authority signed by the General Partner and may conclusively rely on the due authorization of any instrument signed by the General Partner in the name and on behalf of the Partnership or the General Partner.

- (b) <u>Standard of Conduct</u>. The General Partner will exercise its powers and discharge its duties under this Agreement diligently, honestly, and in good faith. Without limitation, in making decisions and managing the Partnership and its Subsidiaries, the General Partner will exercise the standard of care that a prudent general partner of a similar partnership would exercise in similar circumstances. The foregoing is not intended to create any fiduciary responsibilities of the General Partner to the Partnership or any other Partner.
- (c) No Commissions, Rebates. The General Partner shall not accept for its own account in the execution of its duties hereunder, any commissions, reductions, finder's fees or other concessions from tradesmen, suppliers, contractors, insurers, or tenants. If such concessions are received by the General Partner, they shall be remitted to or credited to the Partnership forthwith after receipt.
- (d) Activities of the General Partner. The General Partner shall, and shall cause (by contract or otherwise) the senior management personnel of Cedar Operating Partnership to remain actively involved in the affairs of the Partnership, the Partnership's Subsidiaries and the Portfolio Investments.

6.2 Advisory Board.

- (a) Formation of Advisory Board. The Partnership shall have an advisory board (the "Advisory Board"), the members (each, an "Advisory Board Member") of which shall be comprised of not more than three (3) natural Persons and shall include two (2) representatives of RioCan and one (1) representative of Cedar LP (and one or more alternate Advisory Board Members identified in writing by RioCan or Cedar LP, as applicable, to the other Partners from time to time). None of the Advisory Board Members shall receive any compensation in connection with its position on the Advisory Board.
- (b) Functions of Advisory Board. The functions of the Advisory Board will be to approve such matters as may be required pursuant to the terms of this Agreement to be determined by the "Advisory Board" or the "Partners", other than matters expressly provided in this Agreement to be determined by a Partner acting unilaterally; provided, notwithstanding any other provision in this Agreement to the contrary, that if any provision of this Agreement makes reference to the approval of the Advisory Board Members or the Partners without specifying if unilateral, majority or unanimous approval shall be required, such matter shall be deemed to require the unanimous approval of the Advisory Board Members. Notwithstanding anything to the contrary contained herein, the participation by any representative of a Limited Partner who is an Advisory Board Member in the activities of the Advisory Board shall not be construed to constitute participation by such Limited Partner in the control of the business of the Partnership.
- Meetings of Advisory Board. Regular meetings of the Advisory Board shall be held as and (c) when called by the General Partner or any Advisory Board Member but at least annually beginning after the first full Fiscal Year, upon not less than seven (7) Business Days' prior written notice by the General Partner or any Advisory Board Member to the Advisory Board Members. Special meetings of the Advisory Board may be called by the General Partner or any Advisory Board Member at any time, upon not less than seven (7) Business Days' prior written notice by the General Partner or any Advisory Board Member to the Advisory Board Members, to consider matters for which the consent, approval, review, comment or waiver of the Advisory Board is required by this Agreement or is requested by the General Partner. Advisory Board Members may participate in a meeting of the Advisory Board by means of conference telephone or similar communications equipment by means of which all Persons participating in the meeting can hear each other. A quorum for any meeting of the Advisory Board called to approve a Unanimous Decision shall be three (3) Advisory Board Members and to approve a Majority Decision shall be two (2) Advisory Board Members. All actions to be taken by the Advisory Board shall be by the affirmative vote or written consent of the requisite number of Advisory Board Members needed for the matter in question as provided in this Agreement.

(d) <u>Advisory Board Notices</u>. Notices to the Advisory Board shall be deemed received if sent to each Limited Partner that has appointed an Advisory Board Member.

(e) Joint Venture Investments.

- (i) With respect to any matter that would require a Partnership Decision, the General Partner shall make the applicable election and exercise or waive the rights of the Partnership or any of its Subsidiaries, in each case as a direct or indirect shareholder, member or equity holder of any Joint Venture Investment, only pursuant to the direction of the Advisory Board.
- With respect to each Joint Venture Investment, (A) capital contributions shall be (ii) made by the Partnership or its applicable Subsidiary only to the extent required pursuant to the terms of the applicable Governing Agreements, (B) the General Partner shall cause the Partnership or its applicable Subsidiary, as a direct or indirect shareholder, member or equity holder of any such Joint Venture Investment, to approve any matter and/or take any action only pursuant to the direction of the Advisory Board if the same would constitute a Partnership Decision hereunder to the extent that the Partnership or its applicable Subsidiary shall have the right to approve any such matter or action pursuant to the applicable Governing Agreements, including, without limitation, approving any indemnification claim or vote for dissolution of the applicable Joint Venture Investment, (C) except as required by applicable law or the applicable Governing Agreements, the General Partner shall cause the Partnership or its applicable Subsidiary, as a direct or indirect shareholder, member or equity holder of any such Joint Venture Investment, to distribute all profits received by the Partnership or such Subsidiary to the Partners on a monthly basis in accordance with Article IV, (D) except as required by the applicable Governing Agreements, the General Partner shall not cause or permit the Partnership or its applicable Subsidiary, as a direct or indirect shareholder, member or equity holder of any such Joint Venture Investment, to sell, assign, transfer, convey, gift, exchange or otherwise dispose of any or all of its interest in such Joint Venture Investment without the prior written consent of the Advisory Board if the same would constitute a Partnership Decision hereunder, and (E) except to the extent required by applicable law, the Governing Agreements for the applicable joint venture entity may be amended and/or otherwise modified only if approved by the Advisory Board, other than with respect to changes that are ministerial or otherwise de minimis in nature.

6.3 Partnership Decisions.

- (a) Notwithstanding the provisions of Section 6.1, without the unanimous consent of the Advisory Board Members, in each instance (a "Unanimous Decision"), the Partnership shall not and shall not cause any of its Subsidiaries to:
 - (i) sell (including, without limitation, sell and leaseback), promise to sell, assign, convey, exchange, pledge, transfer, give, dispose, hypothecate or otherwise encumber, directly or indirectly, any Partnership Asset or Portfolio Investment or any material part thereof or material interest therein, other than (i) personal property which may be disposed of or replaced due to wear and tear or obsolescence or otherwise in the ordinary course of business, (ii) easements and other property rights granted in the ordinary course of business (and which do not have a material adverse impact on the value of a Portfolio Investment), and (iii) leases, which shall be governed by Section 6.3(b);
 - (ii) except as expressly provided in the Purchase and Sale Agreement and Section 7.1 hereof, acquire other real or personal property or any direct or indirect interest in another Person, or any material interest therein on behalf of the Partnership or any of its Subsidiaries, either directly or indirectly, other than personal property, easements and other property rights acquired in connection with the ordinary operation of a Portfolio Investment:
 - (iii) other than trade payables incurred in the ordinary course of business, incur debt on behalf of the Partnership or any of its Subsidiaries (or refinance or recast or prepay such debt, other than as required pursuant to the terms of the Financing Documents);
 - (iv) enter into, amend, modify or terminate any Financing Document except for modifications or amendments which are ministerial or otherwise have a de minimis impact on the Partnership or any of its Subsidiaries or otherwise required pursuant to the terms of the Financing Documents;
 - (v) except as expressly provided in Article IX, admit any Person as a Partner or, except as provided in Section 2.2(e) or as required by the applicable Governing Agreements of a joint venture, admit any Person as a partner or member of any of the Partnership's Subsidiaries;
 - (vi) demolish or redevelop any Portfolio Investment and approve the plans and specifications for any redevelopment on account of such demolition or redevelopment;

- (vii) guarantee the debts of any other Person (other than wholly-owned Subsidiaries);
- (viii) make any loans to any Person (other than wholly-owned Subsidiaries of the Partnership);
- (ix) cause the Partnership or any of its Subsidiaries to make any distributions of cash or property except as provided in this Agreeement;
- terminate any Property Management Agreement other than for a "Cause Event" or consent to the assignment by the Property Manager of its interest in any Property Management Agreement (except as otherwise expressly permitted thereunder with respect to a designation by the Property Manager);
- merge or consolidate the Partnership or any of its Subsidiaries with or into another Person:
- (xii) execute and deliver any document which is prohibited under the Delaware Act, this Agreement or any Financing Document;
- (xiii) amend, modify or terminate this Agreement;
- (xiv) take any action not in furtherance of the stated purposes or intended business of the Partnership as set forth in this Agreement;
- take any action under applicable bankruptcy, insolvency or similar laws with respect to the bankruptcy or insolvency of the Partnership or any of its Subsidiaries;
- (xvi) enter into any Related Party Transaction, except as provided in Section 6.8(b);
- (xvii) enter into any Governing Agreement (other than on a form previously approved for use by all of the Partners or the Advisory Board), or amend or modify any Property Management Agreement, Approved Investment Structure, or Governing Agreement of any of the Partnership's Subsidiaries in any manner, in each case (other than any Property Management Agreement), except for modifications or amendments which are ministerial or otherwise have a de minimis impact on the Partnership or any of its Subsidiaries, or are otherwise required pursuant to the terms of any agreements entered into by the Partnership or any of its Subsidiaries in accordance with the terms of this Agreement (such as, by way of example only, changes to Governing Agreements required to comply with the terms of any Financing Documents);
- (xviii) take any action which would cause the REIT not to qualify as a "real estate investment trust" within the meaning of Section 856 of the Code;
- (xix) take any action on behalf of any Subsidiary of the Partnership (including a Joint Venture Investment, unless non-discretionary) which, if taken by the Partnership, would constitute a Unanimous Decision; or
- (xx) approve any other matter set forth in this Agreement requiring unanimous approval of the Advisory Board Members or the Partners, as applicable.
- (b) Notwithstanding the provisions of Section 6.1, without the consent of a majority of the Advisory Board Members, in each instance (a "Majority Decision"), the Partnership shall not and shall not cause any of its Subsidiaries to:
 - approve any budget for the Partnership, any its Subsidiaries or any Portfolio Investment or, once approved, modify any Approved Budget;
 - (ii) make any single Capital Expenditure or group of Capital Expenditures in any Fiscal Year in respect of any single Portfolio Investment (including with respect of any redevelopment thereof), except as provided for in the applicable Approved Budget for such Portfolio Investment;
 - (iii) make any single expenditure or group of expenditures (other than Capital Expenditures) in any Fiscal Year in respect of any single Portfolio Investment exceeding (1) a line item of the applicable Approved Budget by more than fifteen percent (15%) or (2) the aggregate amount of such Approved Budget by more than ten percent (10%) (exclusive of increases attributable to temporary timing differences arising in the ordinary course of business which the General Partner reasonably expects will be reversed over time);
 - set aside funds for reasonable reserves, anticipated contingencies and working capital
 in excess of \$100,000 in the aggregate for any single Portfolio Investment, other than
 as required by law or contract (including Financing Documents) or in accordance the
 applicable Approved Budget;

- approve any leasing plan for any Portfolio Investment or, once approved, modify in any material respect any Approved Leasing Plan;
- (vi) (w) enter into, terminate (including evict), modify or amend any lease of space at any Portfolio Investment for an area in excess of 10,000 square feet of the rentable area of the improvements on the property, or (x) enter into, modify or amend any lease of space at any Portfolio Investment if such lease, modification or amendment would set the net effective rent for such space below ninety seven percent (97%) of the net effective rent for such space provided in the applicable Approved Budget or Approved Leasing Plan, or such lease (or amendment or modification thereof) is not otherwise materially in accordance with the Approved Leasing Plan or (y) enter into, modify or amend any lease of space at any Portfolio Investment if such lease, modification or amendment is neither substantially in the standard form of lease for such Portfolio Investment (with commercially reasonable changes thereto) nor is otherwise on commercially reasonable terms, or (z) approve the plans and specifications for the initial tenant and/or landlord work (or any major renovation) with respect to any lease for an area in excess of 10,000 square feet of the rentable area of the improvements on the property; provided, notwithstanding the foregoing, that no such termination, modification, amendment or approval described in the foregoing clauses (w), (x), (y) or (z) shall be a Majority Decision to the extent the same is expressly required (or the Property Owner shall not have approval rights, in the case of clause (z)) under an existing lease or if such amendment or modification is ministerial or otherwise de minimis in nature (or if such tenant improvements are de minimis in nature, with respect to clause (z));
- (vii) initiate any action, suit, arbitration, or litigation ('Litigation') on behalf of the Partnership or any of its Subsidiaries, except any Litigation initiated in the ordinary course of business or which could reasonably be expected to result in payment to the Partnership or any of its Subsidiaries of \$100,000 or less;
- (viii) settle any Litigation except any Litigation which is covered in full by an insurance policy which is in effect (other than for any deductible which may apply) or that shall result in the payment by the Partnership or any of its Subsidiaries of amounts in excess of \$50,000 to the counterparty in such Litigation;
- (ix) settle or adjust any insurance claim or condemnation action with respect to any single Portfolio Investment that individually or, with respect to a series of related claims in any Fiscal Year with respect to such Portfolio Investment, in the aggregate, exceeds \$100.000:
- (x) approve any audited financial statements of the Partnership;
- (xi) object to the determination of, or thereafter approve on behalf of the Advisory Board, the selection of an independent third party appraiser to determine, the "Fair Market Value" of any Partnership Asset as provided in the definition of such term set forth in Section 1.8 of this Agreement;
- approve the plans and specifications for any single project for common area improvements with respect to a Portfolio Investment that would require Capital Expenditures of \$100,000 or more;
- (xiii) approve any material change to an insurance program, it being acknowledged and agreed that the insurance program in place as of the date of this Agreement is acceptable to the Advisory Board;
- (xiv) take any action on behalf of any Subsidiary of the Partnership (including a Joint Venture Investment, unless non-discretionary) which, if taken by the Partnership, would constitute a Majority Decision; or
- (xv) approve any other matter set forth in this Agreement requiring majority approval of the Advisory Board Members or the Partners, as applicable.
- (c) All requests for approval of a Partnership Decision shall be made by the General Partner or any Advisory Board Member in writing and shall be accompanied by (x) pertinent information regarding such proposed Partnership Decision, and (y) a description of the Partnership Decision proposed to be taken by the Partnership and the basis on which the General Partner or Advisory Board Member recommends taking the proposed Partnership Decision action (a "Consent Notice"). Each Consent Notice shall also specify the date by which the Advisory Board Member shall respond to such Consent Notice, which date shall be not less than ten (10) days after delivery thereof to the Advisory Board. If any Advisory Board Member shall not deliver a written response to a proposed Partnership Decision prior to the date specified in the Consent Notice pertaining thereto, then such Advisory Board Member shall be deemed not to have consented to such Partnership Decision.

6.4 Duties and Conflicts.

- (a) The Partners, in connection with their respective duties and responsibilities hereunder, shall at all times act in good faith and, except as expressly set forth herein, any decision or exercise of right of approval, consent, disapproval or deferral of approval by a Partner is to be made by such Partner pursuant to the terms of this Agreement in good faith. Except for reimbursement of the General Partner's expenses pursuant to and in accordance with the terms of this Agreement, or as otherwise agreed to in writing by the Partners, no Partner or any partner, officer, shareholder or employee of any Partner shall receive any salary or other remuneration for its services rendered pursuant to this Agreement.
- (b) Subject to the terms of Article VII, each Partner recognizes that the Partners have or may have other business interests, activities and investments, some of which may be in conflict or competition with the business of the Partnership (or any of its Subsidiaries) and that the other Partners are entitled to carry on such other business interests, activities and investments. No Limited Partner shall be obligated to devote all or any particular part of its time and effort to the Partnership and its affairs.
- Except as set out in Article VII, any Partner or Affiliate thereof may engage in or possess an (c) interest in any other business ventures of any nature or description, independently or with others, similar or dissimilar to the business of the Partnership (or any of its Subsidiaries), and neither the Partnership (nor any of its Subsidiaries) nor any Partner shall have any rights by virtue of this Agreement or the relationship created hereby in or to any other ventures or activities engaged in by any Partner or Affiliate thereof, or to the income or proceeds derived therefrom, and the pursuit of such ventures or activities by any Partner or its Affiliate shall not be deemed wrongful or improper, even to the extent the same are competitive with the business activities of the Partnership (or any of its Subsidiaries). Except as set out in Article VII, no Partner or Affiliate thereof shall be obligated to present any particular investment opportunity to the Partnership (or any of its Subsidiaries) even if such opportunity is of a character which, if presented to the Partnership (or any of its Subsidiaries), could be taken by the Partnership (or any of its Subsidiaries), and except as set out in Article VII, any Partner or Affiliate thereof shall have the right to take for its own account (individually or as a partner, partner or fiduciary) or to recommend to others any such particular investment opportunity.

6.5 Partnership Counsel.

To the extent that the General Partner deems necessary, the Partnership shall retain one or more law firms to be the Partnership's legal counsel (the "Partnership Counsel"). The fees and expenses of the Partnership Counsel shall be a Partnership expense. Nothing herein shall restrict the Partnership Counsel from acting as counsel to any Partner or any Affiliate of such Partner (at the expense of such Partner or Affiliate), but Partnership Counsel may not represent such Partner or any Affiliate of such Partner in any dispute involving any other Partner or the Partnership (or any of its Subsidiaries).

6.6 Exculpation/Indemnification.

- (a) <u>Limited Partners and Advisory Board Members.</u>
 - (i) Under the laws of the State of Delaware, to the extent that, at law or in equity, any Limited Partner has any duties (including, without limitation, fiduciary duties) and liabilities relating thereto to the Partnership or to the other Partners, such duties are hereby waived and eliminated and superceded by the provisions of this Agreement.
 - (ii) No Limited Partner shall be liable to the Partnership or to any other Partner for any act performed or omitted to be performed by it on behalf of the Partnership (or any of its Subsidiaries) provided such act or omission was taken in good faith, and did not constitute fraud, gross negligence or willful misconduct (including, without limitation, an intentional material breach of the terms of this Agreement).
 - (iii) The Limited Partners shall be indemnified, defended and held harmless by the Partnership from and against any and all expenses (including reasonable attorneys' fees), losses, damages, liabilities, charges and claims of any kind or nature whatsoever including the cost of seeking to enforce this indemnification right (collectively "Indemnified Losses"), incurred by them in their capacities as Limited Partners, arising out of or incidental to any act performed or omitted to be performed by any one or more of the Limited Partners in good faith in their capacities as Limited Partners and/or in connection with the business of the Partnership (or any of its Subsidiaries), including any act or omission constituting ordinary negligence of such Limited Partners, provided that such act or omission did not constitute fraud, gross negligence or willful misconduct (including, without limitation, an intentional material breach of the terms of this Agreement).
 - (iv) The Partnership and the other Partners shall be indemnified and held harmless by each Limited Partner from and against any and all Indemnified Losses arising out of or incidental to any act or omission taken in bad faith by such Limited Partner, or any fraudulent act, gross negligence, or willful misconduct (including, without limitation, a n intentional material breach of the terms of this Agreement) performed or committed by such Limited Partner.

(v) No Advisory Board Member, nor the Limited Partner such Advisory Board Member represents, nor any other Protected Person, shall be liable to any Partner or the Partnership by virtue of such Advisory Board Member acting as an "Advisory Board Member" hereunder, and each of the foregoing Persons shall be indemnified, defended and held harmless by the Partnership from and against any and all Indemnified Losses incurred by them by virtue of such Advisory Board Member acting as an "Advisory Board Member" hereunder; provided, notwithstanding the foregoing, that the Limited Partner represented by an Advisory Board Member shall be liable for any fraudulent act, gross negligence or willful misconduct (including, without limitation, an intentional material breach of the terms of this Agreement) performed or committed by such Advisory Board Member. Under the laws of the State of Delaware, to the extent that, at law or in equity, the Advisory Board Members have any duties (including fiduciary duties) and liabilities relating thereto to the Partnership or to the Partners, such duties are hereby eliminated to the fullest extent permitted under such laws.

(b) <u>General Partner</u>.

- (i) Under the laws of the State of Delaware, to the extent that, at law or in equity, the General Partner has any duties (including, without limitation, fiduciary duties) and liabilities relating thereto to the Partnership or to the other Partners, such duties are hereby waived and eliminated and superceded by the provisions of this Agreement.
- (ii) The General Partner shall not be liable to the Partnership or to any Limited Partner for any act performed or omitted to be performed by it on behalf of the Partnership (or any of its Subsidiaries) provided such act or omission was taken in good faith, and did not constitute fraud, gross negligence or willful misconduct (including, without limitation, an intentional material breach of the terms of this Agreement).
- (iii) The General Partner shall be indemnified, defended and held harmless by the Partnership from and against any and all Indemnified Losses incurred by it in its capacity as a General Partner, arising out of or incidental to any act performed or omitted to be performed by it in good faith in its capacity as the General Partner and/or in connection with the business of the Partnership (or any of its Subsidiaries) including, without limitation, any act or omission constituting ordinary negligence of the General Partner, provided that such act or omission did not constitute fraud, gross negligence or willful misconduct (including, without limitation, an intentional material breach of the terms of this Agreement).
- (iv) The General Partner shall indemnify each Limited Partner and the Partnership (each, and "Indemnified Party"), for any Indemnified Losses resulting from any fraudulent act, gross negligence and/or willful misconduct (including, without limitation, an intentional material breach of the terms of this Agreement) by the General Partner.

(c) <u>General</u>.

- (i) All indemnification obligations under this Agreement shall also run to the benefit of any Affiliate of any Partner or any principal, partner, member, manager, shareholder, controlling person, officer, director, agent or employee of any of the aforesaid Persons (each of the foregoing a "Protected Person").
- (ii) The Partnership shall promptly reimburse (or advance, to the extent reasonably requested by a Protected Person other than in connection with Indemnified Losses resulting from claims made by the Partnership or any Partner) each Protected Person for reasonable legal or other expenses (as incurred) of each Protected Person in connection with investigating, preparing to defend or defending any claim, lawsuit or other proceeding relating to any Indemnified Losses for which the Protected Person may be indemnified pursuant to Section 6.6(a) or 6.6(b), as applicable; provided, that such Protected Person executes a written undertaking to repay the Partnership for such reimbursed or advanced expenses if it is finally judicially determined that such Protected Person is not entitled to the indemnification provided by Section 6.6(a) or 6.6(b), as applicable.
- (iii) The provisions of this Section 6.6 shall continue to afford protection to each Protected Person regardless of whether such Protected Person remains in the position or capacity pursuant to which such Protected Person became entitled to indemnification under this Section 6.6 and regardless of any subsequent amendment to or termination of this Agreement.
- (d) The provisions of this Section 6.6 shall survive a termination of this Agreement.

6.7 Cedar LP Obligations

(a) As a material inducement to RioCan to enter into this Agreement, Cedar LP shall guarantee to RioCan, the payment and performance of the obligations of Cedar GP under this Agreement.

- (b) Cedar LP hereby agrees and acknowledges that it is a primary obligor for the obligations of Cedar GP hereunder and not merely a surety and hereby absolutely, irrevocably and unconditionally guarantees the full and punctual payment and performance of such obligations without the necessity for any suit or proceeding of any kind or nature whatsoever brought by the Partnership or any Limited Partner and without the necessity of any notice or demand to which Cedar LP might otherwise be entitled (including, without limitation, diligence, presentment, notice of maturity, extension of time, change in nature or form of Cedar LP or the obligations, acceptance of security, release of security, imposition or agreement arrived at as to the amount of or the terms of the obligations, notice of adverse change in Cedar LP's financial condition an any other fact which might materially increase the risk to Cedar LP).
- (c) The provisions of this Section 6.7 shall solely be for the benefit of RioCan and shall not confer upon any creditor or other third party having dealings with the Partnership or any of its Subsidiaries any right, claim or other benefit.
- (d) The provisions of this Section 6.7 shall survive a termination of this Agreement.

6.8 Transactions with Partners or Affiliates.

- (a) Neither the Partnership nor any of it Subsidiaries may enter into any transaction with any Partner or any of its Affiliates (each, a "Related Party Transaction"), without first obtaining the unanimous written consent of the Advisory Board and satisfying the remaining requirements of this Section 6.8.
- (b) No Related Party Transaction between the Partnership or any of its Subsidiaries on the one hand and any Partner or any Affiliate of any Partner (each, a "Related Party") on the other hand shall be void or voidable solely by reason of such relationship. The entering into of any Related Party Transaction by the Partnership or any of its Subsidiaries shall not subject the participating Related Party or any of its Affiliates, or their respective officers, directors, managers, partners or stockholders to liability to the Partnership, any of its Subsidiaries or any Partner if all of the material facts as to the Related Party Transaction and the nature of any conflict of interest are disclosed or are known to the Advisory Board Members prior to entering into the Related Party Transaction. In furtherance of the foregoing, the Partners acknowledge and agree that Cedar Operating Partnership or an affiliate of Cedar Operating Partnersip, as determined by Cedar GP (provided such affiliate is directly or indirectly whollyowned by Cedar Operating Partnership or CSCI and generally manages the other properties directly or indirectly owned by Cedar Operating Partnership) shall act as Property Manager for each Portfolio Investments pursuant to a Property Management Agreement substantially in the form attached hereto as Exhibit A, and the foregoing Related Party Transactions shall not require the consent of the Advisory Board for so long as Cedar GP remains the General Partner. In the event of a default under a Related Party Transaction between the Partnership and any of the Cedar Partners or any Affiliate or permitted assignee thereof, RioCan shall have the right to enforce the terms of such Related Party Transaction on behalf of the Partnership. In the event that any of the Property Management Agreements with respect to which Cedar Operating Partnership or its Affiliate is the Property Manager is terminated, with respect to any decision to enter into a replacement Property Management Agreement the following procedure shall govern:
 - (i) RioCan shall forthwith submit to the Cedar Partners the names of at least three (3) property managers (the "Nominees") each of whom is at arm's length to RioCan, and is an experienced operator of properties similar to the Portfolio Investments, and the market terms and conditions on which each of the Nominees is prepared to manage the Portfolio Investments (the "Proposed Terms"); and
 - (ii) Cedar shall consider, in good faith acting reasonably, the Nominees and the Proposed Terms and within ten (10) Business Days after RioCan has submitted the names of the Nominees and their respective Proposed Terms, Cedar shall notify RioCan which one of the Nominees it has selected. All selections made in this regard shall be binding upon the Partners.

6.9 Rights of the Limited Partners.

Except as otherwise expressly provided in this Agreement, neither the Advisory Board nor the Limited Partners shall take part in the management or control of the business of the Partnership (or any of its Subsidiaries) or transact any business for or in the name of the Partnership (or any of its Subsidiaries), nor shall any Advisory Board Member or Limited Partner have the power to sign for or bind the Partnership (or any of its Subsidiaries). Except as otherwise expressly provided herein, any exercise by the Advisory Board or the Limited Partners of their rights under this Agreement shall be deemed to be an action affecting the agreement among the Partners and not an action affecting the management or control of the business of the Partnership (or any of its Subsidiaries).

6.10 Expenses.

The General Partner shall be reimbursed for all of its reasonable and actual out-of-pocket expenses (not including any general office overhead) in accordance with the terms of this Agreement and the Approved Budgets.

6.11 Certain Tax Matters.

- (a) Tax Matters Partner. The "Tax Matters Partner" (as such term is defined in Section 6231(a) (7) of the Code) of the Partnership shall be the General Partner. The Tax Matters Partner shall cause to be prepared and filed all returns of the Partnership and each Limited Partner shall take all actions required to authorize and appoint the General Partner as the party with the sole authority to handle all tax matters of the Partnership. The provisions of Section 8.3(f) shall govern tax elections to be made on behalf of the Partnership. The Tax Matters Partner shall comply with the responsibilities outlined in Sections 6221 through 6233 of the Code (including the Regulations promulgated thereunder) and shall have all powers necessary to perform fully in such capacity. The Tax Matters Partner is authorized to represent the Partnership before taxing authorities and courts in tax matters affecting the Partnership and the Partners in their capacity as such and shall keep the Partners informed of any such administrative and judicial proceedings and shall allow the Limited Partners to participate, at their own expense, in such proceedings; provided, that the Tax Matters Partner shall have no right to enter into any settlement agreement or otherwise settle or compromise any matter in its capacity as Tax Matters Partner without the majority consent of the Advisory Board. The Tax Matters Partner shall be entitled to be reimbursed by the Partnership for all costs and expenses incurred by it in connection with any administrative or judicial proceeding affecting tax matters of the Partnership and the Partners in their capacity as such and to be indemnified by the Partnership (solely out of Partnership Assets) with respect to any action brought against it in connection with any judgment in or settlement of any such proceeding (subject to any limitation on the right to indemnification pursuant to Section 6.6 hereof). Any Partner who enters into a settlement agreement with respect to any Partnership item shall notify the Tax Matters Partner of such settlement agreement and its terms within thirty (30) days after the date of settlement. The Tax Matters Partner shall also manage audits of the Partnership conducted by the Internal Revenue Service or any other taxing authority pursuant to the audit procedures under the Code and the Treasury Regulations promulgated thereunder or other applicable law. This provision shall survive any termination of this Agreement.
- (b) Classification as a Partnership. The parties hereto intend the Partnership be classified as a partnership for United States federal, state and local income tax purposes effective as of the date of this Agreement. No Partner shall elect to have the Partnership classified as an association taxable as a corporation for United States federal income tax purposes pursuant to Regulations Section 301.7701-3. The Tax Matters Partner shall, for and on behalf of the Partnership, take all steps as may be required to maintain the Partnership's classification as a partnership for United States federal income tax purposes, including, if necessary, affirmatively filing Internal Revenue Service Form 8832 no later than seventy-five (75) days after the effective date of this Agreement. By executing this Agreement, each of the parties hereto consents to the authority of the Tax Matters Partner to make any such election and shall cooperate in the making of such election (including providing consents and other authorizations that may be required).
- (c) <u>Tax Election.</u> The Partners shall take all actions necessary to cause the REIT to be qualified, operated and maintained as a real estate investment trust for federal, state and, if applicable, local income tax purposes.
- (d) Transparent Entities. The General Partner will use commercially reasonable efforts to cause Portfolio Investments to be owned through entities that are treated as "transparent" for Canadian tax purposes, provided that the use of such entities does not have an adverse impact on the ability of either CSCI or the REIT to qualify as a real estate investment trust under the Code, and provided, further, that the General Partner's obligation under this Section 6.11(d) shall be deemed satisfied with respect to any transaction structure that has been approved by the Advisory Board.

ARTICLE VII INVESTMENT OPPORTUNITIES; NONCOMPETITION AND NONSOLICITATION

7.1 Investment Opportunities.

- (a) Notwithstanding anything to the contrary contained in this Agreement, during the Investment Period, if any opportunity to acquire a freehold, leasehold or indirect ownership interest (i.e., through the purchase of the ownership interests in a property owning entity or its direct or indirect owners) with respect to a Target Investment is identified by, presented or offered to any Partner or any of its Restricted Parties, that such Person desires to pursue, such Partner shall provide notice of such proposed investment to the other Partners and the Advisory Board and such investment shall be offered to the other Partners for investment by the Partnership as a Portfolio Investment in accordance with subsection (b) below. For the purposes of this subsection (a), "Target Investment" means a potential Portfolio Investment that is: (i) primarily a supermarket-anchored retail property or other retail-anchored property; (ii) comprised of at least 50,000 square feet of gross leaseable area, provided that this subparagraph (ii) shall not apply to any supermarket-anchored retail property; (iii) located in Connecticut, Maryland, Massachusetts, New Jersey, New York, Pennsylvania or Virginia; and (iv) not part of a larger portfolio primarily comprised of real estate properties that include properties that are not in one of the States listed in subparagraph (iii) above; provided however, except for the redevelopment of existing supermarket-anchored or other retailanchored properties, a Target Investment for purposes of this Section 7.1 only shall not include any potential Portfolio Investment that requires ground-up construction of substantially all of the improvements to be located on the applicable site. For purposes of this Agreement, "Restricted Parties" shall mean (1) with respect to the Cedar Partners only, CSCI, Cedar Operating Partnership, or any Subsidiary of CSCI or Cedar Operating Partnership (other than the Partnership and its Subsidiaries), and (3) with respect to RioCan only, RioCan REIT, RioCan Private REIT, Inc. or any Subsidiary of RioCan REIT or RioCan Private REIT, Inc. (other than the Partnership and its Subsidiaries); provided, that "Restricted Parties" shall not include any Subsidiary of CSCI, Cedar Operating Partnership, RioCan REIT or RioCan Private REIT, Inc. which is a joint venture existing as of the date of this Agreement pursuant to which CSCI, Cedar Operating Partnership, RioCan REIT or RioCan Private REIT, Inc., as applicable, does not have the right, directly or indirectly, to prevent such joint venture from acquiring a Target Investment or soliciting and hiring employees; and, anything contained herein to the contrary notwithstanding, in no event shall the provisions of this Section 7.1 apply to any Target Investment owned as of the date hereof, in whole or in part, directly or indirectly, by any of CSCI, Cedar Operating Partnership, RioCan REIT or RioCan Private REIT, Inc. (including, without limitation, if a direct or indirect interest in a Target Investment is offered to a Restricted Party pursuant to the exercise of a buy/sell, right of first offer, right of first refusal or similar right provided in the Governing Agreements of any Subsidiaries of such Persons that are joint ventures). For purposes of Section 7.1 only, "Subsidiary" shall mean any Subsidiary of the applicable Person that is Controlled or at least fifty percent (50%) owned, directly or indirectly, by such Person.
- (b) Any notice provided pursuant to subparagraph (a) above shall set out the material terms of the acquisition of the applicable Target Investment then known or in the possession or control of the party sending the notice (the "Referring Party"). The notice shall include a reasonably detailed report related to such Target Investment which includes, based upon information then known or in the possession or control of the Referring Party:
 - an investment summary containing such details and information regarding the acquisition of such Target Investment as set out in Exhibit B annexed hereto;
 - (ii) a good faith estimate of the Initial Real Property Costs to be required in connection with the acquisition of such Target Investment;
 - (iii) a copy of the agreement of purchase and sale (if available), the letter of intent (if applicable), and/or the basic terms and conditions (or permitted parameters of terms and conditions) respecting a proposed offer to be made or agreement of purchase and sale to be entered into in connection with the acquisition of such Target Investment;
 - (iv) a development pro-forma (if applicable) and valuation and returns analysis of such Target Investment, including an ARGUS report (if applicable);
 - a summary of material terms of any proposed financing with respect to the acquisition of such Target Investment, if any, including the status of discussions with respect thereto;
 - (vi) copies of building condition reports and environmental reports (if available) with respect to such Target Investment; and
 - (vii) a list of properties which are directly or indirectly owned or managed by the Referring Party or any of its Restricted Parties which may compete with the Target Investment or would otherwise trigger the restrictions set forth in Section 7.2(a).
- (c) The party receiving the notice provided for in subsection (a) (in each case, the 'Receiving Party'') shall have (i) fifteen (15) days from receipt of such notice, in the case of a Target Investment which has an estimated purchase price which is below \$15,000,000 as set out in such notice or (ii) thirty (30) days from receipt of such notice, in the case of a Target Investment with an estimated purchase price of \$15,000,000 or more as set out in such notice, to elect pursuant to a written notice delivered to the other Partners to cause the Partnership (through one of its Subsidiaries) to acquire such Target Investment. Any failure by the Receiving Party to make such election in writing as and when required in this subsection (c) shall constitute a rejection of such Target Investment.

- (d) During the period of time referred in subsection (c), the Partners will cause their Advisory Board Members to be available to meet at mutually convenient times to discuss the applicable Target Investment. In addition to the information to be provided with the aforesaid notice, the Referring Party shall provide such additional information as may be reasonably requested by the Receiving Party in connection with the applicable Target Investment as shall be in the possession or control of the Referring Party at such time. Any assumptions, analyses and conclusions contained in the reports provided pursuant to subsection (b) shall be without representation or warranty by the Referring Party including, without limitation, any representation or warranty with respect to the realization of incremental value with respect to such Target Investment.
- (e) If the Receiving Party approves a Target Investment within the period of time provided in subsection (c) above, the General Partner shall proceed to cause such Target Investment to be acquired by the Partnership (through one of its Subsidiaries) in accordance with the Approved Investment Structure, provided that (i) the acquisition shall be on terms and conditions materially the same as (or more favorable to the Partnership or its applicable Subsidiary than) those contained in the notice provided for in subsection (a); (ii) all reports of consultants (including environmental, audits and building condition reports) will be provided to the Advisory Board; and (iii) the Partnership (or its applicable Subsidiary) will not proceed to waive any material conditions to its obligation to acquire the applicable Target Investment unless same are approved by a majority of the Advisory Board.
- (f) If the Receiving Party rejects (or is deemed to have rejected in accordance with subsection (c)) any Target Investment, the Referring Party shall have the right to pursue the acquisition of such Target Investment alone or with any other third party; provided that (i) if the terms of any such acquisition become more favorable to the Referring Party in any material respect than those previously disclosed pursuant to the notice provided for in subsection (a), the Referring Party shall be obliged to deliver another notice to the Receiving Party disclosing the terms of such acquisition in accordance with the provisions of this Section 7.1 and each Partner shall comply again with such provisions, and (ii) such investment complies with the requirements of Section 7.2.

7.2 Noncompetition.

- (a) During the Investment Period, each of the Partners agrees, except as otherwise provided herein, not to (and not permit any of its Restricted Parties to), directly or indirectly, within three (3) miles in any direction on any road, street, highway, freeway or other public or private thoroughfare fronting, adjacent, parallel to or providing access to or from each respective Portfolio Investment, own any interest in, manage, operate, develop or control any Target Investment (including, without limitation, any Target Investment to be developed or redeveloped, but excluding any Target Investment that will be included in a portfolio acquisition if such Target Investment constitutes less than ten percent (10%) of the gross leasable area of the entire portfolio and such portfolio acquisition includes not more than two (2) Target Investments). The Partners further acknowledge and agree that the prohibition on ownership of Target Investments described herein shall not constitute, and shall not be construed to constitute, a prohibition on any Person's non-controlling ownership of stock in any publicly traded companies listed on any national stock exchange.
- (b) Each of the Partners acknowledges and agrees that (i) the other Partners would not have agreed to acquire the Portfolio Investments by and through the Partnership and its Subsidiaries without the provisions of this Section 7.2 and each and every provision in this Section 7.2, including without limitation the provisions of subsection (c) below, and (ii) the foregoing territorial and time limitations and restrictive covenants are reasonable and properly required for the adequate protection of the business and affairs of the Partnership and its Subsidiaries, and in the event any such territorial or time limitation is found to be unreasonable by a court of competent jurisdiction, each of the Partners agrees and submits to the reduction of either said territorial or time limitation or both, to such an area or period as the court may determine to be reasonable.
- (c) Each of the Partners acknowledges that the Partnership and/or its Subsidiaries will suffer damages incapable of ascertainment in the event that any of the provisions of Section 7.2(a) hereof are breached and that Partnership and/or its Subsidiaries will be irreparably damaged in the event that the provisions of Section 7.2(a) are not enforced. Therefore, should any dispute arise with respect to the breach or threatened breach of Section 7.2(a), each of the Partners agrees and consents, that in addition to any and all other remedies available to the Partnership or its Subsidiaries, an injunction or restraining order or other equitable relief may be issued or ordered by a court of competent jurisdiction restraining any breach or threatened breach of Section 7.2(a). Each of the Partners agrees not to assert in any such action that an adequate remedy exists at law. All expenses, including, without limitation, reasonable attorney's fees and expenses incurred in connection with any legal proceeding arising as a result of a breach or threatened breach of Section 7.2(a) shall be borne by the losing party to the fullest extent permitted by law and the losing party hereby agrees to indemnify and hold the other party harmless from and against all such expenses.

7.3 Nonsolicitation.

- (a) During the Term, no Partner shall, and each Partner shall cause their respective Restricted Parties not to, without the prior written consent of the other Partners, directly or indirectly, solicit to hire (or cause to leave the employ of such Partner or its Restricted Parties) any employee of such Partner or its Restricted Parties unless such Person ceased to be an employee of such Partner or its Restricted Parties due to such Partner's or such Restricted Parties' termination of such Person, or, in the case of such Person's voluntary termination of employment with such Partner or its Restricted Parties, at least six (6) months has elapsed since such Person's voluntary termination; provided, however, that nothing in this Section 7.3(a) shall prohibit or restrict any Partner or any of its Restricted Parties from soliciting or hiring any such employee pursuant to any general solicitation. In addition, no Partner shall, and each Partner shall cause their respective Restricted Parties not to, directly, or indirectly, during the Term, call on, solicit or service any tenant, subtenant, landlord, licensee, licensor or other business relation of the Partnership or any of its Subsidiaries in order to induce or attempt to induce such Person to (x) cease doing business with the Partnership or any of its Subsidiaries or (y) deal with any competitor of the Partnership or any of its Subsidiaries.
- (b) Each of the Partners acknowledges that the Partnership and/or its Subsidiaries will suffer damages incapable of ascertainment in the event that any of the provisions of Section 7.3(a) hereof are breached and that Partnership and/or its Subsidiaries will be irreparably damaged in the event that the provisions of Section 7.3(a) are not enforced. Therefore, should any dispute arise with respect to the breach or threatened breach of Section 7.3(a), each of the Partners agrees and consents, that in addition to any and all other remedies available to the Partnership or its Subsidiaries, an injunction or restraining order or other equitable relief may be issued or ordered by a court of competent jurisdiction restraining any breach or threatened breach of Section 7.3(a). Each of the Partners agrees not to assert in any such action that an adequate remedy exists at law. All expenses, including, without limitation, reasonable attorney's fees and expenses incurred in connection with any legal proceeding arising as a result of a breach or threatened breach of Section 7.3(a) shall be borne by the losing party to the fullest extent permitted by law and the losing party hereby agrees to indemnify and hold the other party harmless from and against all such expenses.

ARTICLE VIII BOOKS AND RECORDS. REPORTS TO PARTNERS

8.1 Bank Accounts.

Subject to Section 6.3, the General Partner shall have authority to open bank accounts and designate signatories with respect thereto on behalf of the Partnership or any of its Subsidiaries as it shall deem necessary or desirable for the conduct of Partnership's or any of its Subsidiaries' business. One or more individuals designated by the General Partner, from time to time, shall at all times be designated signatories with respect to such bank accounts. The funds of the Partnership and each of its Subsidiaries shall not be commingled with any other funds.

8.2 Books of Account.

The Partnership shall keep books of account and records showing the assets and liabilities, operations, transactions and financial condition of the Partnership, its Subsidiaries and the Portfolio Investments on an accrual basis in accordance with GAAP. The books of account and records of the Partnership, its Subsidiaries and the Portfolio Investments shall at all times be maintained at the principal office of the Partnership. All such books of account and records may be inspected, copied and audited (including, without limitation, internal control testing) by any Partner, its designees or representatives from time to time upon reasonable prior written notice to the General Partner at the office of the Partnership. The General Partner will consult the Limited Partners on all accounting policies and shall, as soon as possible, advise the Limited Partners in writing in advance of any proposed material change. The General Partner will provide to a Limited Partner, upon request, a description of its principal internal controls and results of testing as they relate to the Portfolio Investments and such other matters relating to internal controls as such Limited Partner may reasonably request from time to time.

8.3 Audit and Reports.

- (a) Operating Statements. The General Partner shall, as a Partnership expense, at least once every calendar year have the Partnership's books and records audited by the Accountant. A copy of the annual audited financial statements of the Partnership shall be submitted, promptly after completion, to all Partners and shall include:
 - (i) a balance sheet;
 - (ii) a statement of the income for such year;
 - (iii) a statement of cash flows;
 - (iv) a statement of each Partner's Capital Account, including such Partner's allocations and share of Profits, Losses and Regulatory Allocations pursuant to Section 5.2(a) (viii); and
 - (v) all notes to the financial statements; and

(vi)

The General Partner shall cause such submission to occur not later than ninety (90) days after the end of each Fiscal Year. Each of the items described in clauses (i) through (vi) above shall be prepared in accordance with GAAP. All financial information required to be provided to the Partners or otherwise required hereunder shall be compiled in U.S. Dollars. The Cedar Partners acknowledge that RioCan and its shareholders will be subject to IFRS commencing January 1, 2011 with retroactive impact to January 1, 2010. Consequently, periodic third party appraisals may be necessitated and the General Partner will cause the Property Manager to oversee such process; provided that RioCan shall bear the cost of any appraisals not required under any Financing and any audit expenses in connection therewith, if applicable. In addition, the Cedar Partners will make all reasonable efforts to provide RioCan with the information it requires in connection with the conversion to IFRS.

- (b) Tax Information. Within forty (40) days following the end of the Fiscal Year of the Partnership, the General Partner shall, as a Partnership expense, furnish each Partner with copies of the Partnership's federal partnership Return of Income and other income tax returns, together with each Partner's Schedule K-1 or analogous schedule, in draft form which returns shall be signed by the chief financial officer of Cedar Shopping Centers Inc., if Cedar GP is the Tax Matters Partner. The General Partner shall also provide such other information reasonably requested by a Partner to assist the Partner in preparing its tax returns.
- (c) Quarterly Reports. Within thirty (30) days after the end of each quarter of each Fiscal Year, the General Partner shall cause to be prepared and made available to each Limited Partner through the General Partner's web site:
 - (i) unaudited financial information for the Partnership, including a complete set of financial statements comprising a balance sheet, income statement, cash flow statement and statement of partner's capital and consolidating statements of income and balance sheets:
 - reports for each Portfolio Investment prepared on GAAP basis in accordance with the (ii) Property Management Agreement, Section 10; and
 - (iii) an internal control certificate signed by the chief financial officer of Cedar Shopping Centers Inc., if Cedar GP is the General Partner.
- (d) Partnership Records. The General Partner hereby agrees (a) to cause the Partnership to preserve all financial and accounting records pertaining to the Partnership during the Term and for six (6) years thereafter (in electronic form, at the General Partner's election), (b) to prepare its financial statements in accordance with GAAP, (c) to not take or fail to take any action that would cause the auditor's report of such statements to include any qualifications due to scope of limitations, lack of sufficient component evidential matter, or a departure from GAAP, and (d) that the following items shall be included with the package of information containing the audited financial statements: (i) a schedule of all Partners, (ii) the annual independent auditor's statement, (iii) any management representation letter that the General Partner provides to its auditors in connection with the preparation of the Partnership's financial statements, and (iv) schedule of unadjusted errors.
- Tax Returns. The General Partner shall, as a Partnership expense, use commercially (e) reasonable efforts to cause to be filed all tax returns related to the Partnership, its Subsidiaries and each Portfolio Investment in a timely manner. The General Partner shall use commercially reasonable efforts to provide the Limited Partners with drafts of all tax returns thirty (30) days prior to the date such returns are filed. Each of the Partners shall promptly provide to the General Partner such information as may be in its possession as shall be necessary or appropriate for the preparation of such returns. The Limited Partners shall have no obligation to deliver any document or other instrument to the General Partner or to any other party except to the extent that such document or instrument is otherwise publicly available, and in no event shall any Limited Partner have any obligation to execute any agreement, certificate or other document unless the same is in a form reasonably acceptable to such Limited Partner. No later than forty (40) days after the end of each Fiscal Year of the Partnership, the Partnership shall, as a Partnership expense, furnish the Partners with all necessary tax reporting information required by the Partners for the preparation of their respective federal, state and local income tax returns, including each Partner's pro rata share of income, gain, loss, deductions and credits for such Fiscal Year. The General Partner shall supervise the Accountant in the preparation of the Partnership's tax returns, the cost of which shall be a Partnership expense.
- (f) Elections. Except as otherwise provided in this Agreement, all decisions as to accounting principles, whether for the Partnership's books or for income tax purposes (and such decisions may be different for each such purpose) and all elections available to the Partnership under applicable tax law, shall be made by the Tax Matters Partner (subject to approval by the Advisory Board with respect to any such decision or election that would have a disproportionate material adverse effect on RioCan). Upon the request of any Partner in connection with the transfer of all or part of such Partner's Interest, the Partnership shall make an election under Code Section 754. The General Partner shall not elect to have the Partnership classified as an association taxable as a corporation for federal income tax purposes and shall take any steps required to maintain the Partnership's classification as a partnership for such purposes.

- (g) Internal Reports. The Cedar Partners shall, at no cost or expense to the Cedar Partners, cooperate with RioCan in good faith in connection with the preparation of internal reports required to be prepared by or on behalf of RioCan or RioCan REIT, including providing readily available information to RioCan in connection therewith.
- (h) Portfolio Investments. As soon as reasonably practicable after the consummation of the acquisition of a Portfolio Investment other than an Initial Property (taking into account the time reasonably necessary to record the closing of a transaction in the appropriate registry(s), as applicable), but in any event not more than thirty (30) days after such acquisition, the General Partner will deliver to the Limited Partners a letter prepared by the General Partner, that (i) states that the transaction has been completed, (ii) lists the amount of dollars that were transferred to the seller of the Portfolio Investment, (iii) contains a sources and uses chart detailing how invested capital was obtained and applied and (iv) states that title to the Portfolio Investment has transferred to the applicable Subsidiary of the Partnership.
- (i) <u>REIT Compliance</u>. The General Partner will (i) monitor the compliance of the REIT with the rules governing "real estate investment trusts" as provided in Section 856 of the Code (the "REIT Rules") and (ii) provide quarterly reporting to the Limited Partners, within thirty (30) days after the end of each quarter, regarding the REIT's compliance with the REIT Rules.
- (j) RioCan's Reporting Expenses. Notwithstanding anything to the contrary contained in this Agreement, if RioCan shall request additional information or materials that are not readily available to the General Partner, or the preparation of additional reports not customarily prepared by the General Partner or the Property Manager (if and to the extent it is an Affiliate of the General Partner), the reasonable cost and expense of providing such information, materials and reports shall be paid by RioCan promptly following demand. The provisions of this Section 8.3(j) shall only apply to supplemental or additional information, materials and reports requested by RioCan, and not to the specific information, materials and reports expressly required to be provided under this Agreement.
- (k) <u>Audit Procedures</u>. Audit procedures, as agreed upon with RioCan, are required to be performed by the Accountant not later than thirty (30) days after the end of each Fiscal Year so as to allow for the completion of the audit of RioCan REIT's financial statements.

8.4 Accountants.

The General Partner shall cause the Partnership to retain Ernst & Young or any other recognized and reputable national independent certified public accounting firm selected by the General Partner to be the accountant and auditor for the Partnership and approved by the Advisory Board (the "Accountant"). The fees and expenses of the Accountant shall be a Partnership expense.

8.5 Annual Budget.

The General Partner shall prepare and deliver to the other Partners an annual budget for the Partnership, each of its Subsidiaries and/or each Portfolio Investment for each Fiscal Year, not later than ninety (90) days prior to the commencement of each Fiscal Year. All budgets will be reforecasted on a quarterly basis. Budgeted income will be prepared in accordance with GAAP.

8.6 Accounting Fee.

The Partnership shall pay to Cedar GP \$25,000 per annum fee for each Portfolio Investment directly or indirectly owned by the Partnership for costs related to the specific reports required to be produced hereunder and under the Property Management Agreements; provided, that the foregoing per annum fee payable for each Portfolio Investment owned directly or indirectly by the Partnership beyond the tenth (10) Portfolio Investment shall be \$15,000 per annum in respect each such Portfolio Investment and in no event shall Cedar GP be entitled to payment of an aggregate amount under this Section 8.6 in excess of \$500,000 per annum.

ARTICLE IX TRANSFER; WITHDRAWAL; REMOVAL OF GENERAL PARTNER

9.1 Transfers Generally.

Except as otherwise expressly set forth in this Article IX, no Partner may sell, assign, pledge, transfer, give, hypothecate or otherwise encumber any direct or indirect interest in the Partnership or any of its Subsidiaries (any such sale, assignment, pledge, transfer, gift, hypothecation, encumbrance or acquisition being hereinafter referred to as a "Transfer"), without the prior written consent of all of the other Partners, which may be granted or withheld in the sole and absolute discretion of such other Partners. For the purposes hereof, "Transfer" shall include a change in ownership of any Partner. Any Transfer of any direct or indirect interest in the Partnership or any of its Subsidiaries in contravention of this Article IX shall be null and void and shall be deemed a material breach of the terms of this Agreement, and the other Partners shall have all the rights and remedies available under this Agreement and applicable law. Notwithstanding anything to contrary contained in this Agreement, (x) transfers of direct or indirect interests in CSCI, Cedar Operating Partnership and RioCan REIT shall be permitted without the consent of any Partner and without the obligation to comply with the further provisions of this Article IX, but subject, however, to the rights of Partners in the event of a Change of Control, and (y) a one time transfer of either (A) forty-nine percent (49%) or a lesser amount of the direct or indirect interests in RioCan to a single Institutional Investor or (B) RioCan's entire Interest to a U.S. entity that is wholly owned and Controlled by RioCan REIT and a single Institutional Investor and at least fifty-one percent (51%) owned, directly or indirectly, by RioCan REIT and not more than forty-nine percent (49%) owned, directly or indirectly, by such Institutional Investor, shall be permitted without the consent of the Cedar Partners, subject to the remaining requirements of this ArticleIX and the rights of the Cedar Partners in the event of a Change of Control with respect to RioCan; provided, that simultaneously with any assignment under the foregoing clause (B), RioCan shall cause such U.S. entity to assume all of the obligations of RioCan under the Purchase and Sale Agreement (including, without limitation, the obligation to pay the Earn-Out Proceeds (as defined therein) to Cedar Operating Partnership as and

9.2 Succession by Operation of Law/Prorations/Cooperation.

If any direct or indirect interest in the Partnership or any of its Subsidiaries is Transferred or proposed to be Transferred pursuant to this Article IX, the parties hereto agree to reasonably cooperate with each other in good faith to structure such Transfer to avoid or minimize transfer fees to lenders and any transfer, deed or similar taxes due in connection therewith and, if so desired, to avoid termination of the Partnership for Federal income tax purposes. All expenses of the Partnership, including transfer taxes (if any), legal, accounting and general audit expenses, occasioned by the sale, assignment or transfer by a Partner of any direct or indirect interest in the Partnership or any of its Subsidiaries or the death, insanity, incompetence or Bankruptcy of a Partner, shall be paid by such Partner or, as applicable, by the transferee of such direct or indirect interest, promptly upon demand thereof, as a condition to the effectiveness of such Transfer.

9.3 General Conditions Applicable to Transfers.

- (a) Notwithstanding anything in this Agreement to the contrary (including but not limited to any of the other sections of this Article IX), except as set forth in clause (x) of Section 9.1, in no event shall (i) any Transfer be made, recognized or consented to by the Partners or deemed effective unless such Transfer will not constitute or result in a material violation or default under any Financing Document or (ii) any direct or indirect interest in the Partnership or any of its Subsidiaries be Transferred to a Person who is the subject of any pending bankruptcy proceedings, or to an individual Person who is a minor or who otherwise lacks legal capacity, and any attempt to effect a Transfer to such a Person shall be void and of no effect and shall not bind the Partnership or (iii) any direct or indirect interest in the Partnership or any of its Subsidiaries be Transferred to a Person (A) named on any list of Persons and governments issued by OFAC pursuant to Executive Order 13224, as in effect on the date hereof, or any similar lists publicly issued by OFAC or any other department or agency of the United States of America ("OFAC Lists"), (B) included in, owned by, controlled by, knowingly acting for or on behalf of, knowingly providing assistance, support, sponsorship, or services of any kind to, or otherwise knowingly associated with any of the Persons referred to or described in the OFAC Lists, or (C) who has knowingly conducted business with or knowingly engaged in any transaction with any Person named on any of the OFAC Lists.
- (b) In the event that any filing, application, approval or consent is required in connection with any Transfer, whether by any Governmental Authority or other third-party, the transferring Partner shall promptly make such filing or application or obtain such approval or consent, at its sole expense.
- (c) Notwithstanding anything to the contrary contained in this Agreement, each Partner shall be an entity organized under the laws of the United States.
- Notwithstanding anything to the contrary contained in this Agreement (including but not (d) limited to the other sections of this Article IX), no Transfer of all or any portion of any Partner's Interest shall be binding upon the other Partners or the Partnership, and the Partnership shall be entitled to treat the record owner of any Interest as the absolute owner thereof in all respects, unless and until (i) true copies of the instruments of transfer executed and delivered pursuant to or in connection with such Transfer shall have been delivered to the General Partner, (ii) the transferee shall have delivered to the General Partner an executed and acknowledged assumption agreement pursuant to which the transferee assumes all the obligations of the transferor arising and accruing from and after the date of such Transfer under, and agrees to be bound by all the provisions of, this Agreement, (iii) the transferee shall have executed, acknowledged and delivered any instruments required under any applicable laws to effect such Transfer and, if applicable, its admission to the Partnership, and (iv) the transferee shall have executed and delivered such other instruments, documents and agreements reasonably required by the General Partner in connection with such Transfer which are consistent with the other terms hereof including, without limitation, a favorable opinion of counsel reasonably satisfactory to General Partner that such Transfer shall not constitute a violation of the Securities Act of 1933, as amended, or of any law or statute of any state and shall have no materially adverse federal income tax impact on the Partnership. Upon compliance with the provisions of this Section 9.3(d) any Person who acquires an Interest in a transaction permitted by this Article IX shall, unless otherwise provided in this Agreement, be admitted as a Partner. Except as otherwise set forth herein, upon the execution and delivery of such assumption agreement, the transferor shall have no further obligation hereunder after the date of the Transfer except that the transferor shall remain primarily liable for all accrued obligations (as of the date of Transfer) of the transferor under this Agreement, notwithstanding any Transfer pursuant to this Article IX.
- (e) Except as otherwise expressly provided herein, all reasonable costs and expenses incurred by the Partnership in connection with any Transfer of an Interest and, if applicable, the admission of a Person as a Partner hereunder, shall be paid by the transferor. Upon compliance with all provisions hereof applicable to any transferee of an Interest becoming a Partner, all Partners hereby agree to execute and deliver such reasonable amendments hereto as are necessary to constitute such person or entity a Partner of the Partnership.
- (f) If any Person acquires all or any part of the Interest of a Partner in violation of this Article IX whether by operation of law, judicial proceeding, or other manner not expressly permitted hereunder, such Person shall have no rights under this Agreement with respect to the Interest so acquired.

(g) If a Transfer of an Interest occurs at any time other than the end of a Fiscal Year, the various items of Partnership income, gain, deduction, loss, credit and allowance as computed for United States federal income tax purposes shall be allocated between the transferor Partner and the transferee Partner in accordance with Section 706 of the Code and the Regulations promulgated thereunder, and the transferor Partner agrees to reimburse the Partnership for any incidental accounting fees and other expenses incurred by the Partnership in making such allocation.

9.4 Buy Sell Rights.

- Either Cedar LP and Cedar GP acting collectively, on the one hand, and RioCan, on the other hand ("Buy Sell Offeror"), shall have the right from time to time to effect the provisions of this Section 9.4 at any time during the Buy Sell Exercise Period by delivering written notice (the "Buy Sell Notice") to RioCan or the Cedar Partners, respectively ("Buy Sell Offeree") (A) of its or their intention to effect the provisions of this Section 9.4(a), (B) designating its or their determination (which shall be made in its or their sole discretion) of the fair market value of all Partnership Assets (the "Partnership Asset Price"), and (C) designating its estimate of the Buy Sell Applicable Purchase Price with respect to each Partner's Interest calculated as if the closing were to occur on the date of the Buy Sell Notice. For the purposes hereof, the "Buy Sell Applicable Purchase Price" payable to any Partner that sells its Interest pursuant to this Section 9.4 shall be the amount (as adjusted as provided below) that such selling Partner would receive as of the applicable calculation date if the Partnership Assets were sold at the Partnership Asset Price and the Partnership and its Subsidiaries were liquidated in accordance with the provisions of subsection 10.2(c) (without giving effect to clause (ii) thereof, and assuming no prepayment premiums, penalties or similar charges shall be due in connection with the repayment of all Financings in connection with such sale). Any disputes as to the Buy Sell Applicable Purchase Price shall be resolved by the Accountant and such determination shall be binding on the Partners. The Partners agree to promptly provide the Accountant with all information necessary to resolve such dispute and shall instruct the Accountant to resolve such dispute as expeditiously as possible.
- (b) Upon receipt of the Buy Sell Notice given pursuant to Section 9.4(a) hereof, Buy Sell Offeree shall then be obligated either to:
 - purchase the Interests of Buy Sell Offeror for cash at a price equal to the Buy Sell Applicable Purchase Price; or
 - (ii) sell its Interests to Buy Sell Offeror for cash at a price equal to the Buy Sell Applicable Purchase Price (the Interests being sold pursuant to this Section 9.4 are hereinafter referred to as the "Buy Sell Interests").

Buy Sell Offeree shall give written notice of its election to Buy Sell Offeror within thirty (30) days after receipt of the Buy Sell Notice (the date of election being the "Buy Sell Election Date"). Failure of Buy Sell Offeree to give Buy Sell Offeror notice within such time shall be a conclusive election under subsection (b)(ii) above.

Within ten (10) Business Days after Buy Sell Offeree's election or deemed election under subsection 9.4(b), the Partner(s) purchasing the Buy Sell Interests (the "Purchasing Partner") shall deposit with the Escrow Agent in cash an amount equal to the greater of (I) Five Hundred Thousand Dollars (\$500,000) and (II) an amount equal to five percent (5%) of the Buy Sell Applicable Purchase Price ("Buy Sell Deposit"). If the Purchasing Partner shall fail to deposit the Buy Sell Deposit within such ten (10) Business Day period, the Purchasing Partner shall be in default hereunder, the other Partner(s) (the "Selling Partner") shall have all remedies available at law or in equity, and the Selling Partner shall have the right, exercisable by delivery of written notice to the Purchasing Partner within ten (10) days of the expiration of such ten (10) Business Day period, to purchase (pursuant to the terms of this Section 9.4) the Interests of the Purchasing Partner for cash at a price equal to ninety percent (90%) of the Buy Sell Applicable Purchase Price and on a date which is not less than thirty (30) days and not more than one hundred twenty (120) days from the Buy Sell Election Date (as selected by the (former) Selling Partner upon not less than ten (10) days notice to the (former) Purchasing Partner). If the Selling Partner does not elect to purchase the Interests of the Purchasing Partner, the rights of the Partners under this Section 9.4 shall be as they were prior to the delivery of the applicable Buy Sell Notice, except that the Purchasing Partner shall lose its right to initiate the buy sell procedures pursuant to this Section 9.4 for a period of eighteen (18) months following the date of the Buy Sell Notice. The charges of the Escrow Agent shall be paid by the Partnership. The Escrow Agent shall hold the Buy Sell Deposit in an interest bearing account pursuant to a written agreement among the Selling Partner, the Purchasing Partner and the Escrow Agent, which agreement shall be satisfactory to such parties in the exercise of their respective reasonable discretion and shall provide, among other things, that the Escrow Agent shall not commingle the Buy Sell Deposit with any other funds. In the event of a closing pursuant to the terms of this subsection 9.4(c), the Buy Sell Deposit, together with any interest earned thereon, shall be credited against the Buy Sell Applicable Purchase Price and paid to the Selling Partner. In the event of a default by the Purchasing Partner in its obligation to purchase the Buy Sell Interests pursuant to, and in accordance with, the terms of this subsection 9.4(c) (other than the failure of the Purchasing Partner to make the Buy Sell Deposit as aforesaid), the Buy Sell Deposit, and any interest thereon, shall be paid to the Selling Partner by the Escrow Agent promptly following written request therefor as the Selling Partner's sole and exclusive remedy, except that the Selling Partner shall have the right, exercisable by delivery of written notice to the Purchasing Partner within thirty (30) days of the Partnership's receipt of the Buy Sell Deposit, to purchase (pursuant to the terms of this Section 9.4) the Interests of the Purchasing Partner for cash at a price equal to ninety-five percent (95%) of the Buy Sell Applicable Purchase Price. If the Selling Partner does not elect to purchase the Interests of the Purchasing Partner, the rights of the Partners under this Section 9.4 shall be as they were prior to the delivery of the applicable Buy Sell Notice, except that the Purchasing Partner shall lose its right to initiate the buy sell procedures pursuant to this Section 9.4 for a period of eighteen (18) months following the date of the Buy Sell Notice. If the Selling Partner shall default in any of its obligations under this subsection 9.4(c), the Buy Sell Deposit, and any interest earned thereon, shall be returned to the Purchasing Partner promptly following written request therefor, the Purchasing Partner shall have all other remedies available to it at law or in equity (including, without limitation, an action for specific performance), and the Selling Partner shall lose its right to initiate the buy sell procedures pursuant to this Section 9.4 for a period of eighteen (18) months following the date of the Buy Sell Notice. Upon deposit by the Purchasing Partner of the Buy Sell Deposit with the Escrow Agent as aforesaid, (i) a binding contract shall be deemed to exist between the Selling Partner and the Purchasing Partner with respect to the Buy Sell Interests, and (ii) the closing shall be held pursuant to an escrow arrangement acceptable to the Partners in the exercise of their reasonable judgment on a Business Day selected by the Purchasing Partner not less than thirty (30) days and not more than one hundred twenty (120) days from the Buy Sell Election Date. The Purchasing Partner shall pay the Buy Sell Applicable Purchase Price (less the Buy Sell Deposit and any interest earned thereon and as adjusted as provided herein) by wire transfer of immediately available federal funds to an account designated in writing by the Selling Partner. At the closing, (A) the Selling Partner shall deliver to Purchasing Partner an assignment of all of the Buy Sell Interests, which such assignment shall be free and clear of all legal and equitable claims (other than the legal and equitable claims, if any, of the Purchasing Partner pursuant to this Agreement) and all liens and encumbrances (other than liens and encumbrances under this Agreement and Financing Documents that shall remain in full force and effect following the closing), and (B) the Purchasing Partner shall deliver to the Selling Partner an assumption of the Selling Partner's obligations under this Agreement arising from and after the date of such assignment.

(c)

- (d) At the closing, (A) all expenses of the Partnership and its Subsidiaries due in connection with a Transfer of the Buy Sell Interests pursuant to Section 9.4(c), including, without limitation, any transfer, controlling interest or other tax, and any prepayment premium or lender transfer fees which are actually due and payable in connection with such Transfer, shall be paid by the Purchasing Partner, (B) the Accountant shall close the books of the Partnership and each of its Subsidiaries as of the closing date, and all items of the Partnership's and each of its Subsidiary's income and expense shall be apportioned in calculating Net Cash Flow (and such other items that are customarily apportioned between sellers and purchasers of real properties shall be apportioned) as of 11:59 p.m. of the day preceding the closing date, (C) Net Cash Flow earned through the closing date and Net Proceeds of a Capital Transaction received prior to the closing date shall be distributed in accordance with the provisions of Article IV, which provisions shall survive the closing pursuant hereto for purposes of making or correcting any closing adjustments, (D) the Buy Sell Applicable Purchase Price (calculated as of the closing date) shall be (x) increased by the aggregate amount of all additional Capital Contributions made by the Selling Partner in the period between the date of the Buy Sell Notice and the closing date (excluding additional Capital Contributions made for payment of ordinary operating expenses), (y) decreased by any amounts of Net Proceeds of a Capital Transaction received by the Partnership with respect to the sale or disposition of any portion of the Portfolio Investments during the period between the date of the Buy Sell Notice and the closing date and distributed to the Selling Partner pursuant to the terms hereof, and (z) adjusted to account for, and fully repay, all outstanding Default Loans (and any accrued and unpaid interest thereon), (E) the Partners shall execute all amendments to fictitious name, partnership or similar certificates necessary to effect the withdrawal of the Selling Partner from the Partnership and, if applicable, the termination of the Partnership, and (F) the Purchasing Partner shall use diligent efforts to secure the release from all lenders (without releasing any claim the Partnership or any of its Subsidiaries may have against the applicable guarantor) of outstanding guaranties in connection with any Financings executed by the Selling Partner or its Affiliates; provided that an entity reasonably acceptable to the Selling Partner shall, pursuant to an agreement in form and content reasonably acceptable to the Selling Partner, defend, indemnify and hold harmless the Selling Partner and its Affiliates, as the case may be, for any claims that arise under such outstanding guaranties for events occurring after the close of the sale of the Selling Partner's Interest if the Purchasing Partner is not able to procure any such release.
- (e) The Partners shall cooperate with each other to effectuate a transfer of the Buy Sell Interests in a manner that will minimize taxes (including, without limitation, transfer taxes) and, if applicable, loan assumption fees; including, without limitation, structuring any such transfer as an entity transfer of the applicable Subsidiaries of the Partnership.
- (f) At the closing, the Selling Partner and the Purchasing Partner shall execute an agreement acceptable to the Selling Partner and the Purchasing Partner in the exercise of their reasonable judgment whereby (X) each shall represent and warrant to the other that each is duly organized, validly existing, has the necessary corporate power and authority to consummate the subject transactions and requires no consents which have not been obtained and (Y) the Selling Partner shall represent to the Purchasing Partner that the Selling Partner is the owner of the Buy Sell Interests free and clear of all liens and encumbrances (other than liens and encumbrances under this Agreement and Financing Documents that shall remain in full force and effect following the closing) and that the Transfer is being made free and clear of all legal and equitable claims (other than the legal and equitable claims of the Purchasing Partner pursuant to this Agreement).
- (g) The Purchasing Partner may, at its option, cause the Buy Sell Interests to be acquired by one or more of Purchasing Partner's designees; provided that any such assignment of the Purchasing Partner's rights hereunder for purposes of accomplishing such purchase by any such designee shall not relieve the Purchasing Partner of any obligation or liability with respect thereto.
- (h) Each Partner agrees that it shall be reasonable and cooperate with the other Partners, including, without limitation, executing any documents which may be reasonably required, in order to consummate the transactions contemplated by this Section 9.4.
- For purposes of the terms of this Section 9.4, Cedar LP and Cedar GP shall be deemed to be one Partner and shall act collectively except solely to the extent that the interests of each are to be transferred to different purchasers.
- (j) Notwithstanding anything to the contrary set forth herein, in the event any rights under Section 9.5 shall be exercised prior in time to the exercise of any rights under this Section 9.4, the rights under Section 9.5 shall supersede any other right existing pursuant to this Section 9.4 (and no Partner shall be entitled to exercise any right hereunder until such time as the procedure under Section 9.5 has been terminated or consummated).

- (a) Anything contained in Section 9.1 of this Agreement to the contrary notwithstanding, if, during the Buy Sell Exercise Period, either Cedar LP and Cedar GP acting collectively, on the one hand, or RioCan, on the other hand (the "ROFR Offeror") desires to sell its or their entire Interest(s) (the "ROFR Interest") to a third party pursuant to a ROFR Third Party Offer (any such sale being hereinafter referred to as a "ROFR Sale"), the ROFR Offeror shall give prompt written notice (the "ROFR Offer Notice") to RioCan or the Cedar Partners, respectively (the "ROFR Offeree") following the ROFR Offeror's receipt of any ROFR Third Party Offer. The ROFR Offer Notice shall include a true and complete copy of the ROFR Third Party Offer. The ROFR Offer Notice shall disclose the identity of such third party, including the principals thereof, and evidence that such third party possesses the financial means to close the contemplated transaction as required under Section 9.5(c) hereof. The ROFR Offer Notice shall be deemed an offer by the ROFR Offeror to sell the ROFR Interest to the ROFR Offeree or the ROFR Offeree's designee on the terms and conditions and for the purchase price set forth in the ROFR Third Party Offer, as modified by this Section 9.5.
- (b) The ROFR Offeree shall, within six (6) months after its receipt of the ROFR Offer Notice, give written notice (a "ROFR Response Notice") to the ROFR Offeror electing to invoke one of the two options described in subparagraphs (i) and (ii) below. If the ROFR Offeree shall fail to send a ROFR Response Notice within such six (6) month period, or shall fail in the ROFR Response Notice to elect one of the two options described in subparagraphs (i) and (ii) below, the ROFR Offeree shall be deemed to have elected to invoke the option described in subparagraph (ii) below.
 - (i) The ROFR Offeree may elect in the ROFR Response Notice to purchase the ROFR Interest. In such event, (A) the economic terms and conditions of the ROFR Third Party Offer shall govern such purchase (including, without limitation, the purchase price, apportionments, and payment of transfer taxes and other closing costs, but excluding remedies upon default) and (B) the non-economic provisions of Section 9.4(c) through (i) of this Agreement (including, without limitation, the timing of closing, deliveries, required consents and remedies upon default) shall govern such purchase, mutatis mutandis, as if the same were an election to purchase a Partner's Interest pursuant to Section 9.4, provided that: (1) the ROFR Offeree shall be the "Purchasing Partner", (2) the ROFR Offeror shall be the "Selling Partner", (3) the ROFR Interest shall be the "Buy Sell Interests", (4) the purchase price set forth in the ROFR Third Party Offer shall be the "Buy Sell Applicable Purchase Price", (5) the closing of such purchase shall occur on the first Business Day that is at least sixty (60) days following the ROFR Offeror's receipt of the ROFR Response Notice (subject to adjournment for not more than sixty (60) days in the aggregate to obtain all required third party consents to such purchase, including, without limitation, the consent of each lender under a Financing), and (6) in no event shall such purchase be conditioned upon the repayment of any Financing or the satisfaction or release of any Financing Document.
 - (ii) The ROFR Offeree may elect not to purchase the ROFR Interest pursuant to subparagraph (i) above. In such event, the ROFR Offeror shall be permitted to sell the ROFR Interest to the third party identified in the ROFR Offer Notice in accordance with the terms of the ROFR Third Party Offer, in all but de minimis respects, including, without limitation, the closing date set forth therein. If the ROFR Sale does not occur on such terms on or prior to the closing date set forth in the ROFR Third Party Offer, the ROFR Interest shall again become subject to the right of first refusal provisions of this Section 9.5.
- For purposes of this provision, a "ROFR Third Party Offer" shall mean an offer by a bona (c) fide third party not affiliated with the ROFR Offeror and having the financial means to close the underlying acquisition to purchase the ROFR Interest, as evidenced by an executed and binding purchase agreement or letter of intent that (i) contains the material terms and conditions of such offer, (ii) is subject to the terms and conditions of this Agreement, (iii) provides that (A) the purchase price shall be payable only in the form of cash, (B) the ROFR Interest shall be delivered free and clear of all liens and encumbrances other than any Financings, and no other property or assets shall be included in such ROFR Sale, and (C) the closing shall be a particular date that is not later than ninety (90) days after the date on which the ROFR Response Notice is given, and (iv) is accompanied by a deposit in the form of a certified check in the amount of at least five percent (5%) of the total purchase price. Notwithstanding the foregoing, (1) the Cedar Partners shall not be permitted, without RioCan's prior written consent (which may be withheld in its sole and absolute discretion), to consummate a ROFR Sale with any third party that is not (or is not wholly owned by) a Person that is regularly engaged in the business of owning, managing or operating commercial real estate properties, (2) RioCan shall not be permitted, without the prior written consent of the Cedar Partners (which may be withheld in their sole and absolute discretion) to consummate a ROFR Sale with any third party that is not an Institutional Investor, (3) no Partner shall be permitted, without the prior written consent of the other Partners (which may be withheld in their sole and absolute discretion) to consummate a ROFR Sale with any third party that is (or is an Affiliate of) the owner of any retail space that, if directly acquired by any Partner, would violate the provisions of Section 7.2 of this Agreement, and (4) each ROFR Sale shall be subject to the provisions of Section 9.3 hereof.
- (d) Notwithstanding anything to the contrary set forth herein, in the event any rights under Section 9.4 shall be exercised prior in time to the exercise of any rights under this Section 9.5, the rights under Section 9.4 shall supersede any other right existing pursuant to this Section 9.5 (and no Partner shall be entitled to exercise any right hereunder until such time as the procedure under Section 9.4 has been terminated or consummated).

Upon the occurrence of a Bankruptcy Event or any other occurrence with respect to a Partner of any event which under the Delaware Act causes the Partner to cease to be a partner of a limited partnership (a "Withdrawal Event"), the Partner affected by such Withdrawal Event shall, unless the other Partners shall otherwise consent within ninety (90) days of such Withdrawal Event, be deemed to have withdrawn as a Partner on the expiration of such ninety (90) day period. In the event that a Partner is deemed to have withdrawn from the Partnership pursuant to this Section 9.6, then such Partner (a "Withdrawn Partner") shall continue to have the rights of an assignee of its Interest which was not admitted as a Partner and shall not be entitled to participate in the management of the Partnership or to vote, approve or consent to any matter for which the vote, approval or consent of any Partners is required. Unless the Partners (other than the Withdrawn Partner) otherwise agree, the Partnership shall not terminate or dissolve upon the occurrence of a Withdrawal Event, provided (to the extent required by any Financing Document) that in the event that the Partnership has two or more General Partners at least one of which is solvent, the Partners shall not agree to terminate or dissolve the Partnership upon the occurrence of a Withdrawal Event. No Partner shall withdraw or retire from the Partnership without the prior written consent of all of the other Partners, except in connection with a Transfer of its entire Interest as expressly permitted under and in accordance with the terms of this Agreement. In furtherance of the foregoing, each Partner hereby waives any and all rights such Partner may have to withdraw and/or resign from the Partnership pursuant to Sections 17-602 and 17-603 of the Delaware Act and hereby waives any and all rights such Partner may have to receive the fair value of such Partner's Interest upon such resignation and/or withdrawal pursuant to Section 17-604 of the Delaware Act, and such Partner shall continue to hold its Interest in accordance with the provisions hereof.

9.7 Death or Incompetency of an Individual Partner.

Upon the death or legal incompetency of an individual Limited Partner (including a substituted Limited Partner), his or her legally authorized personal representatives shall have all of the rights of a Limited Partner for the purpose of settling or managing his or her estate, and shall have such power as the decedent, incompetent, bankrupt or insolvent individual Limited Partner possessed hereunder to make an assignment of his or her interest in the Partnership in accordance with the terms hereof. No such representative shall be admitted as a Limited Partner in the Partnership except in compliance with the provisions of this Article IX.

9.8 General Partner's Withdrawal Rights.

If at any time the Partnership shall have more than one General Partner, a General Partner may withdraw as a General Partner of the Partnership upon obtaining the written consent of all of the other Partners. From and after the effective date of any such withdrawal, the withdrawing General Partner shall automatically cease to serve as the General Partner of the Partnership and such General Partner's Interest shall be deemed to be converted to a limited partnership interest in the Partnership and all references in this Agreement to the "General Partner" shall be deemed to be references to the remaining General Partner only. Except as provided in Article IX, the General Partner may not voluntarily withdraw from the Partnership or dissolve or liquidate.

9.9 Intentionally Omitted.

9.10 Removal of General Partner.

- (a) Generally. The General Partner may be removed upon at least seven (7) days' prior written notice given by a Limited Partner to the General Partner if a Cause Event has occurred, at which time the applicable General Partner shall automatically cease to serve as the General Partner of the Partnership, such General Partner's Interest shall be deemed to be converted to a limited partnership interest in the Partnership and all references in this Agreement to the "General Partner" shall be deemed to be references to the remaining General Partner as provided in Section 9.8 or the replacement General Partner elected pursuant to Section 9.10(a).
- (b) <u>Election of a Replacement General Partner</u>. If the General Partner shall be removed or has withdrawn (or is deemed to have withdrawn), a successor General Partner shall be admitted as a General Partner if the following terms and conditions are satisfied;
 - the admission of such Person shall have been consented to by the Limited Partner that is not an affiliate of the former General Partner;
 - (ii) the Person shall have accepted and agreed to be bound by all the terms and provisions of this Agreement by executing a counterpart hereof and such other documents or instruments as may be required or appropriate in order to effect the admission of such Person as a General Partner as of the effective date of the removal or withdrawal of the former General Partner and the newly admitted General Partner is authorized to and shall continue the business of the Partnership without dissolution;
 - (iii) a certificate evidencing the admission of such Person as a General Partner shall have been filed as provided in the Delaware Act; and
 - (iv) if Cedar GP is the General Partner that has been removed or withdrawn, the replacement General Partner must be a direct or indirect wholly-owned subsidiary of RioCan or RioCan REIT.
- (c) <u>Suspension of Capital Calls</u>. In the event that the General Partner is removed as general partner of the Partnership, the obligation of the Partners to make Capital Contributions to the Partnership shall be suspended until such time as a new general partner has been approved by the Limited Partner that is not an affiliate of the former General Partner, and such new general partner is admitted as the General Partner in accordance with Section 9.10(b).

10.1 Dissolution.

Except as hereinafter provided to the contrary, the Partnership shall be dissolved and its business wound up upon the happening of any of the following events (each, a "Dissolution Event"), whichever shall first occur:

- (a) The sale, condemnation or other disposition of all or substantially all of the Partnership Assets and the receipt of all consideration therefor except that if non-monetary consideration is received upon such disposition the Partnership shall not be dissolved pursuant to this clause until such consideration is converted into money or money equivalent;
- (b) subject to Sections 9.6 and 9.10, upon the dissolution or withdrawal of the General Partner;
- subject to Sections 9.6 and 9.10, upon the occurrence of any Bankruptcy Event with respect to the General Partner;
- (d) at any time that there is no General Partner or any Limited Partners unless the remaining Partners take the necessary action pursuant to Section 17-801(3) or (4) of the Delaware Act, as applicable, to continue the Partnership; and
- (e) the occurrence of any other event which causes dissolution of a limited partnership under the Delaware Act, unless the Partners agree to continue the Partnership pursuant to the Delaware Act.

The General Partner promptly shall notify each of the other Partners in writing of the occurrence of a Dissolution Event.

10.2 Termination.

Notwithstanding any other provision of this Agreement, in all cases of dissolution of the Partnership, the business of the Partnership shall be wound up and the Partnership terminated as promptly as practicable thereafter, and each of the following shall be accomplished:

- (a) The Liquidating Partner shall cause to be prepared (i) statements setting forth the assets and liabilities of the Partnership as of the date of dissolution and as of the date of complete liquidation, a copy of such statements shall be furnished to all of the Partners and (ii) a report in reasonable detail of the manner or disposition of assets.
- (b) The Partnership Assets shall be liquidated by the Liquidating Partner as promptly as possible, but in an orderly and businesslike and commercially reasonable manner. The Liquidating Partner may, in the exercise of its business judgment and if commercially reasonable, determine to defer the sale of all or any portion of the Partnership Assets if deemed necessary or appropriate to realize the fair market value of any such Partnership Assets.
- (c) The proceeds of sale and all other Partnership Assets shall be applied and distributed as follows and in the following order of priority:
 - (i) To the payment of (x) the debts and liabilities of the Partnership and its Subsidiaries (including any outstanding amounts due on any recourse indebtedness encumbering any Portfolio Investment, or any part thereof) and (y) the expenses of liquidation.
 - (ii) To the setting up of any reserves which the Liquidating Partner shall determine in its commercially reasonable judgment to be reasonably necessary for contingent, unliquidated or unforeseen liabilities or obligations of the Partnership, any of its Subsidiaries or the Partners arising out of or in connection with the Partnership or any of its Subsidiaries. Such reserves may, in the commercially reasonable discretion of the Liquidating Partner, be paid over to a national bank or national trust company selected by the Partners and authorized to conduct business as an escrow agent to be held by such bank or trust company as escrow agent for the purposes of disbursing such reserves to satisfy the liabilities and obligations described above, and at the expiration of such period distributing any remaining balance as provided in clause (iii) below.
 - (iii) The balance to the Partners in accordance with their respective Percentage Interests.

Distributions pursuant to the preceding clause (iii) shall be made by the end of the Fiscal Year during which the dissolution of the Partnership occurs (or, if later, within ninety (90) days of such dissolution). To the fullest extent permitted by applicable law, the Partners hereby waive any rights to distributions under Section 17-604 of the Delaware Act.

(d) The Liquidating Partner shall cause the filing of the Certificate of Cancellation pursuant to Section 17-203 of the Delaware Act and shall take all such other actions as may be necessary to terminate the Partnership.

10.3 Liquidating Partner.

- (a) The term "Liquidating Partner" shall mean (i) the General Partner in the case of a termination of the Partnership pursuant to clause (a) of Section 10.1 hereof, (ii) Cedar GP in the case of a termination of the Partnership pursuant to clause (e) of Section 10.1 hereof if RioCan shall be the Partner causing the termination event pursuant to said clause, (iii) Cedar LP in the case of a termination of the Partnership pursuant to clause (b) or (c) of Section 10.1 hereof if the General Partner was RioCan or an Affiliate of RioCan, (iv) RioCan in the case of a termination of the Partnership pursuant to clause (e) of Section 10.1 hereof if Cedar LP or Cedar GP shall be the Partner causing the termination event pursuant to said clause, (v) RioCan in the case of a termination of the Partnership pursuant to clause (b) or (c) of Section 10.1 hereof if the General Partner was Cedar GP, and (vi) the last remaining Partner (or its personal representative or nominee) in the case of a termination of the Partnership pursuant to clause (d) of Section 10.1 hereof.
- (b) Without limiting the foregoing, the Liquidating Partner shall, upon the dissolution and upon completion of the winding up of the affairs of the Partnership, file appropriate certificate(s) to such effect in the proper governmental office or offices under the Delaware Act as then in effect. Notwithstanding the foregoing, each Partner, upon the request of the Liquidating Partner, shall promptly execute, acknowledge and deliver all such documents, certificates and other instruments as the Liquidating Partner shall reasonably request to effectuate the proper dissolution and termination of the Partnership, including the winding up of the business of the Partnership.

10.4 Partnership Assets Reserved and Pending Claims.

- (a) Assets Reserved. If, upon a Dissolution Event, there are any Partnership Assets that, in the judgment of the Liquidating Partner, cannot be sold without sacrificing a significant portion of the value thereof or where such sale is otherwise impractical at the time of the Dissolution Event, such Partnership Assets may be retained by the Partnership if the Liquidating Partner determines that the retention of such Partnership Assets is in the best interests of the Partners and such Partnership Assets shall not be considered for purposes of computing Capital Accounts upon winding-up and amounts distributable pursuant to Section 10.2(c). Upon the sale of such Partnership Assets or a determination by the Liquidating Partner that circumstances no longer require their retention (but in no event more than (2) two years after the Dissolution Event), such Partnership Assets (at their Fair Market Value, as determined in good faith by the Liquidating Partner) or the proceeds of their sale shall be taken into account in computing Capital Accounts on winding-up and amounts distributable pursuant to Section 10.2(c) and distributed in accordance with such value.
- (b) Pending Claims. If there are any claims or potential claims (including potential Partnership expenses in connection therewith) against the Partnership (either directly or indirectly, including potential claims for which the Partnership might have an indemnification obligation) for which the possible loss cannot, in the judgment of the Liquidating Partner, be reasonably ascertained, then such claims shall initially be taken into account in computing Capital Accounts upon winding-up and distributions pursuant to Section 10.2(c) at an amount estimated by the Liquidating Partner to be sufficient to cover any potential loss or liability on account of such claims (including such potential Partnership expenses), and the Partnership shall retain funds (or assets) determined by the Liquidating Partner acting reasonably as a reserve against such potential losses and liabilities, including expenses associated therewith. The Liquidating Partner may in its reasonable discretion obtain insurance or create escrow accounts or make other similar arrangements with respect to such losses and liabilities. Upon final settlement of such claims (including such potential Partnership expenses) or a determination by the Liquidating Partner that the probable loss therefrom can be definitively ascertained, such claims (including such potential Partnership expenses) shall be taken into account in the amount at which they were settled or in the amount of the probable loss therefrom in computing Capital Accounts on winding-up and amounts distributable pursuant to Section 10.2(c)

10.5 No Redemption.

The Partnership may not acquire, by purchase, redemption or otherwise any Interest of any Partner.

10.6 Governance.

Notwithstanding a dissolution of the Partnership, until the termination of the business of the Partnership, the affairs of the Partners, as such, shall continue to be governed by this Agreement. The Liquidating Partner shall be subject to the same restrictions on transactions with related parties or involving conflicts of interest as applied prior to the dissolution of the Partnership, including but not limited to the consent requirements set forth herein of any such transaction. The Liquidating Partner shall also be required to perform its duties under this Agreement using the same standard of care that would be required of the Liquidating Partner if the Liquidating Partner were acting as the General Partner.

10.7 Return of Capital.

No Partner shall have any right to receive the return of its Capital Contribution or to seek or obtain partition of assets of the Partnership, other than as expressly provided in this Agreement.

ARTICLE XI INTENTIONALLY OMITTED

ARTICLE XII CONFIDENTIALITY

12.1 Disclosure.

Notwithstanding any terms or conditions in this Agreement to the contrary, but subject to restrictions reasonably necessary to comply with federal or state securities laws, any person may disclose to any and all persons, without limitation of any kind, the tax treatment and tax structure of the transaction and all materials of any kind (including opinions or other tax analyses) that are provided relating to such tax treatment and tax structure. For the avoidance of doubt, this authorization is not intended to permit disclosure of the names of, or other identifying information regarding, the participants in the transaction, or of any information or the portion of any materials not relevant to the tax treatment or tax structure of the transaction.

12.2 Confidentiality.

- (a) Confidentiality. Each of the Partners shall, and shall direct those of its directors, officers, partners, members, employees, attorneys, accountants, consultants, trustees, Affiliates and advisors (the "Representatives") who have access to Confidential Information to, keep confidential and not disclose any Confidential Information without the express consent, in the case of Confidential Information acquired from the Partnership or with respect to the Partnership (or from, or with respect to, any Subsidiary of the Partnership or any Portfolio Investment) or, in the case of Confidential Information acquired from the other Partners or their Representatives, such other Person, unless:
 - such disclosure shall be required by applicable law, governmental rule or regulation, court order, administrative or arbitral proceeding or by any regulatory authority having jurisdiction over the Person required to make such disclosure;
 - (ii) such disclosure of Confidential Information relating to the Partnership or any any of its Subsidiaries or any Portfolio Investment is reasonably required in connection with any proposed assignment, sale or other disposition of all or any part of an Interest in the Partnership (e.g., to an Institutional Investor pursuant to Section 9.1 or 9.5) (a "Proposed Transfer"); provided, that with respect to the use of any Confidential Information in any Proposed Transfer referred to in this clause (ii), reasonable advance notice shall be given to the Person whose information will be disclosed so that it may require any proposed transferee to enter into a confidentiality agreement with terms substantially similar to the terms of this Section 12.2(a) prior to the disclosure of such Confidential Information.
- (b) Confidential Information. "Confidential Information" shall mean any confidential, non-public information related to the activities, as applicable, of the (i) Partnership, the General Partner, any Subsidiary of the Partnership and the respective Affiliates of each of the foregoing, (ii) any Partner and/or its Affiliates or (iii) any Portfolio Investment, provided, however, that the parties acknowledge that this Agreement and the Purchase and Sale Agreement may be included in an SEC filing and a filing required by the Ontario Securities Commission and shall not be Confidential Information.
- (c) <u>Disclosure of Confidential Information</u>. In the event that any Person bound by the terms of this Section 12.2 or any Representative of such Person is required to disclose any Confidential Information, such Person will provide the applicable Partner with prompt written notice so that such Person, as applicable, may seek a protective order or other appropriate remedy, and such Person required to disclose the Confidential Information will use reasonable efforts (but without expense to such Partner) to cooperate with the Person whose information is required to be disclosed, as applicable, in any effort any such Person undertakes to obtain a protective order or other similar remedy. In the event that such protective order or other remedy is not obtained, the disclosing Partner and its Representatives will furnish only that portion of the Confidential Information that is required and will exercise all reasonable efforts to obtain reasonably reliable assurance that the Confidential Information will be accorded confidential treatment.
- (d) <u>Limited Partner Disclosure Restrictions</u>. Except as otherwise required by applicable law, each of the Partners agrees that it will not, and it will cause its Affiliates, the Partnership, the Partnership's Subsidiaries and their respective other Affiliates not to, without the prior written consent of the other Partners, (i) use in advertising, publicity, or otherwise the name of such other Partners or their Affiliates, or (ii) represent, directly or indirectly, that any product or any service provided by any of the foregoing has been approved or endorsed by such other Partners or their Affiliates. This provision shall survive termination of the Partnership.

12.3 Additional Information/Deliveries.

In addition to any reports required by the express terms of this Agreement, the General Partner will provide to each Limited Partner any additional information and reports regarding the operations of the Partnership and any Portfolio Investment or Subsidiary as such Limited Partner may reasonably request including, without limitation, such additional reports and deliveries set forth on Schedule 7.4 attached hereto. In addition, the General Partner agrees to work with each Limited Partner and its shareholders, employees, agents and/or authorized consultants to provide on a periodic basis, to the extent not already provided to such Limited Partner pursuant to this Agreement, and upon the written request of such Limited Partner, data regarding the performance of the Partnership and any Portfolio Investment or Subsidiary, the valuation of the Partnership Assets and total returns, including time-weighted returns, of such Limited Partner's investment in the Partnership. The General Partner further agrees that, at the request of any Limited Partner, upon reasonable prior written notice and during reasonable business hours, it will meet with such Limited Partner and any of its shareholders, its employees, agents and/or authorized consultants at the offices of the Partnership to review the Partnership's performance and the valuation of the Partnership Assets. Upon the reasonable request of any Limited Partner, the General Partner shall provide to such Limited Partner, without representing or warranting to such Limited Partner the accuracy thereof, good faith calculations of anticipated investment returns determined on a nominal basis. Notwithstanding anything to the contrary contained in this Agreement, if any Limited Partner shall request additional information or materials that are not readily available to the General Partner (or a Property Manager that is the General

Partner's Affiliate), or the preparation of additional reports not customarily prepared by the General Partner (or a Property Manager that is the General Partner's Affiliate), the reasonable cost and expense of providing such information, materials and reports shall be paid by such Limited Partner promptly following demand. The provisions of this Section 12.3 shall only apply to supplemental or additional information, materials and reports requested by a Limited Partner, and not to the specific information, materials and reports expressly required to be provided under this Agreement.

ARTICLE XIII POWER OF ATTORNEY

Each of the Limited Partners hereby irrevocably constitutes and appoints the General Partner, or any successor General Partner, its true and lawful attorney-in-fact with the power and authority to act in such Limited Partner's name and on his behalf in his place and stead, upon five (5) Business Days notice to such Limited Partner, to make, execute, acknowledge, file and record the following documents:

- (a) Amendments to this Agreement as required by the laws of the State of Delaware, or by any other state, including amendments required for the admission or substitution of a Limited Partner, the admission or substitution of a General Partner, and the continuation of the business of the Partnership after the withdrawal or removal of a General Partner;
- (b) Any cancellation of this Agreement as required by the laws of the State of Delaware upon dissolution or termination of the Partnership;
- (c) Amendments to the Certificate as required under the laws of the State of Delaware, or the laws of any other state in which such Certificate (and amendments) are required to be filed or recorded;
- (d) All such other instruments, documents and certificates which may from time to time be required by the laws of the State of Delaware, the United States of America or any other jurisdiction which the Partnership shall determine to do business in accordance with the terms of this Agreement, or any other political subdivision or agency thereof, to effectuate, implement, continue and defend the validity and existence of the Partnership; and
- (e) Any business certificate, fictitious name certificate, certificate of limited partnership, amendment thereto or other instrument or document of any kind necessary to accomplish the business, purposes and objectives of the Partnership.

The power of attorney hereby granted to the General Partner is a special power of attorney coupled with an interest, is irrevocable, and shall survive the death of any Limited Partners that are individuals. This power of attorney may be exercised by the General Partner for each Limited Partner by listing all of the Limited Partners executing any instrument with a signature of the General Partner acting as attorney-in-fact for all of them. In addition, this power of attorney shall survive the delivery of an assignment by a Limited Partner of the whole or any portion of its Interest; except that where the transferee of a Limited Partner has been approved by the General Partner for admission to the Partnership as a substitute Limited Partner, the power of attorney shall survive the delivery of such assignment for the sole purpose of enabling the General Partner to execute, acknowledge, and file any instrument necessary to effect such substitution.

ARTICLE XIV AMENDMENTS; WAIVER

14.1 Amendments; Waiver.

Any amendment to this Agreement shall require the unanimous written consent of the Advisory Board Members.

ARTICLE XV MISCELLANEOUS

15.1 Further Assurances.

Each Partner agrees to execute, acknowledge, deliver, file, record and publish such further reasonable certificates, amendments to certificates, instruments and documents, and do all such other reasonable acts and things as may be required by law, or as may be required to carry out the intent and purposes of this Agreement so long as any of the foregoing do not increase any Partner's obligations hereunder or decrease any Partner's rights hereunder.

15.2 Notices.

All notices, demands, consents, approvals, requests or other communications which any of the parties to this Agreement may desire or be required to give hereunder (collectively, "Notices") shall be in writing and shall be given by personal delivery (including by hand or reputable international courier service) or facsimile or United States or Canada, as applicable, registered or certified air mail (postage prepaid, return receipt requested) addressed as hereinafter provided; provided, however, that any Notice given by facsimile shall also be given by personal delivery or United States or Canada, as applicable, registered or certified air mail. Except as otherwise specified herein, the time period in which a response to any notice or other communication must be made, if any, shall commence to run on the earliest to occur of (a) if by personal delivery, the date of receipt, or attempted delivery, if such communication is refused; (b) if given by facsimile, the date on which such facsimile is transmitted and confirmation of delivery thereof is received if received before 5:00 p.m. on a Business Day, or otherwise on the next Business Day; and (c) if sent by mail (as aforesaid), the date of receipt or attempted delivery, if such mailing is refused. Until further notice, notices and other communications under this Agreement shall be addressed to the parties listed below as follows:

(i) If to the Partnership, Cedar GP or Cedar LP, to:

Cedar Shopping Centers, Inc. 44 South Bayles Avenue Port Washington, NY 11050 Attention: Leo S. Ullman Facsimile: (516) 767-6497

with a copy to:

Stroock & Stroock & Lavan LLP 180 Maiden Lane New York, New York 10038 Attention: Steven Moskowitz, Esq. Facsimile: (212) 806-6006

(ii) If to RioCan, to:

Yonge Eglinton Centre 2300 Yonge Street Suite 500, P.O. Box 2386 Toronto, Ontario M4P 1E4

Attention: Rags Davloor Facsimile: (416) 866-3020

and to:

Yonge Eglinton Centre 2300 Yonge Street Suite 500, P.O. Box 2386 Toronto, Ontario M4P 1F4

Attention: Jonathan Gitlin Facsimile: (416) 866-3020

and to (if prior to December 20, 2009):

Goodmans LLP 250 Yonge Street, Suite 2400 Toronto, Ontario M5B 2M6 Attention: Juli Morrow Facsimile: (416) 979-1234

or to (if on or after December 20, 2009):

Goodmans LLP 333 Bay Street, Suite 3400 Bay Adelaide Centre, West Tower Toronto, Ontario M5H 2S7 Attention: Juli Morrow Facsimile: (416) 979-1234

Any Partner may designate another addressee (and/or change its address) for Notices hereunder by a Notice given pursuant to this Section. Copies of all Notices required to be sent by a Partner to the Partnership under the terms of this Agreement shall also be sent to each Partner in accordance with the terms hereof.

15.3 Applicable Law.

THIS AGREEMENT, THE RIGHTS AND OBLIGATIONS OF THE PARTIES HERETO, AND ANY CLAIMS OR DISPUTES RELATING THERETO SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF DELAWARE (BUT NOT INCLUDING THE CHOICE OF LAW RULES THEREOF).

15.4 Headings, Etc.

All titles or captions contained in this Agreement are inserted only as a matter of convenience and for reference and in no way define, limit, extend, or describe the scope of this Agreement or the intent of any provision hereof.

15.5 Gender.

All pronouns and any variations thereof shall be deemed to refer to the masculine, feminine, and neuter, singular and plural, as the identity of the party or parties may require.

15.6 Successors and Assigns.

This Agreement shall be binding upon the parties hereto and their respective executors, administrators, legal representatives, heirs, successors and assigns, and shall inure to the benefit of the parties hereto and, except as otherwise provided herein, their respective executors, administrators, legal representatives, heirs, successors and permitted assigns.

15.7 No Waiver.

Except as otherwise expressly provided herein, no delay or omission in the exercise of any power, remedy or right herein provided or otherwise available to a Partner or the Partnership shall impair or affect the right of such Partner or the Partnership thereafter to exercise the same. Any extension of time or other indulgence granted to a Partner hereunder shall not otherwise alter or affect any power, remedy or right of any other Partner or of the Partnership.

15.8 Rule of Construction.

The general rule of construction for interpreting a contract, which provides that the provisions of a contract should be construed against the party preparing the contract, is waived by the parties hereto. Each party acknowledges that such party was represented by separate legal counsel in this matter who participated in the preparation of this Agreement or such party had the opportunity to retain counsel to participate in the preparation of this Agreement but elected not to do so. None of the provisions of this Agreement shall be for the benefit of or enforceable by any creditor of the Partnership or any third party. No Partner shall be obligated personally for any debt, obligation or liability of the Partnership solely by being a Partner of the Partnership. Without the consent of all the Partners, the Partnership shall not do business in or otherwise have contact with any jurisdiction other than the State of Delaware and any State or Commonwealth in which any Portfolio Investment is located, if same would result in any Partner being obligated personally for any debt, obligation or liability of the Partnership solely by reason of being a Partner of the Partnership and exercising its rights under this Agreement and the Delaware Act.

15.9 Severability.

In case any one or more of the provisions contained in this Agreement or any application thereof shall be invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions contained herein and other application thereof shall not in any way be affected or impaired thereby.

15.10 Consents.

Except as otherwise expressly provided herein, any consent or approval to any act or matter required under this Agreement must be in writing and shall apply only with respect to the particular act or matter to which such consent or approval is given, and shall not relieve any Partner from the obligation to obtain the consent or approval, as applicable, wherever required under this Agreement to any other act or matter.

15.11 Entire Agreement.

This Agreement contains the entire agreement between the parties relating to the subject matter hereof and all prior agreements relative hereto which are not contained herein are terminated. Amendments, variations, modifications or changes herein may be made effective and binding upon the parties by, and only by, the setting forth of same in a document duly executed by each party, and any alleged amendment, variation, modification or change herein which is not so documented shall not be effective as to any party. Each of the parties agrees that in the event of a conflict between the terms and conditions of this Agreement and the terms and conditions of any Governing Agreements, the terms and conditions of this Agreement shall control.

15.12 Consent to Jurisdiction.

Any action, suit or proceeding in connection with this Agreement may be brought against any Partner or the Partnership in a court of record of the State of New York, County of New York, or in the United States District Court for the Southern District of New York, each Partner and the Partnership hereby consenting and submitting to the jurisdiction thereof. Service of process may be made upon any Partner or the Partnership, by certified or registered mail, at the address to be used for the giving of notice to such Partner under Section 15.9. Each Partner hereby appoints Corporation Service Company, 80 State Street, Albany, New York 12207 as its agent for service of process, with any fees therefore to be borne by the Partnership. Nothing herein shall affect the right of any Partner to commence legal proceedings or otherwise to proceed against any other Partner or the Partnership in any other jurisdiction or to serve process in any manner permitted by applicable law. In any action, suit or proceeding in connection with this Agreement, each Partner and the Partnership hereby waives trial by jury, and any claim that New York County or the Southern District of New York is an inconvenient forum.

15.13 Counterparts.

This Agreement may be executed in any number of counterparts, and each such counterpart will for all purposes be deemed an original, and all such counterparts shall constitute one and the same instrument.

15.14 Representations and Warranties.

- (a) Cedar LP represents and warrants and covenants as follows:
 - Cedar LP is a limited liability company duly organized, validly existing and in good standing under the laws of the State of Delaware.

- (ii) The execution and delivery of this Agreement and all other documents, instruments and agreements to be executed in connection with the transactions contemplated by this Agreement (the "Transaction Documents") have been duly and validly authorized by all necessary actions of Cedar LP, and shall constitute the legal, valid and binding obligations of Cedar LP enforceable against Cedar LP in accordance with the terms hereof and thereof except as the enforcement thereof may be limited by bankruptcy, insolvency, reorganization, liquidation, receivership, moratorium or other similar laws related to or affecting the enforcement of creditors' rights generally or by general principles of equity, regardless of whether such enforceability is considered in a proceeding in equity or at law.
- (iii) No consent, waiver, approval or authorization of or notice to any other Person (including any Governmental Authority) is required to be made, obtained or given by Cedar LP in connection with the execution and delivery of this Agreement or any other Transaction Document except for those which have been heretofore obtained.
- (iv) Neither the execution or delivery of this Agreement nor any other Transaction Document does or will, with or without the giving of notice, lapse of time or both, (i) violate, conflict with or constitute a default under any term or provision of (A) any agreement to which Cedar LP is a party or by which it is bound, or (B) any judgment, decree, order, statute, injunction, rule or regulation of a Governmental Authority applicable to Cedar LP, or by which it or its assets or properties are bound, or (ii) result in the creation of any lien or encumbrance upon Cedar LP or its assets.
- As of the Closing, Cedar LP is a wholly-owned subsidiary of Cedar Operating Partnership.
- (b) Cedar GP represents and warrants and covenants as follows:
 - Cedar GP is a limited liability company duly formed, validly existing and in good standing under the laws of the State of Delaware.
 - (ii) The execution and delivery of this Agreement and all other Transaction Documents have been duly and validly authorized by all necessary actions of Cedar GP, and shall constitute the legal, valid and binding obligations of Cedar GP enforceable against Cedar GP in accordance with the terms hereof and thereof except as the enforcement thereof may be limited by bankruptcy, insolvency, reorganization, liquidation, receivership, moratorium or other similar laws related to or affecting the enforcement of creditors' rights generally or by general principles of equity, regardless of whether such enforceability is considered in a proceeding in equity or at law.
 - (iii) No consent, waiver, approval or authorization of or notice to any other Person (including any Governmental Authority) is required to be made, obtained or given by Cedar GP in connection with the execution and delivery of this Agreement or any other Transaction Document except for those which have been heretofore obtained.
 - (iv) Neither the execution or delivery of this Agreement nor any other Transaction Document does or will, with or without the giving of notice, lapse of time or both, (i) violate, conflict with or constitute a default under any term or provision of (A) any agreement to which Cedar GP is a party or by which it is bound, or (B) any judgment, decree, order, statute, injunction, rule or regulation of a Governmental Authority applicable to Cedar GP, or by which it or its assets or properties are bound, or (ii) result in the creation of any lien or encumbrance upon Cedar GP or its assets.
 - As of the Closing, Cedar GP is a wholly-owned subsidiary of Cedar Operating Partnership.
- (c) RioCan represents and warrants and covenants as follows:
 - RioCan is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware.
 - (ii) The execution and delivery of this Agreement and all other Transaction Documents have been duly and validly authorized by all necessary actions of RioCan and shall constitute the legal, valid and binding obligations of RioCan enforceable against RioCan in accordance with the terms hereof and thereof except as the enforcement thereof may be limited by bankruptcy, insolvency, reorganization, liquidation, receivership, moratorium or other similar laws related to or affecting the enforcement of creditors' rights generally or by general principles of equity, regardless of whether such enforceability is considered in a proceeding in equity or at law.
 - (iii) No consent, waiver, approval or authorization of or notice to any other Person (including any Governmental Authority) is required to be made, obtained or given by RioCan in connection with the execution and delivery of this Agreement or any other Transaction Document except for those which have been heretofore obtained.

- (iv) Neither the execution or delivery of this Agreement nor any other Transaction Document does or will, with or without the giving of notice, lapse of time or both, (i) violate, conflict with or constitute a default under any term or provision of (A) any agreement to which RioCan is a party or by which it is bound, or (B) any judgment, decree, order, statute, injunction, rule or regulation of a Governmental Authority applicable to RioCan or by which RioCan or its assets or properties are bound, or (ii) result in the creation of any lien or encumbrance upon RioCan or its assets.
- (v) As of the Closing, RioCan is wholly-owned and Controlled by RioCan REIT (except for ownership of a de minimis interest by such outside investors as are necessary for RioCan to qualify as a "real estate investment trust" pursuant to Section 856 of the Code).

15.15 Partnership Name.

If, at any time, the Partnership name shall include the name of, or any trade name used by, a Partner or any of its Affiliates, neither the Partnership nor any other Partner shall acquire any right, title or interest in or to such name or trade name.

15.16 Ownership of Partnership Property.

The interest of each Partner in the Partnership shall be personal property for all purposes. All real and other property owned by the Partnership shall be deemed owned by the Partnership as Partnership property. No Partner, individually, shall have any direct ownership of such property and title to such property shall be held in the name of the Partnership.

15.17 Time of the Essence.

Except as otherwise expressly provided in this Agreement, time shall be of the essence with respect to all time periods set forth in this Agreement.

15.18 Status Reports.

Recognizing that each Partner may find it necessary from time to establish to third parties, such as accountants, banks, mortgagees, prospective transferees of its Interest, or the like, the then current status of performance of the Portfolio Investments and the Partnership hereunder, each Partner shall, within ten (10) Business Days following the written request of another Partner made from time to time, furnish a written statement on the status of the following:

- that this Agreement is unmodified and in full force and effect (or if there have been modifications, that the Agreement is in full force and effect as modified and stating the modifications);
- (b) stating whether or not to the best knowledge of such certifying Partner (i) the requesting Partner is in default in keeping, observing or performing any of the terms contained in this Agreement and, if in default, specifying each such default (limited to those defaults of which the certifying Partner has knowledge), and (ii) there has occurred an event that with the passage of time or the giving of notice, or both, would ripen into a default hereunder on the part of the requesting Partner (limited to those events of which the certifying Partner has knowledge); and
- (c) to the best of the knowledge and belief of the Partner making such statement, with respect to any other matters as may be reasonably requested by the requesting Partner.

Such statement may be relied upon by the requesting Partner and any other Person for whom such statement is requested, but no such statement shall operate as a waiver as to any default or other matter as to which the Partner executing it did not have actual knowledge.

15.19 Waiver of Partition.

Except as otherwise expressly provided for in this Agreement, no Partner shall, either directly or indirectly, take any action to require partition or appraisement of the Partnership or any of its assets or properties or cause the sale of any Partnership assets or property, and notwithstanding any provisions of applicable law to the contrary, each Partner (for itself and its legal representatives, successors and assigns) hereby irrevocably waives any and all right to partition, or to seek, bring or maintain any action for partition, or to compel any sale with respect to its interest in, or with respect to, any assets or properties of the Partnership regardless of the manner in which title to such property may be held, except as expressly provided in this Agreement.

15.20 Calculation of Days

The provisions of this Agreement relative to number of days shall be deemed to refer to calendar days, unless otherwise specified. When the date for performance of any monetary obligation of any Partner falls on a non-business day, such obligation need not be performed until the next-following Business Day.

15.21 Dollar Amounts.

All references in this Agreement to dollar amounts shall be to U.S. Dollars.

15.22 No Third-Party Rights.

Except for the Protected Persons and the rights of such parties expressly created hereby, this Agreement is intended solely for the benefit of the parties hereto and is not intended to confer any benefits upon, or create any

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date first written.

GENERAL PARTNER: _] **GP,** LLC, a Delaware limited liability company Cedar Shopping Centers Partnership, L.P., By: a Delaware limited partnership, its sole member By: Cedar Shopping Centers, Inc., a Maryland corporation, its general partner Leo S. Ullman President LIMITED PARTNERS: _] LP, LLC, a Delaware limited liability company Cedar Shopping Centers Partnership, L.P., By: a Delaware limited partnership, its sole member By: Cedar Shopping Centers, Inc., a Maryland corporation, its general partner Leo S. Ullman President RIOCAN HOLDINGS USA INC., a Delaware corporation By:

EXHIBIT C

PRE-RIOCAN OWNER AGREEMENTS

Property	Exhibit #
Columbus Crossing	C(i)
Franklin Village	C(ii)
Loyal Plaza	C(iii)
Stop & Shop – Bridgeport	C(iv)
Blue Mountain Commons	C(v)
Sunset Crossing	C(vi)
Shaw's Plaza	C(vii)

EXHIBIT D

PROPERTY MANAGEMENT AGREEMENT

BACKGROUND

A.	Owner o	owns	or	ground	leases	certain	land	and	improvements	known	as
], located in [] (t	he "Prop	erty")				

B. Owner desires to retain Agent as its exclusive agent for the purposes of leasing and managing the Property on behalf of Owner, and Agent is willing to act as agent for Owner with respect to the Property, on the terms and conditions of this Agreement as more fully set forth herein.

NOW THEREFORE, in consideration of the agreements and covenants herein contained, and intending to be legally bound hereby, Owner, Agent and RioCan REIT agree as follows:

- 1. <u>Retainer</u>: Owner hereby retains Agent to manage and lease, as the exclusive broker, the Property upon the terms and conditions hereinafter set forth for an initial term of three (3) years from the date hereof unless otherwise extended, renewed or terminated as hereinafter set forth. This Agreement is not one of agency between the Agent for Owner but one with the Agent engaged independently in the business of managing the Property as an independent contractor. All employment arrangements are therefore solely Agent's concern, and Owner shall have no liability with respect thereto. Nothing herein shall create an agency coupled with an interest. All records maintained by Agent with respect to the operation or maintenance of the Property shall, at all times, be and constitute the property of Owner and shall be surrendered to Owner in accordance with the terms hereof, without charge or expense.
- 2 . <u>Standard of Care</u>: The Agent will exercise its powers and discharge its duties under this Agreement diligently, honestly, and in good faith. Without limitation, in making decisions and managing the Property, the Agent will exercise the standard of care that a prudent manager of similar properties would exercise in similar circumstances (the "**Standard of Care**").
- 3 . <u>Duties</u>: The Agent shall, at the Owner's sole cost and expense, carry out and perform the following services in accordance with the Standard of Care, subject to the terms and conditions of this Agreement, including, without limitation, the availability of funds to operate the Property:
- 3.1 Use its best efforts to lease or cause brokers or other agents to lease on behalf of Owner all available space in the Property in accordance with the Approved Budget and Approved Leasing Plan (as such terms are hereinafter defined), provided that approval of the Owner shall be required for all leases which require approval pursuant to Section 6.3(a) of the Limited Partnership Agreement of [CR] L.P. (the "Partnership"), of even date, as amended from time to time (the "Partnership Agreement"). In connection with such efforts the Agent will conduct appropriate credit reviews of prospective tenants, and enter into negotiations and finalize leases with Tenants (as hereinafter defined) wishing to enter into, renew, extend, renegotiate or restructure their Leases (as hereinafter defined) or expand the leased premises;
- 3.2 Diligently to collect rents, additional rents and all other sums due from Tenants when due and, where necessary or appropriate, and except as directed otherwise by Owner, take all such actions as Agent shall deem necessary or advisable to enforce all rights and remedies of Owner under the leases relating to the Property (the "Leases") or to protect the interest of Owner, including, without limitation, the preparation and delivery to tenants under the Leases ("Tenants") of all "late payment", default, and other appropriate notices, requests, bills, demands, and statements. Agent may retain counsel, collection agencies, and such other persons and firms as Agent shall deem appropriate or advisable to enforce, after notification to Owner, by legal action the rights and remedies of Owner against any Tenant in default in the performance of its obligations under a Lease. Agent shall promptly notify Owner of the progress of any such legal action;
- 3.3 To pay from the operating funds of the Property or such other funds as are provided by Owner bills and expenses for the maintenance, repair and operation of the Property, (including all utilities and realty taxes), provided, however, that all expenditures in excess of (a) 115% of any line item of an Approved Budget; or (b) in excess of 110% of the total of an Approved Budget, shall require the prior approval of the Owner. The Agent shall dispute, settle or compromise claims for such expenses and charges where appropriate;
- 3.4 Use commercially reasonable efforts, at Owner's expense, to do or cause to be done all such things as are necessary to ensure material compliance by the Owner as landlord with all its covenants, duties, agreements, obligations, terms and conditions of the contracts and encumbrances affecting the Property, and all Leases and any ground lease (if applicable), including, without limitation, providing annual operating cost statements as required by the Leases (with copies to RioCan REIT if it so requests), in each case, to the extent the Agent shall have knowledge of the same;
- 3.5 To establish and maintain such books of account, records, and other documentation pertaining to the operation and maintenance of the Property as are customarily maintained by managing agents of properties similar in location and size to that of the Property. The Agent shall notify RioCan REIT from time to time upon request of the location of such records. Financial records shall be maintained in accordance with United States generally accepted accounting principles consistently applied ("GAAP") and in sufficient detail to facilitate adequate audit and review thereof. Agent shall prepare or cause to be prepared and file all returns and other reports relating to the Property, other than income tax returns and any reports or returns that may be required of any foreign owner of U.S. real property (except as expressly set forth herein);
- 3.6 Select all lawyers and such other professional advisors necessary or appropriate to be retained by the Agent to advise in connection with the management, operation, repair, maintenance, administration and supervision of the Property and the costs of same shall at all times be subject to the Approved Budget or otherwise approved by the Owner, except as otherwise set out herein;
 - 3.7 To account for all advance deposits of Tenants;
- 3.8 To take such steps as may be necessary to evaluate, and if appropriate appeal, any assessment for realty, and any other business taxes relating to the Property, and subject to the Approved Budget or approval of the Owner, ensure applications for rebates for vacant space (if any) are made;
- 3.9 To refund to Tenants from escrow accounts, funds of the Property or funds provided by Owner, as appropriate, pro rated rents, rebates, allowances, advance deposit refunds, and such other amounts as are legally due Tenants;
- 3.10 To consider and advise the Owner from time to time as to rules and regulations to be made by the Owner with respect to the Property or any additional rules and regulations required to be made under the Leases

for the better or more efficient operation of the Property and to use commercially reasonable efforts to cause the same to be fulfilled by the Tenants;

- 3.11 To collect from Tenants all insurance policies, Tenant insurance certificates, or other evidence of insurance required to be carried by Tenants;
- 3.12 Unless otherwise instructed by Owner, to secure for and on behalf of and at the expense of Owner such insurance, including without limitation, employee dishonesty insurance, fire and extended coverage property insurance, public liability insurance and workers' compensation insurance, as may be deemed by Owner (or any mortgagees) to be necessary or appropriate, in amounts satisfactory to Owner and Agent and naming Owner and Agent as co-insureds and in form and substance satisfactory to Owner, Agent and any mortgagees; provided, however, that if Agent promptly notifies Owner of the insurance so secured on behalf of Owner, and promptly complies with Owner's instructions regarding such insurance, Owner releases and holds Agent harmless of and from any claims, loss, damages and liability of any nature whatsoever based upon or in any way relating to Agent's securing or failure to secure any insurance, or any decision made by Agent with respect to the amount or extent of coverage thereof or the company or companies issuing, brokering or negotiating such insurance, provided that the foregoing shall not affect Agent's obligations under Section 19 hereof;
- 3.13 To respond to complaints and inquiries by Tenants, prospective tenants and others, and to take such corrective actions as Agent deems appropriate;
- 3.14 Use commercially reasonable efforts, at Owner's expense, to ensure material compliance by the Owner with all restrictive covenants, easements, cost-sharing agreements with "shadow anchors" and other title agreements to the extent the Agent shall have knowledge of the same and to use commercially reasonable efforts to monitor compliance by the other party to such agreements of its material obligations under all such agreements;
- 3.15 To contract on behalf of and at the expense of Owner for such supplies and services in reasonable quantities and at reasonable prices as may be appropriate with respect to the Property, and to supervise and administer such contracts, including, without limitation, contracts for utilities, mechanical maintenance (including preventative maintenance), window and facade maintenance and cleaning, metal maintenance, pest control, trash removal, janitorial and maintenance supplies, security, public relations, collection and credit reporting, legal and accounting services, computer services, architectural and engineering services, laundry services, and janitorial or cleaning services. In so contracting, Agent may contract with entities or persons affiliated with it, provided, however, that the rates and charges of the affiliated entity or person are generally competitive and consistent with rates and charges by non-affiliated entities and will obtain a minimum of two (2) competitive bids from nonaffiliated contractors who the Agent is satisfied are able to perform such contracts and duties respecting any contract exceeding Twenty Thousand Dollars (\$20,000.00) annually. Notwithstanding anything to the contrary contained herein, Agent shall not enter into, amend or modify any contract of the type described in this Section 3.15 unless such contract (A) is either (x) contained within the then current Approved Budget or is otherwise approved by Owner or (y) terminable without termination fee, premium or penalty by Owner upon not more than thirty (30) days notice and (B) does not provide or allow for annual consideration payable thereunder in excess of \$100,000;
- 3.16 At the expense of Owner in accordance with the Approved Budget, to provide through Agent's (or its affiliates') employees or third party contractors, all work, labor and services necessary or appropriate to operate, maintain and repair the Property, which employees may include, but are not necessarily limited to, a building executive director or supervisor, building manager, leasing specialist or leasing agent, secretarial and clerical staff, maintenance personnel, porters, laborers, security staff and watchmen. All matters pertaining to the employment, contracting, supervision, compensation, promotion and discharge of such employees or contractors shall be the responsibility of Agent. None of such persons shall be employees of the Owner. All amounts payable to such parties, including without limitation, all salaries, benefits, wages, recruitment, termination, severance and all other employment related costs and expenses for those personnel retained in accordance with this subsection shall be reimbursed to Agent to the extent provided in the applicable line items of the Approved Budget or as otherwise approved by the Owner. The Agent shall be entitled to allocate personnel costs on a fair and equitable basis among the Property and the other properties of which the Agent is the property manager as reflected in an Approved Budget or as otherwise approved by the Owner;
- 3.17 To supervise and coordinate the moving in and moving out of Tenants to accomplish efficient and time saving use of personnel and elevators and so that there will be a minimum of disturbance and inconvenience to the normal operation of the Property, and maintain appropriate public relations with Tenants and prospective tenants;
- 3.18 To prepare and file and/or cause to be prepared and filed necessary forms for insurance, hospitalization, benefits, social security taxes, union dues and contributions and such other forms, documents and returns as may be required by any governmental authority, a collective bargaining agreement, or otherwise with respect to employees and contractors, if applicable, of Agent at the Property and comply with all applicable laws and regulations relating thereto in all material respects. The Agent will keep Owner fully advised in a timely manner of any circumstances known to Agent that could lead to a labour disruption;
- 3.19 To prepare and file or cause to be prepared and filed on behalf of Owner such applications for permits, and/or licenses as may be required for the operation of the Property;
- 3.20 To prepare and, where appropriate, transmit payroll records, accounting reports, vacancy and occupancy reports, delinquency reports, cash flow reports, and disbursement ledgers. Agent may contract with others, including but not limited to entities or persons affiliated with it, or provide its own personnel for the performance of accounting, bookkeeping and computer services in connection with such preparation and transmittal, all without any additional charge to Owner;
- 3.21 Unless otherwise directed by Owner, to institute and prosecute on behalf of Owner such legal actions or proceedings as the Agent deems appropriate; to collect sums due Owner; with Owner's approval, to evict a Tenant, former Tenant or occupant of the Property; to regain possession of the Property or any part thereof; to contest any bill or charge asserted against or with respect to the Property; to defend any administrative or legal action brought against Agent and/or Owner with respect to the Property; with Owner's approval, to commence litigation pertaining to any labor or employment related dispute; to administratively process or litigate

any tax related issue or other issues relating to the Property; to appeal all such proceedings and lawsuits; and to settle or compromise any claims, lawsuits, judgments and proceedings relating to the Property. Notwithstanding the foregoing, Agent shall obtain the consent of Owner prior to initiating, settling or adjusting any action, suit, arbitration, or litigation as and to the extent the same shall constitute a Partnership Decision (as defined in the Partnership Agreement);

- 3.22 To maintain such bank or similar accounts on behalf of Owner as are necessary or appropriate in the operation of the Property, including such reserve, investment, security, escrow and other accounts and to deposit all amounts collected pursuant to Leases in such accounts (or as otherwise required by any applicable financing documents). Such funds are not to be commingled with the Agent's other funds;
- 3.23 To open and maintain accounts on behalf of Owner with such suppliers and vendors as are necessary or appropriate for the efficient operation of the Property;
- 3.24 To join and participate on Owner's behalf in such professional, trade or industry organizations and associations relating to shopping centers as is necessary or appropriate with respect to the operation of the Property;
- 3.25 Use commercially reasonable efforts, at Owner's expense, to ensure that the Owner is in material compliance with all then current federal, state and municipal laws and regulations (including, without limitation, all laws and regulations relating to the environment or any health and safety matters), and/or any permits or authorizations granted thereunder and to notify Owner of any material violations of any laws, orders, rules, or determinations of any governmental authority or agency affecting the Property promptly after such violation or determination is known to Agent and, subject to the other terms and provisions of this Agreement, to propose to Owner and implement at Owner's expense remedies of any such violations. Without limiting the generality of the foregoing, the Agent shall prepare all environmental protocols, policies and procedures as required by law and use reasonable commercial efforts to implement and enforce such protocols, policies and procedures;
- 3.26 To notify Owner of any catastrophe or major loss or damage or other material adverse change with respect to the Property, and to similarly notify all appropriate insurance authorities of the same, promptly upon Agent's knowledge thereof;
- 3.27 If from time to time, the Owner may elect to make physical changes to the Property, the Agent shall conduct to the best of its ability any negotiations with any Tenants which may be required in respect of business interruption, physical changes to premises or relocations (temporary or permanent) and shall consult with the Owner on a regular basis during such required negotiations as to what actions are most desirable. The Agent shall also arrange for the supervision of all work required to be performed by the Owner in order to prepare space in the Property for marketing or occupancy by Tenants or any alterations of the common areas or the systems or structures of the Property, all in accordance with plans and specifications approved by the Owner and the Approved Budget;
- 3.28 The Agent shall be paid a construction supervision fee in the amount of (a) five percent (5%) of the total Hard Construction Costs incurred for all construction work performed by or on behalf of Owner at, in or about the Property from and after the date hereof and (b) one percent (1%) of the total Hard Construction Costs incurred for all construction work performed by or on behalf of a Tenant at, in or about the Property from and after the date hereof on a reverse build-to-suit basis (i.e., whereby a Tenant shall construct improvements for the benefit of Owner which Owner shall retain fee title following the term of the applicable Lease) (each a "Construction Fee"). For the purposes hereof, "Hard Construction Costs" means all actual construction costs incurred, including, without limitation, costs or materials, supplies, equipment, labour and services, but excluding all costs commonly referred to as "soft costs" including, without limitation, architectural, engineering and consultants fees and disbursements, insurance, permits and financing costs and expenses, interest, salaries and the Construction Fee. Construction Fees shall be due and payable monthly on the basis of receipts for Hard Construction Fee shall be payable by Owner in connection with Tenant performed construction; provided, however, Agent shall be entitled to compensation on account of supervising Tenant performed construction if and to the extent such compensation shall be provided for in the applicable Lease;
- 3.29 The Agent shall be responsible for reviewing, or retaining and supervising professionals, to review, any plans or specifications for any Tenant changes or leasehold improvements in premises within the Property required by the provisions of any Lease and for planning and arranging for and monitoring the execution of any such work, which plans and specifications shall be subject to the approval of the Owner. The Agent shall use its reasonable commercial efforts to carry out any obligations of the Owner required to effect alterations or improvements to premises within the Property in connection with the leasing of such premises on a timely basis and in accordance with an Approved Budget so that the Tenant's move into such premises is accomplished in accordance with the applicable Lease. The Agent shall also arrange payment of all inducements and improvements or other allowances as set out in an Approved Budget or as otherwise approved by the Owner to be paid in connection with Leases upon fulfillment of all conditions with respect to such payment and the Owner agrees to fund such payments in a timely manner;
- 3.30 Upon request of Owner, to provide or arrange for such engineering, architectural, design or consulting services with respect to construction, rehabilitation or decorating work or proposed construction, rehabilitation or decorating work at the Property, all such services to be paid for by Owner;
- 3.31 To handle on behalf of Owner the submission to appropriate insurance officials of insurance claims and the settlement thereof, provided however, that with respect to any proceeds or reimbursements with respect to such claim which is in excess of Twenty Five Thousand Dollars (\$25,000), Agent shall be paid a processing fee, in addition to all other fees set forth herein, in an amount equivalent to three percent (3%) of the amount received by Owner with respect to that claim;
- 3.32 To prepare such customary reports, data, presentations, market surveys or other material as Owner reasonably requests in connection with the sale, refinancing, disposition or leasing of the Property;
 - 3.33 To institute at Owner's expense, advertising, marketing and public relations campaigns

pertaining to the Property;

- 3.34 To recommend to Owner, where Agent deems it appropriate, programs for the rehabilitation, remodeling, repairs and marketing of the Property;
- 3.35 To prepare and file on behalf of the Owner all Sales and Use Taxes returns relating to the Property and remit with such returns any net tax owing by the Owner on such returns. The Agent shall deposit any net tax refunds received in the accounts maintained for the Owner;
- 3.36 To use commercially reasonable efforts, at Owner's expense, to cause compliance with all material terms and conditions contained in any mortgage, deed of trust or other security instruments affecting the Property or any document governing the Loan described in Section 22 to the extent the Agent shall have knowledge of the same; and
- 3.37 To perform such other customary services on behalf of Owner with respect to the Property as shall be reasonably requested from time to time by Owner. If Owner and Agent disagree as to which services are customarily performed by property managers as aforesaid, Agent shall not be required to perform such service until resolution of such dispute, and such non-performance shall not be the basis of termination by Owner of this Agreement.
- 4. <u>Structural Changes</u>: Owner expressly withholds from Agent any power or authority to make any structural changes in any building or to make any other major alterations or additions in or to any such building or equipment therein, or to incur any expense chargeable to Owner other than expenses related to exercising the express powers above vested in Agent without the prior written approval of Owner (or any party that Owner shall direct), except such emergency repairs as may be required because of danger to life or property or which are immediately necessary for the preservation and safety of the Property or the safety of the occupants thereof or are required to avoid the suspension of any necessary service to the Property.
- 5 . Payments to Owner: Agent agrees to remit promptly to the account designated by Owner, all receipts received in the prior calendar month with respect to the Property in excess of budgeted operating expenses and reserves. In the event that the amount of costs and expenses incurred by the Agent in the management of the Property exceed the amount held in trust by the Agent for the Owner, the Agent shall thereupon furnish the Owner with an accounting of same.
- 6. Owner to Provide Funds: Owner shall, at all times, provide necessary funds to maintain and operate the Property as efficiently as possible and in a first class manner in keeping with the standards of operations for similarly situated shopping centers in the area. Owner shall advance such funds to Agent no later than thirty (30) days after its receipt from Agent of notice of the necessity for such advance. Owner agrees to provide any anticipated cash deficits thirty (30) days prior to its occurrence.
- Access to Books and Records: The Agent will, at reasonable intervals, and during normal business hours, on reasonable prior written notice, permit the Owner or RioCan REIT or the authorized agents of any of the foregoing to examine all books of account, records, reports and other papers of the Agent, where practicable in electronic format, relating to the services performed by the Agent under this Agreement for the Property and to make copies thereof and to make extracts therefrom. The Agent agrees with the Owner that, if in connection with any financing of the Property, any documentation is required in addition to the documentation provided for in this Article, the Agent shall fully co-operate with the Owner, at Owner's expense, and deliver same to such person or entity as the Owner may designate. No such records shall be destroyed by the Agent within six (6) years of the making thereof without the prior approval of the Owner and RioCan REIT. The Agent shall co-operate with Owner's auditor in the preparation of financial statements and their presentation to the Owner. The Agent shall co-operate with RioCan REIT's internal auditor in relation to internal controls over financial reporting at RioCan REIT's expense.
- 8. <u>Auditors and Appraisers</u>: The Agent shall make available to the Owner, the auditor, and to any appraiser such information and material as and when the same may be reasonably requested by the Owner or RioCan REIT and otherwise give such co-operation as may be reasonably necessary for such persons to carry on their duties. The Agent shall assist in the conduct of any spot or internal audit that RioCan REIT requests to be performed from time to time, at RioCan REIT's expense.
- 9. <u>Budgets</u>: The Agent shall prepare and submit to the Owner for its approval not later than ninety (90) days prior to the commencement of each year in the format required by the Owner with respect to the Property:
 - (a) an operating and cash flow budget for the next ensuing three (3) years to be calculated on a monthly basis for the next ensuing year and on an annual basis for the following two (2) years which budget shall set forth both anticipated gross revenues and operating expenses on an accrual basis in accordance with GAAP and which shall include the assumptions used in its preparation;
 - (b) an annual capital expenditure budget for the next ensuing three (3) years broken down on a monthly basis for the next ensuing year and on an annual basis for the following two (2) years;
 - (c) a leasing plan (which, once approved, shall become the "Approved Leasing Plan" for such year) for the next ensuing three (3) years broken down on a monthly basis for the next ensuing year and on an annual basis for the following two (2) years which shall set forth projected monthly or annual lease revenue for the next ensuing three (3) years for the Property and each Tenant including lease start date, lease termination date, renewal increase assumptions, renewal commencement assumptions, lease amounts, known tenant build-out cost assumptions and any other commentary of significance for each Tenant;
 - (d) commentary summarizing the principal conclusions and assumptions of the Approved Leasing Plan; and

(e) such other financial information as reasonably requested by the Owner.

Each of the foregoing budgets and plans shall be in such detail as the Owner shall reasonably require. Agent shall make such reasonable modifications to each proposed pro forma budget it prepares in accordance with this Section 9 until Owner shall have approved the budget in writing (such budget when approved, the "Approved Budget"). During any period when the Advisory Board (as defined in the Partnership Agreement) shall fail to approve any proposed budget prior to the commencement of the fiscal year to which such budget relates, the Property shall be operated pursuant to the proposed budget for such fiscal year with respect to those portions approved by the Advisory Board and with respect to those portions not approved by the Advisory Board, in accordance with the prior year's Approved Budget (except for non-recurring expenditures which shall be deemed removed from such prior year's Approved Budget) with each non-approved line item increased by three and one-half percent (3.5%) from such prior year's Approved Budget; provided, however, that expenditures by the Agent for Necessary Expenses (as defined in the Partnership Agreement) shall not be limited by amounts set forth in the prior year's Approved Budget.

The Agent shall from time to time as circumstances may warrant (but not less often than quarterly) prepare and submit to the Owner for its approval any proposed revisions to any previously Approved Budget, it being understood that where a budget revision is so submitted the previously Approved Budget will not be amended until the revision with respect to which approval of the Owner is sought, is approved by the Owner.

10. Reports:

- 10.1 The Agent covenants and agrees to prepare and submit to the Owner electronically:
- (a) monthly, within fifteen (15) days for preliminary financial results, and twenty (20) days after the end of each month, the following:
 - (i) in the format required by the Owner, a balance sheet, a trial balance, a summary operating statement, a detailed operating statement and a cash flow statement each showing the month's Approved Budget items in comparison with the actual amounts thereof as well as the year to date totals thereof and an annual budget to the end of the year which shall explain any material variances, provide management fees as a separate line item, and include a schedule of fees paid to the Agent for the month;
 - (ii) a statement of arrears of rents on an aged basis, including commentary on any Tenant with material arrears;
 - (iii) a report on status of existing litigation, threatened litigation and contemplated litigation;
 - (iv) a report on Tenant sales in the Property, on a Tenant by Tenant basis, for the previous twenty-four (24) months, obtained in accordance with the provisions of Leases (or if otherwise obtained by the Agent) with respect to all Leases pursuant to which percentage rent is payable, recognizing that amounts of percentage rent are small at most of Agent's properties;
 - a report apprising as to any damage to the Property and the repair and replacement thereof, and as to any anticipated or required change or alteration to the improvements and to the equipment or systems, in each case comprising part of the Property;
 - (vi) a net cash flow report, including a twelve (12) month cash flow forecast, which forecast shall include estimated timing of disbursements to and from the Owner;
 - (vii) a bank reconciliation for the month;
 - (viii) a list of cheques issued for the month;
 - a current rent roll summarizing all Leases in effect in the Property including the relevant rental (including percentage rental where applicable), area occupied, expiry date, any other material conditions and renewal options;
 - (x) a leasing activity report including a comparison to budget;
 - (xi) a capital expenditures report including a comparison to budget;
 - a mortgage payable amortization schedule, together with a summary of the details of any financial covenants and restrictions and the compliance therewith;
 - (xiii) details of all fees payable to the Agent hereunder; and

(xiv) a report on any other matters relating to the Property in the month which the Agent reasonably believes are significant and should be brought to the Owner's attention or which the Owner believes to be significant and has drawn to the attention of the Agent and/or any material workplace health and safety issues relating to the Property and/or any material environmental issues relating to the Property and/or material violations of laws.

all as may be revised from time to time in accordance with the requirements of the Owner acting reasonably together with such other reports and information as may be requested by the Owner from time to time acting reasonably;

- (b) from and after the happening of any damage to or destruction of the Property, on the first Business Day (as hereinafter defined) of each month commencing on the first Business Day of the month following the happening of such damage or destruction and continuing until the first Business Day of the month next following the month in which any work authorized by the Owner shall have been completed, written reports specifying in such detail as the Owner shall require, acting reasonably, a monthly status report as to the repair, replacement or correction of such damage or destruction duly authorized by the Owner;
- (c) in addition to the monthly reports outlined above, after the end of each of the first, second, third and fourth quarters in each year prepare and provide to the Owner, within thirty (30) days of the end of such quarter reports in the format required by the Owner, acting reasonably, in writing including:
 - (i) a report commenting on the financial results for each period;
 - updated tax basis reporting (to allow the calculation of current and deferred taxes on temporary differences);
 - (iii) leasing reports for the Property in the form presently discussed in regular management meetings of Agent including, but not limited to, (A) a leasing report depicting information for new Leases being negotiated, (B) a lease reporting form depicting information for new Leases completed, (C) a leasing expirations and renewals report depicting information for Leases due to expire within twelve months, and (D) a competition report depicting information on competing properties in the area;
 - (iv) a list of non recurring committed expenditures in excess of Twenty Five Thousand Dollars (\$25,000) each and the anticipated date of payment thereof;
 - (v) a fixed asset continuity schedule; and
 - (vi) a report on any other matters pertaining to the Property in the quarter which the Agent reasonably believes are significant and should be brought to the Owners' attention or which the Owner believes to be significant and have drawn to the attention of the Agent and on any material litigation affecting the Property and on any material workplace health and safety issues relating to the Property and on any material, physical or environmental issues relating to the Property, all as may be revised from time to time in accordance with the requirements of the Owner acting reasonably, together with such other reports and information as may be requested by the Owner or RioCan REIT from time to time, acting reasonably.
- 10.2 The Agent covenants and agrees as follows:
- (a) not later than April 30 of each year, to prepare and provide to the Owner and each of the Tenants a reconciliation on account of recoveries from Tenants for operating expenses for the prior year in accordance with the Leases;
- (b) to prepare all budgets and reports in accordance with GAAP; and
- (c) at RioCan REIT's request, to review and approve the purchase price allocation model used in the financial statements prepared pursuant to this Agreement.
- 10.3 The Agent recognizes that RioCan REIT, will be subject to International Financial Reporting Standards ("IFRS") commencing January 1, 2011, with retroactive impact to January 1, 2010. Consequently, periodic third party appraisals may be necessitated and the Agent will oversee such process, provided that RioCan REIT shall pay for the costs of any appraisals not required by credit facilities of Cedar Shopping Centers Inc. (or any subsidiary thereof, or financings on the Property) and any audit expenses in connection therewith, if applicable. The Agent will also use reasonable efforts to provide to RioCan REIT information it requires to convert to GAAP financial information to IFRS.
- 10.4 The Agent will allow RioCan REIT to perform, at its expense, from time to time, internal control auditing of all expenses of the Property (including information technology and other infrastructures as RioCan REIT deems necessary).
- 10.5 The Agent will assist RioCan REIT in preparing any necessary tax certificates for foreign owners.

- 10.6 The Agent agrees that (i) it will deliver to RioCan REIT copies of all reports, statements, plans, budgets and other deliveries to be delivered to the Owner pursuant to Section 9 and this Section 10 at the same time as the same are delivered to the Owner; and (ii) RioCan REIT shall have the same entitlements as are granted to the Owner under Section 9 and this Section 10 to reasonably request any additional information or reporting, or to require reasonable changes in format.
- 10.7 Notwithstanding anything to the contrary contained in this Agreement, if RioCan REIT shall request additional information or materials that are not readily available to Agent, or the preparation of additional reports not customarily prepared by Agent, the reasonable cost and expense of providing such information, materials and reports shall be paid by RioCan REIT promptly following demand. The provisions of this Section 10.7 shall only apply to supplemental or additional information, materials and reports requested by RioCan REIT, and not to the specific information, materials and reports expressly required to be provided under this Agreement.
- 11. <u>Data Transfer</u>: The Agent will provide all the information called for in Sections 9 and 10, and any other information as deemed necessary through the Agent's intranet site or other electronic transfer.
- 12. Fees: Except as otherwise provided for herein, Owner shall pay to Agent a property management fee in an amount equal to three and one-half percent (3.5%) of the gross receipts of the Property (the "Management Fee"). This fee shall be payable in monthly installments from the operating accounts maintained pursuant to Section 3.22 hereof. "Gross receipts" of the Property shall include all rents, percentage rents, tenant charges, reimbursements from Tenants for common area maintenance charges, insurance, utilities and real estate taxes, forfeited deposits and such other amounts as are collected from Tenants and shall exclude sales taxes collected from Tenants, deposits from Tenants (other than forfeited deposits), proceeds from any sale or refinancing of the Property or any portion thereof and proceeds of any lease termination payment, settlement, insurance award (except as provided in Section 3.31) or condemnation award. The Management Fee does not include payment for leasing services, which shall be payable to Agent pursuant to Section 12.2 below.
- 12.1 To the extent that operating revenues of the Property are insufficient to pay the Management Fee in full when due, and to the extent that Agent agrees in writing in advance to defer receipt by it of any part of the Management Fee due it, the amount so deferred shall bear interest at the rate of two (2) percentage points in excess of the "prime rate" or "base rate" from time to time announced by Citibank, N.A., New York New York compounded monthly. Nothing herein contained, however, shall be construed to obligate Agent to defer receipt by it of any Management Fee or other fees whatsoever.
- 12.2 [INSERT FOR BLUE MOUNTAIN AND FRANKLIN VILLAGE ONLY: Subject to the terms and provisions of Section 6(c) of that certain Purchase and Sale Agreement dated as of October 26, 2009, by and between Agent and RioCan Holdings USA Inc.,] Agent or its affiliate shall be the leasing agent for the Property. Owner shall pay Agent or its affiliate a leasing commission for each Lease signed by a Tenant and Owner at any time after October 26, 2009 (a "Leasing Commission") in an amount equal to:
 - (a) five percent (5%) of the Minimum Rent payable during the lesser of (i) the entire primary term of such Lease and (ii) the first five (5) years of the primary term of such Lease, if the premises demised thereunder is less than 5,000 square feet of gross leasable area; and
 - (b) \$3.00 per square foot of gross leasable area, if the premises demised thereunder is 5,000 square feet or more of gross leasable area.

In the event of a lease renewal or extension, the Leasing Commission payable to Agent shall be an amount equal to:

- (a) two and one-half percent (2.5%) of the Minimum Rent payable during the lesser of (i) the entire renewal term of such Lease and (ii) the first five (5) years of the renewal term of such Lease, if the premises demised thereunder is less than 5,000 square feet of gross leasable area; and
- (b) \$1.50 per square foot of gross leasable area, if the premises demised thereunder is 5,000 square feet or more of gross leasable area.

For the purposes hereof, "Minimum Rent" shall mean all fixed rent stated in the Lease in question (regardless of how such rent is denominated), but excluding the following items of additional rent (regardless of how such items are denominated):

- (a) amounts which are above or in addition to the fixed rent, whether payable by Tenant as
 adjustments or otherwise, for realty taxes, cleaning costs, all other operating expense
 escalations or pass-throughs, and percentage rentals (if any);
- (b) amounts paid by Tenant, or melded into Tenant's rental under the Lease, for work performed for Tenant in excess of Landlord's Work as specified in the Lease shall be excluded;
- (c) amounts added to or melded into Tenant's rental under the Lease to reimburse landlord for Tenant's space in another property which landlord agrees to "take over" and credits allowed to Tenant against Lease rental for payments made by Tenant to its landlord(s) to satisfy, cancel or discharge leasehold obligations of Tenant. These payments shall be deducted as allowed or made against Tenant's rental under the Lease;
- (d) amounts agreed to be paid by landlord to landlords of Tenant to satisfy, cancel or discharge Tenant's obligations under its existing leases or agreements and losses incurred in assigning such leases or subletting such space. Such payments shall be deducted when paid and losses shall be deducted when incurred;

- (e) amounts paid for additional cleaning, security and/or other services not commonly supplied to other Tenants of the Property shall be excluded;
- (f) amounts paid by Tenant in connection with Tenant's option to cancel its Lease, if any, shall be excluded; and
- (g) rent concessions, work letter allowances and rent allowances granted to Tenant shall be deducted

In the case of gross rents or semi-gross rents, an appropriate allocation will be made to reflect the portions of such rents which would be equivalent to the minimum rent payable on a net lease.

All Leasing Commissions shall be reduced by 25% if an outside broker is used. Fifty percent (50%) of the amount of any Leasing Commission due hereunder shall be payable to Agent or its affiliate upon the payment by the Tenant to Owner of the first month's rent due under the applicable lease or lease renewal (other than any security deposit) and the lease or renewal being fully executed, and the remaining fifty percent (50%) thereof shall be payable to Agent or its affiliate if and when the Tenant has opened for business (it being acknowledged and agreed that the full Leasing Commission shall be due and payable if the foregoing events shall occur simultaneously). In addition, Owner shall reimburse Agent for the reasonable actual out-of pocket costs of all advertising plans and promotional materials and all reasonable attorneys' fees incurred by Agent in connection with the leasing of any space at the Property.

- 12.3 Upon the sale or transfer, directly or indirectly, of the Property by Owner by deed, or by transfer of all of the ownership interests in Owner or otherwise, Owner shall pay to Agent a disposition fee (a "Disposition Fee") equal to one-half of one percent (0.5%) of the gross sales price paid by the purchaser of the Property; provided, however, that any Disposition Fee payable hereunder shall not exceed \$150,000. The Disposition Fee shall be deemed earned, and, therefore, shall be paid, as and when title (by deed or transfer of ownership interests) to the Property closes and without regard to whether one or more outside brokers were engaged in connection with such sale or transfer. Notwithstanding the foregoing, for so long as Agent or any of its affiliates shall be partners of any ultimate owner of the Owner (Agent or such affiliated partner(s) in their respective capacity as partners of Owner, being "Cedar Affiliated Partner(s)"), Agent shall not be entitled to a Disposition Fee hereunder in the event of any transfer of interests in Owner by and among any of the then existing partners of Owner.
- 12.4 Upon any financing or refinancing by debt, sale and leaseback or other form of financing with respect to the Property arranged by the Agent (other than in connection with (i) any company loan from any partner of Owner to Owner and (ii) any trade payable incurred in the ordinary course of business), Owner shall pay to Agent a financing fee (the "Financing Fee") equal to one-quarter of one percent (0.25%) of the original principal amount of the Financing; provided, however, that fifty percent (50%) of any Financing Fee shall be paid to RioCan REIT if RioCan REIT was the sole procuring party with respect to such financing or refinancing; provided further, that any Financing Fee payable hereunder shall not exceed \$50,000. The Financing Fee shall be deemed earned, and, therefore, shall be paid, as and when the subject financing closes and without regard to whether one or more outside brokers were engaged in connection with such financing.
- 13. Expenses: Owner shall reimburse Agent for reasonable, actual out of pocket expenses including telephone and facsimile charges, postage and express mail service and travel and food expenses incurred by Agent in connection with Agent's on site supervision of the Property by Agent's officers and personnel (evidenced by receipts submitted to Owner). Agent shall not accept for its own account in the execution of its duties hereunder, any commissions, reductions, finder's fees or other concessions from tradesmen, suppliers, contractors, insurers, or tenants. If such concessions are received by the Agent, they shall be remitted to or credited to the Owner promptly after receipt.
- 14. <u>Legal Services</u>: The Agent, on behalf of Owner, shall engage Stuart H. Widowski, Esq., or his successor, as legal counsel to provide legal services for Owner and the Property. Such services shall be provided as required and at a rate of \$275 per hour.

15. <u>Sales and Financing</u>:

- (a) If the Owner executes a listing agreement with a broker for the sale of the Property, the Agent shall co-operate with such broker so that the respective activities of the Agent and broker may be carried on without friction and without material interference with Tenants. The Agent will provide access to the Property during reasonable business hours on reasonable prior written notice.
- (b) If the Owner elects to sell, finance or refinance the Property or its interest therein then upon the written request of the Owner, the Agent agrees at Owner's expense (i) to assist in the setting up of due diligence rooms or websites, copying documents, obtaining consents and delivering notices; (ii) to assist the Owner in the evaluation of the truth, correctness and completeness of any and all facts (collectively, the "Representations") requested by a prospective buyer or lender, in connection with its operation, maintenance, repair, servicing, compliance with governmental requirements and other matters, but without representation or warranty by the Agent; and (iii) to conduct a good faith review of its files to determine which Representations can be made and to disclose to the Owner which Representations cannot be made, but without representation or warranty by the Agent.

- (c) In connection with any sale or financing of the Property, the Agent shall assist in the preparation of any required statement of adjustments and shall provide any ongoing services required in connection with the sale after closing, including, without limitation, collection of receivables, readjustment of the statement of adjustments and tenant billings, subject to the continuation of the term of this Agreement during any such period, including, without limitation, the payment of all fees provided hereunder to the Agent.
- $1\ 6$. <u>Compliance with Laws</u>: In performing its obligations hereunder, Agent shall comply with all applicable federal, state and local laws and regulations.
- Term and Termination: The initial term of this Agreement shall be for a period of three (3) years from the date hereof and this Agreement shall automatically renew from year to year thereafter unless and until terminated by either party upon ninety (90) days' prior written notice thereof. Notwithstanding the foregoing, Owner shall be entitled to terminate this Agreement (with no additional compensation) at any time upon seven (7) days' prior written notice to Agent (which notice may be given by RioCan REIT on behalf of Owner) if a Cause Event shall have occurred and be continuing. Notwithstanding the foregoing, Agent shall be entitled to terminate this Agreement at any time upon seven (7) days' prior written notice to Owner and RioCan REIT if (I) for so long as Agent and the general partner of Owner shall be under common control, the general partner of Owner shall be removed by reason of a "Cause Event" pursuant to the terms of the Partnership Agreement, (II) a Bankruptcy Event shall occur with respect to Owner or (III) Owner is in material default in the performance or observance of any of its covenants or obligations under this Agreement (other than, for so long as Agent and the general partner of Owner shall be under common control, a default by Owner resulting from the wilful default of the general partner of Owner under the Partnership Agreement), which default continues uncured for a period of sixty (60) days after written notice to Owner and RioCan REIT, provided, that if such default is not reasonably susceptible of being cured with such sixty (60) day period and Owner shall have commenced a cure of such default within such sixty (60) day period and is diligently pursuing a cure of such default, Owner shall have such additional time as is reasonably necessary to cure such default. This Agreement shall terminate automatically if:
 - all or substantially all of the Property is condemned or acquired by eminent domain;
 or
 - (ii) all or substantially all of the Property is destroyed by fire or other casualty as a result of which all or substantially all of the Tenants are unable to continue the normal conduct of their business in their respective occupied spaces and are permanently released under their respective Leases from the payment of all rent thereunder; or
 - (iii) all of the Property is sold or ground leased to an unrelated, third party purchaser; or
 - (iv) upon the transfer by all of the Cedar Affiliated Partners of all of their respective indirect interests in the Owner to one or more unaffiliated parties.

For the purposes hereof, (A) "Cause Event" means, with respect to Agent, the occurrence of any of the following events: (a) Agent commits fraud, willful misconduct or gross negligence in the performance of its duties and obligations under this Agreement; (b) Agent is in material default in the performance or observance of any of its covenants or obligations under this Agreement, which default continues uncured for a period of sixty (60) days after written notice to Agent, provided, that if such default is not reasonably susceptible of being cured with such sixty (60) day period and Agent shall have commenced a cure of such default within such sixty (60) day period and is diligently pursuing a cure of such default, Agent shall have such additional time as is reasonably necessary to cure such default; (c) any Bankruptcy Event with respect to Agent; (d) Agent and the general partner of Owner shall not be under common control, or (e) for so long as Agent and the general partner of Owner shall be under common control, the general partner of Owner shall be removed by reason of a "Cause Event" pursuant to the terms of the Partnership Agreement and (B) "Bankruptcy Event" means, with respect to any entity, the occurrence of any of the following events: (i) the filing by it of a voluntary petition in bankruptcy, (ii) an adjudication that it is bankrupt or insolvent unless such adjudication is stayed or dismissed within sixty (60) days, or the entry against it of an order for relief of debtors in any bankruptcy or insolvency proceeding unless such order is stayed or dismissed within ninety (90) days, (iii) the filing by it of a petition or an answer seeking for itself any reorganization, arrangement, composition, readjustment, liquidation, dissolution or similar relief under any statute, law or regulation, (iv) the filing by it of an answer or other pleading admitting or failing to contest the material allegations of the petition filed against it in any proceeding of the nature described in the preceding clause (iii), (v) its seeking, consenting to or acquiescing in the appointment of a trustee, receiver or liquidator of it or of all or any substantial part of its assets, or (vi) the failure within ninety (90) days after the commencement of any proceeding against it seeking reorganization, arrangement, composition, readjustment, liquidation, dissolution or similar relief under any statute, law or regulation, to have the proceeding stayed or dismissed, or the failure within one hundred twenty (120) days after the appointment without its consent or acquiescence of a trustee, receiver or liquidator of it or of all or any substantial part of its assets to have such the appointment vacated or stayed, or the failure within ninety (90) days after the expiration of any such stay to have the proceeding dismissed or the appointment vacated, or (vii) such party generally shall be unable to pay, or shall admit in writing its inability to pay, its debts as they become due.

17.1 Owner shall pay or reimburse Agent for any monies due it under this Agreement for services prior to termination, notwithstanding termination of this Agreement. All provisions of this Agreement that require Owner to have insured or to defend, reimburse or indemnify Agent shall survive any termination and, if Agent is or becomes involved in any proceeding or litigation by reason of having been Owner's Agent, such provisions shall apply as if this Agreement were still in effect. Owner agrees that Agent may withhold funds for thirty (30) days after the end of the month in which this Agreement is terminated to pay bills previously incurred but not yet invoiced, and to close accounts.

- (a) the Agent shall within fifteen (15) days thereafter render a final accounting to the Owner and pay over any balance in the Agent's trust account remaining at the credit of the Owner (subject to Section 17.1 hereof);
- (b) the Agent shall promptly surrender to the Owner all property, documents and information of Owner acquired in the course of or incidental to this Agreement including, but not limited to all lease agreements, computer programs and records and other files, records, contracts and information to the continuing operation of the Property;
- the Agent shall surrender any space occupied by Agent in the Property in connection with its responsibilities under this Agreement;
- (d) the Agent shall assign to the Owner all contracts related to the Property and entered into in accordance with this Agreement, if applicable and assignable, and that were entered into in accordance with this Agreement and the Owner (or a replacement property manager) shall assume all such contracts;
- the Agent shall give notices to Tenants and other persons as Owner may direct with respect to such termination; and
- (f) the Agent shall cooperate fully with Owner and Owner's agents and representatives to effectuate an orderly transition in connection with the management and/or operation of the Property.
- 18. <u>Indemnity</u>:
- 18.1 The Agent shall be indemnified, defended and held harmless by the Owner from and against any and all expenses (including reasonable attorneys' fees), losses, damages, liabilities, charges and claims of any kind or nature whatsoever including the cost of seeking to enforce this indemnification right (collectively "Indemnified Losses"), incurred by Agent arising out of or incidental to any act performed or omitted to be performed by Agent in its capacity as property manager and/or in connection with the Property, including, without limitation, any act or omission constituting ordinary negligence of Agent, provided that such act or omission did not constitute gross negligence, willful misconduct (including, without limitation, an intentional breach of the terms of this Agreement) or fraud.
- 18.2 All indemnification obligations under this Agreement shall also run to the benefit of any affiliate of any Agent or any principal, partner, member, manager, shareholder, controlling person, officer, director, agent or employee of Agent (each of the foregoing, together with Agent, a "**Protected Person**").
- 18.3 The Owner shall promptly reimburse (or advance, to the extent reasonably requested by a Protected Person other than in connection with Indemnified Losses resulting from claims made by Owner) each Protected Person for reasonable legal or other expenses (as incurred) of each Protected Person in connection with investigating, preparing to defend or defending any claim, lawsuit or other proceeding relating to any Indemnified Losses for which the Protected Person may be indemnified pursuant to this Section 18 provided, that such Protected Person executes a written undertaking to repay Owner for such reimbursed or advanced expenses if it is finally judicially determined that such Protected Person is not entitled to the indemnification provided by this Section 18.
- 18.4 The provisions of this Section 18 shall continue to afford protection to each Protected Person regardless of whether such Protected Person remains in the position or capacity pursuant to which such Protected Person became entitled to indemnification under this Section 18 and regardless of any subsequent amendment to or termination of this Agreement.
- 18.5 The Owner shall be indemnified, defended and held harmless by Agent from and against any and all Indemnified Losses incurred by Owner arising out of the gross negligence, willful misconduct (including, without limitation, an intentional breach of the terms of this Agreement) or fraud of Agent.
- 18.6 The provisions of this Section 18 shall survive the expiration or any termination of this Agreement.
 - 19. <u>Insurance</u>:
- 19.1 The Agent shall place and maintain in its capacity as manager and throughout the term of this Agreementunder its corporate insurance policy:
 - comprehensive dishonesty, disappearance and destruction insurance for an amount of not less than \$2,000,000, and
 - (ii) errors and omissions insurance coverage of \$1,000,000.

Any liability insurance policy or policies obtained by or on behalf of Owner shall name the Agent as an additional named insured so as to protect and indemnify the Agent from liability in the same manner and to the same extent as the Owner. All policies of insurance shall contain a waiver of any rights of subrogation which the insurers may have against the Agent whether or not the damage was caused by the act, omission or negligence of the Agent.

19.2 All insurance policies placed pursuant to this Section 19 shall name the Owner as an additional named insured so as to protect and indemnify them in the same manner and to the same extent as the Agent. All

policies of insurance shall contain a waiver of any rights of subrogation of which the insurers may have against the Owner.

- 19.3 Owner and Agent shall each waive any claim for loss or damage against the other and mutually agree to hold each other harmless for loss to the Property to the extent that either party is reimbursed or indemnified by insurance coverage.
- 19.4 The provisions of this Section 19 shall survive the expiration or any termination of this Agreement.
- 20. <u>Violations of Law</u>: Agent will promptly notify Owner of any violations of any requirements of any statute, ordinance, law or regulation of any governmental body or any public authority or official thereof having jurisdiction with respect to the Property known to the Agent and shall promptly use commercially reasonable efforts at Owner's expense to cure such violations and to prevent any civil or criminal liability from being imposed.
- 20.1 In the event (A) it is alleged or charged that the Property or any equipment therein or any act or failure to act by the Owner or its agents with respect to the Property or the sale, rental, or other disposition thereof fails to comply with, or is in violation of, any of the requirements of any provision, statute, ordinance, law, or regulation of any Governmental body or any order or ruling of any public authority or official thereof having or claiming to have jurisdiction thereover, (B) Agent notifies Owner and RioCan REIT of such violation pursuant to Section 20 or any other provision of this Agreement and Owner fails to contest such violation in good faith and/or to commence and diligently prosecute to completion (or permit Agent, at Owner's expense to commence and diligently prosecute) the cure of such violation, and (C) Agent, in its sole and absolute discretion, considers that the action or position of Owner may result in damage or liability to Agent, Agent shall have the right to cancel this Agreement at any time by giving not less than thirty (30) days' prior written notice to Owner and RioCan REIT of its election so to do, which cancellation shall be effective upon the service of such notice. Such notice may be served personally or by United States or Canadian certified mail, and if served by mail shall be deemed to have been served when deposited in the United States or Canadian mail system. Such cancellation shall not release the indemnities of Owner and Agent set forth herein and shall not terminate (i) any liability or obligation of Owner or RioCan REIT to Agent for any payment, reimbursement, or other sum of money then due and payable to Agent hereunder as of the date of such cancellation, or (ii) any obligation of Agent to remit moneys to Owner or to complete its obligations hereunder to the date of such cancellation. Agent shall cooperate with Owner to ensure a smooth and efficient transition to a new managing agent, including but not limited to, prompt delivery of files relating to the Property.
- 21. <u>Confidentiality</u>: The Agent agrees that all confidential information acquired in the course of or incidental to this Agreement including, but not limited to, customer information, trade secrets or other commercial property of the Owner will not be disclosed or communicated in any way by the Agent during the term of this Agreement or thereafter, except to employees of the Agent and its agents, contractors, subcontractors and consultants (provided they agree in writing to be bound by this provision) as required to carry out their duties under this Agreement and as may be required by law. The parties acknowledge that this Agreement may be included in an SEC filing and a filing required by the Ontario Securities Commission, and shall not be confidential information.

2 2 . The Loan: Agent acknowledges that	Owner has obtained a loan from []
("Lender") in the principal amount of up to \$[] (the "Loan"), which is governed by a certain
[] made by Owner for the benefit of	Lender, dated the date hereof (the "Loan Agreement").
For so long as the Loan is outstanding:	

- (a) this Agreement shall be terminable by Lender or its nominee without penalty or premium following the occurrence of an Event of Default (as such term is defined in the Loan Agreement) or by Owner after Lender has notified Owner in writing that Agent is unsatisfactory to Lender, in each case upon thirty (30) days prior written notice to Agent;
- (b) all payments hereunder shall be subject and subordinate in lien and priority of payment to the payment of all principal and interest and all other amounts due under the Loan; and
- (c) Agent shall promptly notify Lender with respect to any default hereunder and promptly deliver to Lender a copy of each notice, report, plan or statement delivered by Agent to Owner hereunder.
- 23. <u>General Provisions</u>: It is expressly agreed by the parties that:
- 23.1 The parties have entered into this Agreement without any inducements, representations, statements, warranties or agreements made by either party other than those expressly stated herein.
- 23.2 This Agreement embodies the entire understanding of the parties with respect to the subject matters stated herein and there are no other understandings or undertakings related to the within subject matters. This Agreement may be modified only by a written agreement signed by the parties hereto. No waiver of any of the provisions of this Agreement shall be binding unless executed in writing by the parties, save and except for any termination of this Agreement that arises under the express terms hereof. No waiver of any of the provisions of this Agreement shall be deemed or shall constitute a waiver of any other provision (whether or not similar) nor shall such waiver constitute a continuing waiver. Failure on the part of a party to complain of any act or failure to act of another party or to declare another party in default, irrespective of how long such failure continues, shall not constitute a waiver by such first mentioned party of its rights hereunder.
- 23.3 The provisions of this Agreement are severable and to the extent that any provision herein is determined by court order, law or rule to be invalid, such invalidity shall in no way affect nor invalidate the other provisions of this Agreement.
 - 23.4 Agent and Owner acknowledge that as of the date of this Agreement Agent is an affiliate of the

general partner of Owner. Any action of Agent hereunder or approval required to be given by the Owner hereunder shall in each case be subject to the approval of the Advisory Board to the extent required pursuant to the express terms of Section 6.3(a) and (b) of the Partnership Agreement. If the approval of the Advisory Board shall not be required pursuant to the express terms of Sections 6.3(a) and (b) of the Partnership Agreement, for so long as Agent shall remain an affiliate of Owner, Agent shall have the authority to make determinations and decisions, and grant approvals, on behalf of Owner. Notwithstanding the terms of Section 23.8 of this Agreement, with respect to notices and information required to be delivered to Owner pursuant to this Agreement, such notices and information shall not be required to be delivered to RioCan REIT unless expressly provided in this Agreement or in the Partnership Agreement.

- 23.5 This Agreement shall be governed by and construed in accordance with the laws of the State of New York.
- 23.6 With respect to any and all disputes under or relating to this Agreement, the parties consent to the exclusive jurisdiction and venue of the Supreme Court of the State of New York, Nassau County and the United States District Court for the Eastern District of New York and the appellate courts with supervisory powers thereover
- 23.7 The parties agree that in any litigation or proceeding commenced by either party against the other, service of process shall be deemed to be effective either by hand delivery thereof or by the mailing thereof via certified mail, postage prepaid, with a proof of mailing receipt validated by the U.S. or Canadian Postal Service constituting the sufficient evidence of service of process.
- 23.8 With respect to any notices that are required or permitted to be made pursuant to this Agreement, they shall be in writing and either delivered personally, sent by United States or Canadian mail or by facsimile (provided that if delivered by facsimile, a confirmation copy of such notice must also be delivered personally or by United States or Canadian mail) addressed as follows:

As to Owner:

Cedar Shopping Centers Partnership, L.P. c/o Cedar Shopping Centers, Inc. 44 South Bayles Avenue, Suite 304 Port Washington, New York 11050 Attention: Leo S. Ullman Facsimile: (516) 767-6497

with a copy to:

RioCan c/o RioCan Real Estate Investment Trust 2300 Yonge Street Suite 500, P.O. Box 2386 Toronto, Ontario M4P 1E4 Attention: Jonathan Gitlin Facsimile: (416) 866-3020

As to RioCan REIT:

RioCan c/o RioCan Real Estate Investment Trust 2300 Yonge Street Suite 500, P.O. Box 2386 Toronto, Ontario M4P 1E4 Attention: Jonathan Gitlin

Facsimile: (416) 866-3020

As to Agent:

Cedar Shopping Centers Partnership, L.P. c/o Cedar Shopping Centers Partnership, L.P. 44 South Bayles Avenue, Suite 304 Port Washington, New York 11050 Attention: Brenda J. Walker

Facsimile: (516) 767-6497

Any party hereto may change its address for notice or facsimile transmission number by notice to the other parties hereto in the manner set forth herein. Any notice, consent or instrument aforesaid, if delivered or sent by facsimile transmission shall be deemed to have been given or made on the date on which it was delivered to such party or if sent by facsimile transaction shall be deemed to be given on the day transmitted if transmitted before 5:00 p.m. on a Business Day and otherwise on the next Business Day or if mailed, shall be deemed to have been given or made on the fifth Business Day following the date on which it was mailed, unless at the time of mailing or within seventy-two (72) hours thereafter, there shall be a strike, labor interruption or lockout in the postal service, in which case, the notice, direction or other instrument as aforesaid shall be delivered by one party to the address of the other.

23.9 This Agreement may not be assigned by Agent without the prior written consent of Owner, provided, however, that Owner consents to Agent's designating a wholly-owned subsidiary or affiliate of Agent to act on behalf of Agent as leasing and rental agent for the Property (so long as such entity remains a wholly-

owned subsidiary or affiliate of Agent). This Agreement shall be binding upon and benefit the parties hereto and their respective successors and permitted assigns.

- 23.10 For purposes of this Agreement, "Business Day" means any day other than Saturday, Sunday or any other day on which banks or savings and loan associations in New York, New York are not open for business.
- 2 3 . 1 1 EACH PARTY HERETO, AFTER CONSULTING OR HAVING HAD THE OPPORTUNITY TO CONSULT WITH COUNSEL, KNOWINGLY, VOLUNTARILY AND INTENTIONALLY WAIVES ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY ACTION BROUGHT WITH RESPECT TO THIS AGREEMENT OR ANY OF THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT OR ANY COURSE OF CONDUCT, DEALING, STATEMENTS (WHETHER ORAL OR WRITTEN) OR ACTIONS OF ANY PARTY TO THIS AGREEMENT. NO PARTY HERETO SHALL SEEK TO CONSOLIDATE, BY COUNTERCLAIM OR OTHERWISE, ANY SUCH ACTION IN WHICH A JURY TRIAL HAS BEEN WAIVED WITH ANY OTHER ACTION IN WHICH A JURY TRIAL CANNOT BE OR HAS NOT BEEN WAIVED.

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF, and intending to be legally bound hereby, the parties have executed this Property Management Agreement as of the day and year first set forth above.

OWNE	<u>R</u>			
[
By:	[]			
	By: Leo S. Ullman President			
AGENT				
CEDAR SHOPPING CENTERS PARTNERSHIP, L.P., a Delaware limited partnership				
Ву:	Cedar Shopping Centers, Inc., a Maryland corporation, its general partner			
Ву:				
RIOCAN REIT				
RIOCAN REAL ESTATE INVESTMENT TRUST				
Ву:				

EXHIBIT E

ALLOTTED CONSIDERATION

Cedar and RioCan agree that the Consideration for the Interests shall be allocated among the assets owned by the Property Owner as of the Closing as follows:

Cash and Cash Equivalents (Class I) Supplies, Prepaid Expenses and Other Current Assets (Class V) Equipment, Furniture and Fixtures (Class V) Real Property Lease and Improvements, and Construction of Improvements in Progress (Class V) Goodwill, Going Concern Value and Other Section 197 Intangibles (Classes VI and VII)

Dollar amount as of the Closing Date Tax basis as of the Closing Date

Tax basis as of the Closing Date Balance

None

Property Allotted Consideration

Columbus Crossing \$19,630,000 Franklin Village* \$43,980,000 \$21,560,000 Loyal Plaza Stop & Shop at Bridgeport \$7,180,000 Blue Mountain Commons

n/a

 Sunset Crossing
 \$7,880,000

 Shaw's Plaza
 \$16,290,000

 TOTAL:
 \$116,520,000

*Includes escrow holdback estimates for 5 spaces (Franklin Village Renewal Lease Spaces) with pending lease renewals with an estimated NOI of \$65,000 and a lease value equal to \$5,467,000. Eighty percent (80%) of the lease value for the 5 renewal tenancies, rounded, equals \$4,380,000. The Allotted Consideration for Franklin Village does not include additional Earn-Out Proceeds related to the Franklin Village Applebee's Space and the Franklin Village New Lease Space.

EXHIBIT F

FORM

ESCROW AGREEMENT

THIS ESCROW AGREEMENT (this "Agreement"), dated as of the [__] day of October, 2009, is among COMMONWEALTH LAND TITLE INSURANCE COMPANY, Two Grand Central Tower 140 East 45th Street, 22nd Floor, New York, NY 10017 ("Escrowee"), CEDAR SHOPPING CENTERS PARTNERSHIP, L.P., a Delaware limited partnership, having an office at 44 South Bayles Avenue, Port Washington, New York 11050 ("Cedar") and RIOCAN HOLDINGS USA INC., a Delaware corporation, having an office c/o RioCan Real Estate Investment Trust, RioCan Yonge Eglinton Centre, 2300 Yonge St., Suite 500, P.O. Box 2386, Toronto, Ontario, M4P 1E4 ("RioCan").

WITNESSETH

WHEREAS, Cedar and RioCan entered into that certain Agreement Regarding Purchase of Partnership Interests (hereinafter referred to as the "<u>Purchase and Sale Agreement</u>"); dated as of the date hereof, for the purchase and sale of the Interests.

WHEREAS, the Purchase and Sale Agreement provides for the terms and conditions applicable to the sale and purchase of the Interests and the performance obligations and rights of Cedar and RioCan; and

WHEREAS, Cedar and RioCan agree, pursuant to the Purchase and Sale Agreement, that Escrowee shall hold, in escrow the Deposit in accordance with the terms and conditions of the Purchase and Sale Agreement and this Agreement.

NOW, THEREFORE, the parties hereto agree as follows:

- Appointment of Agent.
- (a) Cedar and RioCan hereby appoint Escrowee to act as their escrow agent on the terms and conditions hereinafter set forth, and Escrowee accepts such appointment.
- (b) RioCan shall deliver the Deposit to Escrowee pursuant to the wire instructions attached hereto as Exhibit A in accordance with the terms of the Purchase and Sale Agreement. Escrowee agrees to hold the Deposit on behalf of the parties to the Purchase and Sale Agreement, and to apply, disburse and deliver the Deposit as provided in the Purchase and Sale Agreement and this Agreement. In the event of any conflict between the terms and conditions of the Purchase and Sale Agreement and the terms or conditions of this Agreement, as to the obligations of Escrowee, the terms and conditions of this Agreement shall govern and control, and as to Cedar and RioCan, the terms and conditions of the Purchase and Sale Agreement shall control.

Disposition of the Required Deposit.

- (a) Escrowee shall hold the Deposit in an interest bearing segregated account at [JPMorgan Chase Bank, N.A.] which rate of interest need not be maximized. Escrowee shall not commingle the Deposit with any other funds.
- (b) Escrowee shall pay the Deposit in accordance with the terms of the Purchase and Sale Agreement. If, prior to any Closing, either party makes a written demand upon Escrowee for delivery of the Deposit, Escrowee shall give written notice to the other party of such demand. If a written notice of objection to the proposed payment is not received from the other party within seven (7) Business Days after the giving of notice by Escrowee, Escrowee is hereby authorized to deliver the Deposit to the party who made the demand. If Escrowee receives a written notice of objection within said period, then Escrowee shall continue to hold the Deposit and thereafter pay it to the party entitled when Escrowee receives (a) written notice from the objecting party withdrawing the objection, or (b) a written notice signed by both parties directing disposition of the Deposit, or (c) a judgment or order of a court of competent jurisdiction.
- (c) Nothing in this Section 2 shall have any effect whatsoever upon Escrowee's rights, duties, and obligations under Section 3.

3. <u>Concerning Escrowee</u>.

(a) Escrowee shall be protected in relying upon the accuracy, acting in reliance upon the contents, and assuming the genuineness of any notice, demand, certificate, signature, instrument or other document which is given to Escrowee without verifying the truth or accuracy of any such notice, demand,

certificate, signature, instrument or other document;

- (b) Escrowee shall not be bound in any way by any other contract or understanding between Cedar and RioCan, whether or not Escrowee has knowledge thereof or consents thereto unless such consent is given in writing;
- (c) Escrowee's sole duties and responsibilities shall be to hold and disburse the Deposit in accordance with this Agreement and the Purchase and Sale Agreement;
- (d) Upon the disbursement of the Deposit in accordance with this Agreement, Escrowee shall be relieved and released from any liability under this Agreement;
- (e) Escrowee may resign at any time upon at least fifteen (15) Business Days prior written notice to Cedar and RioCan hereto. If, prior to the effective date of such resignation, Cedar and RioCan hereto shall have approved, in writing, a successor escrow agent, then upon the resignation of Escrowee, Escrowee shall deliver the Deposit to such successor escrow agent. From and after such resignation and the delivery of the Deposit to such successor escrow agent, Escrowee shall be fully relieved of all of its duties, responsibilities and obligations under this Agreement, all of which duties, responsibilities and obligations shall be performed by the appointed successor escrow agent. If for any reason Cedar and RioCan shall not approve a successor escrow agent within such period, Escrowee may bring any appropriate action or proceeding for leave to deposit the Deposit with a court of competent jurisdiction, pending the approval of a successor escrow agent, and upon such deposit Escrowee shall be fully relieved of all of its duties, responsibilities and obligations under this Agreement;
- (f) Cedar and RioCan hereby agree to, jointly and severally, indemnify, defend and hold harmless Escrowee from and against any liabilities, damages, losses, costs or expenses incurred by, or claims or charges made against, Escrowee (including reasonable attorneys' fees and disbursements) by reason of Escrowee performing its obligations pursuant to, and in accordance with, the terms of this Agreement, but in no event shall Escrowee be indemnified for its gross negligence, willful misconduct or breach of the terms of this Agreement;
- (g) In the event that a dispute shall arise in connection with this Agreement or the Purchase and Sale Agreement, or as to the rights of Cedar and RioCan in and to, or the disposition of, the Deposit, Escrowee shall have the right to (w) hold and retain all or any part of the Deposit until such dispute is settled or finally determined by litigation, arbitration or otherwise, or (x) deposit the Deposit in an appropriate court of law, following which Escrowee shall thereby and thereafter be relieved and released from any liability or obligation under this Agreement, or (y) institute an action in interpleader or other similar action permitted by stakeholders in the State of New York, or (z) interplead Cedar or RioCan in any action or proceeding which may be brought to determine the rights of Cedar and RioCan to all or any part of the Deposit; and
- (h) Escrowee shall not have any liability or obligation for loss of all or any portion of the Deposit by reason of the insolvency or failure of the institution of depository with whom the escrow account is maintained.

4. <u>Termination</u>.

This Agreement shall automatically terminate upon the delivery or disbursement by Escrowee of the Deposit in accordance with the terms of the Purchase and Sale Agreement and terms of this Agreement, as applicable.

5. Notices.

All notices, demands, consents, reports and other communications provided for in this Agreement shall be in writing, shall be given by a method prescribed in this Section and shall be given to the party to whom it is addressed at the address set forth below or at such other address(es) as such party hereto may hereafter specify by at least seven (7) days' prior written notice.

To Cedar:

c/o Cedar Shopping Centers, Inc. 44 South Bayles Avenue Port Washington, New York 11050 Attention: Leo S. Ullman Facsimile: (516) 767-6497

With a copy to:

Stroock & Stroock & Lavan LLP 180 Maiden Lane New York, New York 10038-4982 Attention: Steven P. Moskowitz, Esq. Facsimile: (212) 806-6006

To RioCan:

Yonge Eglinton Centre 2300 Yonge Street Suite 500, P.O. Box 2386 Toronto, Ontario M4P 1E4 Attention: Rags Davloor Facsimile: (416) 866-3020 With a copy to:

Yonge Eglinton Centre 2300 Yonge Street Suite 500, P.O. Box 2386 Toronto, Ontario M4P 1E4 Attention: Jonathan Gitlin

Attention: Jonathan Gitlin Facsimile: (416) 866-3020

With a copy to:

Goodmans LLP 250 Yonge Street Suite 2400 Toronto, Ontario M5B 2M6 Attention: Juli Morrow

Attention: Juli Morrow Facsimile: (416) 979-1234

To Escrowee:

Commonwealth Land Title Insurance Company Two Grand Central Tower 140 East 45th Street, 22nd Floor New York, New York 10017 Attention: Robert Fitzgerald Engine (212) 986, 2315

Facsimile: (212) 986-3215 Telephone: (212) 973-4809

Any party hereto may change the address to which notice may be delivered hereunder by the giving of written notice thereof to the other parties as provided hereinbelow. Any notice or other communication delivered pursuant to this Section 5 may be mailed by United States or Canadian certified air mail, return receipt requested, postage prepaid, deposited in a United States or Canadian Post Office or a depository for the receipt of mail regularly maintained by the United States Post Office or the Canadian Post Office, as applicable. Such notices, demands, consents and reports may also be delivered (i) by hand or reputable international courier service which maintains evidence of receipt or (ii) by facsimile with a confirmation copy delivery by hand or reputable international courier service which maintains evidence of receipt. Any notices, demands, consents or other communications shall be deemed given and effective when delivered by hand or courier or facsimile, or if mailed only, five (5) Business Days after mailing. Notwithstanding the foregoing, no notice or other communication shall be deemed ineffective because of refusal of delivery to the address specified for the giving of such notice in accordance herewith. The provisions of this Section 5 shall survive the Closings and/or a termination of this Agreement.

Capitalized Terms.

Capitalized terms used herein and not otherwise defined herein shall have the meanings ascribed to them in the Purchase and Sale Agreement.

Governing Law.

THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK APPLICABLE TO AGREEMENTS MADE AND TO BE PERFORMED ENTIRELY WITHIN SUCH STATE, WITHOUT REGARD TO THE CONFLICTS OF LAW PRINCIPLES OF SUCH STATE. THE PARTIES HERETO WAIVE TRY BY JURY IN ANY ACTION OR PROCEEDING ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT.

8. Successors.

This Agreement shall be binding upon and inure to the benefit of the respective successors and permitted assigns of the parties hereto; provided, however, that except as expressly provided herein as to the Escrowee, this Agreement may not be assigned by any party without the prior written consent of the other parties.

9. <u>Entire Agreement</u>.

This Agreement, together with the Purchase and Sale Agreement, contains the entire agreement and understanding between the parties hereto with respect to the subject matter hereof and supersedes all prior agreements and understandings relating to such subject matter.

10. <u>Amendments</u>.

Except as expressly provided in this Agreement, no amendment, modification, termination, cancellation, rescission or supersession to this Agreement shall be effective unless it shall be in writing and signed by each of the parties hereto.

Counterparts and/or Facsimile Signatures.

This Agreement may be executed in any number of counterparts, including counterparts transmitted by facsimile, any one of which shall constitute an original of this Agreement. When counterparts or facsimile copies have been executed by all parties, they shall have the same effect as if the signatures to each counterpart or copy were upon the same documents and copies of such documents shall be deemed valid as originals. The parties agree that all such signatures may be transferred to a single document upon the request of any party. This Agreement shall not be binding unless and until it shall be fully executed and delivered by all parties hereto. In the event that this Agreement is executed and delivered by way of facsimile transmission, each party delivering a facsimile counterpart shall promptly deliver an ink-signed original counterpart of the Agreement to the other party by overnight courier service; provided however, that the failure of a party to deliver an ink-signed original counterpart shall not in any way effect the validity, enforceability or binding effect of a counterpart executed and delivered by facsimile transmission.

12. Severability.

If any provision of the Agreement or the application of any such provision to any person or circumstance shall be held invalid, illegal, or unenforceable in any respect by a court of competent jurisdiction, such invalidity, illegality or unenforceability shall not affect any other provision hereof.

13. <u>EIN</u>.

Cedar shall provide its employer identification numbers to Escrowee promptly following execution and delivery of this Agreement. Each of the parties hereto shall execute and deliver to the others any documents reasonably necessary for establishing the escrow account for the Deposit promptly following request.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the parties have executed and delivered this Escrow Agreement as of the date and year first above written.

ESCROWEE:
COMMONWEALTH LAND TITLE INSURANCE COMPANY
By: Name: Title:
CEDAR:
CEDAR SHOPPING CENTERS PARTNERSHIP, L.P., a Delaware limited partnership
By: Cedar Shopping Centers, Inc., a Maryland corporation, its general partner
By: Leo S. Ullman President
RIOCAN:
RIOCAN HOLDINGS USA INC., a Delaware corporation
By: Name: Title:

EXHIBIT G

FORM OF ASSIGNMENT AND ASSUMPTION AGREEMENT

THIS ASSIGNMENT AND ASSUMPTION AGREEMENT (this "Agreement") is made as of the _____day of ______, 20__, by and between [CEDAR SHOPPING CENTERS PARTNERSHIP, L.P., a Delaware limited partnership] ("Assignor") and [REIT PROPERTY SUBSIDIARY], a Delaware limited partnership ("Assignee").

$\underline{\textbf{W} \, \textbf{I} \, \textbf{T} \, \textbf{N} \, \textbf{E} \, \textbf{S} \, \textbf{S} \, \textbf{E} \, \textbf{T} \, \textbf{H}};$

		hundred percent (100%) of the limited [partnership /			
liability company] interests in [], a [] (the 'Interests')] dated as of [] (the			
"[Partnership/Operating] Agreement");					
October [], 2009 (the "Contract"), between partnership ("Cedar"), and RioCan Holdings U	n Cedar SA Inc.,	t Regarding Purchase of Partnership Interests, dated as of Shopping Centers Partnership, L.P., a Delaware limited a Delaware corporation ("RioCan"), Cedar and RioCan cquire all right, title and interest of Assignor in and to the			
WHEREAS, all capitalized terms used Contract.	l but no	t defined herein shall have the meanings set forth in the			
NOW, THEREFORE, for valuable consideration in hand paid, the receipt and sufficiency of which is hereby acknowledged, the parties hereto agree as follows:					
1. Assignor hereby assigns, conveys, transfers and sets over unto Assignee, without recourse, representation or warranty except as set forth in the Contract, all right, title and interest of Assignor in and to the Interests.					
2. Subject to the terms of the [Pa assignment and assumes all obligations with resp		ip/Operating] Agreement, Assignee hereby accepts such ne Interests.			
3. This Agreement shall be binding upon and inure to the benefit of the parties hereto and each of their respective successors and assigns. None of the provisions of this Agreement shall be for the benefit of or enforceable by any other person or entity.					
		y number of counterparts, and each such counterpart will terparts shall constitute one and the same instrument.			
5. Each party represents and wa duly authorized to do so.	rrants th	nat the individual signing this Agreement on its behalf is			
to time at the request of the other, and without fu	ırther co	that they will execute, deliver and acknowledge from time onsideration, all such further instruments of assignment or nired in order to give effect to the transactions described			
7. This Agreement shall be const. State of Delaware (without regard to principles of		accordance with and governed by the internal laws of the cts of laws).			
[Remainde	er of Pag	ge Intentionally Left Blank]			
IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the day of0					
	ASSIC	GNOR:			
		R SHOPPING CENTERS PARTNERSHIP, L.P., ware limited partnership			
	By:	Cedar Shopping Centers, Inc., a Maryland corporation, its general partner			
		Ву:			
		Leo S. Ullman			
		President			
	ASSIC	GNEE:			
		PROPERTY SUBSIDIARY], ware limited partnership			
	Ву:	[REIT PROPERTY SUBSIDIARY GP], a Delaware limited liability company, its general partner			
		Ву:			
		Leo S. Ullman			
		Authorized Person			

ADDITIONAL MATERIALS

1. COLUMBUS CROSSING

- Super Fresh. Exhibit B (Site Plan) need legible copy
- Old Navy Fully executed copy of Non-Disturbance and Attornment Agreement dated 03/23/2004.

2. FRANKLIN

- Applebees Memorandum Establishing Commencement Date of Lease, date 03/03/1988. In addition, only the first page of the Lease Statement and Agreement dated 02/02/1988 was provided.
- Marshalls -

Waiver re Teppanyaki

- Radio Shack Rider to Extension and Amendment to Lease Agreement dated 12/12/1997.
- Stop & Shop Exhibit AA from the Amendment dated August 7, 1987
- *GNC* Letter dated June 27, 1996.

3. LOYAL PLAZA

- Kmart Amendments dated 08/09/1976, 08/13/1976 and 10/07/1976
 Waivers from Kmart regarding use of pylon sign
- *PLCB* − Ex. B
- Red Lobster A complete and executed copy of a Memorandum of Lease

4. SUNSET

- Giant Food Stores Exhibit B from the Lease.
- *Holiday Hair* Exhibit A of an estoppel dated 12/04/03.
- Premier Tanning Lease Modification Agreement dated 11/14/07, the Consent and Ratification therein is not executed by David Evans, one of the guarantors.

EXHIBIT I

BLUE MOUNTAIN DEVELOPMENT PARCEL

EXHIBIT J

FRANKLIN VILLAGE EARN-OUT SPACE

TENANT	SUITE #	SQ. FT.	NOTES
Franklin Village New L			
Vacant	034	3,908	1,600 for Sally Beauty and 2,300 sf for Five Guys (Executed LOIs)
Vacant	OA201	490	Nurse Staffing (Executed Lease)
Applebees Space			
Applebees	038	5,682	
Franklin Village Renew	al Lease Spa	ice	
L'Equipe	037	2,070	
Olympia Sports	014	3,550	
Radio Shack	004	2,000	
The Men's Wearhouse	026	3,600	
Dress Barn	021	10,150	

FOR IMMEDIATE RELEASE

Contact Information: Cedar Shopping Centers, Inc. Leo S. Ullman, Chairman, CEO and President (516) 944-4525 lsu@cedarshoppingcenters.com

CEDAR SHOPPING CENTERS ANNOUNCES STRATEGIC ALLIANCE WITH RIOCAN REAL ESTATE INVESTMENT TRUST

- Agreements Contemplate Private Placement of Common Stock and Joint Venture for Seven Existing Cedar Properties; Net Proceeds to Cedar estimated at \$100 million –
 - Cedar and RioCan also establish joint venture for new acquisitions -
 - Company Updates Progress on Renewal of Credit Facility -

Port Washington, New York – October 26, 2009 – Cedar Shopping Centers, Inc. ("Cedar" or the "Company") (NYSE: CDR) today announced it has entered into definitive agreements with RioCan Real Estate Investment Trust (TSX: REI.UN) of Toronto, Canada ("RioCan") which provide for the following:

- RioCan will make a \$40 million private placement investment, of 6,666,666 shares of Cedar common stock at \$6.00 per share ("Private Placement") which is conditioned on approval by the NYSE that is expected to occur within 10 days;
- Cedar will grant to RioCan warrants to purchase 1,428,570 shares of Cedar common stock, at an
 exercise price of \$7.00 per share, exercisable over a period of two years;
- Cedar and RioCan will enter into an 80% (RioCan) and 20% (Cedar) joint venture for seven supermarket-anchored properties presently owned by Cedar ("Existing Property Joint Venture");
- Cedar and RioCan will enter into an agreement to acquire primarily supermarket-anchored properties in northeast and mid-Atlantic states during the next two years in amounts anticipated at up to \$500 million in such 80% (RioCan) and 20% (Cedar) joint venture format ("Future Property Joint Venture"); and
- Cedar and RioCan will enter into a standstill agreement with respect to RioCan's ownership of Cedar's common shares for a period of three years.

"We expect the great financial strength of RioCan and its commitment to our Company to be beneficial for both companies in the coming years," commented Leo Ullman, CEO of Cedar. "We believe that this important investment in our Company will lead to the creation of meaningful added value for the benefit of shareholders as we move forward. We have been impressed by RioCan and its management team led by Ed Sonshine. Through many meetings and a great deal of time spent by RioCan with us in its due diligence process, we have come to recognize great symmetry in our conservative approaches to our respective property portfolios at every level of our respective operations."

Ed Sonshine, CEO of RioCan, stated "We are delighted to have been able to conclude the private placement and joint venture arrangements with Cedar Shopping Centers. We have been looking to establish a platform in the United States for future growth of our company and we have been greatly impressed with Cedar's management team, the quality and stability of its property portfolio, and its demonstrated ability to expand and enhance that portfolio. We look forward to a long and mutually successful relationship with Cedar as part of our future growth in the U.S."

Aggregate net proceeds to Cedar will total an estimated \$100 million, which consists of \$40 million from the Private Placement and more than \$60 million from the Existing Property Joint Venture, less closing costs and fees. Closings related to the Existing Property Joint Venture are expected to be completed in the first quarter of 2010. Closing dates for all but two properties are subject to timing of lender consents to the transfer of properties to the Existing Property Joint Venture. Net proceeds from the Private Placement and Existing Property Joint Venture will be used to reduce Cedar's outstanding balances under its secured revolving credit facility for stabilized properties and secured revolving credit facility for development properties.

In addition to the above described transactions, Cedar and RioCan have agreed to enter into a joint venture to acquire additional properties over the next two years. The parties anticipate acquisitions of as much as \$500 million during that period. The properties will primarily consist of supermarket-anchored shopping centers in the northeastern United States. Related to the Future Property Joint Venture, Cedar has granted to RioCan a right of first refusal to participate on a joint venture basis (also 80% RioCan and 20% Cedar) for two years on primarily supermarket anchored properties and other properties in excess of 50,000 square feet to be acquired by Cedar in the states of New York, New Jersey, Pennsylvania, Massachusetts, Connecticut, Maryland and Virginia.

In both the Existing Property Joint Venture and Future Property Joint Venture, Cedar will provide property management, leasing, construction management and financial management services at standard rates. Cedar will also be entitled to certain fees on acquisitions, dispositions, financings and refinancings.

Cedar and RioCan have also entered into a "standstill" arrangement for a period of three years during which RioCan will not be permitted to acquire additional shares of Cedar without consent of the Cedar Board of Directors ("Board"), subject to certain pre-emptive rights granted to RioCan. RioCan will also have the right to designate one director for election to Cedar's Board so long as its ownership interest in Cedar remains at least 9.9%. RioCan's ownership interest as a result of the Private Placement and the warrants, if exercised, will be approximately 15% of Cedar's common shares outstanding.

Cedar also announced today that it has received commitments aggregating more than \$220 million from participating lenders in the renewal of its secured revolving credit facility for stabilized properties. The Company expects to close the renewal at a level of more than \$250 million during the 4^{th} quarter of this year.

Goldman, Sachs & Co. is acting as exclusive financial advisor to Cedar; Stroock & Stroock & Lavan L.L.P. is acting as corporate, real estate and tax counsel to Cedar.

RBC Capital Markets is acting as financial advisor to RioCan. Goodmans LLP and Sidley Austin LLP are acting as legal advisors to RioCan. Davies Ward Phillips & Vineberg LLP and Deloitte LLP are acting as tax advisors to RioCan.

About RioCan Real Estate Investment Trust

RioCan is Canada's largest real estate investment trust with a total capitalization of approximately CDN\$7.8 billion as at September 30, 2009. It owns and manages Canada's largest portfolio of shopping centers with ownership interests in a portfolio of 247 retail properties, including 134 under development, containing an aggregate of over 59 million square feet. For further information, please refer to RioCan's website at www.riocan.com.

About Cedar Shopping Centers, Inc.

Cedar Shopping Centers, Inc. is a fully-integrated real estate investment trust which focuses primarily on ownership, operation, development and redevelopment of "bread and butter" supermarket-anchored shopping centers in coastal mid-Atlantic and New England states. The Company presently owns and operates approximately 13.2 million square feet of GLA at 124 shopping center properties, of which approximately 75% are anchored by supermarkets and/or drugstores with average remaining lease terms of approximately 11 years. The Company's stabilized properties have an occupancy rate of approximately 95%. The Company has also announced a pipeline of seven additional substantially pre-leased primarily supermarket- and drugstore-anchored development properties.

Additional financial and descriptive information on the Company, its operations and its portfolio can be accessed through the Company's website at www.cedarshoppingcenters.com.